FRANCHISE DISCLOSURE DOCUMENT



MTY Franchising USA, Inc. dba *Manchu Wok* a Tennessee corporation 9311 East Via de Ventura Scottsdale, Arizona 85258 (905) 764-7066

www.manchuwok.com www.mtyfranchising.com/brands/manchu-wok/

We grant franchises for MANCHU WOK® Restaurant(s), selling "Chinese and Asian-style" foods and drinks and other menu items, located in street front locations, a Shopping Mall Food Court or in a non-traditional location.

The total investment necessary to begin operation of a Manchu WOK Restaurant franchised business ranges from \$490,050 to \$820,500. This amount includes payments that range from \$453,000 to \$756,500 that must or may be paid to the franchisor or its affiliates.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar-days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. **Note, however, that no governmental agency has verified the information contained in this document.**

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact MTY Franchising USA, Inc., Attn: John Wuycheck, 9311 E. Via de Ventura, Scottsdale, Arizona 85258 and (480) 362-4800.

The terms of your contract will govern your franchise relationship. Don't rely on the disclosure document alone to understand your contract. Read all of your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or an accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "<u>A Consumer's Guide to Buying a Franchise</u>," which can help you understand how to use this disclosure document, is available from the Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

The issuance date of this Franchise Disclosure Document is March 28, 2025.

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibits F or G.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit B includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only <i>Manchu Wok</i> business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be a <i>Manchu Wok</i> franchisee?	Item 20 or Exhibits F or G list current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

<u>Business model can change</u>. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

<u>Renewal</u>. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends</u>. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit A.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About *This* Franchise

Certain states require that the following risk(s) be highlighted:

1. <u>Spousal Liability</u>. Your spouse must sign a document that makes your spouse liable for all financial obligations under the franchise agreement even though your spouse has no ownership interest in the franchise. This guarantee will place both your and your spouse's marital and personal assets, perhaps including your house, at risk if your franchise fails.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

FRANCHISE DISCLOSURE DOCUMENT

TABLE OF CONTENTS

<u>item</u>	<u>Page</u>
ITEM 1 THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES	1
ITEM 2 BUSINESS EXPERIENCE	13
ITEM 3 LITIGATION	16
ITEM 4 BANKRUPTCY	27
ITEM 5 INITIAL FEES	27
ITEM 6 OTHER FEES	29
ITEM 7 ESTIMATED INITIAL INVESTMENT	34
ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	39
ITEM 9 FRANCHISEE'S OBLIGATIONS	46
ITEM 10 FINANCING	47
ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING	47
ITEM 12 TERRITORY	55
ITEM 13 TRADEMARKS	57
ITEM 14 PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	58
ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS	59
ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	59
ITEM 17 RENEWAL, TERMINATION, TRANSFER, AND DISPUTE RESOLUTION	60
ITEM 18 PUBLIC FIGURES	68
ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS	68
ITEM 20 OUTLETS AND FRANCHISEE INFORMATION	70
ITEM 21 FINANCIAL STATEMENTS	73
ITEM 22 CONTRACTS	74
ITEM 23 RECEIPT	74

Exhibits

Α	List of State Regulators/Agents for Service of Process
В	Financial Statements
С	Franchise Agreement
D	Sublease Agreement (subject to amendment to accommodate local laws)
E	Table of Contents of Operating Manual
E-1	In-Store Training Release and Waiver of Liability Agreement
F	Current Franchisees
G	Former Franchisees
Н	State-Specific Disclosure and State-Specific Agreement Amendments
I	Required Lease Term

J K State Effective Dates

Receipt

ITEM 1 THE FRANCHISOR AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

MTY Franchising USA, Inc. is the franchisor. To simplify the language in this Franchise Disclosure Document, "MTY USA," "Manchu Wok," "us," "we," or "Franchisor" means MTY Franchising USA, Inc., the franchisor, doing business as "Manchu Wok." "You" means the person, corporation, partnership or other entity that buys the franchise, the franchisee. If "you" are a business entity, "you" includes shareholders, members or owners of the business to the extent each guaranties or otherwise agrees to perform or be bound by the obligations of the business entity.

The Franchisor, Its Predecessors and Affiliates

We began offering Manchu Wok franchises in the United States in March 2015. MTY Franchising USA, Inc., (formerly known as The Extreme Pita Franchising USA, Inc.), was incorporated in Delaware on March 14, 2001, and converted to MTY Franchising USA Inc., a Tennessee Corporation, as of October 9, 2019. You will see references to our former corporate names in our audited financial statements. Our principal business address in the United States is 9311 East Via de Ventura, Scottsdale, AZ 85258. Our agents for service of process are disclosed in **Exhibit A**.

Manchu Wok is a franchising company which sells and grants franchises for the operation of retail restaurant businesses ("Manchu Wok Franchise" or "Manchu Wok Store") using the name "MANCHU WOK®" in the United States. We are not engaged in any other type of business activity. We have been offering Manchu Wok franchises in the United States since March 2015.

We have two predecessors, a parent and affiliates.

For the Manchu Wok brand, we have two predecessors. Our first predecessor, Manchu WOK Franchising USA Inc. ("MWF") was incorporated in Delaware on August 4, 2004, and offered Manchu Wok franchises from March 2005 through October 2014. Our second predecessor, Manchu WOK (USA), Inc. ("MWUSA"), was incorporated in Delaware on July 25, 2000, and offered Manchu Wok franchises from October 2000 until March 2005. MWUSA also continued to provide support services to a number of Manchu Wok franchisees until December 2014. MWF's and MWUSA's business address was 85 Citizen Court, Unit 9, Markham, Ontario, Canada L6G 1A8.

MTY USA's parent corporation is MTY Franchising Inc. ("MTY Canada"), a Canada corporation formerly known as MTY Tiki Ming Enterprises Inc., having an address at 8150 Route Transcanadienne, Suite 200, Ville Saint-Laurent, Quebec, H4S 1M5, Canada. MTY Canada was initially incorporated under the Canada Business Corporations Act on February 13, 1979, under the name "Matoyee Enterprises Inc." On September 24, 2013, MTY Canada acquired our stock. MTY Canada is a wholly owned subsidiary of MTY Food Group, Inc. ("MTY"), a public corporation listed on the Toronto Stock Exchange. MTY's principal place of business is also 8150 Route Transcanadienne, Suite 200, Ville Saint-Laurent, Quebec, H4S 1M5, Canada.

On September 24, 2013, MTY Canada, acquired our stock. MTY Canada is a wholly owned subsidiary of MTY. MTY Canada is a corporation that was initially incorporated under the Canada Business Corporations Act on February 13, 1979, under the name "Matoyee Enterprises Inc." Its principal place of business is 8150 Route Transcanadienne, Suite 200, Ville Saint-Laurent, Quebec, H4S 1M5, Canada. MTY is a public corporation and its principal place of business is 8150 Route Transcanadienne, Suite 200, Ville Saint-Laurent, Quebec, H4S 1M5, Canada.

MTY Canada currently operates or has franchised over (55) different restaurant concepts and has over 2,500 units under the following brands primarily in Canada and other international countries: Allo Mon Coco, Baton Rouge Steakhouse & Bar, Ben & Florentine, Big Smoke Burger, Café Depot, Casa Grecque, Country Style, Cultures, Dagwoods, Extreme Pita, Giorgio, Jugo Juice, Kim Chi, Koryo, Koya, Kuto Comptoir A Tartares, La Crémière, La Diperie (and Cakes & Shakes by La Dip), Madisons, Manchu Wok, Toujours Mikes, mmmuffins, Mr. Souvlaki, Mr. Sub, Mucho Burrito, Muffin Plus, O'Burger, Pizza Delight, Poke by Sushi Shop, Scores, South St. Burger, Spice Bros, Steak & Frites St. Paul Sukiyaki, Sushi Go, Sushiman, Sushi Shop, Thai Express, Thaizone, The Works, Tiki Ming, Timothy's World Coffee, The COOP Wicked Chicken, Tosto, Turtle Jack's, Tutti Frutti, Valentine, Vanellis, Vie & Nam, Villa Madina, Spice Brothers, Steak Frites and YUZU MTY or one of its subsidiaries is the master licensee for the following brands: TCBY (Canada), Taco Time (Canada), and Au Vieux Duluth Express (Quebec and Ontario, Canada), and Van Houtte (Quebec and Ontario, Canada). MTY Canada also provides support services to Manchu Wok stores in the United States.

Effective March 1, 2018, MTY through the merger of a wholly-owned subsidiary with Imvescor Restaurant Group Inc. ("IRG"), acquired all the outstanding shares of IRG. At closing, IRG operated 5 brands in Canada and had 261 locations in operation.

On May 25, 2016, MTY entered into an agreement with Kahala Brands, Inc., formerly known as Kahala Brands, Ltd., having an address at 9311 East Via de Ventura, Scottsdale, AZ 85258 ("Kahala Brands"), whereby Kahala Brands agreed to merge with a wholly-owned subsidiary of MTY ("Kahala Brands Transaction"). The Kahala Brands Transaction closed July 26, 2016. As of the issuance date, Kahala Brands or one or more of its subsidiaries has franchised or currently operates approximately 2,600 quick service restaurants which consist of the multiple brands further detailed in the chart below.

Effective May 23, 2019, Papa Murphy's Holdings, Inc. ("**Papa Murphy's**"), a Delaware corporation having an address at 8000 NE Parkway Dr. #350, Vancouver, WA 98662, merged with a whollyowned subsidiary of MTY. MTY USA succeeded as the parent company of Papa Murphy's. The current franchisor offers Papa Murphy's unit franchises which sell or offer for sale pizza products. As of November 30, 2024, there were 1,036 franchised (1,001 in the United States, 35 internationally) and 43 company-owned stores in the United States.

Agent for Service of Process

Unless otherwise specified, our registered agent for service of process is CT Corporation System, 300 Montvue Road, Knoxville, Tennessee 37919. We are members in good standing of the International Franchise Association ("**IFA**") and we therefore comply with the IFA's Code of Ethics.

Other Franchises Offered by Franchisor or its Affiliate

MTY USA IS <u>ONLY</u> OFFERING A MANCHU WOK UNIT FRANCHISE UNDER THIS DISCLOSURE DOCUMENT. EACH OF THE FRANCHISES DETAILED BELOW ARE OFFERED BY MTY USA OR ITS AFFILIATE UNDER SEPARATE DISCLOSURE DOCUMENTS FOR EACH BRAND.

We have the following U.S.-based affiliates through common ownership by MTY Food Group, that also offer franchises in the United States and internationally: (1) MTY USA, a Tennessee corporation having an address of 9311 E. Via De Ventura, Scottsdale, Arizona 85258; (2) MTY Canada, a Canada corporation having an address at 8210, route Transcanadienne, Suite 200,

Saint-Laurent, Québec, H4S 1M5, Canada; (3) Kahala Franchising, LLC ("Kahala Franchising"). an Arizona limited liability company with its principal address at 9311 E. Via De Ventura, Scottsdale, Arizona 85258; (4) BF Acquisition Holdings, LLC ("BFAH"), a Delaware limited liability company with a principal address at 9311 E. Via De Ventura, Scottsdale, Arizona 85258; (5) La Salsa Franchise, LLC, a Delaware limited liability company with a principal address at 9311 E. Via De Ventura, Scottsdale, Arizona 85258; (6) Imvescor Restaurant Group Inc. ("IRG") a Canada corporation having an address at 8210, route Transcanadienne, Suite 200, Saint-Laurent, Québec, H4S 1M5, Canada; (7) Papa Murphy's International, LLC (Papa Murphy's), a Delaware limited liability company having an address at 8000 NE Parkway Drive, Suite 350, Vancouver, Washington 98662; (8) VI BrandCo, LLC ("Village Inn"), a Delaware limited liability company having a principal address at 12701 Whitewater Drive, Suite 100, Minnetonka, Minnesota 55343-4164, (9) BQ Concepts, LLC, an Arizona limited liability company having a principal address at 12701 Whitewater Drive, Suite 100, Minnetonka, Minnesota 55343-4164; (10) Wetzel's Pretzels, LLC, a California limited liability company with a principal business office at 35 Hugus Alley, Suite 300, Pasadena, CA 91103; (11) and Famous Dave's of America, Inc., a Minnesota limited liability company having a principal address at 12701 Whitewater Drive, Suite 100, Minnetonka, Minnesota 55343-4164. These affiliates franchise over 50 different concepts.

The following summarizes Manchu Wok and these other quick service restaurant brands as of November 30, 2024, including the type of restaurant business, number of franchised units in operation as of November 30, 2024, and the date MTY USA and/or its current or former affiliates offered franchises in those brands:

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
Extreme Pita	Restaurants serving wrap-style hot and cold pita and wrap sandwiches	1 franchised unit	From March 2001 to July 2014: The Extreme Pita Franchising USA, Inc.; since July 2014: MTY USA
Grabbagreen	Restaurants serving healthy food, juice, smoothies and related products	4 franchised units	Since February 2018 under MTY USA
Ginger Sushi Boutique + Poke Shop	Restaurant serving a variety of sushi menu items and drinks	0 franchised units	From September 2015 under MTY USA

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
Mucho Burrito	Restaurants offering burritos, quesadillas, tacos, nachos, and other assorted food and drinks	1 franchised unit	From January 2010 under Mucho Burrito Franchising USA, Inc.; from March 2019 under MTY USA
Thai Express	Restaurant serving "Thai- style" foods and drinks	7 franchised units (5 in the United States and 2 internationally) (plus 2 company-owned)	From February 2015 under MTY USA
La Diperie	Restaurant serving retail sale of an ice cream product and various dips and toppings	1 franchised unit	From April 2019 under MTY USA
Ben & Florentine	Restaurant serving a superior breakfast & lunch experience	0 franchised units	From December 2018 under MTY USA
Baja Fresh	Restaurant offering a limited menu featuring fresh high quality Mexican-style food products	69 franchised units (67 in the United States and 2 internationally) (plus 6 companyowned units)	October 2016 until July 2017 under Triune, LLC and since then under BF Acquisition Holdings, LLC
La Salsa	Restaurant offering a limited menu featuring fresh high quality Mexican-style food products	5 franchised units	October 2016 under La Salsa Franchise, LLC

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
The Counter	Full service restaurant featuring build-your-own burgers, signature burgers, side dishes, sandwiches, and salads	9 franchised units (8 in the United States and 1 internationally) (plus 3 company-owned units)	December 2017 under CB Franchise Systems, LLC. Then from March 2019 under MTY USA
Built Custom Burgers	Fast casual restaurant featuring build-your-own burgers, signature burgers, side dishes, sandwiches, and salads	7 franchised units (3 in the United States and 4 internationally)	December 2017 under Built Franchise Systems, LLC. Then from March 2019 under MTY USA
sweetFrog	Restaurant offering frozen yogurt using a self-serve delivery format	216 franchised units (206 in the United States which include 9 licensed franchisees plus 10 internationally)	September 2018 under MTY USA
Manchu WOK	Quick service restaurant serving fast and fresh Chinese cuisine	15 franchised units	March 2015: MTY USA
Papa Murphy's	Retail food outlet serving primarily take and bake pizza	1,036 franchised units (1,001 in the United States and 35 internationally) plus 43 company-owned units	From May 2019 Papa Murphy's International LLC
Famous Dave's	Restaurants specializing in authentic, down- home, genuine smoked barbecue	85 franchised units (77 in the United States and 8 internationally) plus 32 companyowned units	From March 1994 under Famous Dave's of America, Inc.

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
Village Inn	Restaurant specializing in pancakes, omelets, skillets, eggs, and other popular breakfast items.	88 franchised units plus 26 company- owned units	From August 2020 under VI BrandCo, LLC
Barrio Queen	Restaurants specializing in authentic Southern Mexican	8 company-owned units	From March 2023 under BQ Concepts, LLC
Champps Kitchen + Bar	sports theme restaurants that provide the public with high-quality food and beverage	2 franchised units plus 1 company-owned unit	From September 1999 to October 2008 under Champps Entertainment, Inc. and From August 2023 under BQ Concepts, LLC
Sauce Pizza / Wine	Restaurants serving wood-fired pizzas, a variety of pasta dishes, and salads	13 company-owned units	March 2024
Wetzel's Pretzels	Restaurant specializing in hand-rolled fresh- baked soft pretzels	406 franchised units (389 in the United States and 17 internationally) plus 36 company-owned units	From April 1996 under Wetzel's Pretzels, LLC

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
Blimpie	Restaurants serving submarine sandwiches and salads	100 franchised units (97 in the United States and 3 internationally) (plus 4 company-owned units in the United States)	From 2006 until 2010 by Kahala Franchise Corp. and since August 2010 under Kahala Franchising
Chicken Strips and Dips	Ghost kitchen concept serving primarily chicken tenders.	3 franchised units	March 2022, Kahala Franchising.
		1,427 franchised units (992 in the United States and 435 internationally)(plus 2 company-owned units).	
Cold Stone Creamery	Restaurants serving super- premium freshly made ice cream, cakes, pies, smoothies, shakes, and other frozen dessert products	105 Cold Stone Creamery franchises also sell Rocky Mountain Chocolate Factory® products and 1 Cold Stone Creamery franchise also sells Tim Hortons® products.	From May 2007 until March 2008 by Cold Stone Creamery, Inc., from March 2008 until March 2010 by Kahala Franchise Corp., and since August 2010 under Kahala Franchising
		Additionally, 13 licensed units.	
Frullati Cafe & Bakery	Restaurants serving sandwiches, salads, smoothies and baked goods	11 franchised units	From 1999 until 2004 by Frullati Franchise Systems, Inc., from 2004 until March 2010 by Kahala Franchise Corp., and since August 2010 under Kahala Franchising

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
Great Steak	Restaurants serving Philadelphia cheesesteak sandwiches, chicken sandwiches and French fries	34 franchised units (24 in the United States and 10 internationally)	From 2004 until March 2010 by Kahala Franchise Corp. and since August 2010 under Kahala Franchising
Johnnie's New York Pizzeria	Restaurants serving New York style pizza, calzones, salads, and related Italian cuisine menu items	2 franchised units	From 2006 until March 2010 by Kahala Franchise Corp. and since August 2010 under Kahala Franchising
Kahala Coffee Traders	Restaurants serving coffee and espresso, tea, baked goods, parfaits, sandwiches and merchandise	5 franchised units. And 1 licensed unit.	November 2011 under Kahala Franchising
Maui Wowi	Store fronts or portable units serving fruit smoothies, Hawaiian coffee and espresso	88 franchised units (80 in the United States and 8 internationally)	Since November 2015 under Kahala Franchising
NrGize Lifestyle Cafe	Cafes serving smoothies, fruit drinks and nutritional supplements	57 franchised units	From 2006 until March 2010 by Kahala Franchise Corp. and since August 2010 under Kahala Franchising
Pinkberry	Restaurants serving frozen yogurt, yogurt drinks, smoothies and frozen desserts	59 franchised units. And 30 licensed units.	From July 2008 until April 2016 under Pinkberry Ventures, Inc. and since June 2016 under Kahala Franchising

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
Planet Smoothie	Restaurants serving smoothies, smoothie bowls, juices and nutritional supplements	162 franchised units (154 in the United States and 8 internationally) Additionally, as of fiscal year end there were 2 Tasti D-Lite outlets.	Since June 2016 under Kahala Franchising
Ranch One	Restaurants specializing in grilled and crispy breaded chicken sandwiches	2 franchised units	From 2001 until 2004 by Ranch *1 Group, Inc., from 2004 until March 2010 by Kahala Franchise Corp., and since August 2010 under Kahala Franchising
Samurai Sam's Teriyaki Grill	Restaurants serving Japanese rice bowls and noodle bowls	12 franchised units	From 2003 until 2004 by SP Franchising, Inc., from 2004 until March 2010 by Kahala Franchise Corp., and since August 2010 under Kahala Franchising
Surf City Squeeze	Juice bars serving smoothies, fruit drinks and nutritional supplements	62 franchised units (plus 1 company- owned unit)	From 1994 until 2004 by Malibu Smoothie Franchise Corp. and Surf City Squeeze Franchise Corp., from 2004 until March 2010 by Kahala Franchise Corp., and since August 2010 under Kahala Franchising

Brand Name	Type of Restaurant Business	Number of Units as of November 30, 2024	Dates unit franchises began being offered by us or our affiliate
TacoTime	Restaurants serving freshly-prepared Mexican food including burritos, taco, quesadillas and nachos	222 franchised units (97 franchised in the United States and 125 internationally) (plus 2 company-owned units) Additionally, there are 78 licensed units.	From 2003 until 2004 by Taco Time International, Inc., from 2004 until March 2010 by Kahala Franchise Corp., and since August 2010 under Kahala Franchising

The Business

A Manchu Wok restaurant ("Manchu Wok Restaurant") is a retail quick service restaurant selling Chinese and Asian-style foods and drinks, and other menu items related to the Manchu Wok concept. A Manchu Wok Restaurant serves its customers by operating a uniform system consisting of high standards of service, the use of consistent quality products, and in accordance with the business format created and developed by us and our affiliates ("System"). We authorize you to use the Marks (as defined below) to operate a Manchu Wok Restaurant

Manchu Wok restaurants are located in street front locations, shopping mall food courts or in non-traditional locations such as highway facilities, airport concourses, professional sports facilities, recreational and entertainment facilities, casinos, college campuses, military installations, governmental or institutional locations, supermarkets and department stores where dining facilities are generally located in a common area and a private/separate dining facility is not required to be maintained. Each Manchu Wok Restaurant offers a selection of Chinese cuisine such as sweet and sour pork, crispy chicken wings, chicken balls and mixed vegetables, soft drinks and other complementary items and beverages. Like most quick-service restaurants, customers order, receive and pay for their food purchases at the counter. There is no wait service.

The Restaurant Industry

Throughout the United States, the food service industry is highly competitive with constantly changing market conditions, and is characterized by a profusion of operators, including well financed and highly sophisticated national and regional chains. You will compete with other restaurants, fast food outlets, supermarkets and other food retailers located in your venue or market area which to some extent includes grocery and convenience stores that sell various prepared food products. Some of your competitors may include Manchu Wok restaurants operated by other franchisees or by our affiliates. The extent to which you may succeed at any particular location cannot be predicted. Because of the highly competitive nature of the business involved, successful operation of a Manchu Wok restaurant will depend in part upon the best

efforts, capabilities, management, and efficient operation by the franchisee; as well as the general economic trend and other local marketing conditions.

Manchu Wok Restaurants compete in the quick service segment of the restaurant industry. You will cater to all persons desiring a guick and moderately priced meal, regardless of age or ethnic background. You will have to compete with other fast food restaurant chains such as hamburger, pizza and Mexican fast food chains, as well as other Chinese and Asian-style fast food chains and numerous local Chinese and Asian style restaurants including, perhaps, other restaurants licensed or owned by us or by the Company. The quick-service segment is a highly competitive and developed market, which can be significantly affected by many factors, including changes in local, regional or national economic conditions, changes in consumer tastes, consumer concerns about the nutritional quality of quick service food and increases in the number of, and particular locations of, competing quick service restaurants. Investing in a Manchu Wok Restaurant is a speculative risk and there are no warranties for your success. You will also face normal business risks that could have an adverse effect on your Manchu Wok Restaurant. These include industry developments, such as pricing policies of competitors, supply and demand, and changing consumer tastes. Another potential risk factor is dependence on personnel, the loss of whom could cause adverse effects. Our ability to fulfill our obligations under the Franchise Agreement with you depends in part on our present and future financial condition. Litigation risks also exist, including product liability litigation against you or us because of food-related illnesses and future litigation which may not be foreseeable.

Affiliates That Provide Products or Services to Franchisees

Kahala Management, L.L.C. ("Kahala Management"), an affiliate of ours, is an Arizona limited liability company that provides administrative, legal, accounting, sales, POS phone support, real estate and marketing support services to us. Kahala Holdings, LLC ("Kahala Holdings") and Kahala Restaurants, L.L.C., ("Kahala Restaurants") are affiliates of ours that also own and operate company-owned outlets detailed above. Neptune Equipment Services, LLC, an Arizona limited liability company ("Neptune Equipment"), is an affiliate of ours that is an approved retailer of equipment that sells, distributes, and coordinates logistics of equipment, menu boards, interior and exterior signage, and smallwares to franchisees and licensees.

On August 8, 2022, MTY Franchising USA, Inc. and its wholly owned subsidiary Grill Merger Sub, Inc. ("Merger Sub") entered into an agreement with BBQ Holdings, Inc. ("BBQ"), a Minnesota corporation, providing for the acquisition of BBQ by MTY Franchising USA, Inc., consisting of a tender offer (the "Offer") for all of the outstanding shares of BBQ common stock, followed by a subsequent merger of Merger Sub with and into BBQ (the "BBQ Merger"), with BBQ surviving the BBQ Merger as a wholly-owned subsidiary of MTY Franchising USA, Inc. This transaction included the rights to franchise and/or operate the Barrio Queen, Famous Dave's, Village Inn, Bakers Square, Granite City Food and Brewery, Real Urban BBQ, Craft Republic Bar & Grill, Champps Kitchen + Bar, Fox & Hound, and Tahoe Joe's Famous Steakhouse brands. BBQ, through its subsidiary, VI BrandCo, LLC a Delaware limited liability company, offers Village Inn restaurants. As of November 30, 2024, there were 114 Village Inn restaurants (including franchised and company-owned) in the United States. BBQ, through its subsidiary, Famous Dave's of America, Inc., a Minnesota corporation, offers Famous Dave's restaurants. As of November 30, 2024, there were 109 Famous Dave's restaurants (including franchised and company-owned) in the United States, and 8 franchised restaurants internationally. With the merger closing that took place on September 27, 2022, MTY Franchising USA, Inc. is the parent company of both VI BrandCo, LLC and Famous Dave's of America, Inc.

On December 8, 2022, MTY Franchising USA, Inc. and its wholly owned subsidiary Twisted Merger Sub, Inc. ("WP Merger Sub"), a Delaware corporation, entered into an agreement with COP WP Parent, Inc. ("COP"), a Delaware corporation, providing for the acquisition of COP by MTY Franchising USA, Inc., consisting of a merger of WP Merger Sub with and into COP (the "COP Merger"), with COP surviving the COP Merger as a wholly-owned subsidiary of MTY Franchising USA, Inc., under the name Twisted Merger Sub, Inc. In November 2023, MTY Franchising USA, Inc. became the parent company of Wetzel's Pretzels, LLC ("Wetzel's Pretzels"), a California limited liability company. This transaction included the rights to franchise and/or operate the Wetzel Pretzel's brand, through Wetzel's Pretzels. As of November 30, 2024, there were 425 Wetzel Pretzels restaurants (including franchised and company-owned) in the United States and 17 franchised restaurants internationally. With the merger closing that took place on December 8, 2022, MTY Franchising USA, Inc. is the parent company of Wetzel's Pretzels, LLC.

On December 15, 2022, MTY Franchising USA, Inc. via its wholly owned subsidiary Sauce Restaurants, LLC ("Sauce Restaurants"), an Arizona limited liability company, simultaneously signed and closed an asset purchase agreement with Sauce, LLC, an Arizona limited liability company, Sauce Holdings, LLC, a Delaware limited liability company, and several other of their affiliates (collectively, "Sauce Sellers"), providing for the acquisition of the assets of Sauce Sellers by Sauce Restaurants. This transaction included the rights to operate (and ultimately franchise should Sauce Restaurants so desire) the Sauce Pizza and Wine brand. As of November 30, 2024, there were 13 Sauce Pizza and Wine restaurants (all of which were company-owned) in the United States. BBQ Holdings, Inc., a subsidiary of MTY Franchising USA, Inc., is the direct parent company of Sauce Restaurants.

Some company-owned Manchu Wok Restaurants may be leased by either us or our affiliates (collectively, "Leasing Affiliates"). In these situations, our Leasing Affiliates enter into a direct lease with the property owner ("Master Lease") for the location of the Manchu Wok Restaurant, and then Sublease the location to the franchisee for that particular Manchu Wok Restaurant. In most other cases, you will enter into a lease for the premises of your Manchu Wok Restaurant directly with the property owner.

Regulations

Your Manchu Wok Restaurant will be subject to various federal, state and local health and sanitation laws that apply to restaurant operations. The business of operating a **Manchu Wok Restaurant** is subject to all of the laws, codes and regulations (referred to below generally as "laws") normally applicable to retail businesses. These include: (1) federal, (2) state, and (3) in most instances, city, county, parish, borough, municipality or other local laws.

To operate a Manchu Wok Restaurant, in some cases you may need to obtain a liquor license. State and local laws, regulations and ordinances vary significantly in the procedures, difficulty and cost associated with obtaining a license to sell liquor, the restrictions placed on the manner in which liquor may be sold, and the potential liability imposed by dram shop laws involving injuries, directly and indirectly, related to the sale of liquor and its consumption. You will need to understand and comply with those laws in operating a Manchu Wok Restaurant.

The United States Department of Agriculture and the Food and Drug Administration regulate the manufacture, labeling and distribution of food products.

Many local or state jurisdictions require food service permits for those preparing, handling and serving food to the public. You and your employees may be required to pass a test or other certification process to obtain such a permit. There may also be local ordinances and regulations governing food storage, preparation and serving.

Your Manchu Wok Restaurant is subject to local food and health permits and inspection laws. Health laws are intended in part to reduce food borne illnesses and may cover such issues as: requiring employees to take a test and obtain a license as a food service worker, having accessible sinks, bathrooms for certain size establishments, inspections for cleanliness and compliance, equipment cleaning, storage and packaging, size of facilities, allowed foods, refrigeration, etc.

Federal. Examples of other federal laws affecting many small businesses are wage and hour, occupational health and safety, equal employment opportunity, taxes, hazardous materials communication to employees, hazardous waste and environmental, and the Americans With Disabilities Act.

State. State laws may cover the same topics as federal laws. Examples of other state laws affecting many small businesses include environmental, occupational health and safety, fire, taxes, health, and building and construction laws.

Local. Local laws may cover the same topics as federal and state laws. Examples of other local laws affecting many small businesses include health and sanitation, building codes, fire codes, permits, and waste disposal.

The foregoing are examples of some, but not all of the laws that may be applicable to the franchised business described in the Franchise Disclosure Document. The Franchise Agreement places the responsibility for complying with all applicable laws and regulations upon you, the franchisee. You should research these requirements before you invest.

In addition, your business will be subject to various employment regulations concerning wage rates, mandated employee benefits, employment taxes, tip reporting, worker safety, unemployment compensation, workers' compensation, teenage labor practices, disabled employees and discrimination in employment practices. Environmental regulations will impact your restaurant operations, particularly in the area of sewer discharge. You will be subject to the Americans with Disabilities Act which prohibits practices that discriminate against physically and mentally challenged individuals regarding access to public accommodations and employment opportunities. Due to the permitted sale of alcoholic beverages, you will need to obtain all required liquor licenses and liquor liability (dram shop) insurance in addition to the required insurance policy. You must report and pay various sales, excise, property, income, use and inventory taxes. Additionally, you must obtain various licenses and permits at the federal, state and local level in order to conduct business including, but not limited to, those related to food preparation. The laws in your state or municipality may be more or less stringent, but failure to comply with the above governmental regulations could have a material adverse impact on you and your business. You should examine these laws before purchasing a Manchu Wok franchise.

ITEM 2 BUSINESS EXPERIENCE

References to titles and positions for the persons listed in this Item 2 may be assigned to MTY, MTY Canada, MTY USA, Kahala Brands, or any one or more affiliated companies.

Chief Executive Officer: Eric Lefebvre

Mr. Lefebvre was promoted and became the Chief Executive Officer of MTY effective November 2018. Prior to that, Mr. Lefebvre was the Chief Financial Officer of MTY since June 2012, and was Vice President of Finance of MTY from November 2009 until June 2012.

Chief Financial Officer: Renee St-Onge

Ms. St-Onge was promoted to Chief Financial Officer of MTY effective November 2018. Prior to that, Ms. St-Onge was with MTY as Controller since 2012.

Chief Legal Officer: Jenny Moody

Ms. Moody has been with the Kahala Brands' Legal department since June 2010. In August 2012 she was named Corporate Counsel, in September 2013 International Counsel, in September 2016 Deputy General Counsel, and in November 2019 General Counsel. In August 2023, Ms. Moody assumed her current role.

Vice President of Training and Customer Service: Kerri Kudla

Ms. Kudla joined the Cold Stone Creamery training team in June 2002. She became Director of Operations & Training Development in January 2009 and was promoted to Senior Director of Operations & Training Development in October 2012. In June 2014, Ms. Kudla assumed her current role.

Chief Operating Officer of Quick Service Restaurants: Jason Brading

Mr. Brading joined MTY in 2014 as Brand Vice President for the MR.SUB brand, and saw the steady addition of many MTY brands under his supervision until 2018 when Mr. Brading was promoted to the role of Chief Operating Officer where he leads the Quick Service Division of restaurants for MTY.

Brand Leader (ManchuWOK): Jeff Roop

Mr. Roop joined ManchuWOK as Director of Operations in June 2022 and assumed the role of Senior Director of Operations in March 2023 and later Brand Leader. For 9 years prior to this he was Director of Operations.

Franchise Sales

Senior Vice President of Development: John Wuycheck

Mr. Wuycheck has served as Kahala Brands' Senior Vice President of Development since September 2014.

Vice President of Franchise Development: Jay Goldstein

Mr. Goldstein has held his current role since May of 2009. Prior to this, he served as Senior Director of Operations, then Senior Director of Development for Kahala Franchising and has worked with the Cold Stone Creamery brand since October 2005.

Director of Franchise Development: Doug Merenda

Mr. Merenda joined Kahala Brands' sales team in October 2015 as a Director of Franchise Development and was a franchise broker for Kahala Brands from April 2015 through October 2015.

<u>Vice President of Business Development – USA: Ramin (Ray) Zandi</u>

Mr. Zandi is the Vice President of Business Development – USA, a position he has held since February 2009.

Franchise Development Manager: Traci Zandi

Ms. Zandi has assisted our Development team in awarding franchise units since October 2016. She also served as Office Manager and Executive Assistant for Extreme Pita from April 2011 until October 2016.

Vice President of Franchise Development: Peter Tsafoulias

Mr. Tsafoulias is the Vice President of Franchise Development, a position he has held since March of 2018. Prior to that, he served as the Director of Franchising of IRG from January 2008 until March 2018.

Senior Director of Franchise Sales: Shemar Pucel

Ms. Pucel joined the company in November 2020 as Director of Franchise Sales and was promoted to Senior Director of Franchise Sales in February 2022. From October 2018 to August 2020, Ms. Pucel was a Director of Development for sweetFrog Premium Frozen Yogurt & Samurai Sam's. From October 2015 to October 2018, Ms. Pucel was with SFF, LLC (formerly SweetFrog Enterprises, LLC) where she first served as Manager of Franchise Marketing and Development and was promoted to Director of Franchise Marketing and Development in August 2016.

Manager of Franchise Development: Marilyn Bower

Ms. Bower joined Kahala Brands as Manager of Franchise Development in February 2025. Previously, she held similar roles at Gold Spectrum and SunMed from 2022 through 2024. Prior to that, Ms. Bower was a Customer Service Representative with Farmers Insurance Agency in Bradenton, Florida from 2020 through 2022.

<u>Head of Development – Wetzel's Pretzels: Jon Fischer</u>

Mr. Fischer serves as Head of Development - Wetzel's Pretzels as of March 2023. Previously, Mr. Fischer served as the Wetzel's Pretzels Chief Development Officer from October

2019. In his role, Mr. Fischer is responsible for Wetzel's Pretzels franchise development, including real estate development for franchised outlets. Prior to that, Mr. Fischer held a variety of Vice-President positions, including in real estate and development, at Papa Murphy's International from August 2014 to October 2019.

Sr. Director of Franchise Sales: Adam Lueras

Mr. Lueras became the Director of Franchise Sales for Wetzel's Pretzels effective March 2020 and was promoted to Sr. Director of Franchise Sales in February 2023. Mr. Lueras previously served as a Franchise Development Director with Jackson Hewitt Tax Service in Jersey City, New Jersey from March 2019 through November 2019. Prior to that, Mr. Lueras served as a Franchise Development Director with Wyndham Hotel Group in Parsippany, New Jersey from March 2016 through July 2018. Prior to that, Mr. Lueras served as a National Sales Manager with Avalara Software in Seattle, Washington from February 2014 through March 2016.

<u>Director of Franchise Sales – Non-Traditional: Ross Duggal</u>

Mr. Duggal became the Director of Franchise Sales – Non-Traditional for Wetzel's Pretzels effective July 2022. Mr. Duggal previously served as the Senior Director of Business Development for OLM Foods located in Sioux Falls, South Dakota from June 2020 through July 2022. Prior to that, Mr. Duggal served as the Director of Business Development for Chester's International in Birmingham, Alabama from March 2015 through May 2020.

Sr. Franchise Sales Manager: Diana Krankl

Ms. Krankl has held her current role since March of 2024, and prior to that was the Franchise Sales Manager for Wetzel's Pretzels effective September 2021. Ms. Krankl previously owned and operated D's Superblends, a food truck business in Los Angeles, California from July 2016 through March 2020, where Ms. Krankl's responsibilities included training, customer service, human resources, quality control, event booking, maintaining and strengthening partner relationships, marketing, bookkeeping, inventory management, design development as well as menu and recipe development.

ITEM 3 LITIGATION

LITIGATION INVOLVING FRANCHISOR, PREDECESSORS AND AFFILIATES

<u>Concluded Arbitration and Litigation Involving The Extreme Pita Franchising USA, Inc. predecessor in interest to MTY Franchising USA, Inc.</u>

Purav Enterprises, L.L.C., Balwant Bahia, and Paramjit Samra v. The Extreme Pita Franchising USA, Inc., EP Development, Inc., and Feisal Ramjee; Superior Court of the State of Washington for King County, Case No. 15-2-15120-7.

On June 22, 2015, Purav Enterprises, L.L.C., Balwant Bahia, and Paramjit Samra (collectively "Plaintiffs"), filed a complaint against The Extreme Pita Franchising USA, Inc., EP Development, Inc. and Feisal Ramjee (collectively "Defendants"). Plaintiffs alleged (i) violations under the

Franchise Investment Protection Act in the State of Washington ("FIPA"); (ii) misrepresentation by the Area Developer of the financial performance of the franchise, omissions of mandatory and material information and inherently misleading information that were material factors in the Plaintiff's purchase of the franchise; and (iii) the Area Developer was not a registered broker in the State of Washington. Plaintiffs sought: (i) rescission of the franchise agreement, the corresponding personal guarantee and related agreements; (ii) treble damages under FIPA; and (iii) costs and attorney's fees. The parties entered into a settlement agreement on March 11, 2016, in which Defendants paid Plaintiffs the sum of \$20,000. The matter was dismissed on March 16, 2016.

Concluded Arbitration and Litigation Involving Kahala Franchising, L.L.C.

KOHO, Inc. v. Kahala Franchising, L.L.C.; Superior Court of the State of California for the County of Los Angeles; Case No.: BC572565.

On or about February 17, 2015, Koho, Inc. ("Koho") filed a Complaint against Kahala Franchising, L.L.C. ("Kahala") alleging: (i) breach of contract; (ii) unjust enrichment; and (iii) declaratory relief. Koho sought: (i) no less than \$540,000 in special and general damages; (ii) litigation costs; (iii) prejudgment interest; (iv) reasonable attorney's fees; and (v) declaratory relief. On or about May 5, 2015, Kahala filed a Cross-Complaint against Koho; Heeyong Kyle Chung; and Hannah Kim; alleging: (i) breach of contract; (ii) unjust enrichment; (iii) disgorgement and restitution; (iv) frauddeceit and concealment; (v) negligent misrepresentation; (vi) conversion; (vii) negligence; and (viii) declaratory relief. Kahala sought: (i) breach of contract damages in amount according to proof; (ii) disgorgement and restitution on Unjust Enrichment cause of action; (iii) judicial determination that Kahala is permitted and entitled to set off amounts owed to it by Koho with funds in its possession which would otherwise have been due Koho; (iv) costs; (v) attorneys' fees; (vi) punitive damages; and (vii) prejudgment interest; (viii) any other relief the Court deems just and proper. On or about June 15, 2015, Koho filed a Notice of Hearing on Demurrer and Demurrer to Kahala's Cross-Complaint; Memorandum of Points and Authorities; Declaration of Daniel D. Hoffman and Exhibits in Support Thereof. Koho filed its Notice of Motion and Motion to Strike Certain Portions of Kahala's Cross Complaint; Memorandum of Points and Authorities on June 17, 2015. On October 16, 2015, the Court overruled Koho's Demurrer to Kahala's Cross Complaint in its entirety and denied their motion to strike Kahala's punitive damages claims; the Court sustained the Demurrer as to Hannah Kim. Mediation was held on May 3, 2016, which failed to yield a settlement between the two parties. On May 5, 2016, Kahala dismissed the claims against Koho and Kyle Chung because they satisfied the debts owed to Kahala, and, therefore, the claims were moot. Hannah Kim was subsequently awarded attorney's fees on June 15, 2016, in the amount of \$10,233. A bench trial commenced on June 15, 2016, and ended on June 16, 2016. Upon the conclusion of Koho's case, Kahala presented its case-in-chief and moved for judgment pursuant to Code of Civil Procedure section 631.8. The Court granted Kahala's Judgment as Koho failed to establish the requisite elements of "breach" and "damages" on the three causes of action asserted in the Complaint. On July 18, 2016, the Court awarded Kahala attorneys' fees in the amount of \$205,000. On September 22, 2016, Koho filed a Notice of Filing of Notice of Appeal and requested that Kahala participate in a mediation to resolve the outstanding award to Kahala. Koho failed to post an appeal bond. On February 13, 2017, Kahala commenced its self-help pursuant to Section 8(i) and (j) of the ARA and began withholding 100% of the Area Representative fees to which Koho would have otherwise been entitled. On June 19, 2017, the parties entered into a settlement agreement whereby Kahala repurchased Koho's Area Developer

territory for the sum of \$75,000 and forgave the remaining damages owed in the amount of \$130,000.

<u>Texas Nrgize #1, Inc. v. Kahala Franchising, L.L.C. and Kahala Holdings, L.L.C.</u>; 67th <u>Judicial District Court, Tarrant County, Texas; Civil Action No.: 067-272652-14 subsequently removed to United States District Court for the Northern District of Texas; Case No.: 4:14-cv-544-Y.</u>

On or about June 18, 2014, Texas Nrgize #1, Inc., an Nrgize franchisee ("Plaintiff"), filed a Petition and Request for Disclosure against Kahala Franchise Corp and Kahala Holdings, L.L.C. (collectively "Defendants") alleging (i) violations of the Texas Business Opportunities Act. Tex. Bus. & Comm. Code §§ 51.001 and the Texas Deceptive Trade Practices Consumer Protection Act, Tex. Bus. & Comm. Code §17.46, and (ii) Breach of Contract and Warranties. Plaintiff sought: (i) economic damages in excess of \$200,000, plus treble damages and pre- and post-judgment interest at the maximum rates allowed by law; (ii) attorneys' fees and costs; and (iii) such other relief to which the Plaintiff may be justly entitled. On July 16, 2014, Defendants filed a Notice of Removal to the United States District Court for the Northern District of Texas. On July 24, 2014, the judge executed the Order Granting the Unopposed Motion to Substitute Parties and Changing Case Style. Kahala Franchising, L.L.C. ("Defendant") was substituted in as a defendant instead of Kahala Franchise Corp. On July 28, 2014, Defendant filed a Motion to Transfer Pursuant to 28 U.S.C. §1404(A) and Brief in Support. This motion sought an order to transfer the litigation to the United States District Court for the District of Arizona pursuant to the parties' forum selection clause contained in the franchise agreement. On August 1, 2014, Defendant filed its Answer, Counterclaim and Third Party Claim. The Counterclaim was against Plaintiff and the Third Party Claim was filed against Duane W. Martin, Argentina Saldivar, and Margena Wood ("Third Party Defendants"). Defendant alleged: (i) Breach of Franchise Agreement against Plaintiff, and (ii) Breach of Guaranty against Third Party Defendants. Defendant sought: (i) judgment against Plaintiff and the Third Party Defendants in an amount to be proven at trial; (ii) attorneys' fees; (iii) costs pursuant to A.R.S. §§12-341 and 12-341.01 and the parties' contractual agreements; and (iv) any other relief the Court deemed fit. Plaintiff filed its Response to Defendant's Motion to Transfer Pursuant to 28 U.S. C. §1404 (A) and Brief in Support on August 18, 2014, then filed its Answer to Defendant's Counterclaim on August 22, 2014. On August 28, 2014, Defendant filed its Reply in Support of Motion to Transfer Pursuant to 28 U.S. C. § 1404(A) and Brief in Support. On September 24, 2014, the United States District Court for the Northern District of Texas Court requested that each party submit a supplemental brief to benefit the Court in resolving the Defendant's Motion to Transfer. Mediation was held on September 29, 2014, but the parties failed to come to an agreement. On October 17, 2014, Third Party Defendants filed their Answer and Counterclaim. Third Party Defendants alleged: (i) violations of the Texas Business Opportunities Act, Tex. Bus. & Comm. Code §§ 51.001 and the Texas Deceptive Trade Practices Consumer Protection Act, Tex. Bus. & Comm. Code §17.46, and (ii) breach of contract and warranties. Third Party Defendants sought: (i) economic damages in excess of \$200,000, plus treble damages under the Code and pre- and-post judgment interest at the maximum rates allowed by law; (ii) attorney's fees; (iii) costs; and (iv) any other general or special relief that the Court deemed fit. On October 24, 2014, Defendant filed its Supplemental Briefing In Support of Its Motion to Transfer Pursuant to 28 U.S.C. §1404(A). On November 10, 2014, Defendant filed its Answer to the Third Party Counterclaims. On November 13, 2014, Defendant filed its Notice of Dismissal Without Prejudice as to Third Party Defendant Argentina Saldivar only. On November 14, 2014, Plaintiff filed its Response to Defendant's Supplemental Briefing In Support of Their Motion to Transfer Pursuant to 28 U.S.C. §1404(A). On November 26, 2014, Defendant filed its Reply to Plaintiff's Response to Defendant's Supplemental Briefing in Support of Its Motion to Transfer Pursuant to 28 U.S.C. §1404(A). On February 24, 2015, the Court granted Defendant's Motion to Transfer the case to the United States District Court of the District of Arizona; Phoenix Division;

Case No.: CV15-0337 PHX DGC. In April 2015, Plaintiff moved to compel arbitration which was ultimately denied by the Court. The parties participated in a mediation in August 2015, which was unsuccessful. In December 2015, the parties executed a settlement agreement in which Defendant paid Plaintiff the sum of \$35,000. The parties filed a Stipulation to Dismiss With Prejudice on December 18, 2015.

<u>Concluded Arbitration, and Litigation Involving SFF, L.L.C., successor in interest to SweetFrog Enterprises, L.L.C.</u>

<u>Urquieta Sweet Frog, L.L.C. and Ana Urquieta v. SweetFrog Enterprises, L.L.C. d/b/a SFF, L.L.C.,</u> American Arbitration Association; Case No. 01 14 0001 8086.

On December 23, 2014, Urquieta Sweet Frog, L.L.C. and Ana Urquieta, a former sweetFrog franchisee and its owner (collectively "Plaintiffs"), filed a Demand for Arbitration against SweetFrog Enterprises, L.L.C. ("Defendant"). Plaintiffs alleged: (i) Defendant engaged in fraud; (ii) unfair practices; and (iii) deceptive actions. On February 2, 2015, Defendant timely filed an Answer and Counterclaim and denied all allegations, and further asserted a counterclaim against Plaintiffs for unpaid royalties. This matter was settled in December of 2015. Under the settlement, Defendant agreed to pay Plaintiffs \$300,000 and the parties executed mutual releases.

<u>SFF, L.L.C. v. Carmel Village Yogurt Company L.L.C.; City of Richmond, Virginia Circuit Court; Case No. CL16-3927.</u>

On August 29, 2016, SFF, L.L.C. ("Plaintiff") filed a lawsuit against three sweetFrog franchisee entities, Carmel Village Yogurt Company L.L.C. ("Defendant Carmel YC"), Huntersville Yogurt Company, L.L.C. ("Defendant Huntersville YC"), and Mooresville Yogurt Co, L.L.C. ("Defendant Mooresville YC"), and their main member, Steve Anto ("Defendant Anto"); (all named Defendants collectively referred to as, "Anto Defendants"). Plaintiff alleged: (i) Defendant Carmel YC breached its franchise agreement through its unauthorized closure of its franchised shop; and (ii) as a result of the breach, Plaintiff had the contractual right to terminate Defendant Carmel YC's franchise agreement and cross terminate the franchise agreements of Defendant Huntersville YC and Defendant Mooresville YC. Plaintiff sought: (i) declaratory judgment that the three franchise agreements had terminated; (ii) specific performance of the Anto Defendants' post-termination obligations; (iii) damages for past due fees; (iv) lost future royalties in excess of \$116,000; and (v) recovery of its attorneys' fees. In response to Plaintiff's complaint, Anto Defendants denied Plaintiff's claims and asserted counterclaims against Plaintiff and alleged: (i) Defendant Carmel YC's franchise agreement was unenforceable and, alternatively, that Plaintiff was in breach of Defendant Carmel YC's franchise agreement due to Plaintiff's allowance of another franchisee to open a shop within three miles of Defendant Carmel YC's shop. Anto Defendants sought: (i) a declaratory judgment that they were not in default of their franchise agreements; (ii) damages of not less than \$425,000; and (iii) recovery of their attorneys' fees. Plaintiff denied Anto Defendants' claims and filed a demurrer and pleas in bar seeking to have those claims dismissed. Prior to the court hearing and ruling on Plaintiff's motion, the parties entered into an agreement pursuant to which: (i) the parties acknowledged the valid termination of Defendant Carmel YC's franchise agreement; (ii) Defendant Carmel YC transferred the assets of its business to Plaintiff and Plaintiff paid Defendant Carmel YC \$25,000; (iii) Plaintiff reinstated Defendant Huntersville YC's and Defendant Mooresville YC's terminated franchise agreements; and (iv) Plaintiff granted Defendant Anto the right to develop a new sweetFrog shop at a mutually acceptable location on or before November 8, 2018. The case was dismissed with prejudice on December 15, 2016.

<u>Concluded Arbitration and Litigation Involving Fresh Enterprises, L.L.C. successor in interest to BF Acquisition, L.L.C.</u>

<u>Fresh Enterprises, L.L.C. v. Ledang Investment Group, L.L.C., Vincent Tienn Le, Ho Tien Le and Hue This Dang Superior Court of the State of California, County of Santa Clara, Case No. 1-13-CV-257219.</u>

On July 2, 2013, Fresh Enterprises, L.L.C., as successor-in-interest to Baja Fresh Westlake Village, Inc. ("Plaintiff"), filed a complaint against Ledang Investment Group, L.L.C.; Vincent Tien Le, Ho Tien Le and Hue Thi Dang (collectively "Defendants Ledang" or "Cross Claimants Ledang") for: (i) implied indemnity; (ii) equitable indemnity; (iii) express indemnity; (iv) breach of contract; (v) declaratory relief seeking unspecified damages; (vi) indemnification; (viii) a judgment of unlawful detainer; and (ix) declaration that Defendants Ladang were obligated to reimburse Plaintiff for various expenses. On January 6, 2014, Cross Claimants Ledang filed a Cross-Complaint against Plaintiff, Baja Fresh Westlake Village, L.L.C., Triune Corporation and National Franchise Sales, Inc. (collectively "Counter Defendants") for: (i) breach of contract; (ii) breach of covenant of good faith and fair dealing; (iii) negligent misrepresentation; and (iv) intentional misrepresentation. On February 19, 2014, Counter Defendants filed a Motion to Compel Arbitration, which was granted. The disputes between the parties were then arbitrated before the American Arbitration Association (Case Number 72-20-1400-0126). On February 2, 2015, the Arbitrator issued an award in favor of Cross Claimants Ledang in the amount of \$660,620.84. The parties entered into a Settlement and Release Agreement on July 20, 2015, under which Counter Defendants paid the Cross Claimants Ledang the sum of \$585,000 and the matter was dismissed with prejudice.

Concluded Arbitration and Litigation Involving Famous Dave's of America, Inc.

<u>Desert Ribs, L.L.C., Famous Gracie, L.L.C., Famous Freddie, L.L.C., Famous George, L.L.C. and Famous Charlie, L.L.C. v. Famous Dave's of America, Inc., American Arbitration Association, Minneapolis, Minnesota, Case No. 01 16 0000 8549.</u>

On March 14, 2016, the franchisees for the Famous Dave's® Restaurants in Chandler, Peoria, Mesa and Gilbert, Arizona ("Claimants") filed a Demand for Arbitration against Famous Dave's alleging that Famous Dave's (1) violated the Minnesota Franchise Act ("MFA"), (2) breached the implied covenant of good faith and fair dealing under the Famous Dave's® Franchise Agreements with Claimants (the "Franchise Agreements"), and (3) breached certain express provisions of the Franchise Agreements. Claimants sought damages of \$2,984,098, and a permanent injunction prohibiting Famous Dave's from engaging in discriminatory conduct in violation of the MFA. On July 20, 2016, the arbitrators ruled in partial favor of the pre-hearing motion filed by Famous Dave's by dismissing Claimants' MFA claims against Famous Dave's. Upon the dismissal of the MFA claims, Claimants voluntarily dismissed their remaining claims against Famous Dave's and entered into a confidential settlement agreement and mutual release (the "Settlement Agreement"), dated August 22, 2016, with Famous Dave's. The Settlement Agreement included the following material terms: (i) the territorial rights granted to Claimants in the Franchise Agreements were modified; (ii) the managing member of Claimants (the "Consultant") entered into a consulting agreement with Famous Dave's that provided for the design, development and build-out of a counter-service/line-service prototype barbecue restaurant concept (the "Prototype") and the payment of a consulting fee of \$410,000 to the Consultant in installments over a three-year period; and (iii) Claimants entered into a right of first offer agreement with

Famous Dave's granting to a Claimant the first right to enter into an area development agreement with Famous Dave's for the development of seven Prototype restaurants in a reserved territory in Arizona.

Famous Dave's of America, Inc. v. Allan Gantes; John Gantes; M Mart 1, L.L.C.; Kurt Schneiter; Shoreline FD Investors, L.L.C.; SR El Centro FD, Inc. SR Long Beach FD, Inc.; SR Palmdale FD, Inc.; SR Restaurant Holdings Group, Inc.; SR Simi Valley FD, Inc.; SR Tracey FD, Inc.; Superior Court of the State of California, County of Los Angeles, Central Division, Case No. BC589329.

On July 24, 2015, Famous Dave's of America, Inc. filed a lawsuit against Allan Gantes; John Gantes; M Mart 1, L.L.C.; Kurt Schneiter; Shoreline FD Investors, L.L.C.; SR El Centro FD, Inc.; SR Long Beach FD, Inc.; SR Palmdale FD, Inc.; SR Restaurant Holdings Group, Inc.; SR Simi Valley FD, Inc.; SR Tracey FD, Inc., (collectively referred to herein as "Defendants") (SR El Centro FD, Inc.; SR Long Beach FD, Inc.; SR Palmdale FD, Inc.; SR Restaurant Holdings Group, Inc.; SR Simi Valley FD, Inc.; SR Tracey FD, Inc. will herein be referred to as the "SR Defendants"), the former franchisees for the Famous Dave's® Restaurants in El Centro, Long Beach, Palmdale, Simi Valley, and Tracy, California, and others (collectively referred to herein as "Restaurants") based in part on the continued operation of the Restaurants as Famous Dave's® Restaurants using Famous Dave's Marks and Restaurant System after the termination of their Franchise Agreements by Famous Dave's for failure to cure breaches of the Franchise Agreement, including the failure to pay the Royalty and Marketing Fund Fees due under the Franchise Agreements, within the prescribed cure period after receipt of written notice, in violation of the post-termination obligations of the Franchise Agreements. Famous Dave's alleged Lanham Act violations, including federal trademark infringement, federal trademark dilution, federal unfair competition and false advertising, and federal trade dress dilution; trademark infringement, trademark dilution, unfair competition and false advertising under California law; common law trademark infringement; breach of the Franchise Agreements; breach of the implied covenant of good faith and fair dealing; and intentional interference with contract. Famous Dave's sought injunctive relief to enjoin Defendants from continuing to use the Marks and Restaurant System and enforcing compliance with the post-termination obligations of the Franchise Agreements, and also sought damages in an amount that was to be determined at trial, reasonable attorneys' fees, interest and costs of suit. On March 11, 2016, the SR Defendants filed its First Amended Cross Complaint against Famous Dave's. The SR Defendants alleged that (i) Famous Dave's breached its obligations cited in the Franchise Agreements by failing to provide certain marketing support and access to customer contact data, vendors, internet reporting and support; (ii) Famous Dave's failed to provide operations and preferred practices training to the SR Defendants designated representative (iii) breach of the covenant of good faith and fair dealing; and (iv) breach of fiduciary duty. The SR Defendants sought: (i) compensatory damages in an amount not less than twenty million dollars; (ii) punitive damages; (iii) costs; (iv) attorney's fees. On September 7, 2016, Famous Dave's filed its Answer to the first amended cross complaint vehemently denying the allegations alleged by the SR Defendants, On September 29, 2018, the parties agreed to enter into a confidential settlement agreement and a mutual release of claims (the "El Centro Settlement Agreement"), which contained the following material terms: (i) SR Defendants received \$75,000 towards the payment of their attorney's fees contingent on the de-imaging of the Restaurants and the sale of its Major Assets (as defined in the Franchise Agreement) relating to the El Centro restaurant to Gantes; (ii) Allan Gantes paid to SR Restaurant Holdings Group, Inc. \$7,500 upon the closing of the sale of the Major Assets; (iii) the SR Defendants agreed to de-identify the Restaurants in Long Beach, California and Tracy, California; (iv) notices (sent via email and mailer) were provided to the "Pig Club" customers of the Long Beach, California and Tracy, California Restaurants to opt into notices from Q restaurants; and (v) Famous Dave's consented

to the sale of the Major Assets owned by SR El Centro FD, Inc. to an entity wholly owned by Gantes or an affiliate subject to Famous Dave's review of the purchase agreement which review would not be withheld or unreasonably delayed, and satisfaction of all conditions to the Assignment stated in the Franchise Agreement. Upon completion of the sale the aforementioned \$75,000 would be paid to the SR Defendants; (vi) the SR Defendants no longer could hold themselves out as Famous Dave's franchisees and (vii) all franchise agreements that were the subject of the lawsuit were terminated. The Request for Dismissal with prejudice was entered on November 26, 2018.

Concluded Arbitration and Litigation Involving VI BrandCo, L.L.C.

In re: Restaurants Acquisition I, L.L.C. (Giuliano vs. W. Craig Barber et. al. United States Bankruptcy Court for the District of Delaware on December 2, 2015 (Case No. 15-12406 (KG)).

On December 1, 2017, the Chapter 7 trustee in the Restaurants Acquisition I, L.L.C. ("RAI") bankruptcy proceeding filed suit in the United States Bankruptcy Court against our Chief Executive Officer W. Craig Barber, our Chief Concept Officer – Family Restaurant Division Robert Langford and companies owned jointly by them alleging avoidance, fraudulent transfer, breach of contract and breach of fiduciary duty in connection with their executive roles with RAI and as members of Dynamic Management Company, LLC related to the RAI's ownership and operation of Black-Eyed Pea and Dixie House restaurants. On March 6, 2019, Barber and Langford each settled with the Chapter 7 trustee by each agreeing to pay to the trustee and estate separate payments totaling \$150,000 each over a three-year period.

Concluded Arbitration and Litigation Involving Wetzel's Pretzels, L.L.C.

<u>Pretzelsdallas1, Inc. v. Wetzel's Pretzels, L.L.C.; Los Angeles, California; AAA Case No.01-19-0002-9326.</u>

On or about July 19, 2017, we entered into a franchise agreement with Pretzelsdallas1, Inc. (then known as Triple Scoops, Inc.). On or about September 11, 2019, Pretzelsdallas1, Inc., ("Claimant/Counter Respondent") a franchisee, filed a demand for arbitration against Wetzel's Pretzels, L.L.C .("Respondent/Counter Claimant") with the American Arbitration Association in which it alleged claims for (i) recission and restitution for intentional misrepresentation, (ii) recission and restitution for negligent misrepresentation and (iii) violation of the California Corporations Code §31201. Claimant/Counter Respondent sought \$368,837 in damages. On or about February 26, 2020, Respondent/Counter Claimant filed an answer and counterclaim against Claimant/Counter Respondent. Respondent/Counter Claimant alleged: (i) breach of contract- franchise agreement; (ii) breach of contract - sublease agreement; and (iii) breach of guaranty. Respondent/Counter Claimant sought: (i) judgment on its claims for payments on the franchise agreement in the amount of \$133,600; (ii) judgment on its claims for payments on the sublease agreement in the amount of \$14,8520; judgment in its favor on its claims for payments on the guaranty in the amount of \$148,450; (iv) interest on the sums; (v) attorney's fees; and (vi) any other relief the court deems fit. On September 24, 2021, the parties entered into a settlement agreement, in which a mutual release of all claims was agreed to, and Respondent/Counter Claimant paid Claimant/Counter Respondent the sum of \$125,000. The matter was subsequently dismissed with prejudice.

Concluded Arbitration and Litigation Involving Papa Murphy's International, L.L.C.

DTD Pizza LLC, Brian Watson, Alton Spears, LMP Enterprises LLC, Pizza Enterprises LLC, Alan and Denise Barnett, DOB Enterprises, Inc., Douglas and Lesia Billing, Rob & Bud's Pizza, Robert J. Dickerson Trust UA, Rob Dickerson, 4LM Enterprises, Inc., Jana and Randell Liles, Ben and Kim Mayfield, SEAMS Holdings LLC, Scott and Erica Shelby, Robert Hoersting, PM Savannah LLC, James and Mona King, Hans King, Pizza For 4 Kings Corp, Alamo Quality Pizza I, LLC., Quality Pizza III, LLC., Gerardo Torres, George Knost, Arkel Food Services, LLC., Reece Alexander Overcash, III, Angelo S. Chantilis, Jr., Double AA Partners, LLC., Jeffrey L Comish, John Stalker, and Papa's of Tennessee, LLC. v. Papa Murphy's International LLC, Papa Murphy's Company Stores, Inc., PMI Holdings Inc., Papa Murphy's Intermediate Inc., Papa Murphy's Holdings, Inc., Lee Equity Partners LLC, John D. Barr, Ken Calwell, Thomas H. Lee, Yoo Jin Kim, Benjamin Hochberg, John D. Schafer, Achi Yaffe, Janet Pirus, Victoria Blackwell, Gail Lawson, Dan Harmon, Scott Mullen, Jayson Tipp, Kevin King, Stephen Maeker, Steve Millard, Steve Figiola; Washington Superior Court, Clark County, Case No. 14-2-00904-0.

<u>and</u>

Mitch and Kristen Brink, Brink Holdings Inc., Angela Buchannan, Tim Forester, Z-Axis, Inc., Heather and Gary Nychyk, Bar N Pizza, LLC, John DeMattia, DeMattia LLC, a Texas Limited Liability Company, Harry and Terry Olson, Hot Pizza Inc., Steven Pyatt, Craig Braun, David Mraz, JIM LLC, Philip and Maria Ahn Wilson, Papa South, LLC, Steven and Holly Mead, Thomas Lance, PMG Tampa, LLC, Ilya and Chantal Rubin, Pie in the Sky LLC, Joanna and Glenn Patcha, Alchemy Foods LLC, Ian Hasinoff and Susan Lorimer, Eddrachillis LLC, Cole Kilen, Eye on the Pie LLC, Ann and Harvey Callegan, Just for Fun, LLC, Eugene and Joy Hill, Conn, Edward Turnbull, Turnbull Restaurant Group LP, Turnbull Restaurant Group GP, Conn, LLC, Loralie and Trey Bennett, Pizza Revolution of Fort Walton Beach LLC, Pizza Revolution of Panama City LLC, Pizza Revolution at Tyndall LLC, Steven Terry, Matthew and Cindy Terry, Alice and Douglas Worthington, Thomas Stephenson, Make Dough Enterprises Inc., Jared Richardson, Russell Crader, and Red Rust, LLC, v. Papa Murphy's International LLC, Papa Murphy's Company Stores, Inc., PMI Holdings Inc., Papa Murphy's Intermediate Inc., Murphy's Holdings, Inc., Lee Equity Partners LLC, John D. Barr, Ken Calwell, Thomas H. Lee, , Yoo Jin Kim, Benjamin Hochberg, John D. Schafer, Achi Yaffe, Janet Pirus, Victoria Blackwell, Gail Lawson, Dan Harmon, Scott Mullen, Jayson Tipp, Kevin King, Stephen Maeker, Steve Millard, Steve Figiola; Washington Superior Court, Clark County, Case No. 14-2-01743-3.

These two related actions were commenced in April 2014 and June 2014, respectively, by separate groups of current and former franchisees against us, certain members of our board of managers and executive team, and others in Washington Superior Court (Clark County), alleging misrepresentations involving financial performance representations in ITEM 19 of the franchise disclosure document the franchisees' local marketing obligations, among other things, and brought claims for violation of the Washington Franchise Investment Protection Act ("WFIPA"), fraud, negligent misrepresentation and breach of contract. These two actions were consolidated in September 2014 under Case Number 14-2-00904-0.

Each of the plaintiff groups =entered into settlements with Papa Murphy's in which they dismissed all of their claims against defendants with prejudice and the action was dismissed in June 2020. The settlements are as follows: (1) one plaintiff group dismissed its claims against Papa Murphy's for no consideration; (2) two plaintiff groups agreed to pay amounts ranging from \$5,000 to \$8,000 to Papa Murphy's and remained in the system; (3) Papa Murphy's agreed to pay one plaintiff group's advertising costs for one year, agreed to allow the franchisee to develop an additional

franchise, and agreed to return the franchisee's initial development fee of \$10,000; (4) another plaintiff group agreed to remain in the system in exchange for Papa Murphy's paying 3.8% of the franchisees' sales towards local advertising for a period of two years and extending the franchise agreement's term for an additional ten years; (5) Papa Murphy's settled with fifteen different plaintiff groups and paid amounts ranging from \$10,000 per group to \$4 million per group; (6) Papa Murphy's agreed to purchase one plaintiff group's nine Papa Murphy's stores at an agreed upon value of the stores' assets plus \$500,000; and (7) Papa Murphy's agreed to purchase seven plaintiff groups' Papa Murphy's stores at an agreed upon value of the stores' assets.

Rob & Bud's Pizza, L.L.C. v. Papa Murphy's International, Inc. and Papa Murphy's International, L.L.C.; United States District Court for the Western District of Washington, Case No. 5:15-cv-05090-TLB.

In spring 2015, Papa Murphy's sent a notice of default to plaintiff for alleged defaults under the plaintiff's franchise agreements. In response, on April 17, 2015, the plaintiff brought an action seeking a declaratory judgment and injunction preventing Papa Murphy's from terminating the franchises. The plaintiff subsequently added claims in the case alleging that Papa Murphy's tortiously interfered with the plaintiff's employees and negligence in how Papa Murphy's handled the plaintiff's customer database, and sought compensatory damages, punitive damages and costs in an unspecified amount. The plaintiff was also a plaintiff in the <u>LMP</u> case described above. The case was dismissed with prejudice as part of a settlement with plaintiff in this case and the <u>LMP</u> case under which Papa Murphy's purchased plaintiff's nine Papa Murphy's stores at an agreed upon value of the stores' assets plus \$500,000.

<u>PUBLIC AGENCY ACTIONS AGAINST</u> MTY USA, AFFILIATES AND/OR THEIR PREDECESSORS

<u>Concluded State Administrative Actions Involving SFF, L.L.C., successor in interest to SweetFrog Enterprises, L.L.C.</u>

In the Matter of SweetFrog Enterprises, L.L.C. f.k.a. Imagination Enterprises, Inc., d/b/a Sweet Frog, Administrative Proceeding Before the Securities Commissioner of Maryland, Case No. 2012-0055.

As a result of an inquiry into the franchise related activities of SweetFrog Enterprises, L.L.C., ("SFE") the Maryland Securities Commissioner ("Commissioner") concluded that grounds existed to allege that SFE violated the registration and disclosure provisions of the Maryland Franchise Law in relation to the offer and sale of certain license agreements. SFE acknowledged that those license agreements constituted franchises as defined under the Maryland Franchise Law. SFE represented that it entered into license agreements with eight Maryland licensees during the time it was not registered to offer and sell franchises in Maryland. On August 29, 2012, the Commissioner and SFE agreed to enter into a consent order whereby SFE, without admitting or denying any violations of the law, agreed to: (i) immediately and permanently cease from the offer and sale of franchises in violation of the Maryland Franchise Law; (ii) file and diligently pursue an application for an initial franchise registration in Maryland relating to the license agreements it offered and sold to Maryland licensees; and (iii) to offer to rescind the license agreements of all Maryland licensees to whom it sold unregistered franchises. We are not aware of any licensees that accepted the rescission and have made a good faith effort to obtain that information.

Concluded State Administrative Actions Involving Predecessor Blimpie Associates, Ltd.

In May 1992, Blimpie Associates, Ltd. ("Blimpie") and Joseph Dornbush (formerly the President of Blimpie) (collectively "Respondents") responded to a claim by the New York Department of Law that it had sold franchises during a period of time when Blimpie's prospectus had not been updated by amendment. Without the admission of any wrongdoing, Respondents consented to the entry of an order in which Respondents agreed: (i) to entry of a judgment enjoining them from further violations of the New York Franchise Sales Act; and (ii) to pay the sum of \$18,000 to the State of New York as an additional allowance. Respondents paid the \$18,000 in May 1992 and executed the consent judgment on August 25, 1992.

<u>Concluded State Administrative Actions Involving Maui Wowi Franchising, Inc., predecessor in interest to Kahala Franchising, L.L.C.</u>

In the Matter of Maui Wowi Franchising, Inc., Before the Securities Commissioner of Maryland, Case No. 2005-0651.

On November 11, 2005, Maui Wowi Franchising, Inc., the predecessor franchisor of the Maui Wowi brand ("MWF"), entered into a Consent Order with the Securities Commissioner of Maryland ("Commissioner") resulting from MWF inadvertently entering into four franchise agreements with Maryland residents after its registration in Maryland expired on June 9, 2004 ("Maryland Franchisees"). The Consent Order required MWF to cease and desist from the offer and sale of unregistered franchises in Maryland; to diligently pursue the completion of its then pending application; to register its Offering Circular in Maryland; to develop and implement new franchise law compliance procedures to ensure future compliance with the registration and disclosure provisions of Maryland Franchise Law; and to enroll an officer and a franchise compliance person in a franchise law compliance training program. Upon notification by the Commissioner, MWF sent to the Maryland Franchisees the registered Offering Circular, a copy of the Consent Order, and a letter notifying the Maryland Franchisees that they could rescind their franchise agreements. At this time, MWF is in full compliance with the Consent Order.

In the Matter of Maui Wowi Franchising, Inc., Before the Securities Commissioner of Maryland, Case No. 2007-0194.

On September 12, 2007, "MWF" entered into a Consent Order with the Maryland Commissioner resulting from MWF inadvertently entering into two franchise agreements with two Maryland residents ("Second Maryland Franchisees") without delivering to them the appropriate Offering Circular. MWF was registered in the State of Maryland at the time of the offer and sale with an Offering Circular containing certain specific information required only by Maryland law. At the same time, MWF used a second form of Offering Circular in other states that did not contain all of the information required by Maryland law. Prior to the execution of the franchise agreements with the Second Maryland Franchisees, MWF accidentally delivered to them the Offering Circular that did not contain the Maryland-specific information. We subsequently reported these mistakes to the Commissioner. The Consent Order required MWF to cease and desist from the offer and sale of franchises in Maryland in violation of the Maryland Franchise Law; to diligently pursue the completion of its then pending application to register its Offering Circular in Maryland; to implement additional compliance measures to ensure future compliance with the Maryland Franchise Law; to employ an approved franchise law compliance training program or trainer to monitor MWF's franchise activities in Maryland for two years; and to reimburse the Maryland Attorney General for its investigation and resolution costs in the total amount of \$2,500. Additionally, MWF was required to provide to the Second Maryland Franchisees the registered Offering Circular, a copy of the Consent Order, and a letter notifying the Second Maryland Franchisees that they have a right to rescind their franchise agreements. The Commissioner and MWF subsequently entered into an Amended Consent Order in which MWF elected to withdraw from the State of Maryland instead of employing a compliance monitor, with the agreement to employ a monitor if MWF was to re-register in the State of Maryland. MWF fully complied with the Amended Consent Order, and subsequently employed a compliance monitor and was granted registration in the State of Maryland.

<u>Concluded State Administrative Actions, Arbitration, and Litigation Involving BF Acquisition Holdings, L.L.C. and/or its predecessors</u>

State of Maryland Determination; Case Number 2012-0073.

In February 2012, the State of Maryland alleged that during the period January 1, 2009 to November 26, 2009, Triune, LLC ("Triune"): (i) did not retain signed acknowledgements of receipt reflecting the dates that its Franchise Disclosure Document was delivered to certain Maryland residents and non-residents; (ii) sold franchises to certain Maryland residents and non-residents without providing them with a copy of a 2009 Franchise Disclosure Document; (iii) sold franchises to certain Maryland residents and non-residents without providing them with a copy of a 2009 Franchise Disclosure Document that contained its 2008 financial statements with a going concern note from its auditors resulting from the unfavorable financial condition of its parent company; and (iv) sold franchises to certain Maryland residents and non-residents without including, or abiding with, a deferral condition in their Franchise Agreements that was imposed upon it by the State of Maryland, all as required by the Maryland Franchise Registration and Disclosure Law (the "Maryland Law") and in violation of the Maryland Law. Without admitting or denying the allegations, in September 2012, Triune voluntarily entered into a Consent Order with the Office of the Attorney General of Maryland and agreed to: (i) not violate the Maryland Law in the future: (ii) pay the Office of the Attorney General the sum of \$50,000 as a civil penalty; (iii) retain copies of all acknowledgments of receipt confirming dates that prospective Maryland franchisees received any Maryland Franchise Disclosure Documents; (iv) comply with the disclosure and antifraud provisions of the Maryland Franchise Law and the record keeping and escrow requirements of the Code of Maryland Regulations; and (v) send a copy of the Consent Order to certain Maryland franchisees.

State of Virginia Determination; Case Number SEC-2012-00027.

In February 2012, the Division of Securities and Retail Franchising of the State Corporation Commission (the "Commission") alleged that during 2009 Triune, LLC ("Triune"): (i) offered or sold franchises in Virginia in 2009 that were not registered under the Virginia Retail Franchising Act (the "Virginia Act"); (ii) offered or sold franchises in Virginia without disclosing that it was not registered to do so; (iii) failed to provide material information regarding the parent company's unfavorable financial condition and the potential impact that it could have on Triune as stated in a going concern note in its 2008 financial statements from its auditors; and (iv) failed to provide a prospective franchisee with a copy of its Franchise Disclosure Document as required by rule or order of the Commission at least 14 calendar days before the prospective franchisee signed a binding agreement or made any payment to it in connection with the sale or offer to sell a franchise in Virginia. Without admitting or denying the allegations, on November 26, 2012, Triune voluntarily entered into a Settlement Order with the Commission and agreed: (i) to not violate the Virginia

Act in the future; (ii) to pay Virginia the sum of \$25,000 as a penalty and the sum of \$5,000 to defray the Commission's costs of investigation; (iii) to offer certain Virginia franchisees a refund of their initial franchise fees; and (iv) to send a copy of the Settlement Order to certain Virginia franchisees.

Lawsuits Filed by Franchisor Kahala Franchising, L.L.C. Against Franchisees During Fiscal Year December 1, 2023 through November 30, 2024

Suit for Breach of Contract

Kahala Franchising, L.L.C. v. All About Food, Inc. and Chu Yup Lee a/k/a Michale Lee; In the Circuit Court of the Nineteenth Judicial Circuit Lake County, Illinois; Case No.: 2024LA00000001.

Suit for Forcible Entry and Detainer

Cold Stone Creamery Leasing Company, Inc. v. JRF, Inc.; Iowa District Court for Dallas County; Case No.: SCSC050015.

Other than these actions, no litigation is required to be disclosed in this Item.

ITEM 4 BANKRUPTCY

No bankruptcy information is required to be disclosed in this Item.

ITEM 5 INITIAL FEES

All fees and payments described in this ITEM 5 are not refundable unless otherwise noted.

You must pay us a \$30,000 initial franchise fee ("Initial Franchise Fee") when you sign your Franchise Agreement. The Initial Franchise Fee is not refundable except, as described below in this ITEM 5, if we terminate the Franchise Agreement because you do not successfully complete our Initial Training (as defined in ITEM 11 D). We expect that these fees will be uniformly applied in the coming year.

The Initial Franchise Fee is not refundable unless we terminate your Franchise Agreement because you or your Manager fail two attempts to successfully complete Initial Training, you comply with the refund conditions in the Franchise Agreement, and the parties have signed a termination agreement that will include mutual releases. Under these circumstances, we will refund your Initial Franchise Fee, less \$10,000 plus the out-of-pocket expenses (estimated to be between \$2,000-\$4,000) that we incurred in performing our obligations under the Franchise Agreement (including costs associated with training and the leasing and development of your Manchu WOK® Restaurant).

When you sign your Franchise Agreement, you must also pay us a \$10,000 Grand Opening Marketing and Promotion Fee as defined in your Franchise Agreement. This money will be used to conduct a marketing and promotional campaign for your grand opening. The payment is not refundable.

When you sign the Franchise Agreement with us for the premises at which you will operate your Manchu WOK® Restaurant ("**Premises**"), you must pay us a \$8,000 lease administration fee ("**Lease Administration Fee**") if we get involved in the premises lease negotiation (whether you lease the premises directly or from us). This Lease Administration Fee is not refundable for any reason.

If the landlord requires Manchu WOK or you to enter into a lease that has an initial term of greater than 10 years, then, at our option, we may require you to extend the original term of your Franchise Agreement accordingly and, in such case, your Initial Franchise Fee will be increased by an amount equal to \$3,000 for each year (or part year) in excess of 10 years.

We will supervise project management services related to the construction and improvement of the Premises for your Manchu WOK Restaurant. You will pay us for all expenses associated with this task and we, in turn, will pay the contractors and suppliers that perform the work. We estimate that the cost to construct the leasehold improvements, including the purchase and installation of equipment, fixtures and furnishings, will range from \$303,000 to \$454,000. This estimate assumes that you build your Manchu WOK Restaurant according to our standard design guidelines under normal market conditions. You also must pay us a development and construction fee ("**Development Construction Fee**") equal to the greater of \$2,500 or 3% of the total cost of the development of the Premises for Manchu WOK store construction projects in North America (up to a maximum of \$15,000); this fee is payable when the construction is completed.

You will pay us the Architect Fee ranging from \$18,000 to \$20,000 based on our past two years' experience. This includes engineering drawings that may be purchased from us, you landlord, or an architect. These fees are non-refundable.

If you would like to hire your own architect, a written approval must be obtained from us and all drawings must be prepared in conformance with our specifications and submitted to us for review and approval. You will pay us a \$2,000 Plan Review Fee for the reviewing process.

We reserve the right to require you to make a \$30,000 lump sum deposit for your construction of Manchu WOK® Restaurant ("Construction Deposit"). If required, the Construction Deposit will be paid to us by you upon signing the Franchise Agreement and will be deductible on the final payment you make to your contractor.

Additional Fees if You Sublease the Premises from Us

We have the option to require that you either: (1) lease the space for your Premises directly from the landlord and sign a Lease Addendum (Note 1) with us and the landlord; or (2) sublet the space from us after we have obtained the prime lease.

If you sublease the Premises from us, the below additional fees will be applied when you sign the Franchise Agreement and sublease agreement:

Fee	Amount
Security Deposit (three months) (Note 2)	\$15,000 to \$75,000
Sublease Fee (Note 3)	\$500 to \$2,500

Note 1 – Our Lease Addendum is attached as **Schedule D** to your Franchise Agreement.

Note 2 - You will also have to pay us a security deposit equal to three month's rent under the sublease ("Security Deposit") when you sign the Franchise Agreement (unless our prime lease requires a higher security deposit, in which case you will have pay the higher amount and your costs will be higher).

Note 3 - You must also pay us or landlord a one-time fee ("**Sublease Fee**") ranging from \$500 to \$2,500 when you sign the sublease agreement.

ITEM 6 OTHER FEES

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS
Royalty	Traditional 7% of weekly gross sales Non-Traditional: 6% of weekly gross sales	Payable on Thursday each week based upon the seven day period ending on the immediately preceding Sunday night	Gross sales includes all the revenues from your Restaurant but not any sales tax you pay.
Marketing Fee	1% of weekly gross sales	Payable at the same time as the Royalty	You must contribute this amount to our advertising fund.
Regional Cooperative Marketing Contribution	Your proportionate share of regional and local advertising and public relations programs will not exceed an amount that, when added to the amount you pay for your Marketing Fee, is greater than 5% of your gross sales over any 12 month period.	Monthly as required by your regional cooperative.	Payable only if a cooperative is formed in your area.
Successor Franchise Fee	\$15,000	Upon signing the Successor Franchise Agreement (if offered)	This fee is only payable if you are offered the opportunity to sign a Successor Franchise Agreement. The Successor Franchise Term is for 5 years.
Transfer Fee	\$10,000	When you apply for our consent to the transfer of your franchise	One of the conditions to transferring your franchise is paying us this transfer fee
Security Deposit on Lease or Sublease (if	Equal to three month's gross rent on the	When you sign the lease or sublease	We estimate that the Security Deposit will

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS
applicable) (as also noted in Item 5, above)	Premises unless the primary lease requires a higher amount.		range from \$15,000 to \$75,000.
Lease Administration Fee (if applicable) (as also noted in Item 5 above)	\$8,000	When you enter into, modify or renew your Lease or Sublease	
Rent	\$5,000 to \$25,000 per month	Per the lease or sublease	
Sublease Fee (if applicable) (as noted in Item 5 above)	\$500 to \$2,500	At the time you sign a sublease agreement with us.	If you sublease the Premises from us.
Special Assistance Fee	\$500 a day per person providing assistance plus all associated costs and expenses.	When invoiced	If you require special assistance and we think it appropriate to render it and we can accommodate your request, we will provide you with the assistance.
Transfer Escrow Fee	As charged by third- party escrow agent.	As agreed with escrow agent.	Payable at closing when you transfer your franchisee.
Development Construction Fees	The greater of: \$2,500 or 3% of the total development cost for Manchu WOK store construction projects in North America, up to a maximum of \$15,000.	When Invoiced (when the construction is completed).	Every 5 years, you must renovate, alter, and/or upgrade your premises, fixtures, and equipment as may be required by the landlord or as we may require, so your Restaurant reflects the then-current image of the Manchu WOK system. This may entail the complete remodeling and upgrading of your Restaurant. For any renovation, alteration or upgrading, you must pay us a Renovation Development and Construction Fee.

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS
Training for Replacement Manager	\$3,000 plus attendees' travel, lodging and food expenses, as well as all other expenses.	Before training	All attendees must satisfactorily complete Initial Training.
Relocation Charge	\$500	At signing of relocation amendment to Franchise Agreement	Payable if we approve the relocation of your store.
Administration Fee for Franchisee Programs	Up to 15% of gross costs	Upon demand	You must participate in any programs we establish to benefit the franchisees such as, for example, lease and rental payment administration, insurance plan(s), realty tax payment administration, etc. You must pay us a reasonable administration fee not to exceed 15% of the gross amount of the payments we receive for these programs.
Supplemental or Refresher Training \$3,000 plus attendees' travel, lodging and food expenses, as well as all other expenses.		Before training	We may provide additional training, including assistance with any new system developments. If we make Supplemental or Refresher Training mandatory, you must attend.
Interest	Charged on all outstanding amounts from due date at a rate of the lesser of: (i) bank prime plus 5% per year; or (ii) the highest rate permitted by law	When default of any payment due to us or any affiliate. Payable upon demand.	You pay interest calculated per this section on any and all monies due to us from time to time and which are not paid when due.
Supplier Inspection	Varies	On demand	If you propose a new supplier, and we inspect the supplier or test the supplier's

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS
			products, we may charge you or the supplier an amount up to the reasonable cost of inspection and the actual cost of the test.
Digital Menu Software Monthly Service Fee	\$49 to \$60	Monthly	Payable to the system provider at the beginning of each month
POS System Software Subscription	\$50 to \$70 / POS	Monthly	Payable to the system provider at the beginning of each month
Optional for POS System Maintenance Warranty	Terminal \$250 to \$300 for 3 years; Printer \$110 to \$150 for 3 years	When incurred	Amount incurred will depend on the number of terminals and printers in your POS System
Audit	Amounts due by reason of revenue understatement, plus interest at the rate of the lesser of 18% a year (or the highest rate permitted by law) computed from the date payment was due to the date payment is received), as well as the full cost of the inspection or audit, including the charges of any independent accountant or other professionals we retained for the audit or inspection and the travel expenses, room, board and compensation of our employees and those of any independent	Within 10 days after receipt of the inspection or audit report.	We can inspect and audit your books and records. If we discover an understatement of gross sales for any period, you must pay us the Royalty and other sums due on account of the understatement. Further, if the inspection or audit is made necessary because you failed to furnish reports, financial statements, tax returns or schedules or any other documentation required, or if an understatement of gross sales for any period is determined to be greater than 2%, you must pay us these amounts.

TYPE OF FEE	AMOUNT	DUE DATE	REMARKS
Indemnification	An amount equal to our damages, losses or expenses	On demand	You must indemnify us if we incur any losses, damages, or expenses as a result of your ownership or operation of your franchised restaurant, any violation of your Franchise Agreement, or any of your acts or omission.
Dishonored Check Fee	\$25 for each check or draft returned for non- sufficient funds	On demand	This fee is charged for each dishonored payment and/or dishonored electronic funds transfer.
Damages for Breach of Non-Compete Obligations under the Franchise Agreement	Will vary under the circumstances.	As incurred	Payable to Franchisor
Non-participation Fee	\$100 per day if you fail or refuse to participate in any required national, local, regional, seasonal, promotional or other program, initiative and campaign or in any new or modified product or service test or offering.	Upon failing or refusing to participate	Payable to us.
Document Administration Fee	\$500	As incurred	Applicable if we must prepare an amendment to your franchise documents.
Early Termination Fee	The average monthly Royalty and Advertising Fees paid for any consecutive 12 month period within the preceding 48 month period multiplied by the number of months remaining in the term of the Franchise Agreement, and the product is divided by 2.	30 days prior to the early closing of the restaurant	You must provide us with 90 days prior written notice of the termination of your Franchise Agreement.

(Please review the above table in conjunction with all of the notes that follow.)

Notes:

The fees listed in this ITEM 6 are non-refundable, uniformly imposed by us, and are payable to us unless otherwise stated. None of the above fees are refundable except that the Security Deposit is refundable to the extent that it is not needed to pay any arrearages in lease or sublease payments.

ITEM 7 ESTIMATED INITIAL INVESTMENT

YOUR ESTIMATED INITIAL INVESTMENT

Type of expenditure	Amount (low)	Amount (high)	Method of payment	When due	To Whom Payment is to be made
Initial Franchise Fee	\$30,000	\$30,000	Lump sum	When you sign your Franchise Agreement	Us
Leasehold Improvements and Design Costs incl. signage (2)	\$220,000	\$340,000	Lump sum	Commonly, 90% before completion	Us
Equipment, Fixtures, and Furnishings ⁽²⁾	\$89,000	\$126,000	Lump sum	Commonly, 90% before completion of construction and 10% upon completion	Us
POS System & Installation (3)	\$4,000	\$10,000	Lump sum	Upon Ordering	Vendor
Construction Deposit (4)	\$30,000	\$30,000	Lump sum	When you sign your Franchise Agreement	Us
Development Construction Fee (5)	\$2,500	\$15,000	Lump sum	Commonly, 100% upon completion	Us
Lease Administration Fee	\$8,000	\$8,000	Lump sum	When you sign your lease or sublease	Us
Security Deposit ⁽⁶⁾	\$15,000	\$75,000	Lump sum	When you sign your lease or sublease	Us

Type of expenditure	Amount (low)	Amount (high)	Method of payment	When due	To Whom Payment is to be made
Sublease Fee	\$500	\$2,500	Lump sum	If you sign your sublease, at the time when you sign	Us or landlord
Rent (7)	\$15,000	\$75,000	Lump sum	Monthly	Us or landlord
Nonfood, Inventory and Supplies	\$15,000	\$20,000	Lump sum	Commonly, 50% before construction and 50% upon completion	Contractors and vendors
Initial Food, Inventory and Other Costs (8)	\$8,000	\$10,000	Lump sum	As incurred	Suppliers
Grand Opening, Marketing and Promotion Fee	\$10,000	\$10,000	Lump sum	When you sign your lease or sublease	Us
Initial Training Expenses (10)	\$5,900	\$7,500	Lump sum	As incurred	Vendors, hotels, and restaurants
Architect Fees	\$18,000	\$25,000	Lump sum	When you sign your Franchise Agreement	Us
Business License, and Permits (11)	\$150	\$500	Lump sum	Upon issuance	Local government
Additional Funds, (incl. insurance and Initial phase of 3 months) (12)	\$15,000	\$30,000	As incurred	As incurred	Us, landlord, vendors, contractors, and suppliers
Digital Menu System ⁽¹³⁾	\$4,000	\$6,000	Lump Sum	Upon Ordering	Vendor
Total (14)	\$490,050	\$820,500			

Notes:

Please note that we do not offer direct or indirect financing to you for any items. The availability and terms of financing from other sources will likely depend on factors such as the

availability of financing generally, your creditworthiness, and the policies of lending institutions. These fees are not refundable (except as discussed in Note 1 below).

- (1) <u>INITIAL FRANCHISE FEE</u>. The Initial Franchise Fee is not refundable unless we terminate your Franchise Agreement because you or your Manager fail two attempts to successfully complete Initial Training and the parties have signed a termination agreement that will include mutual releases. Under these circumstances, we will refund your Initial Franchise Fee, less \$10,000 plus the out-of-pocket expenses (estimated to be between \$2,000-\$4,000) that we incurred in performing our obligations under the Franchise Agreement (including costs associated with training and the leasing and development of your Manchu WOK® Restaurant).
- LEASEHOLD IMPROVEMENTS AND EQUIPMENT. The estimate is for the leasehold (2) improvements and equipment you will need to operate the Store, such as proper wiring and plumbing, floor covering, wall covering, partitions, ventilation, heat, lighting, storefront counter, painting, refrigerators, wok ranges, steam tables, fryers, sinks, walk-in freezer, steamers, signage, menuboards, smallwares and miscellaneous other items. You will need to obtain the exact equipment we specify and we may specify specific suppliers from which you may be required to purchase the equipment. The amount spent for equipment, fixtures, trade fixtures, and other fixed assets will vary for each Store depending upon the Store's size, style, and the volume of products to be offered in the Store. The amounts noted on the chart are for stores that range in size from 535 to 1,127 square feet. These costs also assume that the landlord provides you with a standard "Vanilla Shell" prior to the date you begin your leasehold improvements. The costs to complete your leasehold improvements will vary widely and may be significantly higher than projected in this table, depending on such factors as the city or town in which you propose to operate, property location, population density, economic climate, prevailing interest rates and other financing costs, the conditions of the property, current building code requirements, current permit requirements, implementation and enforcement of the Americans With Disabilities Act in particular jurisdictions, and the extent of alterations required for the property. You should investigate all of these costs in the area where you wish to establish your Manchu Wok Restaurant and the costs associated with modifying the specific property in which you choose to locate your Manchu Wok Restaurant. In some limited cases, these costs have added more than \$300,000 to the cost of opening a Manchu Wok Restaurant. Your costs will also be higher if your landlord does not provide you with a Vanilla Shell finish or if you select a larger location. We will procure the equipment, fixtures, trade fixtures, and other fixed assets used in the Store, and you will pay us the cost of equipment.
- (3) <u>POS</u>. We require our franchisees to purchase a –Vivonet Point of Sale (POS) system consisting of a computer hardware and software platform (including peripheral hardware devices such as touch screen terminals, printers, card readers, cash drawers, router, battery back-up, etc.). The cost for the POS system and the requisite high speed Internet service will vary based on the size of your Store and the number of POS stations required for the Store.
- (4) <u>CONSTRUCTION DEPOSIT</u>. The Construction Deposit will be paid by you to us for any new Manchu WOK construction projects. We reserve the right to waive or reduce the Construction Deposit for any upgrade refurbishment or renovation at existing Manchu WOK Restaurants.

DEVELOPMENT CONSTRUCTION FEE. You will need to construct improvements, or (5) "build out," the premises at which you will operate the Store. Generally, you will take the premises in "vanilla box" condition (e.g., primed drywall ready to be painted, but without improvements). We will supervise the construction/development/improvement of your Manchu WOK Restaurant. You will pay us all of the costs associated with this task and we, in turn, will pay the contractors, architects, designers and other vendors. As we disclose in Item 8, once you sign the lease for your Manchu Wok restaurant, you must retain certain professionals with whom you will enter into an engagement agreement directly, including: a) our designated architectural firm, or another approved architect, at our discretion, to prepare your construction documents; b) our designated kitchen design and installation specialist, or a kitchen design and installation specialist approved by us, c) our designated kitchen equipment supplier and installer, or a kitchen equipment supplier and installer approved by us, and d) our designated millwork company or a millwork company approved by us to provide a design services proposal as well as a proposal for millwork and interior finish elements fabrication and installation.

The low estimate assumes that you have the requisite expertise to handle bidding and construction administration on your own. The high estimate assumes that you retain the services of a third party to handle bidding and construction administration. We can recommend service providers who can provide these services, but we do not require you to retain a specific firm. You must obtain our prior approval before hiring a construction manager who is not a firm that we currently recommend.

You will also pay a Development Construction Fee equal to the greater of \$2,500 or 3% of the total development costs for Manchu WOK store construction projects in North America, up to a maximum of \$15,000. Costs are likely to vary, and may be much higher if you wish to establish your Store in an area special requirements apply (for example, architectural requirements). The table reflects a range for this fee based on our estimate of the fees that you are likely to incur, given the estimated costs for leasehold improvements and equipment that are provided in Note 2 above. The development construction fee will not exceed \$15,000.

- (6) <u>SECURITY DEPOSIT AND SUBLEASE FEE</u>. If you lease the Premises from us: (a) a Security Deposit equal to three months' Gross Rent will be paid to us by you; and (b) a Sublease Fee ranging from \$500 to \$2,500.
- (7) RENT. The rent estimate is for three months, as explained below. If you do not own a location for your Store, you must purchase or lease a space. Store locations and sizes vary by shopping centers, office towers, airports, casinos, etc. Manchu WOK Traditional stores vary in size from 535-825 square feet, with approx. 675 square feet as the average (excluding some atypical locations). The figures in the estimate are calculated on the following assumptions: (a) you will have to pay three months' rent; (b) for space in the range of 535-825 sq. ft.; (c) at \$7.75 to \$21.40 per square foot per year. This estimate also assumes that the landlord will abate pre-opening rent (if that is not the case, then your costs will be higher). If the site you choose is larger or has a higher rental cost, then your costs will be higher than those in the chart. Rent varies considerably from market to market, and from location to location within each market. Rents may vary beyond the range that we have provided, based on factors such as market conditions in the relevant area, the type and nature of improvements needed to the premises, the size of the Store, the terms of the lease, and the desirability of the location.

- (8) <u>INVENTORY</u>. Items of inventory which you are required to obtain from us or from our distributors are paid for at standard prices and terms. All items of inventory which you obtain from sources of your own choosing are paid for directly to the supplier of those inventory items at prices agreed upon by you and the supplier. Start-up inventory of products and supplies will vary based on expected volume of business and size of storage areas in the building. This estimate is for the initial inventory only.
- (9) <u>GRAND OPENING MARKETING AND PROMOTION FEE</u>. With this money, we will prepare, coordinate and conduct a Grand Opening Marketing and Promotion/Advertising Program for your Manchu WOK Restaurant.
- (10) <u>INITIAL TRAINING EXPENSES</u>. The amount represents the estimated reasonable travel, lodging, meals, and auto for 24 days of training for two individuals. The cost you incur will vary depending upon factors such as the distance traveled, mode of transportation, per diem expenses actually incurred, and the number of persons who will attend training. If you send more than two persons to attend training, we estimate that the additional cost, on a per person basis, will range from \$2,500 to \$3,000.
- (11) <u>BUSINESS LICENSES AND PERMITS</u>. Local, municipal, county, and state regulations vary on what licenses and permits are required by you to operate. These fees are paid to governmental authorities before starting business.
- (12) ADDITIONAL FUNDS. This estimates your initial start-up expenses such as payroll cost, utilities, high speed Internet access fees, digital menu-board maintenance fees, POS polling fees, etc. to the extent that these costs are not covered by sales revenue. This does not include a draw or salary for any owner including you. New businesses often generate a negative cash flow. These numbers are estimates only and are not guaranteed in any manner by us and there is no assurance that additional working capital will not be necessary during this start-up phase or after. Your costs may vary depending on factors including: to what degree you follow Manchu WOK's suggested methods and procedures, your management skill, experience and business acumen, local economic conditions, local market for products and services, competition and your financing costs (if any) for your Manchu WOK Restaurant. The initial start-up phase is generally three months but can be a longer period. We relied on our and our franchisees' experience in opening franchisee-owned Manchu WOK Restaurants when making this estimate.

This estimate also includes legal, accounting and miscellaneous other professional fees that you may incur before you open for business, including (among other things) to assist you in reviewing the Franchise Agreement. Your actual costs may vary, for example, depending on the degree to which you rely upon your advisors, type of financing, and the permitting process in your city.

Your credit history could impact the amount (and cost) of funds needed during the startup phase. If you have no credit history or a weak credit history suppliers may give you less favorable lending and payment terms, which might increase the amount of funds you will need during this period.

You will need to have staff on-hand before opening to prepare the Store for opening, training, orientation, and related purposes. We estimate that you will need approximately 250 hours of staff time over three days, at \$10 per hour, to get ready for your opening. If

prevailing wage rates in your area change (for example, if the local, state, or federal minimum wage is raised), your costs may be higher.

- (13) MENU SYSTEM. We require our franchisees to use Digital Menu System provided by our approved vendor- UNOApp in Canada. The system consisting of digital screens, media player, mounting brackets, HDMI cables, etc. The cost for the POS system and the requisite high-speed Internet service will vary based on the size of your Store and the number of digital screens required for the Store.
- (14) <u>TOTAL.</u> All amounts exclude federal, state and local taxes. We relied on our own experience and information provided to us by our franchisees when preparing these figures. You should review these estimates on your own, preferably with a business advisor of your own choosing.

The figures in the chart and the explanatory notes are only estimates. Your actual costs may vary considerably, depending, for example, on factors such as: local economic conditions; the local market for the store; timing of your Store opening; the prevailing wage rate; competition; the sales level achieved during the initial period of operation; and your management and training experience, skill, and business acumen. You should review these figures carefully with a business advisor, accountant or attorney before making any decision to purchase a franchise.

You should take into account the cash outlays and probable losses that you may incur while you are trying to get established. Extensive start-up costs may be involved, depending upon your circumstances.

ITEM 8 RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

To ensure that high and uniform standards of quality and service are maintained, you must operate your Manchu WOK® Restaurant in strict conformity with our methods, standards and specifications we may periodically designate in the Confidential Manual (as defined in Item 11 below) or otherwise in writing. We do not provide material benefits to franchisees based on their purchase of particular products or services. We strictly regulate the sale of edible and non-edible products which may be sold in your Manchu WOK® Restaurant. You may not make or offer or sell any products (whether edible or non-edible) or services that we have not approved in writing.

You must buy all Approved Products used or offered for sale at your Manchu WOK Restaurant only from suppliers that we have approved in writing. "Approved Products" means all food and beverage ingredients and products, packaging and serving materials, paper goods, utensils, supplies, and other items of inventory, uniforms, marketing or promotional materials, Fixtures, Equipment, as well as all other items required for or used in any manner in the operation of your Manchu WOK Restaurant, both in inventory and as finished goods for sale.

Additionally, you must buy all Proprietary Products as well as Approved Products and packaging bearing our trademarks only from us, our affiliates, or our approved suppliers. Our Proprietary Products include certain menu items that are produced according to the secret recipes, standards, and specifications owned by us or by our affiliates. We (or our affiliates) will derive revenue from your purchases of the Approved Products and Proprietary Products. Although we, nor any of our affiliates are not currently an approved supplier of any Approved Products or Proprietary Products as of the date of this Franchise Disclosure Document, we reserve the right to do so in the future.

For some products we may instead of approving or designating suppliers, provide specifications and approve specific items, rather than the supplier of those items. In this case you may buy the approved item from any source you choose.

You must maintain in sufficient supply, and may offer for sale only the items, products, and services that we have approved in writing for sale at your Manchu WOK® Restaurant. You must offer all types of items and products that we designate, and you may not change the items at your Manchu WOK® Restaurant unless we have given our prior written approval. You must use only the ingredients, preparation methods, and techniques that we specify. You may not deviate from our standards (for example, standards relating to approved products, food preparation, ingredients and recipes) without our prior written consent. If we disapprove of any items, products, or services, in our discretion, you must stop selling and offering those items and services for sale, and remove them from your store.

You must buy only approved food items, ingredients, supplies, packaging, materials, and other (edible and non-edible) products used or offered for sale at your Manchu WOK® Restaurant. In some cases these products may be purchased only from suppliers (including manufacturers, distributors, and other sources) who we have approved in writing as meeting our then-current standards. Our approval of suppliers is conditioned on factors such as their adequate quality controls and capacity to supply your needs promptly and reliably, and whose approval will enable the Manchu WOK® Restaurant system, in our sole opinion, to take advantage of marketplace efficiencies. We will have the right to revoke our approval at any time. We also have the right to appoint only one supplier for any particular item (which may be us or one of our affiliates).

In some cases where specific approved products are available from more than one source, we will not designate an approved supplier of that approved item. We will instead designate the approved item and you must buy only the approved item, with no substitutions, but you may buy that approved item from any supplier of your choosing.

We estimate that your purchases or leases from designated or approved suppliers will represent 80% of the total cost to you of opening your Manchu WOK® Restaurant and 80% of the total purchases and leases for the continuing operation of your Manchu WOK® Restaurant. We estimate that your total purchases/leases in accordance with our specifications where we have not designated or approved a supplier will represent 20% of your total costs of establishing your Manchu WOK® Restaurant and 20% of the total cost of the continuing operations of your Manchu WOK® Restaurant. All designated or approved suppliers are subject to change at any time.

If you want to buy products (other than Proprietary Products) from an unapproved supplier, you must submit to us a written request asking for our approval of the proposed supplier. You may not buy from any supplier unless we have given our prior written approval of that supplier. Before we will approve a supplier, we will have the right to inspect the supplier's facilities, and ask that samples from the supplier be delivered to us or to an independent laboratory that we designated for testing. Either you or the supplier may have to pay a charge not to exceed the reasonable cost of the inspection and the actual cost of the test. We may also require that the supplier comply with other requirements that we may deem appropriate, such as payment of licensing fees, as well as reasonable continuing inspection fees and administrative costs. We reserve the right, at our option, to periodically re-inspect the facilities and products of any approved supplier and to revoke our approval if the supplier fails to continue to meet any of our criteria. Nothing in the Franchise Agreement may be construed to require us to approve any particular supplier, nor to require us to make available to prospective suppliers, standards and specifications for formulas that we, in our sole discretion, deem confidential. We formulate criteria

in order to maintain the highest level of quality and may revise our criteria periodically. Depending upon the type of product for which approval is sought, or for which a new approved supplier is proposed, we anticipate providing our response to the request within one to three months after receipt of the request and the accompanying information.

Our criteria for supplier approval are set out in our internal specifications and are not communicated to our franchisees.

If you want to have an unapproved edible product, unapproved non-edible product, or unapproved packaging, approved for sale in your Manchu WOK® Restaurant, you must submit to us a written request asking for our approval using our defined processes. You may not buy or make any additional products unless we have given you our prior written approval to make or sell that product. Our process and criteria for product approval is set out in the Confidential Manual.

You must allow us to enter your Manchu WOK® Restaurant to conduct inspections. We may conduct these inspections at any time. On notice from us you must immediately take the actions needed to correct any deficiencies found during the inspection.

You must purchase or lease the goods, services, fixtures and equipment, and inventory needed to operate your Manchu WOK® Restaurant in accordance with specifications we set and/or from the suppliers as we may designate. These specifications may include the quality and appearance of the product. Except as described below with respect to the build-out of your Premises, neither we nor any of our affiliates is presently an approved or exclusive supplier. We may, however, designate ourselves or any affiliates as approved or exclusive suppliers of any required goods or services at any time in the future.

You must purchase all the furniture, equipment and fixtures and your opening inventory for your Manchu WOK® Restaurant from suppliers we designate. If we or our designated suppliers cannot furnish any equipment or fixtures, then you may purchase the item from another supplier approved by us in writing. In such event, our approval will not be unreasonably withheld.

In addition, we also will supervise the construction for your Manchu WOK Restaurant, and we reserve the right to approve of or designate the contractors and suppliers who will perform the construction work. You will pay us for all expenses associated with this task and we, in turn, will pay the contractors and suppliers that perform the work. You also must pay us a Development Construction Fee equal to the greater of \$2,500 or 3% of the total cost of construction for construction projects for "Manchu WOK" store construction projects in North America, up to a maximum of \$15,000.

You must, at your own cost and expense, use our designated and approved third party design architect, as detailed in the Confidential Manual, to prepare the initial design drawings for your Franchised Business. Except for the design architect designated and approved by us, no other architect may be used by you for the design of your restaurant. You must also, at your own cost and expense, retain a licensed architect of record to prepare the permitted construction set of drawings.

Specifications for food and paper items are contained in the Manchu WOK® "Recipe Manual". Certain products and supplies must be purchased from our approved distributors. Soft drink syrup and bottled beverages must be purchased through Pepsi Cola. Certain other products, including without limitation MW uniforms, MW breaded chicken, MW 3-compartment paper plate, MW 20 oz & 32 oz cold drink cups, MW printed napkins, MW marinated beef, MW marinated

chicken, MW veggie & chicken eggrolls, MW soup mix, MW fortune cookies, MW Battered Pork, MW Lo Mein Noodle, MW Premade Honey garlic Sauce, MW Premade Orange Sauce, MW Premade Base Sauce, MW Premade Sweet and Sour Sauce and MW 20 oz food pail, must be purchased from a distributor or supplier we designate or authorize. The list of these products may be revised by us periodically by adding, substituting or removing products. We expect to receive vendor allowances from the suppliers and/or distributors of these products based on the volume of your purchases. During our most recent fiscal year which ended on November 30, 2024, these vendor allowances averaged approximately 1% to 5% of franchisees' purchases. We expect to negotiate purchase arrangements with suppliers (including price terms) for the benefit of all MANCHU WOK® Restaurants, including those owned by franchisees. We do not provide material benefits (e.g., renewal or additional franchises) to a franchisee based on the use of designated or approved suppliers. Failure to buy from approved suppliers is, however, an act of default under your Franchise Agreement.

We will conduct a grand opening marketing campaign on your behalf, including providing various marketing materials and services.

MTY USA and our subsidiaries are approved suppliers of goods and services, including lease review services, leasehold improvements, equipment, furniture, freight services, signage, engineering drawings and architectural layout drawings. Neptune Equipment, an affiliate of Kahala Franchising is currently one of the approved suppliers of certain equipment, menu boards, furniture, wall graphics, computer hardware and smallwares. You are not required to purchase from them and they are not the only approved supplier. MTY USA is not a supplier of any products or services. Kahala Management, an affiliate of MTY USA, is currently the only approved service provider of phone support maintenance for the software and hardware of the POS system ("POS Help Desk Phone Support Maintenance"). You are required to purchase the POS Help Desk Phone Support Maintenance from Kahala Management.

Kahala Management is also an approved service provider of real estate services, including negotiating a term sheet, negotiating the lease terms, and locating a site for a franchisee upon request from a franchisee. You may, but are not required, to use Kahala Management for real estate services.

We may, at our option, require you to enter into a lease for the restaurant premises directly with the landlord ("Direct Lease"). You and your attorneys must negotiate the terms of the Direct Lease. We have no liability to you regarding the terms or negotiations of the Direct Lease. The Direct Lease for the premises is subject to our final approval. You and the landlord under the Direct Lease must sign our approved Lease Addendum (See Exhibit I: Required Lease Terms (Lease Addendum to Lease Agreement)). If we require you to enter into a Direct Lease, you must provide us with a copy of the Direct Lease and the Lease Addendum for our approval at least 10 days before you sign the Direct Lease, and you must provide us with a copy of the executed Direct Lease and Lease Addendum within 10 days after signing. If you do not provide us with all of the required documents, we will not approve your Direct Lease.

A non-traditional restaurant will generally be located inside an existing primary business. As a result, if you open and operate a non-traditional restaurant, there may not be a Sublease between you and our Leasing Affiliate. If the circumstances make a Sublease appropriate, then you would sign a Sublease with our Leasing Affiliate.

You should carefully read the Direct Lease, or the Sublease and Master Lease, and, if applicable, the sublicense agreement. The Sublease provides that a default under your Franchise

Agreement is a default under the Sublease. It also provides that you will not raise any defense or counterclaim in any action we or our Leasing Affiliates commence, other than a defense of payment.

We also receive periodic vendor allowances or other consideration from our designated or approved suppliers. We retain these payments to partially compensate us for our ongoing efforts in establishing and maintaining quality sources of supply, in evaluating potential new suppliers, and in monitoring and evaluating approved suppliers and upstream manufacturers to insure that those suppliers and manufacturers meet our quality and performance standards. We do not presently participate in any purchasing or distribution cooperatives, but we reserve the right to do so in the future.

Pursuant to the mergers as fully described in Item 1, the total revenues and expenses of Kahala Franchising and its subsidiaries and affiliates have been consolidated with MTY USA's, as reflected in the audited consolidated financial statements presented in this disclosure document. For the year ending November 30, 2024, MTY USA and its subsidiaries, as of such date, derived revenues from the sales of products, services, and vendor allowances in the amount of \$51,714,322, which was approximately 8.5% of MTY USA's total consolidated recognized revenue in the amount of \$597,538,000.

During our last fiscal year, our consolidated revenue received \$30,372,764 of the \$51,714,322 from such vendors. Additional other revenues (for example, revenue from miscellaneous fees and expenses from franchisees) in the amount of \$5,680,438 were also received by MTY USA and its subsidiaries during the last fiscal year.

During our last fiscal year, Neptune Equipment earned a total of \$1,670,243 of the \$51,714,322. MTY USA on a consolidated basis with its subsidiaries, earned revenue from POS help desk support maintenance services and the sale of POS equipment, in a total amount of \$1,845,877 of the \$51,714,322.

We may, in our sole discretion, establish one or more strategic alliances or preferred vendor programs with one or more suppliers who are willing to supply all or some Restaurants with some or all of the products and/or services that Restaurants are authorized to offer to customers. If we do so, it may limit and/or require you to use suppliers other than those that you would otherwise use, and/or limit the number of approved suppliers with whom you may do business.

None of our officers owns an interest in any companies that are vendors or suppliers to the Manchu WOK System.

You must lease (or otherwise acquire) the premises for your MANCHU WOK® Restaurant and build-out your restaurant in accordance with our specifications and guidelines. Accordingly, the exterior and interior design and construction of your restaurant and all your signage, menuboards (including digital menuboards) fixtures, and equipment (including your POS System) must conform to our specifications.

We must approve the site for your Restaurant. You must acquire or enter a lease for the site that we have approved containing certain provisions that we require. If you lease the space for the Premises, we may require you either to: (a) lease the space for your Premises directly and sign a lease as well as a Lease Addendum; or (b) sign a lease directly with us. If the landlord refuses to enter a lease with you despite your best efforts to have it do so, and will only sign a

primary lease with us, we will exercise reasonable commercial efforts to enter a lease on terms satisfactory to us. If we are successful, we will sublease the premises to you under a net lease, which will pass-through all the expenses under the primary lease to you. The form of Sublease Agreement we will require you to sign is attached as **Exhibit D** to this Franchise Disclosure Document. However, in certain circumstances, a landlord may require you to sign a different form of sublease agreement.

You must acquire an approved Point of Sale and Electronic Cash Register System (POS System), and Digital Menu System from a vendor approved by us. Depending on the size of your Manchu WOK® Restaurant and/or your anticipated volume of business, additional terminals or hardware might be required. We will also require you to obtain and maintain high-speed Internet connectivity for your MANCHU WOK® Restaurant.

You must utilize our approved mobile application ("App") and online food ordering service (including any third-party delivery order integration) and may not use any other store-specific App or online ordering service. Olo is a Franchisor-approved online ordering vendor, as of this Disclosure Document's issuance date. You may also utilize third-party delivery services, through a provider of your choice, unless we require you to use a specific third-party(ies) and provided that if required and/or if you choose to participate in such third-party delivery services, you may be required to utilize a point-of-sale integration directed by us.

You must join our pre-approved electronic funds transfer program called ACH to pay your weekly royalties and any other amounts you authorize via electronic funds transfer. We pay all bank charges for the ACH program.

You must obtain all insurance we require and obtain it from an insurer having an A.M. Best's financial strength rating of "A-VIII" or better. Your insurance must: (i) insure the particular Franchisee listed on your Franchise Agreement; (ii) name us and our parents, subsidiaries, affiliates, directors, officers, and employees as additional insured; (iii) contain a waiver by the insurance carrier of all subrogation rights against us and our affiliates for casualty losses; (iv) provide that we will receive by an endorsement 30 days' prior written notice of cancellation; and (v) provide that failure by franchisee to comply with any term, condition or provision of the contract, or other conduct by franchisee, will NOT void or otherwise affect the coverage afforded us. Before you may open your Manchu WOK restaurant, at the time you sign your lease, annually at least 10 days prior to renewal of your insurance coverage, and at any other time upon our request, you must provide a copy of your certificate of insurance to us which meets our requirements.

Our minimum insurance coverage requirements (subject to change, including to increase) are below. You need to evaluate if your particular business will require greater coverage or other types of insurance. For example, we strongly recommend that you consult with an insurance broker to discuss whether your particular lease/situation requires and/or should obtain additional common types of insurance (including without limitation, employment practices liability, business interruption, umbrella insurance and cyber liability/data breach insurance coverage). Such insurance may significantly increase your premiums but may also save you money in the long run. We make no representation that the minimum coverage that we specify will be sufficient for your business. You will pay your insurance premiums directly to your insurance broker or to the insurance company issuing the policy.

TYPE OF COVERAGE	LIMITS/SPECIFICATIONS
General Liability	\$1,000,000 Bodily Injury/Property Damage Per Occurrence / \$2,000,000 Aggregate
Building Improvements and Betterments	100% of Full Replacement Cost – No Coinsurance (minimum of \$100,000)
Business Personal Property	100% of Full Replacement Cost – No Coinsurance – Special Form or equivalent (minimum of \$100,000)
Spoilage	\$5,000
Flood, Earthquake and Volcanic Eruption	Subject to Territory Limitations – required if in a designated Flood Zone
Workers' Compensation and Employer's Liability Insurance	As required by law
Hired and Non-Owned Automobile Liability	\$1,000,000 Combined Single Limit per accident

You must always keep the required insurance coverage in force, and you must comply with any changes we make periodically to our insurance requirements. Upon 30 days' notice to you, we may require you to increase and/or otherwise change the minimum coverage of the insurance referred to above including to reflect identification of special risks, changes in law or standards of liability, higher damage awards or other relevant changes in circumstances. In the event you fail to obtain or maintain the required insurance coverage, we reserve the right, but are not obligated to, obtain the required insurance on your behalf and charge the insurance premium to you.

We must approve all your marketing materials before you use them; in addition all of your computer hardware and software must conform to our specifications and be acquired from an approved vendor.

We consider a variety of factors when determining whether to renew or grant additional franchises. Among the factors considered are compliance with the requirements described above.

ITEM 9 FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the franchise and other agreements. It will help you find more detailed information about your obligations in these agreements and in other items of this Disclosure Document.

Franchise Agreement and Sublease

	OBLIGATION	SECTION IN AGREEMENT	DISCLOSURE DOCUMENT ITEM
a.	Site selection and acquisition/lease	Sections 4.1 and 4.2	Items 6, 7, 8 and 11
b.	Pre-opening purchases/leases	SECTION 4 and Section 6.8	Items 5, 7, 8, 10, and 11
C.	Site development and other preopening requirements	SECTION 4	Items 7, 8 and 11
d.	Initial and ongoing training	Sections 5.2 and 6.20	Items 5, 6, 7 and 11
e.	Opening	Section 4.5	Item 11
f.	Fees	SECTION 3; Sublease Sections III and V	Items 5, 6, 7 and 8
g.	Compliance with standards and policies/Confidential Manual	Sections 2.4, 2.5, and SECTION 6	Item 11
h.	Trademarks and proprietary information	SECTION 9 and Section 11.1	Items 13 and 14
i.	Restrictions on products/services offered	Section 6.4	Items 8 and 16
j.	Warranty and customer service requirements	Section 6.1	Not Applicable
k.	Territorial development and sales quotas	Not Applicable	Not Applicable
I.	Ongoing product/service purchases	Section 6.4	Item 8
m	Maintenance, appearance and remodeling	Section 6.6; Sublease Section IV	Item 6
n.	Insurance	SECTION 7; Sublease Section IV	Items 6 and 8
Ο.	Marketing	Sections 3.5, 5.5, 5.6 and 6.13	Items 5, 6 and 11
p.	Indemnification	Section 13.2; Sublease Section IX	Item 6
q.	Owner's participation/management/staffing	Section 6.2	Items 11 and 15
r.	Records/reports	SECTION 8; Sublease Section VI	Items 6 and 11

	OBLIGATION	SECTION IN AGREEMENT	DISCLOSURE DOCUMENT ITEM
s.	Inspections/audits	Sections 6.7 and 8.3; Sublease Section IV	Items 6 and 11
t.	Transfer	SECTION 12; Sublease Section VII	Items 6 and 17
u.	Renewal	Section 2.2	Items 6 and 17
V.	Post-termination obligations	Sections 10.2, 10.3, 10.4, 10.5, 10.6, 10.7, 10.8, 10.9, 10.10 and 10.12	Item 17
W.	Non-competition covenants	SECTION 11	Item 17
Χ.	Dispute resolution	SECTION 14	Item 17
y.	Personal guarantee	SECTION 15	Item 15
Z.	Cross defaults	SECTION 10; Section 10.1(j)	Not Applicable

ITEM 10 FINANCING

We do not offer direct or indirect financing. We do not guarantee your note, lease or other obligations.

ITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, MTY USA is not required to provide you with any assistance.

A. Pre Opening Obligations

We will assist you in the building and opening of your Manchu WOK® Restaurant as follows: (all section numbers refer to the Franchise Agreement)

- We will review and approve the site you have selected; Sec 4.1 and 4.2
- We will provide building plans, a finish schedule and a contact list for approved or designated suppliers of furniture fixtures, equipment and inventory; Sec 4.4
- We will designate suppliers who will sell to you the furniture, fixtures, equipment and opening inventory; Sec 4.8
- If required by the landlord of the site you select, we will enter into a head lease and sublease the leased premises to you; Sec 4.3
- We will conduct a Grand Opening Marketing campaign and Promotional program for you using the \$10,000 Grand Opening Marketing and promotion fee you pay to us: Secs 3.5.d and 5.5

- We will provide you with a loan only of copies of our Confidential Operations Manual, Recipe Manual and Local Store Marketing Manual (collectively, the "Confidential Manual"); Sec 5.1
- We will provide an Initial Training Program for you and one of your employees as designated by you. One of your trainees must be your store Manager who will be responsible for the operation and management of your Manchu WOK[®] Restaurant. Sec 5.2

B. Continuing Obligations

We will assist you in the ongoing operations of your Manchu WOK® Restaurant as follows:

- We will provide updates and revisions to the Confidential Manual periodically as we deem necessary or advisable; Sec 5.1
- We will provide ongoing training in methods and techniques both in your Manchu
 WOK Restaurant and /or at locations we designate; Sec 5.2
- We will provide ongoing operational advice and guidance to you concerning the use of the Manchu WOK Restaurant System; Sec 5.3
- At your request and expense, we will provide special assistance to you as we deem necessary or advisable in the circumstances. Sec 5.4

We have the right to delegate to others the performance of any pre-opening or continuing obligations to franchisees.

C. Site Selection and Construction

- i) You must choose the location for your Manchu WOK® Restaurant. We must approve such location and will do so if it meets our minimum site-approval criteria. We will notify you of our approval or disapproval of a proposed site within 30 days of receipt of your proposal. Our approval of any location means only that it meets our minimum criteria for a Manchu WOK® Restaurant. Franchisees generally will select a site that we have approved before they sign their Franchise Agreements. If, however, you do not, you must select an acceptable site and enter a lease that we have approved within 30 days after all parties have signed your Franchise Agreement. The consequences of your failure to comply with your development obligations (for example, by failing to timely select an acceptable site) is that it is an Event of Default under your Franchise Agreement, for which we may terminate your Franchise Agreement.
- ii) We will review and must approve the site you have selected for your Manchu WOK® Restaurant. Although we devote significant time and research to identifying prospective locations, we do not represent or guarantee that your approved location will be successful, make a profit, or attain any level of revenues. We have developed specific criteria to evaluate each prospective location. Manchu WOK® Restaurants are generally located in street front locations, Shopping Mall Food Courts or in non-traditional locations. non-traditional locations include highway facilities, airport concourses, professional sports facilities, recreational and entertainment facilities, casinos, college campuses, military installations, governmental or institutional locations, supermarkets and department stores. In addition to analyzing demographic information for each prospective location, we consider factors such as location, parking, population density, traffic

patterns, accessibility and visibility. We also consider rental rates, lease duration and other lease terms in analyzing prospective locations.

- iii) If the landlord of the approved site refuses to enter into a lease with you despite your best efforts to have it do so, and will only sign a primary lease with us, we will exercise reasonable commercial efforts to enter into a head lease with the landlord on terms acceptable to us, and if we do, we will sublease the site to you under our standard sublease agreement. We will supervise constructing and improving the premises, in return for which you must pay us the Development Construction Fee.
- iv) We will supervise the construction of leasehold improvements and, typically, will select the contractors and suppliers who do the work.
- v) You must open your Restaurant within 180 days of the date of the Franchise Agreement. The opening of your Restaurant may be affected by the weather, finding of a location, negotiation of a lease, the location and condition of the proposed store, delays in obtaining construction and food-related permits, the construction schedule for the Restaurant, and the construction schedule of the landlord to build or renovate the base building. The consequences of your failure to do so is that it is an Event of Default under your Franchise Agreement, for which we may terminate your Franchise Agreement. The typical length of time between the signing of the Franchise Agreement or the first payment of any consideration for a Manchu WOK® Restaurant and the opening for business of a new Manchu WOK® Restaurant is 180 days.
- vi) Every 5 years, you must renovate, alter, and/or upgrade your premises, fixtures, and equipment as may be required by the landlord or as we may require, so your Restaurant reflects the then-current image of the Manchu WOK system. This may entail the complete remodeling and upgrading of your Restaurant. For any renovation, alteration or upgrading, you must pay us a Renovation Development and Construction Fee.

D. Training – Initial, Refresher, Supplemental and Other Ongoing

We will provide a mandatory pre-opening training course in the Manchu WOK® restaurant system ("Initial Training") for two (2) of the employees of your Manchu WOK Restaurant who you designate. Your Manager must be one of the trainees. You must be the Manager if you are an individual; if you are a corporation, one of the shareholders who owns at least a 10% equity interest must be the Manager. The consequences of your or your Manager's failure to successfully complete Initial Training after two attempts is that it is an Event of Default under your Franchise Agreement, for which we may terminate your Franchise Agreement. If you previously operated a Manchu WOK Restaurant, then, before opening your Restaurant, we will provide refresher training in food preparation techniques and other aspects of the physical operation of a Manchu WOK Restaurant ("Refresher Training") to two persons you designate (one of whom must be your Manager for this location). We may provide further training assistance as we consider necessary ("Supplemental Training"), including assistance with any new developments in the Manchu WOK Restaurant System such as store generation upgrades or matters that may be described in revisions made to the Confidential Manual. All persons attending the training must complete it to our satisfaction. Initial Training, including all requisite materials, are provided to you; the cost of this Initial Training is included in your Initial Franchise Fee. For Refresher Training. Supplemental Training and any training other than Initial Training, or training required on a transfer, you will pay \$3,000 for all persons attending the training. In addition, you shall pay for all personal expenses incurred by the persons attending any training (including, travel, lodging, and meal expenses). Only one Initial Training is included in the Initial Franchise Fee. In the event you

or your Manager fails to pass the Initial Training, you shall be required to pay \$3,000 for subsequent Initial Training. Your Manager must attend and satisfactorily complete all training. The consequences of your Manager's failure to do so is that it is an Event of Default under your Franchise Agreement, for which we may terminate your Franchise Agreement. If at any time or times we designate any Supplemental Training to be mandatory, you or your Manager must attend the training.

ii) Initial Training is usually for 20 working days and averages 8 hours per day. Typically, you must complete training approximately one month before opening your Manchu WOK® Restaurant.

Our training programs are under the general supervision of Mr. Tony Lei, Regional Manager of USA. Mr. Lei has been employed by us or predecessor/affiliated companies in the food service business since 1992. Mr. Lei will be assisted by other corporate employees to conduct the paperwork training and for subjects related to the employees' specialized knowledge. These employees' experience in their specific areas ranges from 5 - 20 years. On the job training currently is supervised by Mr. William Tung. Mr. Tung has been employed by us or predecessor/affiliated companies since 1999 and was a District Manager for Province of Ontario in Canada for five years before assuming his current responsibilities as corporate trainer in 2020 On the job training will be held at an assigned Manchu WOK® restaurant, under the guidance of the restaurant's store manager. Training includes the review of the Confidential Manual and the management of a MANCHU WOK® Restaurant. Your Manager and your designated employee must complete Initial Training to our satisfaction through their demonstrated knowledge and understanding of both the Confidential Manual and the bookkeeping and inventory control system. The subjects covered and approximate hours of classroom and on-the-job training are described below:

TRAINING PROGRAM

Column 1	Column 2	Column 3	Column 4
Subject	Hours of Paperwork Training	Hours of on-the-job training	Location
In-Store Training		120	Training restaurant or other location as designated by us
New Owner Training	40		Training restaurant, KTEC (Kahala Training & Education Center) in Scottsdale, AZ, virtually, or other location designated by us

Instructional materials consist of the Confidential Manual, Training Manual and standard forms. We reserve the right to select an alternative trainers and training location, in Canada or the United States, at any time.

You and anyone taking in-store Training (including employees of franchisees) must sign the In-Store Training Release and Waiver of Liability Agreement which is attached to this Franchise Disclosure Document as **Exhibit E-1**.

- iii) You, your Manager and any other employees we designate must attend and participate in (at your expense) management seminars and training and refresher courses and Manchu WOK® and/or industry conventions, including the Annual Franchisee Convention (or regional meetings in any year in which the Annual Franchisee Convention is not held), and participate in ongoing aptitude testing that we conduct. We will not compel you, your Manager, and your other employees to attend more than two seminars, conventions or other courses in any calendar quarter. This restriction does not apply to Initial Training or any other training that we may require your Manager or any replacement Manager to attend. Please note, at our sole discretion, the Training Program for a non-traditional Manchu WOK may differ slightly from the Training Program outlined above.
- iv) To the extent we consider it appropriate or reasonably necessary, we will provide continuing advice and guidance to you concerning how to use the Manchu WOK® Restaurant System and in the operation of your restaurant business including advice and guidance regarding formulating and implementing local marketing and promotional programs, purchasing goods and other supplies, establishing and maintaining our administrative, bookkeeping, accounting and general operating procedures, hiring and training employees and providing results of research on market trends, where practicable.
- v) To the extent that we consider it appropriate and necessary and have available personnel and can otherwise accommodate your request, we will furnish additional or special assistance to you (in addition to the assistance referred to above) to resolve any specific problems you encounter when operating your Manchu WOK Restaurant. If we send our personnel to your Manchu WOK Restaurant, you must pay us \$500 per day (or part of each day) for each person we provide to render these additional or special services, together with all reasonable costs we incur on account of travel, lodging and meals for all personnel provided.

E. Manuals

We will provide you (paper, electronic or other format) with a loan only of copies of our Operations Manual, which consists of our confidential Operations Manual, Recipe Manual and Local Store Marketing Manual. The Table of Contents of the Operations Manual is attached to this Franchise Disclosure Document as **Exhibit E**. This Operations Manual is confidential and remains our property. We may periodically modify the Operations Manual or change the format in which the Operations Manual is presented. Currently, the Operations Manual has a total of 259 pages.

F. Marketing – Initial/Ongoing/Local/Regional/National

i) We will conduct a Grand Opening Marketing campaign and Promotional program for you using the \$10,000 Grand Opening Marketing and Promotion Fee you pay to us. For the Grand Opening Marketing and Promotional Program, we furnish merchandising material, including "Now Open" signs, "directional" signs and a grand opening banner. We also give you various coupons and marketing fortune cookies that contain a promotional offer in lieu of the traditional fortune. The costs associated with any discount attributable to any coupons used by your customers or any free food and beverages that you provide your customers are your responsibility.

- ii) At our option, we will provide ongoing marketing and promotional programs and suggestions/guidance for marketing to assist you in the marketing and promotion of your Manchu WOK restaurant.
- iii) All Franchisees must contribute a Marketing Fee equal to 1% of weekly gross sales to the marketing and sales promotion fund (the "Fund") (gross sales includes all the revenues from your Restaurant but not any sales tax you pay). We may co-mingle Marketing Fees contributed to the Fund with marketing funds established by our affiliates, principally those affiliates operating in the United States. We will use the Fund for formulating, developing, and conducting programs and policies for marketing, sales promotional matters, and marketing research, communication, and development; purchasing advertising or marketing rights or services in any media; administrating any media-buying co-operative we may establish; and developing and administering marketing and promotional activities for the Manchu WOK® System. We may retain and use the Fund monies in the manner as we consider appropriate.
- iv) Through the Fund, we will provide you with marketing materials, including ad slicks, promotional materials, customer programs and logo art work. The Fund may disseminate advertising in electronic, print, and other media, including newspapers, magazines, handbills and flyers. The Fund may advertise in national, regional, and local media.
- v) We administer the Fund. We use our in-house marketing department (which is employed to work on marketing and promotional materials for the Manchu WOK® restaurant concept, including Mall and non-traditional locations) to create and place marketing programs. We may administratively aggregate the monies in the Fund and pay ourselves out of the Fund an annual administration fee of 15% of the total amounts paid to the Fund. Monies expended by the Marketing Fund shall be used to promote the Manchu WOK® restaurant concept and not to sell additional franchises.
- vi) Fund expenditures during our most recent fiscal year ending November 30, 2024, fell into the following categories:

Production	0.0%
Media Placement	48.47%
Administrative	3.71%
Other	47.81%
TOTAL	100%

The "other" expenses included public relations, research, concept development and communications. None of the Fund was used for the solicitation of franchises.

- vii) You may develop and use your own marketing materials; however, we must approve all the materials, in writing, before you use them. We will approve or disapprove of your proposed marketing materials within 10 working days after you submit them for evaluation. Any marketing materials or programs which you may develop become our property without requirement of payment to you.
- viii) The Fund will collect marketing fees from all franchisee and company-owned Manchu WOK® Restaurants. All Company-owned outlets contribute to the Fund on the same basis as franchisees. Some non-traditional locations are exempt from payment of marketing contributions. The Fund is not required to spend any amount on marketing in the area where your Manchu WOK® Restaurant is located.

- ix) We have no fiduciary responsibility to you on our management of the Fund. The Fund is not audited, and the financial statements for the Fund and accounting of the Fund are not available to franchisees.
- x) Any funds not spent in the year in which they are collected will be spent in the next year.
- xi) There is presently no marketing council. However we and our affiliates intend to establish a Franchisee Advisory Council comprising initially of appointed Franchisee representatives representing all Manchu WOK® restaurants to discuss matters of common interest and make recommendations concerning Manchu WOK® restaurant operations, procedures, marketing etc. to us. It is anticipated that this Franchisee Advisory Council will eventually become a body whose representatives will be elected annually by all Manchu WOK® Franchisees. We and our affiliates will carefully consider any and all recommendations made by the Franchisee Advisory Council but we are not bound to accept or institute any recommendation(s) or proposal(s) made by the Franchisee Advisory Council.
- xii) In addition to contributing to the Fund and fulfilling any marketing obligations under your Lease or Sublease, you must use and display all marketing and promotional materials (including banners, signs and point-of-purchase materials) as we may provide. Any marketing program you may conceive or develop becomes our sole property.
- xiii) We and our affiliates have the exclusive right to use our trademarks (both registered and unregistered) and licensed items for e-commerce purposes and for use on the World Wide Web (or in any other electronic medium, including for example social media and social networking sites). You may not use the licensed items or our trademarks on the World Wide Web. You must not operate or advertise, market, or otherwise promote your Manchu WOK® Restaurant on the Internet. You may not register any domain name containing the words "Manchu WOK" or any similar words, or establish, operate, or participate in a Web site on which the words appear.
- xiv) We reserve the right to create a regional marketing cooperative in any Designated Market Area ("**DMA**") (DMAs are local television market areas designated by the Nielsen division of the Television Bureau of Advertising, Inc.) and to establish its rules and regulations. If and when we request, you must become a member of the cooperative for the DMA in which some or all of your Trade Area is located. In no event will your Manchu WOK® Restaurant be required to be a member of more than one cooperative. The cooperative must be organized, governed, and operated in the manner we prescribe. The cooperative will have the right to require you contribute to the cooperative as well as determine the amount of your contribution, but the contribution, when combined with the amount you must contribute to the Fund, will not exceed, in any 12 month period, 5% of your gross sales. The cooperative need not operate from written governing documents nor need it prepare annual or periodic financial statements available for your review. You may obtain an accounting of advertising expenditures during the prior fiscal year by making a written request for it.
- xv) The Fund is not a trust fund. We have no fiduciary duty to any franchisee in connection with the collection or expenditure of Fund monies or any other aspect of the Fund's operations.
- xvi) A telephone directory advertisement must be approved by us in advance.
- xvii) You must display pamphlets and other promotional/advertising materials as we may specify and including materials that solicit new franchisees that we may provide to you.

xviii) At our request and at your expense, you must participate in our System marketing and Restaurant monitoring programs including Internet-based programs, such as various e-mail programs (such as e-mail list building), messaging, newsletters, customer retention and acquisition programs, social media campaigns and tracking surveys, campaign tracking, mystery shopper programs, and health and safety monitoring services. We reserve the right to periodically add, delete or modify System marketing and Restaurant programs.

G. POS System

You must purchase and use Vivonet cash register system ("**POS System**") inclusive of necessary software. The POS System includes devices that (i) record cumulative sales and provide sequential customer count recording on a daily and cumulative basis and that cannot be turned back or reset; and (ii) permit electronic polling of all cash register information. We have the right to poll (i.e., take copies of information from) the POS System at any time we choose and use it for any purpose whatsoever. All required hardware and software associated with this system must be obtained from Vivonet Acquisition Ltd., 1188 W Georgia St. #1790, Vancouver, BC V6E 4A2.

For your POS System, you must acquire and maintain high-speed Internet connectivity. We estimate that the cost of purchasing required hardware and software, including installation, will range from \$4,000 to \$8,000, and the cost of optional maintenance warranty is range from \$250 to \$300 for three years per screen terminal and \$110 to \$150 for 3 years per Printer. You would also need to pay for a monthly software subscription fee range from \$50 to \$70 per POS.

Neither we nor any of our affiliates is obligated to provide ongoing maintenance, repairs, upgrades or updates to your POS System.

H. Menu System

You must purchase and use the Digital Menu System inclusive of necessary hardware provided by our approved vendor- UNOApp, 670 Caledonia Road, 2nd Floor, Toronto, Ontario, M6E 4V9.

To maintain Manchu Wok brand standards, we shall have sole control over the displayed contents including, but not limited to, National menu offerings, pricing, advertising, promotions and videos. All other content specific to your location (Local advertising, Mangers Specials, etc.) must be obtain approval from Marketing department and will be uploaded directly from UNOApp. In order to run and sustain the UNOApp menu board system, you must acquire and maintain high—speed Internet connectivity. As of the date of this Franchise Disclosure Document, we are currently using 4 X 49" LG screen system as our standard menu displays. We estimate that the cost of purchasing this system with required software & hardware including initial content set up, will range from \$4,000 to \$6,000 depending on the size and number of screens in your menu system. In addition to the cost of hardware, a monthly fee of \$49 to \$60 will be applicable. This is a software licensing and servicing fee.

ITEM 12 TERRITORY

You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

The Franchise is granted for a specific location, and you must operate your Manchu WOK® Restaurant only at the location that we have approved ("Approved Location") within the territory; you may not operate at any other location and which will be described in the Franchise Agreement, and you may not use other channels or methods of distribution, including the Internet (or any existing or future form of electronic commerce), to conduct sales outside of your Franchise Territory. In some instances, the Approved Location will be described to include a portion of a facility at which the Restaurant premises is located, such as a specific terminal at an airport or an entire shopping mall. You may not move or otherwise change the location of your Manchu WOK® Restaurant without our approval of the proposed new site. If your relocation is caused because your lease or sublease terminates or expires, or if the Restaurant premises are destroyed, condemned or otherwise rendered unusable, we will consent to the relocation but only if (a) we approve the new site: (b) the relocation is at your expense: (c) you reimburse us and/or affiliates for all reasonable costs and expenses which we incur that are associated with your relocation; and (d) you agree to indemnify us and our affiliates against, and reimburse us or them for, all loss, liability, costs and expenses incurred in connection with your relocation. We have the right to terminate your franchise if you do not find an acceptable new site within 30 days or do not open your Restaurant at the new location within 3 months. For relocations based on any other reason, (a) we have the absolute right to withhold our consent; and (b) if we do consent, you must pay us all fees that a new franchisee would pay us, including the then-current initial franchise fee.

So long as you are not in default of your Franchise Agreement, neither we nor our affiliates will open or operate, or grant a third party the right to operate, another Manchu WOK Restaurant at your Approved Location.

If the Lease for your Restaurant contains provisions respecting geographic restrictions on development of additional Restaurants, we and our affiliates will respect such Lease provisions.

- We and our affiliates reserve all rights not expressly granted in your Franchise Agreement, including the right other than within your Restaurant premises to:
- establish, acquire, develop, and license or franchise other systems different from the Manchu WOK[®] System licensed by your Franchise Agreement, within or outside the Approved Location, without offering or providing you any rights in, to, or under the other systems;
- establish other franchises and company or affiliate-owned outlets or other channels of distribution selling or leasing similar products or services under different trademarks; and
- sell, through dissimilar channels of distribution, such as retail grocery stores, the Internet, catalogue sales, telemarketing, or other direct marketing sales, under any terms that we consider appropriate, Manchu WOK® products and services similar or identical to those authorized for your Manchu WOK® Restaurant and using the trademarks we permit you to use.

Except as discussed above, neither we nor our affiliates have as at the date of this Franchise Disclosure Document any specific plan or intention to operate, or franchise the operation of, any business similar to or competitive with, the Manchu WOK® Restaurant franchise being offered in this Franchise Disclosure Document, but it is possible that we or an affiliate may do so within the next year. Your Franchise Agreement does not and will not grant to you any options, right of first refusal, or similar rights to acquire additional franchises within or outside of the Approved Location.

Your territorial rights, if any, do not depend on achieving a certain sales volume, market penetration, or other contingency (provided however that you must not be in default of any provision contained in your Franchise Agreement).

Except as described above, we have no contractual right to modify any territorial right that we grant.

We are not restricted from soliciting or accepting orders from, or selling products or services using any trademarks, to, consumers located at any location (other than your Restaurant premises), including orders, products or services offered from other distribution channels, such as the Internet, catalogues, telemarketing, or other direct marketing sales. If we do so, we will not pay any compensation to you.

You may only serve retail customers at your Restaurant premises. You are not restricted from soliciting customers or accepting orders from customers residing at any location; however, you are prohibited from providing off-premises sales or services such as off-site delivery. In other words, all products sold by your Restaurant must be sold at your Restaurant premises, and to customers who are physically present at the Restaurant premises. You also may not solicit customers at or from any location through use of any other channels of distribution.

On May 25, 2016, MTY entered into an agreement with Kahala Brands, whereby Kahala Brands agreed to merge with a wholly-owned subsidiary of MTY. The transaction closed on July 26, 2016. As of the date of this Franchise Disclosure Document, Kahala Brands or one or more of its subsidiaries has franchised or currently operates eighteen (18) different restaurant concepts and is also the master franchisee for two (2) of those restaurant concepts in various countries.

Certain of the restaurant concepts franchised or sub-franchised by us or our affiliates offer goods or services that are the same or similar to those that you will offer under the Manchu Wok brand. Our affiliate offers the same or similar goods or services under the *Samurai Sam's* trademark. The *Samurai Sam's* concept is franchised in the United States by Kahala Brands subsidiary Kahala Franchising, LLC. Since you will not be granted an exclusive territory, any conflicts that may arise between you and a *Samurai Sam's* franchisee regarding territory or Franchisor support that cannot be resolved through negotiation between the parties shall be resolved through arbitration.

ITEM 13 TRADEMARKS

Franchisees receive the right to operate their MANCHU WOK® Restaurant(s) under the federally registered trademark MANCHU WOK®. You will also be permitted to operate your MANCHU WOK® Restaurant under any other registered or unregistered trade names, trademarks, service marks, logotypes and other commercial symbols (collectively with MANCHU WOK®, the "Marks"), that Manchu WOK, now or in the future, considers appropriate to use for your MANCHU WOK® Restaurant. The terms "Marks" or "Trademarks" apply to unregistered trademarks of MANCHU WOK® as well as to registered Trademarks. We own the Marks that will be licensed to you through the Franchise Agreement.

The following Marks are registered with the United States Patent and Trademark Office on the Principal Register.

TRADEMARK	REGISTRATION NUMBER	REGISTRATION DATE
MANCHU WOK	1,279,286	May 22, 1984

All required affidavits have been filed with the U.S. Patent and Trademark Office and the registrations for these marks have become incontestable.

There are no applications pending with the United States Patent and Trademark Office at this time.

You must use all trade names, trademarks, service marks, logotypes and commercial symbols in full compliance with rules established by us. You are prohibited from using any name or mark as part of your corporate name or with a prefix, suffix or from modifying words, terms, designs or symbols which may form a part of any of our Marks. In addition, you may not use any of our Marks for the sale of any unauthorized products or services or in any other manner not explicitly authorized by us in writing in our sole discretion. Should you develop any marketing material(s) or program(s) for the use in your Manchu WOK® Restaurant at any time, we shall be deemed for all purposes to be the owner of such material or program and shall have the right to license the use of such new material/program to others.

There are no presently effective determinations of the United States Patent and Trademark Office, Trademark Trial and Appeal Board, the trademark administrator of any state or any court, nor are there any pending infringement, opposition or cancellation proceedings or any pending material litigation involving the use or ownership of the Marks. No currently effective litigation affects our use or ownership rights in any mark. No currently effective agreement limits our right to use or license the use of our Marks.

If you become aware of any infringement of, or challenge to, using any of the Marks, you must promptly notify us and we will take action as we deem appropriate in our sole discretion.

We or our affiliate will defend you against all claims of trademark infringement for use of the Marks and hold you harmless from any of the claims if you have complied with your Franchise Agreement, promptly given us written notice of any claim, grant us complete authority to defend against the claim, and otherwise cooperate with us in accordance with our instructions. We reserve the right to control any litigation related to the Marks and we have the sole right to decide to pursue or settle any infringement actions related to the Marks. We may conduct the legal action (or settle same) on the terms as we consider appropriate, in our sole discretion and we shall bear all costs of defense of any of the claim or suit, provided that you have satisfied the conditions described in this paragraph.

If, at any time, we consider it advisable to modify or discontinue the use of any of the Marks and/or use one or more additional or substitute names or marks for reasons including the rejection of any pending registration or revocation of any existing registration of any of the Marks, or the superior rights of senior users you must do so at your sole expense within 30 days of our request. We are liable solely to reimburse you for your reasonable direct printing and signage expenses incurred to modify or discontinue the use of the Mark and substitute a different Mark. The reimbursable expenses do not include any expenditures you make to promote a modified or substitute Mark.

You must not directly or indirectly contest our right to use the Marks, trade secrets or techniques that are part of the Manchu WOK® System.

We know of no infringing uses of any of the Marks that could materially affect your use of the Marks.

ITEM 14 PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

You do not receive the right to use any item covered by a patent or copyright, but you will become licensed only to use the proprietary information contained in the Confidential Manual. We have not filed an application for copyright registration of the Confidential Manual. However, we claim a copyright and assert that the information described in the Confidential Manual is proprietary. You must tell us when you learn of unauthorized use of this proprietary information. We are not obligated to take any action, but will respond to this information as we think appropriate.

Our affiliate, Manchu WOK Holdings, Inc., has licensed to us the right to use the proprietary information for the franchise system. The term of this license agreement is indefinite but is terminable by either party if the other party i) commits a material breach which remains uncured for 30 days; ii) commits an act of insolvency; or iii) repeatedly commits defaults. If the license agreement is terminated, the trademark licensor will assume all of our obligations under your Franchise Agreement with respect to the proprietary information for the franchise system with respect to the proprietary information which they licensed to us. Neither the license agreement nor any other agreement currently in effect significantly limits our right to use, or license the use of, the proprietary information in the Confidential Manual in any manner material to the franchise.

You may use the trade secrets and confidential information we reveal to you solely for the purpose of conducting the Business in accordance with the provisions of your Franchise Agreement. You may divulge the trade secrets and confidential information to your employees, but only on a "need-to-know" basis and in circumstances that will maintain its confidentiality. You must have your directors, officers, members, managers, shareholders, employees and agents maintain the confidentiality of the confidential information and trade secrets, and must obtain from them written agreements to maintain the confidentiality, in a form we prescribe.

ITEM 15 OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

- i.) You must have at least two people who have successfully completed Initial Training operate your Manchu WOK® Restaurant. One of them must be your Manager.
 - ii.) If you are an individual, you must be the Manager.
- iii.) If you are a corporation, you must designate one of your shareholders who owns at least 10% of your equity in the corporation to be the Manager.
- iv.) Your Manager must devote his or her full-time efforts to on-premises supervision of your Restaurant.
- v.) Your Manager must have the authority to receive and implement all directions we give him or her and have the authority to represent you in all dealings with us.
- vi.) Before his or her appointment as Manager, you must submit in writing the name and qualifications of any proposed Manager, or replacement Manager, together with any other pertinent information that we request for our review and written approval of the proposed Manager.
 - vii.) Without our written approval, you may not appoint any person a Manager.
- viii.) We reserve the right to withhold our consent to the appointment of any person as manager until that person shall have successfully completed all training we deem necessary.
- ix.) The Manager and all other employees who operate your Manchu WOK® Restaurant shall be required to enter our standard Confidentiality and Non-competition Agreement.
- x.) We require that all shareholders, members, managers, or other owners of a corporate franchisee personally guarantee the franchisee's payments and observance and performance to us and our affiliates of all of your obligations. All the Franchisee's legal and beneficial owners must sign the guarantee in the form attached as Schedule H to the Franchise Agreement.

ITEM 16 RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

You may sell and provide only products and services that conform to our standards and specifications. You also will have certain obligations to offer for sale particular items. You may only offer and sell products at your Manchu WOK® Restaurant that we have approved in writing for sale at your Manchu WOK® Restaurant. We have the right, without limit, to change the types of authorized products and services.

You may only serve retail customers at your Restaurant premises, for personal carry-out consumption, and for delivery service in a manner that complies with our standards. You are not restricted from soliciting customers, or accepting orders from customers, residing at any location; however, you are prohibited from providing off-premises sales or services, such as, for example,

catering or off-site delivery. In other words, all products sold by your Restaurant must be sold at your Restaurant premises and to customers who are physically present at your Restaurant premises. You also may not solicit customers at or from any location through use of other channels of distribution, such as the Internet, catalogues, telemarketing, or other direct marketing.

Except as described above, you are not restricted by the Franchise Agreement or any other custom or practice of ours or of our affiliates with respect to products or services which you may offer or sell or with respect to customers to whom you may sell.

ITEM 17 RENEWAL, TERMINATION, TRANSFER, AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists certain important provisions of the franchise and related agreements. You should read these provisions in the agreements attached to this disclosure document.

	PROVISION	SECTION IN FRANCHISE AGREEMENT	SUMMARY*
a.	Length of the franchise term	Section 2.1	If you are purchasing a new or existing non-operating Manchu Wok Restaurant, the term is (i) 10 years from the date the restaurant opens to the public if you own the property, or enter into a lease directly with the landlord or other third party, or (ii) the term of the sublease if you enter into a sublease with MTY affiliate, excluding any extensions and renewal options. If you are purchasing an existing and operating Manchu Wok Restaurant, the term is (i) 10 years from the effective date of the franchise agreement if you own the property, or enter into a lease directly with the landlord or other third party, or (ii) the term of the sublease if you enter into a sublease with a MTY affiliate, excluding any extensions or renewal options. If you are purchasing a Manchu Wok restaurant that will be co-branded into another MTY or MTY affiliate brand Restaurant, the term of your franchise agreement for this brand will equal the remaining term of the existing franchise agreement for the store which this brand will be co-branded into, so both franchise agreements will expire concurrently
b.	Renewal of the term.	Section 2.2	You do not have any rights to operate the franchised business beyond the Term, subject to any state laws to the contrary;

PROVISION		SECTION IN FRANCHISE AGREEMENT	SUMMARY*
			Franchisor reserves the right to offer a successor term as provided for in the Franchise Agreement and below
C.	Requirements for you to renew.	Section 2.2	If the Term of the Franchise Agreement expires, we reserve the right to offer you the opportunity to extend your rights to operate the franchised business by entering into a successor franchise agreement ("Successor Franchise Agreement") upon such terms as we are offering new franchisees when your Term is expired. Any such Successor Franchise Agreement may have materially different terms and conditions from the original Franchise Agreement.
d.	Termination by you	Not Applicable	None.
e.	Termination by us without cause	Not Applicable	None.
f.	Termination by us with cause	Sections 10.1, 10.2	We may terminate only if you default under your Franchise Agreement.
g.	"Cause" defined- curable defaults	Section 10.2	Including, but not limited to, failing to successfully complete initial training after two attempts; failing to pursue or successfully complete development obligations, if any; failing to commence operations; failing to pay amounts owed to Franchisor; defaulting under the lease or sublease; failing to furnish reports or other documents pursuant to the Franchise Agreement; engaging in conduct that is detrimental to the Franchisor, marks, or similar including failing to pay trade creditors or misusing the marks; breaching any obligation to Franchisor under the Franchise Agreement or other contract.
h.	"Cause" defined- non- curable defaults	Section 10.1	Including, but not limited to, losing the right to the premises; underreporting gross sales; improperly assigning or encumbering the business; breaching the Franchise Agreement's restrictive covenants; receiving 3 or more notices of default from Franchisor during any consecutive 12 month period; intentionally misrepresenting information in connection with the application; becoming insolvent,

PROVISION		SECTION IN FRANCHISE AGREEMENT	SUMMARY *
			committing an act of bankruptcy, or similar acts; abandoning the business; failing to cure default under any agreement with Franchisor; creating an imminent public health or safety risk; misappropriating funds withheld from employee wages; violating or having assets, property, or interests blocked under anti-terrorism laws.
i.	Your obligations on termination/non-renewal	Sections 2.4, 10.2-10.10, 10.12	Obligations include, rescinding contracts as necessary, ceasing use of our trademarks, de-identifying premises, paying amounts due, including for an early termination caused by your default, future royalty fees that would have been paid throughout the remainder of the term, returning manuals and other operations materials, assigning telephone numbers and if Manchu WOK requests, assigning your Lease and/or selling us any of your equipment or fixtures at the price described in your Franchise Agreement (also see "r", below), providing an accounting, complying with confidentiality and non-compete requirements.
j.	Assignment of contract by us	Section 12.1	No restriction on our right to assign.
k.	"Transfer" by you- defined	Section 12.2	You need our prior written consent to transfer of your Franchise Agreement, assets, or stock or ownership interests. Transfers include selling, assigning, charging, or granting a security interest in or otherwise encumbering those items as well as certain actions that may result in change of control if you are a legal entity.
I.	Our approval of transfer by you	Section 12.2	We have the right to approve all transfers, but will not unreasonably withhold approval
m.	Conditions for our approval of transfer	Section 12.2	Transfer fee of \$10,000 paid, new franchisee/transferee satisfactorily completes training, you sign release and settle all accounts with us and trade creditors, no default exists under the current agreement, and bulk-sales laws complied with and updating/refurbishing of your Manchu WOK Restaurant to our then

PROVISION		SECTION IN FRANCHISE AGREEMENT	SUMMARY *
			current standards, and, at our option, transferee will sign our then-current form of franchise agreement (and guarantee) for new franchisees or you will assign existing franchise agreement to transferee plus new guarantees). If you leased the Premises directly from a third party, then the transferee must sign and deliver to us a new Lease Addendum. We reserve the right to require transferring franchisees to use and pay for the services of a third-party escrow agent designated or approved by us.
n.	Our right of first refusal to acquire your business	Section 12.8	We can match any offer for an interest in the Franchise Agreement or your Manchu WOK Restaurant assets, or stock within 30 days of receiving notice of the offer.
0.	Our option to purchase your business	Section 10.6 and	We have the option to purchase your equipment, fixtures, and other assets when your franchise expires or terminates.
p.	Your death or disability	Sections 12.4, 12.6, 12.7	Within 30 days of the death or permanent disability of the franchisee or franchise owner (anyone holding 25% or greater voting share or interest in your Manchu WOK restaurant), that individual's representative must hire a satisfactory replacement; Within six months of the permanent disability or death of the franchisee or franchise owner as defined above, that individual's estate must transfer his or her interests in your Manchu WOK Restaurant (or shares of any corporate owner) or in the Franchise Agreement in accordance with the terms of Section 12. If a satisfactory replacement is not hired within 30 days, Franchisor has the option but not obligation to operate and manage the business for the account of the Franchisee. If a transfer does not occur within six months, or a replacement is not hired, Franchisor may terminate the Agreement upon 10 days' notice.
q.	Non-competition covenants during the term of the franchise	Sections 11.2, and 11.3	No involvement in any restaurant business or retail food outlet that specializes in the sale of Chinese and Asian-style food or that we reasonably consider similar in

	PROVISION	SECTION IN FRANCHISE AGREEMENT	SUMMARY *
			nature, style or otherwise to any Manchu WOK® restaurant or using the Manchu WOK Restaurant System. This also applies to Guarantors and other related parties.
r.	Non-competition covenants after the franchise is terminated or expires	Sections 11.4 and 11.5	For a period of 18 months, no involvement in any restaurant business or retail food outlet located at the Premises, or the premises at which any restaurant or outlet using the Manchu WOK Restaurant System (whether franchisee- or companyowned, or street-front or mall-based) or within 3 miles of the exterior boundaries of any building in which (i) the Premises are located or (ii) any premises at which any restaurant or outlet utilizing the Manchu WOK Restaurant System (whether franchisee-or company-owned, or street-front or mall-based) is then being operated, or is then being constructed or then under contract to be constructed. This also applies to Guarantors and other related parties.
S.	Modification of the agreement	Sections 2.6 and 17.2	Amendments only effective if in writing; but we may change the Manchu WOK Restaurant System, including the Confidential Manual.
t.	Integration/merger clause	Section 17.3 and 17.7	Only the terms of the Franchise Agreement are binding, subject to state law. Any representations or promises made outside of the disclosure document and the franchise agreement may not be enforceable.
u.	Dispute resolution by arbitration or mediation	Section 14	Any dispute that cannot be resolved by, or is not subject to, mediation shall be settled by arbitration administered by the American Arbitration Association in the county and state hwre the Franchised Business is located, before a single arbitrator who is a licensed U.S. attorney with at least five years of franchise law experience. This does not apply to a dispute where we bring an action for an express obligation to pay monies, declaratory relief, preliminary or permanent equitable relief, any action for

PROVISION		SECTION IN FRANCHISE AGREEMENT	SUMMARY*
			dame to Franchisor's goodwill, proprietary information, trademarks, or for fraudulent conduct by this franchisee, or where a delay in the dispute resolution process may adversely affect the public. Disputes must be conducted on an individual basis only. Any issue regarding the validity or applicability of the class action waiver must be decided by a court. If the waiver is deemed unenforceable, arbitration will not apply.*
V.	Choice of forum	Section 14.1	Arbitration takes place in the county and state where the Franchised Business is located. For certain excluded disputes (e.g. IP, money owed, injunctive relief, urgent matters), the Franchisor may bring an action in any court having jurisdiction, and the Franchisee waives all objections to venue and jury trial.
w.	Choice of law	Section 17.13	Except to the extent governed by the United States trademark laws or the franchise laws of any state, the laws of the State where the Franchised Business is located govern all rights obligations of the parties under this Agreement without regard to conflict of law, and any arbitration shall take place before a sole arbitrator in the County of the State where the Franchised Business is located.

Notes:

^{*} Applicable state law may require additional disclosures related to the information in this item. These additional disclosures, if any, appear in an addendum. See **Exhibit H** of this disclosure document.

<u>Sublease</u>

This table lists certain important provisions of the Sublease. You should read these provisions in the agreements attached to this disclosure document.

	Provision	Section in Sublease	Summary
a.	Term of the Sublease	1.1 and 14.2	Ends one day before expiration of the Lease.
b.	Renewal or extension of the term		
C.	Requirements for you to renew or extend		
d.	Termination by you	Not Applicable	You have no right to terminate the Sublease.
e.	Termination by us without cause	11.1 and 12.1	The Sublease will terminate if our Leasing Affiliate cannot obtain necessary written consent of Landlord or if the Lease is terminated
f.	Termination by us with cause	10.1 and 14.1	Our Leasing Affiliate may terminate the Sublease for cause.
g.	"Cause" defined – curable defaults	10.1.1 – 10.1.2	Curable defaults include: (i) failure to pay any amounts due; (ii) failure to comply with any non-monetary covenants, agreements, provisions, or conditions of the Sublease, except for those which have no cure period.
h.	"Cause" defined – defaults that cannot be cured	10.1.3 – 10.1.11	Non-curable defaults include: (i) bankruptcy, insolvency, or similar acts or events; (ii) removal, sale or disposal of goods, merchandise, and/or chattels outside of the ordinary course of business; (iii) failure to continually operate the business pursuant to the Sublease and Franchise Agreement; (iv) seizure of goods or chattels; (v) a receiving order is made or a receiver is appointed; (vi) improper subleasing, assignment, transfer, or encumbrance of the premises or Sublease; (vii) intentional falsification of reports required by the Sublease or Franchise Agreement; (viii) termination of the Franchise Agreement or other applicable agreement; (x) three or more defaults within a 12-consecutive-month period, whether or not cured.

	Provision	Section in Sublease	Summary
i.	Your obligations on termination/nonrenewal	4.1.6 and 10.1.11	Our Leasing Affiliate may enter and take possession of the Premises and remove all persons or property from the Premises; you are not relieved of certain obligations under the Sublease.
j.	Assignment of agreement by us		
k.	"Transfer" by you – defined	7.1	Approval of Leasing Affiliate is required to assign, transfer, sublet, permit others to use or occupy, mortgage, or encumber the Sublease, Premises, or any equipment, fixture, or furniture. If an entity, any change in ownership or control is an assignment.
I.	Our approval of transfer by you	7.1 and 14.5.12	Prior written consent of Leasing Affiliate, at its sole discretion, is required. Approval of landlord may be required.
m.	Conditions for our approval of transfer	7.1	Conditions include: (i) proposed transferee must fully comply with all requirements and conditions of the Franchise Agreement; (ii) You, and Guarantor, must remain jointly and severally liable for all obligations of the Sublease and Franchise Agreement; (iii) proposed transferee must enter into agreement to be bound to the terms and conditions of the Sublease and Franchise Agreement; and (iv) You, Guarantor, and the proposed transferee must jointly and severally indemnify and hold our Leasing Affiliate harmless for related costs. Subject to applicable law, our Leasing Affiliate has arbitrary discretion to approve a proposed transfer.
n.	Our right of first refusal to acquire your business	Not Applicable	None.
0.	Our option to purchase your business	Not Applicable	None.
p.	Your death or disability	Not Applicable	None
q.	Non-competition covenants during the term of the franchise	Not Applicable	None
r.	Non-competition covenants after the franchise is terminated or expires	Not Applicable	None

	Provision	Section in Sublease	Summary
S.	Modification of the agreement	14.5.2	No amendment or modification is valid or binding unless it is in writing, refers specifically to the Sublease, and is signed by the party as to whom enforcement is sought.
t.	Integration/merger clause	14.2 and 14.4	The Sublease and Franchise Agreement, including any contemplated or referenced agreements, contain the entire agreement of the parties. The Lease is incorporated into the Sublease <i>mutatis mutandis</i> .
u.	Dispute resolution by arbitration or mediation	Not Applicable	None
٧.	Choice of forum	Not Applicable	None
W.	Choice of law	Not Applicable	None

ITEM 18 PUBLIC FIGURES

We and our affiliates do not use any public figure to promote our franchise.

ITEM 19 FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying: or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

The average gross sales amounts contained in the table below pertain to the historic performance of all franchised Manchu Wok stores located in the United States that were existing outlets that did not open for the first time during the past fiscal year and reported sales. The time period measured was December 1, 2023, through November 30, 2024.

	Average Gross	Median Gross	Number of	Number of stores that	Percentage of stores	Number of stores that	Percentage of stores
	Sales	Sales	Stores	met or exceeded the Average Gross Sales	that met or exceeded the Average Gross Sales	met or exceeded the Median Gross Sales	that met or exceeded the Median Gross Sales
All franchised stores	\$1,895,019	\$1,124,717	15	8	53%	8	53%

Notes:

- 1. As of November 30, 2024, there were 15 franchised stores open in the United States. 15 stores were franchised, existing locations that did not open for the first time during the past fiscal year, were located in the United States, and were not temporarily closed as of fiscal year end.
- 2. The information is based upon gross sales, which means the total revenue derived from the sales of goods and services less sales tax, discounts, allowances and returns, of franchised, stores.
- 3. Some outlets have sold this amount. Your individual results may differ. There is no assurance that you'll sell as much.
- 4. The above table and information were prepared from (i) weekly sales reports provided to us by franchisees. We have not audited the information provided by the franchisees.
- 5. We will provide to you the spreadsheets as written substantiation for the representations made in this Item 19 upon reasonable request.

Other than the preceding financial performance representation, we do not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing outlet, however, we may provide you with the actual records of that outlet. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Kahala Franchising, L.L.C., 9311 E. Via De Ventura, Scottsdale, Arizona 85258, (480) 362-4800, the Federal Trade Commission, and the appropriate state regulatory agencies.

ITEM 20 OUTLETS AND FRANCHISEE INFORMATION

Table No. 1 Systemwide Outlet Summary For fiscal years 2022-2024

OUTLET TYPE	YEAR	OUTLETS AT THE START OF THE YEAR	OUTLETS AT THE END OF THE YEAR	NET CHANGE
Franchised				
	2022	21	18	-3
	2023	18	15	-3
	2024	15	15	0
Company - Owned				
	2022	0	0	0
	2023	0	0	0
	2024	0	0	0
Total Outlets				
	2022	21	18	-3
	2023	18	15	-3
	2024	15	15	0

Table No. 2
Transfers of Outlets from Franchisees to New Owners (other than the Franchisor)
For fiscal years 2022-2024

STATE	YEAR	NUMBER OF TRANSFERS
Total		
	2022	0
	2023	0
	2024	0

Table No. 3 Status of Franchised Outlets For fiscal years 2022-2024

STATE	YEAR	OUTLE TS AT START OF YEAR	OUTLE TS ADDED	TERMIN ATIONS	NON- RENEW ALS	REACQ UIRED BY FRANC HISOR	CEASE D OPERA TIONS - OTHER REASO NS	OUTLE TS AT END OF THE YEAR
Alabama								
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2024	1	0	0	0	0	0	1
California								
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2024	1	1	0	0	0	0	2
Florida								
	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
	2024	3	0	0	0	0	0	3
Illinois								
	2022	3	0	0	0	0	0	3
	2023	3	0	0	0	0	0	3
	2024	3	0	0	0	0	0	3
Kansas								
	2022	1	0	0	0	0	1	0
	2023	0	0	0	0	0	0	0
	2024	0	0	0	0	0	0	0
Missouri								
	2022	1	0	0	1	0	0	0
	2023	0	0	0	0	0	0	0
	2024	0	0	0	0	0	0	0

Nevada								
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	1	0
	2024	0	0	0	0	0	0	0
New Jersey								
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	1	0
	2024	0	0	0	0	0	0	0
NI V								
New York	2222							4
	2022	1	0	0	0	0	0	1
	2023	1	0	0	0	0	0	1
	2024	1	0	0	0	0	0	1
Texas								
Texas	2022	4	0	0	0	0	0	4
		4	0	0		0	1	
	2023				0			3
	2024	3	0	0	0	0	0	3
Washington								
J	2022	4	0	0	1	0	0	3
	2023	3	0	0	0	0	0	3
	2024	3	0	0	0	0	1	2
-								
Total				_				
	2022	21	0	0	2	0	1	18
	2023	18	0	0	0	0	3	15
	2024	15	1	0	0	0	1	15

Table No. 4
Status of Company-Owned Outlets
For fiscal years 2022-2024

STATE	YEAR	OUTLETS AT START OF YEAR	OUTLETS OPENED	OUTLETS REACQUI RED FROM FRANCHI SEE	OUTLETS CLOSED	OUTLETS SOLD TO FRANCHI SEE	OUTLETS AT END OF THE YEAR
-------	------	-----------------------------------	-------------------	---	-------------------	--------------------------------------	-------------------------------------

Minnesota							
	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2024	0	0	0	0	0	0
Total							
	2022	0	0	0	0	0	0
	2023	0	0	0	0	0	0
	2024	0	0	0	0	0	0

Table No. 5
Projected Openings as of November 30, 2024

State	Franchise Agreements Signed but Outlet Not Opened	Projected New Franchised Outlet in Next Fiscal Year	Projected New Company-Owned Outlet in the Next Fiscal Year	Total
Total	0	0	0	0

The name, address, and telephone number of all of our franchisees as of November 30, 2024, is set forth in **Exhibit F**.

The name and last known address and telephone number of every franchisee who has left our Manchu WOK® system for any reason including, without limitation, termination, cancellation, non-renewal or otherwise voluntarily or involuntarily ceased to do business under a Manchu WOK Franchise Agreement, during our most recently completed fiscal year or has not communicated with us within 10 weeks of the date of this Franchise Disclosure Document appears at **Exhibit G**.

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system. In some instances, current and former franchisees sign provisions restricting their ability to speak openly about their experience with the franchise system. You may wish to speak with current and former franchisees, but be aware that not all such franchisees will be able to communicate with you.

As of November 30, 2024, there are no Manchu WOK franchisee associations in existence regardless of whether or not they use our trademark.

ITEM 21 FINANCIAL STATEMENTS

Attached as **Exhibit B are** our audited consolidated financial statements for the years ended November 30, 2024, 2023, and for the fiscal years ended on November 30, 2023, and 2022.

ITEM 22 CONTRACTS

A copy of MTY Franchising USA, Inc.'s current Franchise Agreement and Sublease Agreement (Sublease subject to amendment to accommodate local laws) are attached as **Exhibits C** and **D**, respectively. A copy of the In-Store Training Release and Waiver of Liability Agreement is attached to this Franchise Disclosure Document as **Exhibit E-1**.

ITEM 23 RECEIPT

Exhibit K contains two detachable pages acknowledging the receipt of the Franchise Disclosure Document by you. One copy is for your records, and one copy must be signed and dated by you and returned to us.



EXHIBIT A

Manchu WOK

LIST OF STATE REGULATORS / AGENTS FOR SERVICE OF PROCESS

EXHIBIT A

LIST OF STATE ADMINISTRATORS AND AGENTS FOR SERVICE OF PROCESS

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
CALIFORNIA	Department of Business Oversight One Sansome Street, Suite 600 San Francisco, CA 94104 415-972-8559 1-866-275-2677	Commissioner of Business Oversight 320 West 4th Street, Suite 750 Los Angeles 90013-2344 1-866-275-2677
CONNECTICUT	Securities and Business Investment Division Connecticut Department of Banking 260 Constitution Plaza Hartford, CT 06103 860-240-8230	Connecticut Banking Commissioner Same Address
FLORIDA	Department of Agriculture & Consumer Services Division of Consumer Services Mayo Building, Second Floor Tallahassee, FL 32399-0800 850-245-6000	Same
GEORGIA	Office of Consumer Affairs 2 Martin Luther King Drive, S.E. Plaza Level, East Tower Atlanta, GA 30334 404-656-3790	Same
HAWAII	State of Hawaii Business Registration Division Securities Compliance Branch Dept. of Commerce and Consumer Affairs 335 Merchant Street, Room 205 Honolulu, HI 96813 808-586-2722	Hawaii Commissioner of Securities Same Address
ILLINOIS	Franchise Division Office of the Attorney General 500 South Second Street Springfield, IL 62706 217-782-4465	Illinois Attorney General Same Address
INDIANA	Securities Commissioner Indiana Securities Division 302 West Washington Street, Room E 111 Indianapolis, IN 46204 317-232-6681	Indiana Secretary of State 201 State House 200 West Washington Street Indianapolis, IN 46204
IOWA	Iowa Securities Bureau Second Floor Lucas State Office Building Des Moines, IA 50319 515-281-4441	Same

STATE	STATE ADMINISTRATOR	AGENT FOR
STATE KENTUCKY	STATE ADMINISTRATOR	SERVICE OF PROCESS Same
KENTUCKT	Kentucky Attorney General's Office Consumer Protection Division	Same
	1024 Capitol Center Drive	
	Frankfort, KY 40602	
	502-696-5389	
LOUISIANA	Department of Urban & Community Affairs	Same
LOUISIANA	Consumer Protection Office	Carrie
	301 Main Street, 6th Floor	
	One America Place	
	Baton Rouge, LA 70801	
	504-342-7013 (gen. info.) 504-342-7900	
MAINE	Department of Business Regulations	Same
1717 XII V III	State House - Station 35	
	Augusta, ME 04333	
	207-298-3671	
MARYLAND	Office of the Attorney General	Maryland Securities Commissioner
MARCI EARL	Securities Division	Same Address
	200 St. Paul Place	Carrie / Idanose
	Baltimore, MD 21202	
	410-576-6360	
MICHIGAN	Michigan Department of Attorney General	Michigan Department of Commerce
	Consumer Protection Division	Corporations and Securities Bureau
	Antitrust and Franchise Unit	Same Address
	525 W. Ottawa Street	
	G. Mennen Williams Building, 1st Floor	
	Lansing, MI 48913	
	517-373-7117	
MINNESOTA	Minnesota Department of Commerce	Minnesota Commissioner of Commerce
	85 7 th Place East, Suite 280	Same Address
	St. Paul, MN 55101	
	651-539-1500	
NEBRASKA	Department of Banking and Finance	Same
	1526 K Street, Suite 300	
	Lincoln, NE 68508	
	P.O. Box 95006	
	Lincoln, Nebraska 68509-5006	
	Tele: 402-471-2171	
NEW HAMPSHIRE	Attorney General	Same
	Consumer Protection and Antitrust Bureau	
	State House Annex	
	Concord, NH 03301	
	603-271-3641	

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
NEW YORK	Office of the New York State Attorney	
	General	One Commerce Plaza
	Investor Protection Bureau	99 Washington Avenue, 6 th Floor
	Franchise Section	Albany, NY 12231-0001
	28 Liberty Street New York, NY 10005	518-473-2492
	212-416-8236 Phone	
	212-416-6042 Fax	
NORTH CAROLINA	Secretary of State's Office/Securities Division	
	2 South Salisbury Street	Secretary of State's Office
	Raleigh, NC 27601	Same Address
	919-733-3924	
NORTH DAKOTA	North Dakota Securities Department	North Dakota Securities Commissioner
	600 East Boulevard Avenue	Same Address
	State Capitol, Fifth Floor	
	Bismarck, ND 58505-0510	
OHIO	701-328-4712; Fax: 701-328-0140 Attorney General	Same
ОПІО	Consumer Fraud & Crime Section	Same
	State Office Tower	
	30 East Broad Street, 15th Floor	
	Columbus, OH 43215	
	614-466-8831 or 800-282-0515	
OKLAHOMA	Oklahoma Securities Commission	Same
	2915 Lincoln Blvd.	
	Oklahoma City, OK 73105	
	405-521-2451	
OREGON	Department of Insurance and Finance	Director
	Corporate Securities Section	Department of Insurance and Finance
	Labor and Industries Building	Same Address
	Salem, OR 96310	
	503-378-4387	
RHODE ISLAND	<u> </u>	Director, Rhode Island Department of
	Regulation	Business Regulation
	Securities Division	Same address
	John O. Pastore Center – Building 69-1	
	1511 Pontiac Avenue Cranston, RI 02920	
	401-222-3048	
SOUTH CAROLINA	Secretary of State	Same
300 III CAROLINA	P.O. Box 11350	Carrie
	Columbia, SC 29211	
	803-734-2166	
SOUTH DAKOTA	Division of Insurance	Director of the South Dakota
	Securities Regulation	Division of Securities
	124 S. Euclid, Suite 104	Same Address
	Pierre, SD 57501	
	605-773-4823	

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF PROCESS
TEXAS	Secretary of State Statutory Documents Section P.O. Box 12887 Austin, TX 78711-2887 512-475-1769	Same
UTAH	Utah Department of Commerce Consumer Protection Division 160 East 300 South (P.O. Box 45804) Salt Lake City, UT 84145-0804 TELE: 801-530-6601 FAX:801-530-6001	Same
VIRGINIA	State Corporation Commission Division of Securities and Retail Franchising Tyler Building, 9 th Floor 1300 E. Main Street Richmond, VA 23219 804-371-9051	Clerk of the State Corporation Commission Tyler Building, 1st Floor 1300 E. Main Street Richmond, VA 23219 804-371-9733
WASHINGTON	Department of Financial Institutions Securities Division 150 Israel Rd S.W. Tumwater, WA 98501 360-902-8762	Director, Dept. of Financial Institutions Securities Division 150 Israel Rd S.W. Tumwater, WA 98501
WISCONSIN	Wisconsin Dept. of Financial Institutions Division of Securities 345 W. Washington Avenue, 4th Floor Madison, WI 53703 608-266-8557	Wisconsin Commissioner of Securities Same Address



EXHIBIT B

Manchu WOK FINANCIAL STATEMENTS

Consolidated financial statements of MTY Franchising USA, Inc.

For the years ended November 30, 2024 and 2023

Independent auditor's report	1-2
Consolidated statements of operations and comprehensive (loss) income	3
Consolidated statements of changes in stockholder's equity	4
Consolidated balance sheets	5-6
Consolidated statements of cash flows	7-8
Notes to the consolidated financial statements	9-34

×



Report of Independent Auditors

To the Management and Stockholder of MTY Franchising USA, Inc.

Opinion

We have audited the accompanying consolidated financial statements of MTY Franchising USA, Inc. and its subsidiaries (the Company), which comprise the consolidated statements of operations and comprehensive (loss) income for the years ended November 30, 2024 and 2023, the consolidated statements of changes in stockholder's equity for the years ended November 30, 2024 and 2023, the consolidated balance sheets as of November 30, 2024 and 2023, and the consolidated statements of cash flows for the years then ended, including the related notes (collectively referred to as the consolidated financial statements).

In our opinion, the accompanying consolidated financial statements present fairly, in all material respects, the financial position of the Company as of November 30, 2024 and 2023, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audit in accordance with auditing standards generally accepted in the United States of America (US GAAS). Our responsibilities under those standards are further described in the Auditors' Responsibilities for the Audit of the Consolidated Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for one year after the date the consolidated financial statements are available to be issued.



Auditors' Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditors' report that includes our opinion. Reasonable assurance is a high level of assurance, but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with US GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with US GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit
 procedures that are appropriate in the circumstances, but not for the purpose of expressing an
 opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is
 expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

Montréal, Canada January 31, 2025

¹ CPA auditor, public accountancy permit No. A125677

Pricewaterhouse Coopers UP

Consolidated statements of operations and comprehensive (loss) income

Years ended November 30, 2024 and 2023 (In thousands of US dollars)

	Notes	2024	2023
	_	\$	\$
			Adjusted ⁽¹⁾
Revenue	19	597,538	606,617
Costs and expenses			
Operating expenses	20	490,300	498,484
Depreciation – property, plant and equipment	7	13,469	11,358
Amortization – intangible assets	8	15,765	19,213
Impairment charge – property, plant and equipment	7 & 10	6,150	169
Impairment charge – operating lease right-of-use assets	6	849	_
Impairment charge – intangible assets and goodwill	8, 9 & 10	37,452	4,063
Interest expense	22	53,768	53,977
Management fees charged by parent company	25	2,101	2,249
	_	619,854	589,513
Other income (expenses)			
Interest income		12,145	2,255
Loss on disposal of property, plant and equipment and intangible assets		(376)	(1,843)
Restructuring	21	(1,342)	_
Gain on extinguishment of debt		97	_
Gain on contingent consideration from a business acquisition	3		1,600
Loss on de-recognition/lease modification of operating lease liabilities	6	(41)	(736)
		10,483	1,276
(Loss) income before income taxes	-	(11,833)	18,380
Income tax expense (recovery)	23		
Current		9,918	7,807
Deferred		(9,195)	(6,405)
	(-	723	1,402
Net (loss) income and comprehensive (loss) income	_	(12,556)	16,978

⁽¹⁾ Certain comparative figures have been reclassified to conform with current year presentation. Lease income in comparative period of \$26.3 million was reclassed from operating expense to revenue.

Consolidated statements of changes in stockholder's equity

Years ended November 30, 2024 and 2023

(In thousands of US dollars, except number of common stock issued)

	Common stock issued	Common stock value	Retained earnings	Total stockholder's equity
	-	\$	\$	\$
Balance as at November 30, 2022	15	179,154	56,804	235,958
Net income and comprehensive income		***************************************	16,978	16,978
Balance as at November 30, 2023	15	179,154	73,782	252,936
Net loss and comprehensive loss		-	(12,556)	(12,556)
Balance as at November 30, 2024	15	179,154	61,226	240,380

MTY Franchising USA, Inc. Consolidated balance sheets

As at November 30, 2024 and 2023 (In thousands of US dollars)

		2024	2023
	Notes	\$	\$
Assets			
Current assets			
Cash		12,834	21,138
Restricted cash	2	250	271
Accounts receivable	4	28,755	29,461
Inventories		7,182	6,414
Assets held for sale	5, 7 & 8	3,116	1,668
Current portion of loans receivable		384	272
Receivable from ultimate parent	13 & 25	204,389	148,828
Prepaid expenses and deposits		8,039	8,386
Other current assets		4,834	3,333
Income taxes receivable	23	4,402	4,919
		274,185	224,690
Loans receivable		66	151
Contract cost asset		4,277	3,696
Other assets		2,106	1,896
Property, plant and equipment	7	53,675	63,599
Operating lease right-of-use assets	6	190,303	187,074
Intangible assets	8	577,327	602,168
Goodwill	9	346,490	376,175
		1,448,429	1,459,449
Liabilities			
Current liabilities			
Accounts payable		14,808	15,878
Accrued liabilities		26,837	28,769
Gift card liability	11	109,252	104,858
Promotional funds payable		10,523	14,690
Current portion of operating lease liabilities	6	38,340	37,080
Current portion of deferred revenue and deposits	12	8,093	7,775
Advance from parent company	13 & 25	4,417	2,249
Advances from companies under common control	13 & 25		42
Current portion of holdback payable	15		912
, ,		212,270	212,253

MTY Franchising USA, Inc. Consolidated balance sheets (continued)

As at November 30, 2024 and 2023

(In thousands of US dollars)

		2024	2023
	Notes	\$	\$
Liabilities (continued)			
Long-term loan from company under common control	14	708,363	705,683
Other liabilities		_	159
Liabilities held for sale	5, 7 & 8	2,116	
Operating lease liabilities	6	156,073	153,413
Deferred revenue and deposits	12	28,170	24,753
Deferred income taxes	23	101,057	110,252
		1,208,049	1,206,513
Stockholder's equity			
Common stock	17	179,154	179,154
Retained earnings		61,226	73,782
		240,380	252,936
		1,448,429	1,459,449

Approved by the Board on January 31, 2025	
	, Directo

Consolidated statements of cash flows

Years ended November 30, 2024 and 2023 (In thousands of US dollars)

		2024	2023
	Notes	\$	\$
Occupation and Man			
Operating activities			
Net (loss) income and comprehensive (loss) income		(12,556)	16,978
Items not affecting cash:			
Depreciation – property, plant and equipment	7	13,469	11,358
Amortization – intangible assets	8	15,765	19,213
Interest expense	22	53,768	53,977
Loss on disposal of property, plant and equipment and intangible assets		376	1,843
Impairment charge – property, plant and equipment	7 & 10	6,150	169
Impairment charge – operating lease right-of-use assets	6	849	:
Impairment charge – intangible assets and goodwill	8 & 10	37,452	4,063
Gain on contingent consideration from a business acquisition	3		(1,600)
Loss on de-recognition/lease modification of operating lease liabilities	6	41	736
Deferred income tax recovery		(9,195)	(6,405)
		106,119	100,332
Interest paid		(53,768)	(53,977)
Changes in non-cash working capital items		11 11 11 11 11	
Accounts receivable		2,874	(4,133)
Inventories		(925)	198
Prepaid expenses and deposits		(278)	(1,140)
Loans receivable		(27)	1,410
Other current assets		(1,502)	(412)
Income taxes		(303)	(1,842)
Accounts payable		(1,070)	(911)
Accrued liabilities		(1,817)	(5,853)
Promotional funds payable		(4,166)	(2,045)
Gift card liability		4,546	7,521
Deferred revenue and deposits		3,735	1,088
Other		713	1,512
Net cash provided from operating activities	ļ .	54,131	41,748

Supplemental cash flow information

Consolidated statements of cash flows (continued)

Years ended November 30, 2024 and 2023 (In thousands of US dollars)

		2024	2023
	Notes	\$	\$
Investing activities			
Net cash outflow on acquisitions	3		(221,180)
Net cash acquired through business acquisitions	3		6,884
Additions to property, plant and equipment	7	(10,963)	(13,250)
Additions to intangible assets	8	(318)	(565)
Proceeds on disposal of intangible assets	7	230	_
Proceeds on disposal of property, plant and equipment		692	358
Proceeds on disposal of assets held for sale		1,614	22-0
Net cash used in investing activities		(8,745)	(227,753)
Financing activities			
Receivable from ultimate parent and parent company	13	(55,562)	(221,985)
Receivable from companies under common control		2,638	405,836
Repayment of holdback payable	15	(787)	:
Net cash (used in) provided from financing activities	I	(53,711)	183,851
Net decrease in cash		(8,325)	(2,154)
Cash, beginning of year		21,409	23,563
Cash, end of year	1	13,084	21,409

24

The accompanying notes are an integral part of the consolidated financial statements.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

1. Nature of operations

MTY Franchising USA, Inc. (the "Company" or "MTY USA") was incorporated on March 14, 2001. The Company operates, develops and franchises restaurants under a multitude of different banners in the United States of America (the "US").

2. Significant accounting policies

Basis of presentation

The accounting policies of the Company are in accordance with accounting principles generally accepted in the US ("US GAAP"). The Company uses the US dollar as its functional and reporting currency, and tabular amounts are rounded to the nearest thousand (\$000) except when otherwise indicated. MTY USA is a wholly owned subsidiary of MTY Franchising Inc. which is a subsidiary of the ultimate parent of MTY Food Group Inc.

The preparation of the consolidated financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from those estimates.

Presented below are those policies considered particularly significant:

Basis of consolidation

The consolidated financial statements reflect the financial position and operating results of the Company, including wholly owned subsidiaries and investees that we control.

The principal subsidiaries of the Company are as follows:

Principal subsidiaries	Percentage of equity interest
	%
BF Acquisition Holdings, LLC	100
BQ Concepts, LLC	100
Famous Dave's of America Inc.	100
Kahala Franchising, LLC	100
La Salsa Franchise, LLC	100
Papa Murphy's International, LLC	100
VI Brandco, LLC	100
Wetzel's Pretzels, LLC (Note 3)	100

Revenues and expenses of subsidiaries are included in the consolidated statement of operations and comprehensive income (loss) from the effective date of acquisition. The subsidiaries are consolidated from the acquisition date until the date on which the Company ceases to control them.

All intercompany transactions, balances, revenues and expenses are eliminated in full upon consolidation.

Business combinations

The Company accounts for acquired businesses using the acquisition method of accounting in accordance with Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") 805, Business Combinations. The consideration transferred for the acquisition is the fair values of the assets transferred, the liabilities incurred, and the equity interest issued. The consideration transferred includes the fair value of any asset or liability resulting from a contingent consideration arrangement. Acquisition-related costs are expensed as incurred. Identifiable assets acquired and liabilities and contingent liabilities assumed in a business combination are measured initially at their fair value at the acquisition date.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Business combinations (continued)

Goodwill is measured as the excess of the purchase price over the estimated fair values of the net assets acquired. If, after reassessment, the net of the acquisition date amounts of the identifiable assets acquired and liabilities assumed exceeds the sum of the consideration transferred, the amount of any non-controlling interests in the acquiree and the fair value of the acquirer's previously held interest in the acquiree (if any), the excess is recognized immediately in the statement of operations as a bargain purchase gain.

Goodwill reflects how the acquisition will impact the Company's ability to generate future profits in excess of existing profits. The consideration paid mostly relates to combined synergies, related mainly to revenue growth. These benefits are not recognized separately from goodwill as they do not meet the recognition criteria for identifiable intangible assets.

When the consideration transferred by the Company in a business combination includes assets or liabilities resulting from a contingent consideration arrangement, the contingent consideration is measured at its acquisition date fair value and included as part of the consideration transferred in a business combination.

When a business combination is achieved in stages, the Company's previously held equity interest in the acquiree is remeasured to fair value at the acquisition date (i.e. the date when the Company obtains control) and the resulting gain or loss, if any, is recognized in the statement of operations. Amounts arising from interests in the acquiree prior to the acquisition date that have previously been recognized in income and other comprehensive income are reclassified to the statement of operations where such treatment would be appropriate if that interest were disposed of.

If the initial accounting for a business combination is incomplete by the end of the reporting period in which the combination occurs, the Company reports provisional amounts for the items for which the accounting is incomplete.

Functional currency

The functional currency of the Company and its subsidiaries is the US dollar. The Company translates monetary assets and liabilities that are denominated in currencies other than the US dollar at the exchange rates prevailing at the end of the reporting period; non-monetary assets denominated in foreign currencies are translated using the exchange rate prevailing at the transaction date; all revenue and expense items denominated in foreign currencies are translated at the exchange rate prevailing at the transaction date. All foreign exchange gains and losses are reported in the statement of operations.

Revenue recognition

Revenue is recognized upon the transfer of control of promised goods or services to customer in an amount that reflects the consideration the Company expects to receive for those goods or services:

Revenue from franchise locations

- Royalties are based either on a percentage of gross sales as reported by the franchisees or on a fixed monthly fee. They are recognized on an accrual basis in accordance with the substance of the relevant agreement, as they are earned.
- ii) Promotional fund contributions are based on a percentage of gross sales as reported by the franchisees. Corresponding promotional fund transfers are presented directly on the consolidated balance sheets. The Company is not entitled to retain these promotional fund payments received and is obligated to transfer these funds to be used solely for use in promotional and marketing-related costs for specific restaurant banners. The Company sometimes charges a fee for the administration of the promotional funds.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Revenue recognition (continued)

Revenue from franchise locations (continued)

- iii) Initial franchise fees are recognized on a straight-line basis over the term of the franchise agreement as the performance obligation relating to franchise rights is fulfilled. Amortization begins once the restaurant has opened.
- iv) Upfront fees related to master license agreements are recognized over the term of the master license agreements on a straight-line basis.
- v) Renewal fees and transfer fees are recognized on a straight-line basis over the term of the related franchise agreement.
- vi) The Company earns rent revenue on certain leases it holds; the Company's policy is described below.
- vii) Revenue from equipment sale and retail sales are recognized upon transfer of control, generally upon shipment of the equipment or goods. This revenue is recorded in resale material and retail sales.
- viii) The Company recognizes breakage income proportionately as each gift card is redeemed, based on the historical redemption patterns of the gift cards. The Company also charges various program fees to its franchisees as gift cards are redeemed. Notably, this does not apply to gift card liabilities assumed in a business acquisition, which are accounted for at fair value at acquisition date.
- ix) The Company receives considerations from certain suppliers. Fees are generally earned based on the value of purchases during the period. Agreements that contain an initial upfront fee, in addition to ongoing fees, are recognized on a straight-line basis over the term of the respective agreement. Supplier contributions are recognized as revenue as they are earned and are recorded in franchising revenue.
- x) The Company earns e-commerce fees, which includes point-of-sale ("POS") support fees and transaction fees for purchase made through one of the Company's brands' e-commerce platforms. POS supports fees are received quarterly in advance and are recognized over the period they cover. Transaction fees are recognized when the food items purchased from a store are delivered or picked up by customers.

Revenue from corporate-owned locations

Revenue from corporate-owned locations is recorded when goods are delivered to customers.

Contract cost asset

The Company recognizes incremental costs of obtaining a contract as an asset if they are expected to be recoverable, unless their amortization period would be less than one year, in which case they are expensed as incurred. The costs are amortized to operating expenses over the term of the related franchise agreement.

Leasing

In accordance with ASC 842, the Company determines if an arrangement is or contains a lease at contract inception and recognizes a right-of-use asset and a lease liability at the lease commencement date. Leases with an initial term of 12 months or less but greater than one month are not recorded on the balance sheet for select asset classes.

The lease liability is measured at the present value of future lease payments as of the lease commencement date. The right-of-use asset recognized is based on the lease liability adjusted for prepaid and deferred rent and unamortized lease incentives. An operating lease right-of-use asset is amortized on a straight-line basis over the lease term and is recognized as a single lease cost against the operating lease liability. A finance lease right-of-use asset is amortized on a straight-line basis, with interest costs reported separately, over the lesser of the useful life of the leased asset or lease term.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Leasing (continued)

Operating lease expense is recognized on a straight-line basis over the lease term and is included in Operating expenses. Variable lease payments are expensed as incurred. The Company uses its incremental borrowing rates as the discount rate for its leases, which is equal to the rate of interest the Company would have to pay on a collateralized basis to borrow an amount equal to the lease payments under similar terms. The lease terms for all the Company's leases include the contractually obligated period of the leases, plus any additional periods covered by Company options to extend the leases that the Company is reasonably certain to exercise. Certain leases provide that the lease payments may be increased annually based on the fixed rate terms or adjustable terms such as the Consumer Price Index. Future base rent escalations that are not contractually quantifiable as of the lease commencement date are not included in the lease liability.

Lease expense is comprised of operating and finance lease costs, short-term lease costs, and variable lease costs, which primarily include common area maintenance, real estate taxes, and insurance for the Company's real estate leases.

The Company enters into leases for franchised and corporately-owned locations, offices, and equipment in the normal course of business.

The Company as lessee

The Company recognizes operating lease liabilities with corresponding operating lease right-of-use assets, except for short-term leases and leases of low value assets, which are expensed on a straight-line basis over the lease term. The Company's leases are all classified as operating leases. The amortization of the operating lease right-of-use asset and interest expense related to the operating lease liability are recorded together as the lease expense to produce a straight-line recognition effect in the consolidated statement of operations. Under ASC 842, operating lease right-of-use assets are tested for impairment in accordance with ASC 360, Property, Plant and Equipment.

The Company as lessor

When the Company enters into a sublease arrangement as an intermediate lessor, the Company accounts for the head lease and the sublease as two separate contracts. All the subleases of the Company are classified as operating subleases by reference to the operating lease right-of-use asset arising from the underlying asset. For operating subleases, the Company recognizes an operating right-of-use asset relating to the head lease and recognizes a deferred rent asset or liability in the sublease. As the intermediate lessor, the Company retains the operating lease liability on the head lease in its consolidated balance sheet. During the term of the sublease, the Company recognizes both lease income on the sublease and lease expense on the head lease.

Income taxes

The Company accounts for income taxes pursuant to ASC 740, Income Taxes ("ASC 740"). Deferred tax assets and liabilities are recorded for differences between the financial statement and tax basis of the assets and liabilities that will result in taxable or deductible amounts in the future based on enacted tax laws and rates. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be realized. Income tax expense is recorded for the amount of income tax payable or refundable for the period increased or decreased by the change in deferred tax assets and liabilities during the period.

The Company recognizes the financial statement benefit of a tax position only after determining that the relevant tax authority would more likely than not sustain the position following an audit. For tax positions meeting the more-likely-than-not threshold, the amount recognized in the financial statements is the largest benefit that has a greater than 50 percent likelihood of being realized upon ultimate settlement with the relevant tax authority.

Expected Credit loss

The Company currently uses an expected credit loss ("ECL") model for its trade receivables, which permits the uses of the lifetime expected loss provision for all trade receivables and also incorporates forward-looking information. Lifetime ECL represents the ECL that will result from all probable default events over the expected life of a financial instrument.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Assets held for sale

Assets are classified as held for sale when management with the appropriate authority commits to a plan to sell the assets, the assets are available for immediate sale, the assets are actively marketed at a reasonable price, the sale is probable within a year, and certain other criteria met. Assets held for sale primarily include Company-owned stores and, in some instances, associated trademarks and right-of-use assets, when the Company has committed to their sale as part of an approved plan. Assets designated as held for sale are held at the lower of the net book value or fair value less costs to sell. Depreciation is not charged against property, plant and equipment or right-of-use assets classified as assets held for sale.

Property, plant and equipment

Land and buildings held for use in the production or supply of goods or services, or for administrative purposes, are stated in the consolidated balance sheet at their historical costs less accumulated depreciation (buildings) and accumulated impairment losses. Cost includes expenditures that are directly attributable to the acquisition of the asset, including any costs directly attributable to bringing the asset to a working condition for its intended use.

Equipment, leasehold improvements, rolling stock and computer hardware are stated at cost less accumulated depreciation and accumulated impairment losses.

Depreciation is recognized so as to write off the cost or valuation of assets less their residual values over their useful lives, using the straight-line method. The estimated useful lives, residual values and depreciation methods are reviewed at the end of each year, with the effect of any changes in estimate accounted for on a prospective basis.

An item of property, plant and equipment is derecognized upon disposal or when no future economic benefits are expected to arise from the continued use of the asset. Any gain or loss arising on the disposal or retirement of an item of property, plant and equipment is determined as the difference between the sales proceeds and the carrying amount of the asset and is recognized in the statement of operations.

Depreciation is based on the following terms:

Buildings Straight-line 25 to 50 years
Equipment Straight-line Three to 10 years
Leasehold improvements Straight-line Term of the lease
Rolling stock Straight-line Five to seven years
Computer hardware Straight-line Three to seven years

Intangible assets

Intangible assets acquired separately

Intangible assets with finite useful lives that are acquired separately are carried at cost less accumulated amortization and accumulated impairment losses, if applicable. Amortization is recognized on a straight-line basis over their estimated useful lives. The estimated useful lives and amortization methods are reviewed at the end of each year, with the effect of any changes in estimate being accounted for on a prospective basis. Intangible assets with indefinite useful lives that are acquired separately are carried at cost less accumulated impairment losses, if applicable.

Intangible assets acquired in a business combination

Intangible assets acquired in a business combination and recognized separately from goodwill are initially recognized at their fair value at the acquisition date.

Subsequent to initial recognition, intangible assets having a finite life acquired in a business combination are reported at cost less accumulated amortization and accumulated impairment losses, if applicable, on the same basis as intangible assets that are acquired separately. Intangible assets having an indefinite life are not amortized and are therefore carried at cost reduced by previous impairment losses, if applicable.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Intangible assets (continued)

Derecognition of intangible assets

An intangible asset is derecognized on disposal, or when no future economic benefits are expected from use or disposal. Gains or losses arising from derecognition of an intangible asset, measured as the difference between the net disposal proceeds and the carrying amount of the asset, are recognized in the statement of operations when the asset is derecognized.

The Company currently carries the following intangible assets in its books:

Franchise rights

The franchise rights acquired through business combinations are recognized at the fair value of the estimated future cash inflows related to the acquisition of franchises. The franchise rights are generally amortized on a straight-line basis over the term of the agreements which typically range between 10 to 20 years.

Trademarks

Trademarks acquired through business combinations are recognized at their fair value at the time of the acquisition and are not amortized. Trademarks are determined to have an indefinite useful life based on their brand recognition and their ability to generate revenue through changing economic conditions with no foreseeable time limit.

Other

Included in other intangible assets are purchased software and liquor licences, which are being amortized over their expected useful life on a straight-line basis.

Impairment of long-lived assets other than goodwill

The Company continually reviews whether events or circumstances subsequent to the acquisition of any long-lived assets, including intangible assets with finite and infinite useful lives, have occurred that indicate the remaining estimated useful lives of those assets may warrant revision or that the remaining balance of those assets may not be recoverable. If events and circumstances indicate that the long-lived assets should be reviewed for possible impairment, the Company uses projections to assess whether future cash flows on an undiscounted basis related to the assets exceed the recorded carrying amount of those assets to determine if an asset is impaired. Should an impairment be identified, a loss would be recorded to the extent that the carrying value of the impaired assets exceeds their fair values as determined by valuation techniques appropriate in the circumstances that could include the use of similar cash flow projections on a discounted basis. The reporting units of indefinite intangible assets are individual brands, comprised of franchise rights, trademarks, and perpetual licenses.

Impairment of goodwill

For the purposes of impairment testing, goodwill is allocated to the unit or group of units ("reporting unit") that are considered to represent the lowest level within the group at which the goodwill is monitored for internal management purposes. As of September 1, 2024, the Company reassessed the reporting units based on a strategic realignment and merger of business units in the US. With the change in management team and the overall change in decision making and overall synergies brought from the merger of these units, it was determined that the merger US Goodwill Unit A and D would be merged and that US Goodwill Unit B and C would be merged.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Impairment goodwill (continued)

As at November 30, 2024, goodwill is allocated as follows:

Goodwill unit description

	2024	2023
US Goodwill Unit A	A group of reporting units comprised of all the brands, excluding Papa Murphy's and the brands acquired with BBQ Holdings, Inc. ("BBQ Holdings")	A group of reporting units comprised of all the brands, excluding Papa Murphy's and the brands acquired with BBQ Holdings, Inc. ("BBQ Holdings")
US Goodwill Unit B	A group of reporting units comprised of the BBQ Holdings brands and the Papa Murphy's brand	One reporting unit comprised of Papa Murphy's brand
US Goodwill Unit C		A group of reporting units comprised of the BBQ Holdings brands
US Goodwill Unit D		One reporting unit comprised of Wetzel's Pretzels brand

Impairment of goodwill (continued)

Goodwill is tested for impairment on an annual basis (September 1 for the Company) and between annual tests if an event occurs or circumstances change that would more likely than not reduce the fair value of a reporting unit below its carrying value. Fair value is determined using a discounted cash flow methodology with a risk adjusted weighted average cost of capital.

Cash and restricted cash

Cash and restricted cash include cash on hand and short-term investments, if any, with maturities upon acquisition of generally three months or less or that are redeemable at any time at full value and for which the risk of a change in value is not significant. As at November 30, 2024, cash and restricted cash included \$250 of restricted cash (2023 – \$271) that is required as part of guarantees on certain lease commitments.

Inventories

Inventories are measured at the lower of cost and net realizable value. Costs of inventories are determined on a first-in-first-out basis and include acquisition costs and other costs incurred to bring inventories to their present location and condition.

Market value represents the current replacement cost, provided that the cost does not exceed the net realizable value or is not less than the net realizable value reduced by a normal profit margin.

Contingencies

Litigation, disputes and closed stores

Provisions for the expected cost of litigation, disputes and the cost of settling leases for closed stores, with the exception of operating lease liabilities already recorded pursuant to ASC 842, are recognized when it becomes probable the Company will be required to settle the obligation, at management's best estimate of the expenditure required to settle the Company's obligation.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Contingencies (continued)

Contingent liabilities acquired in a business combination

Contingent liabilities acquired in a business combination are initially measured at fair value at the acquisition date. At the end of subsequent reporting periods, such contingent liabilities are measured at the higher of the amount that would be recognized, and the amount initially recognized.

Financial instruments

The Company's financial instruments consist of cash, restricted cash, accounts receivable, loans receivable, receivable from company under common control, receivable from ultimate parent, accounts payable, accrued liabilities, promotional funds payable, advance from parent company, advances from companies under common control, long-term loan from company under common control and holdbacks payable. Unless otherwise noted, it is management's opinion that the Company is not exposed to significant interest, currency or credit risks arising from these financial instruments. The fair values for cash, restricted cash, accounts receivable, loans receivable, receivable from company under common control, receivable from ultimate parent, accounts payable, accrued liabilities, promotional funds payable, advance from parent company and advances from companies under common control approximate their carrying values due to their immediate or short-term maturities, unless otherwise noted. The long-term loan from company under common control, advances from ultimate parent and holdbacks payable are measured at amortized cost using the effective interest method.

Promotional funds

Pursuant to the franchise agreements, franchisees must pay a fee to the promotional funds. These amounts are collected by the Company in its capacity as agent and must be used for promotional and advertising purposes, since the amounts are set aside to promote the respective banners for the franchisees' benefit. The promotional funds collected, and the related expenditures are reported on a gross basis in the consolidated statements of operations and comprehensive income. To the extent that promotional funds received exceed the related promotional expenditures, the excess contributions will be recorded in accounts payable or accrued liabilities.

Cash held pursuant to the promotional funds received are classified as unrestricted cash as there are no legal restrictions on the use of these funds; however, the Company intends to use these funds solely to support the promotional funds rather than to fund its ongoing operations. As at November 30, 2024, promotional funds were in a net liability position amounting to \$7,694 (2023 – net liability position of \$11,357).

Subsequent events

Subsequent events were evaluated through the date that the consolidated financial statements were issued, which was January 31, 2024.

Estimates and assumptions

Business combinations

For business combinations, the Company must make assumptions and estimates to determine the purchase price accounting of the business being acquired. To do so, the Company must determine the acquisition date fair value of the identifiable assets acquired, including such intangible assets as franchise rights and master franchise rights, trademarks, step-in rights and liabilities assumed. Among other things, the determination of these fair market values involves the use of discounted cash flow analyses and future system sales growth. Goodwill is measured as the excess of the fair value of the consideration transferred including the recognized amount of any non-controlling interest in the acquiree over the net recognized amount of the identifiable assets acquired and liabilities assumed, all measured at the acquisition date. These assumptions and estimates have an impact on the asset and liability amounts recorded in the consolidated balance sheets on the acquisition date. In addition, the estimated useful lives of the acquired amortizable assets, the identification of intangible assets and the determination of the indefinite or finite useful lives of intangible assets acquired will have an impact on the Company's future statement of operations.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

2. Significant accounting policies (continued)

Estimates and assumptions (continued)

Goodwill and indefinite-lived intangible assets

The fair value calculation includes estimates of revenue growth, which are based on past performance and internal projections for the intangible asset group's forecasted growth, and royalty rates, which are adjusted for our particular facts and circumstances. The discount rate is selected based on the estimated cost of capital that reflects the risk profile of the related business. These estimates are highly subjective, the ability to achieve the forecasted cash flows used in our fair value calculations is affected by factors such as the success of strategic initiatives, changes in economic conditions, changes in our operating performance and changes in our business strategies.

Contingencies

The Company makes assumptions and estimations based on its current knowledge of future disbursements it will have to make in connection with various events that have occurred in the past and for which the amount to be disbursed and the timing of such disbursement are uncertain at the date of producing its consolidated financial statements. This includes contingencies for onerous contracts, litigations and disputes and other contingencies.

Gift card liabilities

Management is required to make certain assumptions in both the prorated recognition based on redemption pattern and remoteness recognition of gift card breakage. The significant estimates are breakage rate and the redemption patterns.

3. Business acquisitions

I) Sauce Pizza and Wine (2023)

On December 15, 2022, the Company completed the acquisition of the assets of Sauce Pizza and Wine, an operator of fast casual restaurants operating in the state of Arizona in the US. As of the date of the acquisition, Sauce Pizza and Wine was operating 13 corporate-owned restaurants. The purpose of the transaction was to diversify the Company's range of offerings in the US.

The transaction included a purchase price totaling \$11,165 and a holdback on acquisition of \$798, as detailed below. The resulting aggregate cash outflow in connection with the Sauce Pizza and Wine acquisition was \$9,927.

	2023
	\$
Consideration paid:	
Purchase price	11,165
Working capital	(401)
Cash	23
Discount on non-interest-bearing holdback	(39)
Total consideration	10,748
Cash	(23)
Holdback	(798)
Net consideration paid/cash outflow	9,927

3. Business acquisitions (continued)

I) Sauce Pizza and Wine (2023) (continued)

The final purchase price allocation is as follows:

	2023
Net assets acquired:	\$
Current assets	
Cash	23
Inventories	183
Prepaid expenses and deposits	193
	399
Property, plant and equipment	3,821
Operating lease right-of-use assets	7,262
Intangible assets – Trademark	4,140
Goodwill (1)	3,658
	19,280
Current liabilities	
Accrued liabilities	78
Gift card liability	1,086
Current portion of operating lease liabilities	1,218
	2,382
Operating lease liabilities	6,150
	8,532
Net purchase price	10,748

⁽¹⁾ Goodwill is deductible for tax purposes.

Total expenses incurred related to acquisition costs amounted to \$160.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

3. Business acquisitions (continued)

II) Wetzel's Pretzels (2023)

On December 8, 2022, the Company completed the acquisition of all of the issued and outstanding shares of COP WP Parent, Inc. ("Wetzel's Pretzels"), a franchisor and operator of quick service restaurants operating in the snack category across 25 states in the US, as well as in Canada and Panama. As of the date of the acquisition, Wetzel's Pretzels was operating 328 franchised and 38 corporate-owned restaurants. The purpose of the transaction was to diversify the Company's range of offerings in the US.

The transaction included a purchase price totaling \$210,189, as detailed below. The resulting aggregate cash outflow in connection with the Wetzel's Pretzels acquisition was \$203,328. The transaction consideration also includes \$3,000 held in escrow contingent on the execution of several lease contracts within 12 months of acquisitions. As of December 8, 2023, only a portion of the contracts were executed and therefore \$1,600 was released from escrow and recorded as a gain in the statement of profit and loss.

2023
210,189
210,189
(6,861)
203,328

3. Business acquisitions (continued)

II) Wetzel's Pretzels (2023) (continued)

The final purchase price allocation is as follows:

	2023
Net assets acquired:	\$
Current assets	
Cash	6,861
Accounts receivable	1,005
Inventories	265
Current portion of loans receivable	45
Prepaid expenses and deposits	757
Income taxes receivable	1,371
	10,304
Loans receivable	594
Property, plant and equipment	5,082
Operating lease right-of-use assets	21,931
Intangible assets – Franchise rights	35,600
Intangible assets – Trademarks	71,700
Goodwill (1)	118,447
	263,658
Current liabilities	
Accounts payable	911
Accrued liabilities	4,979
Promotional funds payable	431
Current portion of operating lease liabilities	936
Current portion of deferred revenue and deposits	67
Income taxes payable	547
	7,871
Operating lease liabilities	20,995
Deferred revenue and deposits	939
Deferred income taxes	23,664
	53,469
Net purchase price	210,189

⁽¹⁾ Goodwill is deductible for tax purposes.

Total expenses incurred related to acquisition costs amounted to \$320.

4. Accounts receivable

Details of accounts receivable are as follows:

	2024	2023
	\$	\$
Total accounts receivable	30,676	31,738
Less: Allowance for credit losses	(1,921)	(2,277)
Total accounts receivable, net	28,755	29,461
Of which:		
Not past due	23,612	25,787
Past due for more than one day but no more than 30 days	1,736	430
Past due for more than 31 days but no more than 60 days	695	208
Past due for more than 61 days	2,712	3,036
Total accounts receivable, net	28,755	29,461
	2024	2023
	, , , , , , , , , , , , , , , , , , ,	\$
Allowance for credit losses, beginning of year	(2,277)	(2,013)
Recovery (provision)	322	(733)
Additions through business acquisition		(374)
Reversal of amounts previously written off	(33)	(2)
Write-offs	67	845
Allowance for credit losses, end of year	(1,921)	(2,277)

5. Assets and liabilities held for sale

Assets and liabilities held for sale as at November 30, 2024 and 2023 are stated at fair value less costs to sell. During the reporting period, the Company designated certain assets and liabilities related to a casual dining brand as held for sale. This decision reflects management's formal commitment to a plan to divest these assets, which include both Company-owned stores and associated intangible assets.

The assets reclassified as held for sale primarily consist of inventories, prepaid expenses, deposits, right-of-use assets (Note 6), property plant, and equipment (Note 7) and intangible assets (Note 8). The liabilities included in the carrying value are the gift card payable and the lease liability (Note 6). This reclassification on November 30, 2024 led to an impairment charge of \$849 to right-of-use-asset, \$505 to property, plant and equipment and \$1,088 to trademarks in Reporting Unit A.The total carrying amount reclassified as held for sale is comprised of assets of \$3,116 and liabilities of \$2,116 resulting in a net amount of \$1,000.

In November 2023 assets held for sale comprised of one locations leasehold improvement, land and building that were transferred from property, plant and equipment (Note 7). They did not meet the definition of assets held for sale at the acquisition date of BBQ Holdings.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

6. Leases

Operating lease right-of-use assets

The following table provides the net carrying amounts of the operating lease right-of-use assets by class of underlying asset and the changes in the years ended November 30, 2024 and 2023:

	Offices and stores	Other	Total
	\$	\$	\$
Balance as at November 30, 2022	165,208	433	165,641
Additions	7,115	160	7,275
Additions through business acquisitions (Note 3)	29,274	-	29,274
Depreciation expense	(40,972)	(278)	(41,250)
De-recognition/lease modifications	26,139	(5)	26,134
Balance as at November 30, 2023	186,764	310	187,074
Additions	12,844	90	12,934
Assets held for sale (Note 5)	(953)		(953)
Depreciation expense	(39,781)	(139)	(39,920)
Impairment charge	(849)	** <u>*</u>	(849)
De-recognition/lease modifications	32,047	(30)	32,017
Balance as at November 30, 2024	190,072	231	190,303

The Company recorded sublease income from its operating lease right-of-use assets amounting to \$23,431 (2023 – \$24,789).

6. Leases (continued)

Operating lease liabilities

The following table provides the net carrying amounts of the operating lease liabilities and the changes in the years ended November 30, 2024 and 2023:

	2024	2023
	\$	\$
Operating lease liabilities, beginning of year	190,493	167,716
Additions	12,934	7,284
Additions through business acquisitions (Note 3)	-	29,380
Transfer to assets held for sale (Note 5)	(1,848)	-
Lease renewals and modifications	35,264	29,624
Lease terminations	(3,316)	(3,554)
Other adjustments	(329)	(201)
Interest expense	10,731	10,449
Payments	(49,516)	(50,205)
Operating lease liabilities, end of year	194,413	190,493
Recorded in the consolidated balance sheets as follows:		
	2024	2023
	\$	\$
Current portion	38,340	37,080
Long-term portion	156,073	153,413
	194,413	190,493

Maturity analysis

The following table sets out a maturity analysis of lease payments, showing the undiscounted lease payments to be paid after November 30, 2024:

	Leases	Expected sublease income
	\$	\$
2025	49,565	22,430
2026	43,786	18,679
2027	37,267	14,647
2028	30,325	10,977
2029	22,736	6,760
Thereafter	50,978	10,130
Total undiscounted lease payments	234,657	83,623
Less: Unearned finance income	(40,244)	_
Total present value of lease liabilities and expected sublease income	194,413	83,623

7. Property, plant and equipment

Cost	Equipment	Leasehold improve- ments	Rolling stock	Computer hardware	Land	Building	Total
Cost	Equipment \$	ments \$	SLOCK \$	naroware \$	Land \$	Sullaing \$	10tai
	Ψ	Ψ	Ψ	•	v	J	Ψ
Balance as at November 30,	0.4.000						
2022	21,299	26,495	24	3,998	3,145	3,509	58,470
Additions	6,215	5,978	-	1,007		50	13,250
Additions through business acquisition (Note 3)	2,014	6,242	(8)	(89)	75	(314)	7,920
Disposals	(578)	(155)	_	(125)	_	_	(858)
Impairment (Note 10)	(139)	(16)		(14)	-	-	(169)
Balance as at November 30, 2023	28,811	38,544	16	4,777	3,220	3,245	78,613
Additions	5,100	4,427		1,369	0,220	67	10,963
Disposals	(1,088)	(221)	_	(90)			(1,399)
Impairment (Note 10)	(2,239)	(3,662)		(249)			(6,150)
Transfer to assets held for	No Park						- 8 gal
sale (Note 5)	(469)	(323)	-	(22)		1 N-	(814)
Balance as at November 30,				AT 15 3			K
2024	30,115	38,765	16	5,785	3,220	3,312	81,213
		Leasehold improve-	Rolling	Computer			
Accumulated depreciation	Equipment	ments	stock	hardware	Land	Building	Total
	\$	\$	\$	\$	\$	\$	\$
Balance as at November 30,	4 =0=	4 - 4 -	_				
2022	1,535	1,512	7	914	_	44	4,012
Depreciation Disposals	3,362	7,381	(7)	524	(98	11,358
Balance as at November 30,	(282)	(65)		(9)	\$ 2	***	(356)
2023	4,615	8,828	-	1,429		142	15,014
Depreciation	4,658	8,087	3	531		190	13,469
Disposals	(516)	(115)	_	(68)			(699)
Transfer to assets held for		Y OF		0.0			
sale (Note 5)	(141)	(98)		(7)		_	(246)
Balance as at November 30, 2024	8,616	16,702	3	1,885		332	27,538
2024		10,702	3	1,005		332	21,330
×		Leasehold improve-	Rolling	Computer			
Carrying amounts	Equipment	ments	stock	hardware	Land	Building	Total
	\$	\$	\$	\$	\$	\$	\$
November 30, 2023	24,196	29,716	16	3,348	3,220	3,103	63,599
November 30, 2024	21,499	22,063	13	3,900	3,220	2,980	53,675

8. Intangible assets

Cost	Franchise	T dd	0.11	T . I
Cost	rights	Trademarks	Other	Total
	\$	\$	\$	\$
Balance as at November 30, 2022	181,224	417,870	5,007	604,101
Additions	_	-	565	565
Additions through business acquisitions (Note 3)	34,260	67,830	517	102,607
Disposals		-	(177)	(177)
Impairment (Note 10)	(559)	(3,504)	` <u>—</u>	(4,063)
Balance as at November 30, 2023	214,925	482,196	5,912	703,033
Additions	275		43	318
Disposals	- No. 1		(230)	(230)
Impairment (Note 10)	(1,390)	(6,191)	(186)	(7,767)
Transfer to assets held for sale (Note 5)		(1,222)	(175)	(1,397)
Balance as at November 30, 2024	213,810	474,783	5,364	693,957
Accumulated amortization	Franchise rights	Trademarks	Other	Total
	\$	\$	\$	\$
Balance as at November 30, 2022	80,593	_	1,108	81,701
Disposals	_	_	(49)	(49)
Amortization	18,434	<u></u>	779	19,213
Balance as at November 30, 2023	99,027	<u></u>	1,838	100,865
Amortization	15,124		641	15,765
Balance as at November 30, 2024	114,151		2,479	116,630
Carrying amounts	Franchise	Tradamarka	Othor	Total
Carrying amounts	rights	Trademarks	Other	Total
	\$	\$	\$	\$
November 30, 2023	115,898	482,196	4,074	602,168
November 30, 2024	99,659	474,783	2,885	577,327

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

9. Goodwill

The changes in the carrying amount of goodwill are as follows:

	2024	2023
	\$	\$
Goodwill, beginning of year	424,828	286,713
Amount recognized from business acquisitions (Note 3)		138,115
Goodwill, end of year	424,828	424,828
Accumulated impairment, beginning of year	(48,653)	(48,653)
Impairment (Note 10)	(29,685)	2-
Accumulated impairment, beginning and end of year	(78,338)	(48,653)
Carrying amount	346,490	376,175

10. Impairment - property, plant and equipment, intangible assets and goodwill

The Company performed its annual impairment test as at September 1, 2024, resulting in the recognition of \$6,679 (2023 - \$4,063) of impairment losses on its intangible assets for five of its brands (2023 - four brands). The Company also recorded an impairment of \$1,088 on one of its brands resulting from the reclass to assets held for sale (Note 5).

As of September 1, 2024, the Company reassessed the reporting unit based on a strategic realignment and merger of business units in the US. The goodwill impairment for the reporting unit comprised of Papa Murphy's brand was subject to an impairment test prior to the strategic realignment and this resulted in an impairment of goodwill of \$29,685 (2023 - \$nil).

Additionally, the Company recorded \$6,150 of impairment losses on its property, plant and equipment (2023 - \$169), for a total of \$43,602 (2023 - \$4,232) of impairment charges on its property, plant and equipment, intangible assets and goodwill for the year ended November 30, 2024, which have been recognized in the consolidated statement of operations.

Impairment charges were based on the amount by which the carrying values of the assets exceeded the fair value, determined using expected discounted projected operating cash flows for trademarks and franchise rights.

Impairment by reporting unit for the year ended November 30, 2024:

	Property, plant and equipment	Franchise rights	Trademarks	Other	Goodwill	Total
Ī	\$	\$	\$	\$	\$	\$
Reporting Unit A	1,886	205	2,069			4,160
Reporting Unit B	4,264	1,185	4,122	186	29,685	39,442
	6,150	1,390	6,191	186	29,685	43,602

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

10. Impairment (continued)

Impairment by reporting unit for the year ended November 30, 2023:

		Intangible	assets	
	Property, plant and equipment	Franchise rights	Trademarks	Total
	\$	\$	\$	\$
Reporting Unit A	22	559	3,264	3,845
Reporting Unit B	_	7		-
Reporting Unit C	105		240	345
Reporting Unit D	42	-	-	42
	169	559	3,504	4,232

The key assumptions used are those related to projected operating cash flows, as well as the discount rates. The sales forecasts for cash flows were based on the subsequent fiscal year's budgeted operating results, which were prepared by management and approved by the Board, and internal forecasts for subsequent years, which were prepared by management and developed from the budgeted operating results.

The following table presents the key assumptions used in the Company's impairment tests, as well as the recoverable amounts measured at fair value as at September 1, 2024 and 2023:

		2024				2023
(\$, except percentage data)	Reporting Unit A	Reporting Unit B	Reporting Unit A	Reporting Unit B	Reporting Unit C	Reporting Unit D
Discount rates after tax	10.0%	10.0%	10.5%	10.5%	10.5%	10.5%
Discount rates pre-tax	12.7%	13.0%	13.4%	13.8%	13.7%	13.6%
Recoverable amounts	875,925	604,533	539,407	266,604	313,644	270,246

Long-term growth rates ranging from 0% to 2% (2023 – 0% to 2%) were used in the impairment test for Reporting Unit A. A change of 100 basis points in discount rates in Reporting Unit A would result in additional impairment charges on intangible assets of four brands (2023 – three brands) representing 0.5% (2023 – 0.7%) of the total carrying value of the franchise rights and trademarks in that unit. A change of 100 basis points in discount rates in Reporting Unit A would not result in additional impairment charges on goodwill for the years ended November 30, 2024 and 2023. For Reporting Unit A, an increase of 430 basis points in the discount rate would have resulted in its recoverable amount being equal to its carrying value.

Long-term growth rates of 2.0% were used in the impairment test for Reporting Unit B. A change of 100 basis points in discount rates in Reporting Unit B would result in additional impairment charges on intangible assets of one brand (2023 – nil) representing 0.1% (2023 – nil) of the total carrying value of the franchise rights and trademarks in that unit. A change of 100 basis points in discount rates in Reporting Unit B would not result in additional impairment charges on goodwill for the years ended November 30, 2024 and 2023. For Reporting Unit B, an increase of 130 basis points in the discount rate would have resulted in its recoverable amount being equal to its carrying value.

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

11. Gift card liability

The changes in the carrying amount of the gift card liability are as follows:

2024	2023
\$	\$
104,858	91,453
77,690	52,873
(66,987)	(39,603)
	5,884
(320)	(1,065)
(5,989)	(4,684)
109,252	104,858
	\$ 104,858 77,690 (66,987) — (320) (5,989)

12. Deferred revenue and deposits

	2024	2023
	\$	\$
Franchise fee deposits	33,594	28,948
Unearned rent	1,737	1,794
Supplier contributions and other allowances	932	1,786
	36,263	32,528
Current portion	(8,093)	(7,775)
	28,170	24,753

Deferred revenues consist mostly of initial, transfer and renewal franchise fees paid by franchisees, as well as upfront fees paid by master franchisees, which are generally recognized on a straight-line basis over the term of the related agreement. Deferred revenues also include amounts paid in upfront fees received from agreements with suppliers, which are amortized over the term of the related agreement.

\$7,329 (2023 – \$8,055) of revenue recognized in the current year was included in the deferred revenue balance at the beginning of the year.

The following table provides estimated revenues expected to be recognized in future years related to performance obligations that are unsatisfied as at November 30, 2024:

Estimate for fiscal year:	\$
2025	8,093
2026	4,452
2027	3,660
2028	2,814
2029	2,411
Thereafter	14,833
	36,263

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

13. Receivables and advances from ultimate parent, parent company and companies under common control

The receivable from ultimate parent is primarily comprised of a renewable one-year term loan facility of \$194,988 (2023 – \$148,670). The term loan is unsecured and bears interest at the rate set at the Applicable Federal Rates as issued by the US Internal Revenue Service for short-term instruments. The term loan may be repaid, in whole or in part, at any time, without premium or penalty. MTY USA may also request repayment, in whole or in part, at any time, without penalty.

The advance from ultimate parent is comprised of three loans totaling \$708,363 seen in note 14 (2023 – \$705,683). Those loans are unsecured and each tranche has a different interest rate spanning from non interest bearing to 9.26%. These loans may be repaid, in whole or in part, at any time, without premium or penalty.

The receivable from company under common control and advances from parent company and companies under common control are non-interest bearing and receivable/due on demand with no specified collection/repayment terms.

14. Long-term loan from company under common control

Non interest-bearing loan, due on demand Interest-bearing loan at 5.4%, repayable by November 27, 2026 ⁽¹⁾ Interest-bearing loan at 9.26%, repayable by December 8, 2029

⁽¹⁾ This loan is subject to a maximum debt-to-EBITDA ratio of 6.00:1.00 starting on November 27, 2019, and a minimum EBITDA interest coverage ratio of 2.00:1.00 to be calculated in conjunction with interest payments based on the past 12 months.

15. Holdback payable

Non-interest-bearing holdback on acquisition of Sauce Pizza and Wine, repayable in December 2023 (Note 3)

Non-interest-bearing holdback on acquisition of the assets of two restaurants located in California, repayable within the next 12 months

Current portion of holdback payable

	2024	2023
	\$	\$
	-	837
		75
4 6 6	_	912

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

16. Contingencies

The contingencies for litigation and disputes represent management's best estimate of the outcome of litigations and disputes that are ongoing at the date of the consolidated balance sheet. These contingencies consist of multiple items, a large part of which are insurance claims, including worker's compensation claims, at the Company's corporately owned locations. The timing of the settlement of these contingencies is unknown given their nature, as the Company does not control the litigation timelines.

	2024	2023
	\$	\$
Provision for litigations, disputes and other contingencies, beginning of year	2,849	724
Reversals	(790)	(304)
Amounts used	(9,424)	(8,797)
Additions	10,058	11,226
Provision for litigations, disputes and other contingencies, end of year	2,693	2,849

17. Common stock

		2024		2023
	Shares Issued	\$	Shares issued	\$
Balance, beginning and end of year	15	179,154	15	179,154

18. Financial instruments

In the normal course of business, the Company uses various financial instruments which by their nature involve risk, including market risk and the credit risk of non-performance by counterparties. These financial instruments are subject to normal credit standards, financial controls, risk management as well as monitoring procedures.

Fair value of recognized financial instruments

The Company has determined that the fair value of its financial assets and financial liabilities with short-term maturities approximates their carrying value. These financial instruments include cash, accounts receivable, receivable from ultimate parent, deposits loans receivable, accounts payable, accrued liabilities, promotional funds payable, advances from parent company and from companies under common control and holdbacks.

Risk management policies

The Company, through its financial assets and financial liabilities, is exposed to various risks. The following analysis provides a measurement of risks as at November 30, 2024.

Credit risk

The Company's credit risk is primarily attributable to its trade receivables, loans receivable and deposits. For accounts receivable and loans receivable, the amounts disclosed on the consolidated balance sheets are net of allowances for bad debts, estimated by the Company's management based on past experience and counterparty specific circumstances. The Company believes that the credit risk of accounts receivable is limited for the following reasons:

- The Company's broad client base is spread mostly across the US, which limits the concentration of credit risk,
- The Company accounts for specific bad debt provisions when management considers that the expected recovery is less than the actual account receivable.

The credit risk on cash is limited because the Company invests its excess liquidity in high-quality financial instruments and with credit-worthy counterparties.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

18. Financial instruments (continued)

The credit risk on deposits is also limited as these are mostly with well-established and credit-worthy companies.

Interest rate risk

Interest rate risk is the Company's exposure to increases and decreases in financial instrument values caused by the fluctuation in interest rates. The Company has limited exposure to interest rate risk as its long-term loans from company under common control and ultimate parent as well as its long-term debt have fixed interest rates

19. Revenue

	For the year ended					
	November 30, 2024			11/30/2023 Adjusted (1)		(1)
	Franchising	Corporate	Total	Franchising	Corporate	Total
	\$	\$	\$	\$	\$	\$
Corporate store revenues		330,599	330,599	_	337,937	337,937
Royalties	128,335	7 July Y	128,335	128,461	-	128,461
Franchise fees, transfer fees and master license fees	5,770		5,770	4,996	<u></u>	4,996
Promotional funds	56,563		56,563	56,734	_	56,734
Program allowances	30,373		30,373	32,499		32,499
Breakage income	5,989	() () () () () ()	5,989	4,684	-	4,684
Resale material and retail sales	4,781	_	4,781	5,754		5,754
Rent	25,199		25,199	26,337	_	26,337
Other	9,929	_	9,929	9,215		9,215
	266,939	330,599	597,538	268,680	337,937	606,617

⁽¹⁾ Certain comparative figures have been reclassified to conform with current year presentation. Lease income in comparative period of \$26.3 million was reclassed from operating expense to revenue.

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

20. Operating expenses

	For the year ended					
	November 30, 2024			November 30, 2023 Adju		usted (2)
	Franchising	Corporate	Total	Franchising	Corporate	Total
	\$	\$	\$	\$	\$	\$
Cost of goods sold	2,342	89,624	91,966	2,284	94,465	96,749
Wages and benefits	52,855	118,514	171,369	54,590	115,976	170,566
Advertising, marketing and promotion	443	6,000	6,443	462	5,311	5,773
Rent	27,748	37,925	65,673	31,895	34,105	66,000
Professional and consulting fees and commission	11,544	2,037	13,581	12,054	4,394	16,448
Office, travel, meals and entertainment						
and utilities	13,482	37,998	51,480	13,302	37,713	51,015
Promotional funds	56,563	_	56,563	56,734	_	56,734
Gift card program costs	6,499	_	6,499	6,695	_	6,695
Other (1)	954	25,918	26,872	2,342	25,474	27,816
Bad debt (recovery) expense	(293)	147	(146)	438	250	688
	172,137	318,163	490,300	180,796	317,688	498,484

- (1) Other operating expenses are comprised mainly of other office administration expenses.
- (2) Certain comparative figures have been reclassified to conform with current year presentation. Lease income in comparative period of \$26.3 million was reclassed from operating expense to revenue.

Franchising operations

The franchising business mainly generates revenues from royalties, supplier contributions, franchise fees and rent.

Corporate store operations

Corporate stores generate revenues from the direct sale of prepared food to customers.

21. Restructuring

During the year, the Company initiated a restructuring plan as part of a strategic realignment to streamline operations and improve efficiency. The Company recognized restructuring costs of \$1,342 primarily related to employee severance costs of \$832 with the reminder being in relation to the discontinuation of one of its brands. No additional expenses are expected to be incurred.

22. Interest expense

Interest charged by companies under common control (Note 25) Interest expense

2024	2023
\$	\$
53,768	53,977
53,768	53,977
· · · · · · · · · · · · · · · · · · ·	

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

23. Income taxes

The Company accounts for income taxes in accordance with ASC 740. ASC 740 prescribes the use of the liability method whereby deferred tax asset and liability account balances are determined based on differences between the financial reporting and tax bases of assets and liabilities and are measured using the enacted tax rates. The effects of future changes in tax laws or rates are not anticipated.

Under ASC 740, income taxes are recognized for the following: a) amount of tax payable for the current year and b) deferred tax liabilities and assets for future tax consequences of events that have been recognized differently in the consolidated financial statements than for tax purposes.

Income tax expense (recovery)	2024	2023
	\$	\$
Current tax expense	9,918	7,807
Deferred tax recovery	(9,195)	(6,405)
Total tax expense	723	1,402

The provision for income taxes recorded in the consolidated financial statements differs from the amount which would be obtained by applying the statutory federal income tax rate of 21% (2023 – 21%) to the income for the period as follows:

	2024	2023
	\$	\$
(Loss) income before income taxes	(11,833)	18,380
Income tax (recovery) expense at federal statutory rate	(2,478)	3,860
State and local income taxes net of federal tax benefit	881	2,360
Non-deductible/non-taxable items	5,484	(216)
True-up of prior year tax provision	(3,950)	(2,239)
Rate variation on deferred income tax	724	(102)
Credits generated and used in current year		(2,371)
Other	62	110
Income tax expense	723	1,402

Notes to the consolidated financial statements

Years ended November 30, 2024 and 2023

(In thousands of US dollars)

23. Income taxes (continued)

Components of the net deferred tax asset (liability):

	2024	2023
	\$	\$
Inventories	(275)	(264)
Allowance for credit losses	192	595
Deferred revenue and deposits	4,676	4,725
Gift card liability	2,959	2,947
Accrued liabilities	41,158	33,567
Non-capital losses and other tax credits	1,917	4,570
Other	(1,941)	(1,285)
Operating lease liabilities	48,125	48,406
Total deferred tax assets	96,811	93,261
Deferred costs	(600)	(948)
Property, plant and equipment	(9,632)	(11,490)
Operating lease right-of-use assets	(47,277)	(47,497)
Intangible assets	(140,054)	(143,345)
Holdback payable	(305)	(233)
Total deferred tax liabilities	(197,868)	(203,513)
Net deferred tax liability	(101,057)	(110,252)

24. Supplemental cash flow information

During the year ended November 30, 2024, the Company paid \$10,348 (2023 – \$8,268) in income taxes. Furthermore, there are non-cash items included in the proceeds on disposition of property, plant and equipment amounting to a net liability of \$113 (2023 – net liability of \$1,574). The non-cash items were primarily related to commitments made as part of the disposal of a portfolio of corporately-owned locations.

25. Related party transactions

The Company has transactions in the normal course of business with its ultimate parent, parent company and companies under common control. These transactions were in the normal course of business and measured at the exchange amount, which is the amount of consideration established and agreed to by the related parties.

Consolidated financial statements of MTY Franchising USA, Inc.

For the years ended November 30, 2023 and 2022

Independent auditor's report	1-2
Consolidated statements of operations and comprehensive income	3
Consolidated statements of changes in stockholder's equity	4
Consolidated balance sheets	5-6
Consolidated statements of cash flows	7-8
Notes to the consolidated financial statements	9-36



Report of Independent Auditors

To the Management and Stockholder of MTY Franchising USA, Inc.

Opinion

We have audited the accompanying consolidated financial statements of MTY Franchising USA, Inc. and its subsidiaries (the "Company"), which comprise the consolidated statement of operations and comprehensive income for the year ended on November 30, 2023 and 2022, the consolidated statement of changes in stockholder's equity for the year ended on November 30, 2023 and 2022, the consolidated balance sheets as of November 30, 2023 and 2022, and the consolidated statement of cash flows for the year then ended, including the related notes (collectively referred to as the "consolidated financial statements").

In our opinion, the accompanying consolidated financial statements present fairly, in all material respects, the financial position of the Company as of November 30, 2023 and 2022, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audit in accordance with auditing standards generally accepted in the United States of America (US GAAS). Our responsibilities under those standards are further described in the Auditors' Responsibilities for the Audit of the Consolidated Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for one year after the date the consolidated financial statements are available to be issued.



Auditors' Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditors' report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with US GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with US GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks.
 Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit
 procedures that are appropriate in the circumstances, but not for the purpose of expressing an
 opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is
 expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

Montreal, Canada February 2, 2024

¹ CPA auditor, public accountancy permit No. A125677

Pricewaterhouse Coopers UP

Consolidated statements of operations and comprehensive income

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

	Notes	2023	2022
	7	\$	\$
Revenue	19	580,280	263,686
Costs and expenses			
Operating expenses	20	472,147	194,664
Depreciation – property, plant and equipment	7	11,358	2,480
Amortization – intangible assets	8	19,213	14,631
Impairment charge - property, plant and equipment	7 & 10	169	338
Impairment charge – operating lease right-of-use assets	6	_	247
Impairment charge – intangible assets	8 & 10	4,063	5,643
Interest expense	21	53,977	18,135
Management fees charged by parent company	24	2,249	2,928
	_	563,176	239,066
Other income (expenses)			
Interest income		2,255	2.256
(Loss) gain on disposal of property, plant and equipment and intangible assets			,
		(1,843)	92
Gain on contingent consideration from a business acquisition	3	1,600	_
(Loss) gain on de-recognition/lease modification of operating lease liabilities	6	(736)	20
liabilities	0 _	1,276	2,368
	*		
Income before income taxes	1	18,380	26,988
Income tax expense (recovery)	22		
Current	22	7,807	9,911
Deferred		(6,405)	1.5
Deletied	<u></u>	1,402	(2,496) 7,415
Not income and comprehensive income	-	16,978	
Net income and comprehensive income		10,978	19,573

Consolidated statements of changes in stockholder's equity

Years ended November 30, 2023 and 2022

(In thousands of US dollars, except number of common stock issued)

	Common stock issued	Common stock value	Retained earnings	Total stockholder's equity
	/ 1	\$	\$	\$
Balance as at November 30, 2021	15	179,154	37,231	216,385
Net income and comprehensive income		_	19,573	19,573
Balance as at November 30, 2022	15	179,154	56,804	235,958
Net income and comprehensive income	_	<u> </u>	16,978	16,978
Balance as at November 30, 2023	15	179,154	73,782	252,936

MTY Franchising USA, Inc. Consolidated balance sheets

As at November 30, 2023 and 2022 (In thousands of US dollars)

	4	2023	2022
	Notes	\$	\$
A			
Assets			
Current assets		04.400	00.050
Cash		21,138	23,059
Restricted cash		271	504
Accounts receivable	4	29,461	24,961
Inventories		6,414	6,164
Assets held for sale	5 & 7	1,668	1,563
Current portion of loans receivable		272	461
Receivable from company under common control	13	-	3
Receivable from ultimate parent	13 & 24	148,828	126,323
Prepaid expenses and deposits		8,386	7,728
Other current assets		3,333	3,167
Income taxes receivable	22	4,919	2,827
		224,690	196,760
Loans receivable		151	543
Contract cost asset		3,696	3,127
Other assets		1,896	1,492
Property, plant and equipment	7	63,599	54,458
Operating lease right-of-use assets	6	187,074	165,641
Intangible assets	8	602,168	522,400
Goodwill	9	376,175	238,060
		1,459,449	1,182,481
Liabilities			
Current liabilities Accounts payable		15.878	16,650
Accounts payable Accrued liabilities		28,769	29,565
Gift card liability	11	104,858	91,453
Promotional funds payable		14,690	16,303
Current portion of operating lease liabilities	6	37,080	35,431
Current portion of deferred revenue and deposits	12	7,775	7,571
Advance from parent company	13 & 24	2,249	2,928
Advance from ultimate parent	13	Yes light (198,801
Advances from companies under common control	13 & 24	42	42
Current portion of holdback payable	15	912	_
		212,253	398,744

Consolidated balance sheets (continued)

As at November 30, 2023 and 2022

(In thousands of US dollars)

		2023	2022
	Notes	\$	\$
Liabilities (continued)			
Long-term loan from company under common control	14	705,683	299,850
Other liabilities		159	412
Operating lease liabilities	6	153,413	132,285
Deferred revenue and deposits	12	24,753	23,288
Deferred income taxes	22	110,252	91,944
		1,206,513	946,523
Stockholder's equity			
Common stock	17	179,154	179,154
Retained earnings		73,782	56,804
		252,936	235,958
		1,459,449	1,182,481

Approved by the Board on February 2, 2024	
	, Director

MTY Franchising USA, Inc. Consolidated statements of cash flows

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

		2023	2022
	Notes	\$	\$
Operating activities			
Net income and comprehensive income		16,978	19,573
Items not affecting cash:			
Depreciation – property, plant and equipment	7	11,358	2,480
Amortization – intangible assets	8	19,213	14,631
Interest expense	21	53,977	18,135
Loss (gain) on disposal of property, plant and equipment and intangible		111	
assets		1,843	(92)
Impairment charge – property, plant and equipment	7 & 10	169	338
Impairment charge – operating lease right-of-use assets	6		247
Impairment charge – intangible assets	8 & 10	4,063	5,643
Gain on contingent consideration from a business acquisition	3	(1,600)	 2
Loss (gain) on de-recognition/lease modification of operating lease			
liabilities	6	736	(20)
Deferred income tax recovery	<u>a_</u>	(6,405)	(2,496)
		100,332	58,439
Interest paid		(53,977)	(16,192)
Changes in non-cash working capital items			
Accounts receivable		(4,133)	(213)
Inventories		198	151
Prepaid expenses and deposits		(1,140)	(2,997)
Loans receivable		1,410	329
Other current assets		(412)	(574)
Income taxes		(1,842)	(1,048)
Accounts payable		(911)	(2,511)
Accrued liabilities		(5,853)	(6,759)
Promotional funds payable		(2,045)	(218)
Gift card liability		7,521	7,747
Deferred revenue and deposits		1,088	2,831
Other		1,512	74
Net cash provided from operating activities	7	41,748	39,059

Consolidated statements of cash flows (continued)

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

		2023	2022
	Notes	\$	\$
Investing activities			
Net cash outflow on acquisitions	3	(221,180)	(207,083)
Net cash acquired through business acquisitions	3	6,884	10,431
Additions to property, plant and equipment	7	(13,250)	(1,830)
Additions to intangible assets	8	(565)	(2,950)
Proceeds on disposal of property, plant and equipment		358	326
Net cash used in investing activities	7	(227,753)	(201,106)
Financing activities			
Net advance (to) from ultimate parent and parent company		(221,985)	174,599
Net advance from (to) companies under common control		405,836	(142)
Repayment of holdback payable	15		(7,076)
Net cash provided from financing activities	_	183,851	167,381
Net (decrease) increase in cash		(2,154)	5,334
Cash, beginning of year		23,563	18,229
Cash, end of year	I	21,409	23,563
Supplemental cash flow information	23		

The accompanying notes are an integral part of the consolidated financial statements.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

1. Nature of operations

MTY Franchising USA, Inc. (the "Company" or "MTY USA") was incorporated on March 14, 2001. The Company operates, develops and franchises restaurants under a multitude of different banners in the United States of America (the "US").

2. Significant accounting policies

Basis of presentation

The accounting policies of the Company are in accordance with accounting principles generally accepted in the US ("US GAAP"). The Company uses the US dollar as its functional and reporting currency, and tabular amounts are rounded to the nearest thousand (\$000) except when otherwise indicated. MTY USA is a wholly owned subsidiary of MTY Food Group Inc.

The preparation of the consolidated financial statements in conformity with US GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from those estimates.

Presented below are those policies considered particularly significant:

Basis of consolidation

The consolidated financial statements reflect the financial position and operating results of the Company, including wholly owned subsidiaries and investees that we control.

The principal subsidiaries of the Company are as follows:

Principal subsidiaries	Percentage of equity interest
	%
BF Acquisition Holdings, LLC	100
Built Franchise Systems, LLC	100
CB Franchise Systems, LLC	100
Kahala Brands Ltd.	100
Papa Murphy's Holdings Inc.	100
BBQ Holdings, Inc. (Note 3)	100
Wetzel's Pretzels, LLC (Note 3)	100

Revenues and expenses of subsidiaries are included in the consolidated statement of operations and comprehensive income from the effective date of acquisition. The subsidiaries are consolidated from the acquisition date until the date on which the Company ceases to control them.

All intercompany transactions, balances, revenues and expenses are eliminated in full upon consolidation.

Business combinations

The Company accounts for acquired businesses using the acquisition method of accounting in accordance with Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") 805, Business Combinations. The consideration transferred for the acquisition is the fair values of the assets transferred, the liabilities incurred, and the equity interest issued. The consideration transferred includes the fair value of any asset or liability resulting from a contingent consideration arrangement. Acquisition-related costs are expensed as incurred. Identifiable assets acquired and liabilities and contingent liabilities assumed in a business combination are measured initially at their fair value at the acquisition date.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

2. Significant accounting policies (continued)

Business combinations (continued)

Goodwill is measured as the excess of the purchase price over the estimated fair values of the net assets acquired. If, after reassessment, the net of the acquisition date amounts of the identifiable assets acquired and liabilities assumed exceeds the sum of the consideration transferred, the amount of any non-controlling interests in the acquiree and the fair value of the acquirer's previously held interest in the acquiree (if any), the excess is recognized immediately in profit or loss as a bargain purchase gain.

Goodwill reflects how the acquisition will impact the Company's ability to generate future profits in excess of existing profits. The consideration paid mostly relates to combined synergies, related mainly to revenue growth. These benefits are not recognized separately from goodwill as they do not meet the recognition criteria for identifiable intangible assets.

When the consideration transferred by the Company in a business combination includes assets or liabilities resulting from a contingent consideration arrangement, the contingent consideration is measured at its acquisition date fair value and included as part of the consideration transferred in a business combination.

When a business combination is achieved in stages, the Company's previously held equity interest in the acquiree is remeasured to fair value at the acquisition date (i.e. the date when the Company obtains control) and the resulting gain or loss, if any, is recognized in profit or loss. Amounts arising from interests in the acquiree prior to the acquisition date that have previously been recognized in income and other comprehensive income are reclassified to profit or loss where such treatment would be appropriate if that interest were disposed of.

If the initial accounting for a business combination is incomplete by the end of the reporting period in which the combination occurs, the Company reports provisional amounts for the items for which the accounting is incomplete. Those provisional amounts are adjusted retrospectively during the measurement period or additional assets or liabilities are recognized, to reflect new information obtained about facts and circumstances that existed at the acquisition date that, if known, would have affected the amounts recognized at that date.

Goodwill

Goodwill represents the excess of cost over the net tangible assets and identifiable intangible assets of acquired businesses. Goodwill is carried at cost reduced by previous impairment losses, if any.

Functional currency

The functional currency of the Company and its subsidiaries is the US dollar. The Company translates monetary assets and liabilities that are denominated in currencies other than the US dollar at the exchange rates prevailing at the end of the reporting period; non-monetary assets denominated in foreign currencies are translated using the exchange rate prevailing at the transaction date; all revenue and expense items denominated in foreign currencies are translated at the exchange rate prevailing at the transaction date. All foreign exchange gains and losses are reported in profit or loss.

Revenue recognition

Revenue is recognized upon the transfer of control of promised goods or services to customer in an amount that reflects the consideration the Company expects to receive for those goods or services:

Revenue from franchise locations

- Royalties are based either on a percentage of gross sales as reported by the franchisees or on a fixed monthly fee. They are recognized on an accrual basis in accordance with the substance of the relevant agreement, as they are earned.
- ii) Promotional fund contributions are based on a percentage of gross sales as reported by the franchisees. Corresponding promotional fund transfers are presented directly on the consolidated balance sheets. The Company is not entitled to retain these promotional fund payments received and is obligated to transfer these funds to be used solely for use in promotional and marketing-related costs for specific restaurant banners. The Company sometimes charges a fee for the administration of the promotional funds.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

2. Significant accounting policies (continued)

Revenue recognition (continued)

Revenue from franchise locations (continued)

- iii) Initial franchise fees are recognized on a straight-line basis over the term of the franchise agreement as the performance obligation relating to franchise rights is fulfilled. Amortization begins once the restaurant has opened.
- iv) Upfront fees related to master license agreements are recognized over the term of the master license agreements on a straight-line basis.
- v) Renewal fees and transfer fees are recognized on a straight-line basis over the term of the related franchise agreement.
- vi) The Company earns rent revenue on certain leases it holds; the Company's policy is described below.
- vii) Revenue from equipment sale and retail sales are recognized upon transfer of control, generally upon shipment of the equipment or goods. This revenue is recorded in resale material and retail sales.
- viii) The Company recognizes breakage income proportionately as each gift card is redeemed, based on the historical redemption patterns of the gift cards. The Company also charges various program fees to its franchisees as gift cards are redeemed. Notably, this does not apply to gift card liabilities assumed in a business acquisition, which are accounted for at fair value at acquisition date.
- ix) The Company receives considerations from certain suppliers. Fees are generally earned based on the value of purchases during the period. Agreements that contain an initial upfront fee, in addition to ongoing fees, are recognized on a straight-line basis over the term of the respective agreement. Supplier contributions are recognized as revenue as they are earned and are recorded in franchising revenue.
- x) The Company earns e-commerce fees, which includes point-of-sale ("POS") support fees and transaction fees for purchase made through one of the Company's brands' e-commerce platforms. POS supports fees are received quarterly in advance and are recognized over the period they cover. Transaction fees are recognized when the food items purchased from a store are delivered or picked up by customers.

Revenue from corporate-owned locations

Revenue from corporate-owned locations is recorded when goods are delivered to customers.

Contract cost asset

The Company recognizes incremental costs of obtaining a contract as an asset if they are expected to be recoverable, unless their amortization period would be less than one year, in which case they are expensed as incurred. The costs are amortized to operating expenses over the term of the related franchise agreement.

Leasing

In accordance with ASC 842, the Company determines if an arrangement is or contains a lease at contract inception and recognizes a right-of-use asset and a lease liability at the lease commencement date. Leases with an initial term of 12 months or less but greater than one month are not recorded on the balance sheet for select asset classes.

The lease liability is measured at the present value of future lease payments as of the lease commencement date. The right-of-use asset recognized is based on the lease liability adjusted for prepaid and deferred rent and unamortized lease incentives. An operating lease right-of-use asset is amortized on a straight-line basis over the lease term and is recognized as a single lease cost against the operating lease liability. A finance lease right-of-use asset is amortized on a straight-line basis, with interest costs reported separately, over the lesser of the useful life of the leased asset or lease term.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

2. Significant accounting policies (continued)

Leasing (continued)

Operating lease expense is recognized on a straight-line basis over the lease term and is included in Operating expenses. Variable lease payments are expensed as incurred. The Company uses its incremental borrowing rates as the discount rate for its leases, which is equal to the rate of interest the Company would have to pay on a collateralized basis to borrow an amount equal to the lease payments under similar terms. The lease terms for all the Company's leases include the contractually obligated period of the leases, plus any additional periods covered by Company options to extend the leases that the Company is reasonably certain to exercise. Certain leases provide that the lease payments may be increased annually based on the fixed rate terms or adjustable terms such as the Consumer Price Index. Future base rent escalations that are not contractually quantifiable as of the lease commencement date are not included in the lease liability.

Lease expense for lease payments is recognized on a straight-line basis over the lease term. Lease expense is comprised of operating and finance lease costs, short-term lease costs, and variable lease costs, which primarily include common area maintenance, real estate taxes, and insurance for the Company's real estate leases.

The Company enters into leases for franchised and corporately-owned locations, offices, and equipment in the normal course of business.

The Company as lessee

The Company recognizes operating lease liabilities with corresponding operating lease right-of-use assets, except for short-term leases and leases of low value assets, which are expensed on a straight-line basis over the lease term. The Company's leases are all classified as operating leases. The amortization of the operating lease right-of-use asset and interest expense related to the operating lease liability are recorded together as the lease expense to produce a straight-line recognition effect in the consolidated statement of operations. Under ASC 842, operating lease right-of-use assets are tested for impairment in accordance with ASC 360, Property, Plant and Equipment.

The Company as lessor

When the Company enters into a sublease arrangement as an intermediate lessor, the Company accounts for the head lease and the sublease as two separate contracts. All the subleases of the Company are classified as operating subleases by reference to the operating lease right-of-use asset arising from the underlying asset. For operating subleases, the Company recognizes an operating right-of-use asset relating to the head lease and recognizes a deferred rent asset or liability in the sublease. As the intermediate lessor, the Company retains the operating lease liability on the head lease in its consolidated balance sheet. During the term of the sublease, the Company recognizes both lease income on the sublease and lease expense on the head lease.

Income taxes

The Company accounts for income taxes pursuant to ASC 740, Income Taxes ("ASC 740"). Deferred tax assets and liabilities are recorded for differences between the financial statement and tax basis of the assets and liabilities that will result in taxable or deductible amounts in the future based on enacted tax laws and rates. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be realized. Income tax expense is recorded for the amount of income tax payable or refundable for the period increased or decreased by the change in deferred tax assets and liabilities during the period.

The Company recognizes the financial statement benefit of a tax position only after determining that the relevant tax authority would more likely than not sustain the position following an audit. For tax positions meeting the more-likely-than-not threshold, the amount recognized in the financial statements is the largest benefit that has a greater than 50 percent likelihood of being realized upon ultimate settlement with the relevant tax authority.

Allowance for doubtful accounts

The Company currently uses the simplified expected credit loss ("ECL") model for its trade receivables, which permits the use of the lifetime expected loss provision for all trade receivables and also incorporates forward-looking information. Lifetime ECL represents the ECL that will result from all probable default events over the expected life of a financial instrument.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

2. Significant accounting policies (continued)

Assets held for sale

Assets are classified as held for sale when management with the appropriate authority commits to a plan to sell the assets, the assets are available for immediate sale, the assets are actively marketed at a reasonable price, the sale is probable within a year, and certain other criteria met. Assets held for sale consist primarily of Company-owned stores where the Company has committed to a plan to sell specific stores. Assets designated as held for sale are held at the lower of the net book value or fair value less costs to sell. Depreciation is not charged against property, plant and equipment classified as assets held for sale.

Property, plant and equipment

Land and buildings held for use in the production or supply of goods or services, or for administrative purposes, are stated in the consolidated balance sheet at their historical costs less accumulated depreciation (buildings) and accumulated impairment losses. Cost includes expenditures that are directly attributable to the acquisition of the asset, including any costs directly attributable to bringing the asset to a working condition for its intended use.

Equipment, leasehold improvements, rolling stock and computer hardware are stated at cost less accumulated depreciation and accumulated impairment losses.

Depreciation is recognized so as to write off the cost or valuation of assets less their residual values over their useful lives, using the straight-line method. The estimated useful lives, residual values and depreciation methods are reviewed at the end of each year, with the effect of any changes in estimate accounted for on a prospective basis.

An item of property, plant and equipment is derecognized upon disposal or when no future economic benefits are expected to arise from the continued use of the asset. Any gain or loss arising on the disposal or retirement of an item of property, plant and equipment is determined as the difference between the sales proceeds and the carrying amount of the asset and is recognized in profit or loss.

Depreciation is based on the following terms:

Buildings Straight-line 25 to 50 years
Equipment Straight-line Three to 10 years
Leasehold improvements Straight-line Term of the lease
Rolling stock Straight-line Five to seven years
Computer hardware Straight-line Three to seven years

Intangible assets

Intangible assets acquired separately

Intangible assets with finite useful lives that are acquired separately are carried at cost less accumulated amortization and accumulated impairment losses, if applicable. Amortization is recognized on a straight-line basis over their estimated useful lives. The estimated useful lives and amortization methods are reviewed at the end of each year, with the effect of any changes in estimate being accounted for on a prospective basis. Intangible assets with indefinite useful lives that are acquired separately are carried at cost less accumulated impairment losses, if applicable.

Intangible assets acquired in a business combination

Intangible assets acquired in a business combination and recognized separately from goodwill are initially recognized at their fair value at the acquisition date.

Subsequent to initial recognition, intangible assets having a finite life acquired in a business combination are reported at cost less accumulated amortization and accumulated impairment losses, if applicable, on the same basis as intangible assets that are acquired separately. Intangible assets having an indefinite life are not amortized and are therefore carried at cost reduced by previous impairment losses, if applicable.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

2. Significant accounting policies (continued)

Intangible assets (continued)

Derecognition of intangible assets

An intangible asset is derecognized on disposal, or when no future economic benefits are expected from use or disposal. Gains or losses arising from derecognition of an intangible asset, measured as the difference between the net disposal proceeds and the carrying amount of the asset, are recognized in profit or loss when the asset is derecognized.

The Company currently carries the following intangible assets in its books:

Franchise rights

The franchise rights acquired through business combinations are recognized at the fair value of the estimated future cash inflows related to the acquisition of franchises. The franchise rights are generally amortized on a straight-line basis over the term of the agreements which typically range between 10 to 20 years.

Trademarks

Trademarks acquired through business combinations are recognized at their fair value at the time of the acquisition and are not amortized. Trademarks are determined to have an indefinite useful life based on their brand recognition and their ability to generate revenue through changing economic conditions with no foreseeable time limit.

Other

Included in other intangible assets are purchased software and liquor licences, which are being amortized over their expected useful life on a straight-line basis.

Impairment of long-lived assets other than goodwill

The Company continually reviews whether events or circumstances subsequent to the acquisition of any long-lived assets, including intangible assets with finite useful lives, have occurred that indicate the remaining estimated useful lives of those assets may warrant revision or that the remaining balance of those assets may not be recoverable. If events and circumstances indicate that the long-lived assets should be reviewed for possible impairment, the Company uses projections to assess whether future cash flows on an undiscounted basis related to the assets exceed the recorded carrying amount of those assets to determine if an asset is impaired. Should an impairment be identified, a loss would be recorded to the extent that the carrying value of the impaired assets exceeds their fair values as determined by valuation techniques appropriate in the circumstances that could include the use of similar cash flow projections on a discounted basis. The reporting units of indefinite intangible assets are individual brands, comprised of franchise rights, trademarks, and perpetual licenses.

Impairment of goodwill

For the purposes of impairment testing, goodwill is allocated to the unit or group of units ("reporting unit") that are considered to represent the lowest level within the group at which the goodwill is monitored for internal management purposes. As at November 30, 2023, goodwill is allocated as follows:

	Reporting unit description
Reporting Unit A	A group of units comprised of acquired brands in the US, excluding the Papa Murphy's, BBQ Holdings, Inc. ("BBQ Holdings") and Wetzel's Pretzels brands
Reporting Unit B	One unit comprised of the Papa Murphy's brand
Reporting Unit C	A group of units comprised of the BBQ Holdings brands
Reporting Unit D	One unit comprised of the Wetzel's Pretzels brand

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

2. Significant accounting policies (continued)

Impairment of goodwill (continued)

Goodwill is tested for impairment on an annual basis (August 31 for the Company) and between annual tests if an event occurs or circumstances change that would more likely than not reduce the fair value of a reporting unit below its carrying value. Fair value is determined using a discounted cash flow methodology with a risk adjusted weighted average cost of capital.

Cash and restricted cash

Cash and restricted cash include cash on hand and short-term investments, if any, with maturities upon acquisition of generally three months or less or that are redeemable at any time at full value and for which the risk of a change in value is not significant. As at November 30, 2023, cash and restricted cash included \$271 of restricted cash (2022 – \$504) that is required as part of guarantees on certain lease commitments.

Inventories

Inventories are measured at the lower of cost and market value. Costs of inventories are determined on a first-infirst-out basis and include acquisition costs and other costs incurred to bring inventories to their present location and condition.

Market value represents the current replacement cost, provided that the cost does not exceed the net realizable value or is not less than the net realizable value reduced by a normal profit margin.

Contingencies

Litigation, disputes and closed stores

Provisions for the expected cost of litigation, disputes and the cost of settling leases for closed stores, with the exception of operating lease liabilities already recorded pursuant to ASC 842, are recognized when it becomes probable the Company will be required to settle the obligation, at management's best estimate of the expenditure required to settle the Company's obligation.

Contingent liabilities acquired in a business combination

Contingent liabilities acquired in a business combination are initially measured at fair value at the acquisition date. At the end of subsequent reporting periods, such contingent liabilities are measured at the higher of the amount that would be recognized, and the amount initially recognized less cumulative amortization recognized, if any.

Financial instruments

The Company's financial instruments consist of cash, restricted cash, accounts receivable, loans receivable, receivable from company under common control, receivable from ultimate parent, accounts payable, accrued liabilities, promotional funds payable, advance from parent company, advances from companies under common control, long-term loan from company under common control and holdbacks payable. Unless otherwise noted, it is management's opinion that the Company is not exposed to significant interest, currency or credit risks arising from these financial instruments. The fair values for cash, restricted cash, accounts receivable, loans receivable, receivable from company under common control, receivable from ultimate parent, accounts payable, accrued liabilities, promotional funds payable, advance from parent company and advances from companies under common control approximate their carrying values due to their immediate or short-term maturities, unless otherwise noted. The long-term loan from company under common control, advances from ultimate parent and holdbacks payable are measured at amortized cost using the effective interest method.

Promotional funds

Pursuant to the franchise agreements, franchisees must pay a fee to the promotional funds. These amounts are collected by the Company in its capacity as agent and must be used for promotional and advertising purposes, since the amounts are set aside to promote the respective banners for the franchisees' benefit. The promotional funds collected, and the related expenditures are reported on a gross basis in the consolidated statements of operations and comprehensive income. To the extent that promotional funds received exceed the related promotional expenditures, the excess contributions will be recorded in accounts payable or accrued liabilities.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

2. Significant accounting policies (continued)

Promotional funds (continued)

Cash held pursuant to the promotional funds received are classified as unrestricted cash as there are no legal restrictions on the use of these funds; however, the Company intends to use these funds solely to support the promotional funds rather than to fund its ongoing operations. As at November 30, 2023, promotional funds were in a net liability position amounting to \$11,357 (2022 – net liability position of \$13,278).

Subsequent events

Subsequent events were evaluated through the date that the consolidated financial statements were issued, which was February 2, 2024.

Estimates and assumptions

Business combinations

For business combinations, the Company must make assumptions and estimates to determine the purchase price accounting of the business being acquired. To do so, the Company must determine the acquisition date fair value of the identifiable assets acquired, including such intangible assets as franchise rights and master franchise rights, trademarks, step-in rights and liabilities assumed. Among other things, the determination of these fair market values involves the use of discounted cash flow analyses and future system sales growth. Goodwill is measured as the excess of the fair value of the consideration transferred including the recognized amount of any non-controlling interest in the acquiree over the net recognized amount of the identifiable assets acquired and liabilities assumed, all measured at the acquisition date. These assumptions and estimates have an impact on the asset and liability amounts recorded in the consolidated balance sheets on the acquisition date. In addition, the estimated useful lives of the acquired amortizable assets, the identification of intangible assets and the determination of the indefinite or finite useful lives of intangible assets acquired will have an impact on the Company's future profit or loss.

Goodwill and indefinite-lived intangible assets

The fair value calculation includes estimates of revenue growth, which are based on past performance and internal projections for the intangible asset group's forecasted growth, and royalty rates, which are adjusted for our particular facts and circumstances. The discount rate is selected based on the estimated cost of capital that reflects the risk profile of the related business. These estimates are highly subjective, the ability to achieve the forecasted cash flows used in our fair value calculations is affected by factors such as the success of strategic initiatives, changes in economic conditions, changes in our operating performance and changes in our business strategies.

Contingencies

The Company makes assumptions and estimations based on its current knowledge of future disbursements it will have to make in connection with various events that have occurred in the past and for which the amount to be disbursed and the timing of such disbursement are uncertain at the date of producing its financial statements. This includes contingencies for onerous contracts, litigations and disputes and other contingencies.

Gift card liabilities

Management is required to make certain assumptions in both the prorated recognition based on redemption pattern and remoteness recognition of gift card breakage. The significant estimates are breakage rate and the redemption patterns.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

3. Business acquisitions

I) Sauce Pizza and Wine (2023)

On December 15, 2022, the Company completed the acquisition of the assets of Sauce Pizza and Wine, an operator of fast casual restaurants operating in the state of Arizona in the US. As of the date of the acquisition, Sauce Pizza and Wine was operating 13 corporate-owned restaurants. The purpose of the transaction was to diversify the Company's range of offerings in the US.

The transaction included a purchase price totaling \$11,165 and a holdback on acquisition of \$798, as detailed below. The resulting aggregate cash outflow in connection with the Sauce Pizza and Wine acquisition was \$9,927.

	2023
	\$
Consideration paid:	
Purchase price	11,165
Working capital	(401)
Cash	23
Discount on non-interest-bearing holdback	(39)
Total consideration	10,748
Cash	(23)
Holdback	(798)
Net consideration paid/cash outflow	9,927

(In thousands of US dollars)

3. Business acquisitions (continued)

I) Sauce Pizza and Wine (2023) (continued)

The final purchase price allocation is as follows:

	2023
	\$
Net assets acquired:	
Current assets	
Cash	23
Inventories	183
Prepaid expenses and deposits	193
Topala expenses and deposits	399
Property, plant and equipment	3,821
Operating lease right-of-use assets	7,262
Intangible assets – Trademark	4,140
Goodwill (1)	3,658
	19,280
Current liabilities	
Accrued liabilities	78
Gift card liability	1,086
Current portion of operating lease liabilities	1,218
	2,382
Operating lease liabilities	6,150
	8,532
Net purchase price	10,748

⁽¹⁾ Goodwill is deductible for tax purposes.

Total expenses incurred related to acquisition costs amounted to \$160.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

3. Business acquisitions (continued)

II) Wetzel's Pretzels (2023)

On December 8, 2022, the Company completed the acquisition of all of the issued and outstanding shares of COP WP Parent, Inc. ("Wetzel's Pretzels"), a franchisor and operator of quick service restaurants operating in the snack category across 25 states in the US, as well as in Canada and Panama. As of the date of the acquisition, Wetzel's Pretzels was operating 328 franchised and 38 corporate-owned restaurants. The purpose of the transaction was to diversify the Company's range of offerings in the US.

The transaction included a purchase price totaling \$210,189, as detailed below. The resulting aggregate cash outflow in connection with the Wetzel's Pretzels acquisition was \$203,328. The transaction consideration also includes \$3,000 held in escrow contingent on the execution of several lease contracts within 12 months of acquisitions. As of December 8, 2023, only a portion of the contracts were executed and therefore \$1,600 was released from escrow and recorded as a gain in the statement of profit and loss.

	\$
Consideration paid:	
Purchase price	210,189
Total consideration	210,189
Cash	(6,861)
Net consideration paid/cash outflow	203,328

2023

3. Business acquisitions (continued)

II) Wetzel's Pretzels (2023) (continued)

The final purchase price allocation is as follows:

	2023
	\$
Net assets acquired:	
Current assets	
Cash	6,861
Accounts receivable	1,005
Inventories	265
Current portion of loans receivable	45
Prepaid expenses and deposits	757
Income taxes receivable	1,371
	10,304
Loans receivable	594
Property, plant and equipment	5,082
Operating lease right-of-use assets	21,931
Intangible assets – Franchise rights	35,600
Intangible assets – Trademarks	71,700
Goodwill (1)	118,447
	263,658
Current liabilities	
Accounts payable	911
Accrued liabilities	4,979
Promotional funds payable	431
Current portion of operating lease liabilities	936
Current portion of deferred revenue and deposits	67
Income taxes payable	547
	7,871
Operating lease liabilities	20,995
Deferred revenue and deposits	939
Deferred income taxes	23,664
	53,469
Net purchase price	210,189

⁽¹⁾ Goodwill is deductible for tax purposes.

Total expenses incurred related to acquisition costs amounted to \$320.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

3. Business acquisitions (continued)

III) BBQ Holdings (2022)

On September 27, 2022, the Company completed the acquisition of all of the issued and outstanding common shares of BBQ Holdings. BBQ Holdings is a franchisor and operator of casual and fast casual dining restaurants across 37 states in the US, Canada, and the United Arab Emirates. As of the date of the acquisition, BBQ Holdings was operating 198 franchised and 103 corporate-owned restaurants under 10 different brands. The purpose of the transaction was to diversify the Company's range of offerings in the US as well as to bring proficiency in operating corporate-owned restaurants.

The transaction included a purchase consideration totaling \$182,458, repayment of long-term debt of \$24,625 and early cash settlement of stock options and restricted stock units of \$10,204, as detailed below. The resulting aggregate cash outflow in connection with the BBQ Holdings acquisition was \$207,123.

	As previously		
	reported	Adjustments	2022
	\$	\$	\$
Consideration paid:			
Cash	207,083	_	207,083
Amount paid for early settlement of options	10,164	-	10,164
Cash amount paid for early settlement of options (1)	_	40	40
Total consideration	217,247	40	217,287

3. Business acquisitions (continued)

III) BBQ Holdings (2022) (continued)

The final purchase price allocation is as follows:

	As previously		
	reported	Adjustments	2022
	\$	\$	\$
Net assets acquired:			
Current assets			
Cash	20,595	-	20,595
Accounts receivable (1)	5,847	(537)	5,310
Inventories	3,853	-	3,853
Prepaid expenses and deposits (1)	1,348	271	1,619
Income taxes receivable	895		895
	32,538	(266)	32,272
Loans receivable (1)	143	87	230
Other assets (1)	180	(180)	-
Property, plant and equipment (1)	54,238	(983)	53,255
Operating lease right-of-use assets (1)	79,601	81	79,682
Intangible assets – Franchise rights (1)	8,130	(1,340)	6,790
Intangible assets – Trademarks (1)	121,440	(8,010)	113,430
Intangible assets – Other (1)	1,007	517	1,524
Goodwill (2)	52,484	14,933	67,417
	349,761	4,839	354,600
Current liabilities			
Accounts payable (1)	5,644	113	5,757
Accrued liabilities	17,502	(880)	16,622
Gift card liability (1)	7,609	4,798	12,407
Current portion of operating lease liabilities (1)	12,561	63	12,624
Current portion of deferred revenue and deposits (1)	425	(425)	
	43,741	3,669	47,410
Other liabilities (1)	605	63	668
Operating lease liabilities (1)	67,040	18	67,058
Deferred income taxes (1)	21,128	1,049	22,177
	132,514	4,799	137,313
Net purchase price	217,247	40	217,287

⁽¹⁾ The Company has recorded adjustments to its previously reported preliminary purchase price allocation reported in the annual 2022 financial statements. The adjustments relate to the fair values of accounts receivable, other assets, prepaid expenses and deposits, loans receivable, property, plant and equipment, operating lease right-of-use assets, franchise rights, trademarks, other intangible assets, accounts payable, gift card liability, current portion of deferred revenue and deposits, current portion of operating lease liabilities, operating lease liabilities, deferred income taxes and other liabilities.

Total expenses incurred related to acquisition costs during the year ended November 30, 2023 amounted to \$417 (2022 – \$3,566).

⁽²⁾ Goodwill is deductible for tax purposes.

4. Accounts receivable

Details of accounts receivable are as follows:

	2023	2022
	\$	\$
Total accounts receivable	31,738	26,974
Less: Allowance for credit losses	(2,277)	(2,013)
Total accounts receivable, net	29,461	24,961
Of which:		
Not past due	25,787	21,364
Past due for more than one day but no more than 30 days	430	677
Past due for more than 31 days but no more than 60 days	208	248
Past due for more than 61 days	3,036	2,672
Total accounts receivable, net	29,461	24,961
	2023	2022
	\$	\$
Allowance for credit losses, beginning of year	(2,013)	(2,063)
(Provision) recovery	(733)	97
Additions through business acquisition	(374)	(369)
Reversal of amounts previously written off	(2)	-
Write-offs	845	322
Allowance for credit losses, end of year	(2,277)	(2,013)

5. Assets held for sale

Assets held for sale as at November 30, 2023 and 2022 are stated at fair value less costs to sell and are comprised of one location's leasehold improvements, land and building that were acquired with BBQ Holdings and that were transferred from property, plant and equipment (Note 7). They did not meet the definition of assets held for sale as at the acquisition date of BBQ Holdings.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

6. Leases

Operating lease right-of-use assets

The following table provides the net carrying amounts of the operating lease right-of-use assets by class of underlying asset and the changes in the years ended November 30, 2023 and 2022:

	Offices and		
	stores	Other	Total
	\$	\$	\$
Balance as at November 30, 2021	100,769	103	100,872
Additions	1,948	s s	1,948
Additions through business acquisitions (Note 3)	79,208	393	79,601
Depreciation expense	(28,644)	(63)	(28,707)
Impairment charge	(247)	2 - 2	(247)
De-recognition/lease modifications	12,174	2 — 2	12,174
Balance as at November 30, 2022	165,208	433	165,641
Additions	7,115	160	7,275
Additions through business acquisitions (Note 3)	29,274		29,274
Depreciation expense	(40,972)	(278)	(41,250)
De-recognition/lease modifications	26,139	(5)	26,134
Balance as at November 30, 2023	186,764	310	187,074

The Company recorded sublease income from its operating lease right-of-use assets amounting to \$24,789 (2022 – \$24,414).

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

6. Leases (continued)

Operating lease liabilities

The following table provides the net carrying amounts of the operating lease liabilities and the changes in the years ended November 30, 2023 and 2022:

	2023	2022
	\$	\$
Operating lease liabilities, beginning of year	167,716	101,910
Additions	7,284	1,948
Additions through business acquisitions (Note 3)	29,380	79,601
Lease renewals and modifications	29,624	13,918
Lease terminations	(3,554)	(1,763)
Other adjustments	(201)	282
Interest expense	10,449	3,853
Payments	(50,205)	(32,033)
Operating lease liabilities, end of year	190,493	167,716
Recorded in the consolidated balance sheets as follows:		
	2023	2022
	\$	\$
Current portion	37,080	35,431
Long-term portion	153,413	132,285
	190,493	167,716

Maturity analysis

The following table sets out a maturity analysis of lease payments, showing the undiscounted lease payments to be paid after November 30, 2023:

	Leases	Expected sublease income
	\$	\$
2004	47.000	00.040
2024	47,680	23,049
2025	41,990	19,288
2026	35,406	15,232
2027	29,151	11,325
2028	22,532	7,672
Thereafter	52,003	9,385
Total undiscounted lease payments	228,762	85,951
Less: Unearned finance income	(38,269)	
Total present value of lease liabilities and expected sublease income	190,493	85,951

7. Property, plant and equipment

Cost	Equipment	Leasehold improve- ments	Rolling stock	Computer hardware	Land	Building	Total
	\$	\$	\$	\$	\$	\$	\$
Balance as at November 30,							
2021	3,117	642	40	1,147	-	_	4,946
Additions	596	682	_	538	_	14	1,830
Additions through business							
acquisition (Note 3)	18,275	25,400	_	2,407	3,925	4,231	54,238
Disposals	(522)	(11)	(16)	(94)	-	_	(643)
Impairment (Note 10)	(167)	(171)	_	r - r	_	-	(338)
Transfer to assets held for sale		(47)			(700)	(700)	(4.500)
Balance as at November 30,		(47)	_	· · · · · · · · · · · · · · · · · · ·	(780)	(736)	(1,563)
2022	21,299	26,495	24	3,998	3,145	3,509	58,470
Additions through business	21,200	20,400		0,000	0,140	0,000	00,470
acquisitions (Note 3)	2,014	6,242	(8)	(89)	75	(314)	7,920
Additions	6,215	5,978		1,007	_	50	13,250
Disposals	(578)	(155)	_	(125)	_	-	(858)
Impairment (Note 10)	(139)	(16)	=	(14)			(169)
Balance as at November 30,							
2023	28,811	38,544	16	4,777	3,220	3,245	78,613
		Leasehold improve-	Rolling	Computer			
Accumulated depreciation	Equipment	ments	stock	hardware	Land	Building	Total
	\$	\$	\$	\$	\$	\$	\$
							17
Balance as at November 30,							
2021	1,381	232	20	324		-	1,957
Depreciation	459	1,292	3	682	-	44	2,480
Disposals	(305)	(12)	(16)	(92)			(425)
Balance as at November 30,				_ , , , ,			
2022	1,535	1,512	7	914	_	44	4,012
Depreciation	3,362	7,381	(7)	524		98	11,358
Disposals	(282)	(65)		(9)			(356)
Balance as at November 30, 2023	4,615	8,828	_	1,429		142	15,014
		Leasehold					
	4	improve-	Rolling	Computer			
Carrying amounts	Equipment	ments	stock	hardware	Land	Building	Total
	\$	\$	\$	\$	\$	\$	\$
November 30, 2022	19,764	24,983	17	3,084	3,145	3,465	54,458
November 30, 2022	24,196	29,716	16	3,348	3,220	3,103	63,599
110101111111111111111111111111111111111		20,710	10	0,040	U,ZZU	0,100	00,000

8. Intangible assets

	Franchise			
Cost	rights	Trademarks	Other	Total
	\$	\$	\$	\$
Balance as at November 30, 2021	174,870	300,297	1,050	476,217
Additions	ř-	_	2,950	2,950
Additions through business acquisitions (Note 3)	8,130	121,440	1,007	130,577
Impairment (Note 10)	(1,776)	(3,867)		(5,643)
Balance as at November 30, 2022	181,224	417,870	5,007	604,101
Additions	_	_	565	565
Additions through business acquisitions (Note 3)	34,260	67,830	517	102,607
Disposals	_	_	(177)	(177)
Impairment (Note 10)	(559)	(3,504)		(4,063)
Balance as at November 30, 2023	214,925	482,196	5,912	703,033
Accumulated amortization	Franchise rights	Trademarks	Other	Total
	\$	\$	\$	\$
Balance as at November 30, 2021	66,514		556	67,070
Amortization	14,079	n c	552	14,631
Balance as at November 30, 2022	80,593	_	1,108	81,701
Disposals	10 - C-20	- V	(49)	(49)
Amortization	18,434		779	19,213
Balance as at November 30, 2023	99,027		1,838	100,865
Carrying amounts	Franchise rights	Trademarks	Other	Total
	\$	\$	\$	\$
November 30, 2022	100,631	417,870	3,899	522,400
November 30, 2023	115,898	482,196	4,074	602,168

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

9. Goodwill

The changes in the carrying amount of goodwill are as follows:

	2023	2022
	\$	\$
Goodwill, beginning of year	286,713	234,229
Amount recognized from business acquisitions (Note 3)	138,115	52,484
Goodwill, end of year	424,828	286,713
Accumulated impairment, beginning and end of year	(48,653)	(48,653)
Carrying amount	376,175	238,060

10. Impairment

The Company performed its annual impairment test as at August 31, 2023, resulting in the recognition of \$4,063 (2022 – \$5,643) of impairment losses on its intangible assets for four of its brands (2022 – three brands), following indicators of impairment that were noted.

Additionally, the Company recorded \$169 of impairment losses on its property, plant and equipment (2022 – \$338), and did not record any impairment on goodwill (2022 – nil), for a total of \$4,232 (2022 – \$5,981) of impairment charges on its property, plant and equipment and intangible assets for the year ended November 30, 2023, which have been recognized in the consolidated statement of operations.

Impairment charges were based on the amount by which the carrying values of the assets exceeded recoverable amounts, determined using expected discounted projected operating cash flows for trademarks and franchise rights.

Impairment by reporting unit for the year ended November 30, 2023:

Reporting Unit A	
Reporting Unit B	
Reporting Unit C	
Reporting Unit D	

	assets	Z 151 (\$11)	
Total	Trademarks	Franchise rights	Property, plant and equipment
\$	\$	\$	\$
3,845	3,264	559	22
-	Y - P - 2-1		
345	240		105
42			42
4,232	3,504	559	169

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

10. Impairment (continued)

Impairment by reporting unit for the year ended November 30, 2022:

		Intangible	assets	
	Property, plant and equipment	Franchise rights	Trademarks	Total
	\$	\$	\$	\$
Reporting Unit A	79	1,776	3,867	5,722
Reporting Unit B	259	_	_	259
Reporting Unit C		_	_	
	338	1,776	3,867	5,981

The key assumptions used, where the recoverable amount was measured as a reporting unit's fair value, are those related to projected operating cash flows, as well as the discount rates. The sales forecasts for cash flows were based on the subsequent fiscal year's budgeted operating results, which were prepared by management and approved by the Board, and internal forecasts for subsequent years, which were prepared by management and developed from the budgeted operating results.

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

10. Impairment (continued)

The following table presents the key assumptions used in the Company's impairment tests, as well as the recoverable amounts measured at fair value as at August 31, 2023 and 2022:

				2023		2022
(\$, except percentage data)	Reporting Unit A	Reporting Unit B	Reporting Unit C	Reporting Unit D	Reporting Unit A	Reporting Unit B
Discount rates after tax	10.5%	10.5%	10.5%	10.5%	10.3%	10.3%
Discount rates pre-tax	13.4%	13.8%	13.7%	13.6%	13.1%	13.2%
Recoverable amounts	539,407	266,604	313,644	270,246	515,478	250,715

Long-term growth rates ranging from 0% to 2% (2022 – 0% to 2%) were used in the impairment test for Reporting Unit A. A change of 100 basis points in discount rates in Reporting Unit A would result in additional impairment charges on intangible assets of three brands (2022 – four brands) representing 0.7% (2022 – 0.5%) of the total carrying value of the franchise rights and trademarks in that unit. A change of 100 basis points in discount rates in Reporting Unit A would not result in additional impairment charges on goodwill for the years ended November 30, 2023 and 2022. For Reporting Unit A, an increase of 440 basis points (2022 – 320 basis points) in the discount rate would have resulted in its recoverable amount being equal to its carrying value.

A long-term growth rate of 1.5% (2022 – 1.5%) was used in the impairment test for Reporting Unit B. A change of 100 basis points in discount rates in Reporting Unit B would not result in additional impairment charges on intangible assets or goodwill for the years ended November 30, 2023 and 2022. For Reporting Unit B, an increase of 200 basis points (2022 – 110 basis points) in the discount rate would have resulted in its recoverable amount being equal to its carrying value.

Long-term growth rates of 2.0% were used in the impairment test for Reporting Unit C. A change of 100 basis points in discount rates in Reporting Unit C would not result in additional impairment charges on intangible assets or goodwill for the years ended November 30, 2023. For Reporting Unit C, an increase of 200 basis points in the discount rate would have resulted in its recoverable amount being equal to its carrying value. For the year ended November 30, 2022, the impairment test of Reporting Unit C was performed as at November 30, 2022 and was based on qualitative factors, which did not give rise to any indications of impairment.

A long-term growth rate of 2.0% was used in the impairment test for Reporting Unit D. A change of 100 basis points in discount rates in Reporting Unit D would not result in additional impairment charges on intangible assets or goodwill for the year ended November 30, 2023. For Reporting Unit D, an increase of 160 basis points in the discount rate would have resulted in its recoverable amount being equal to its carrying value.

11. Gift card liability

The changes in the carrying amount of the gift card liability are as follows:

Gift card liability, beginning of year
Activations
Redemptions
Gift card liability acquired and purchase price allocation adjustments (Note 3)
Deferred program fees and other
Gift card breakage recorded
Gift card liability, end of year

2023	2022
\$	\$
91,453	76,097
52,873	37,978
(39,603)	(25,168)
5,884	7,609
(1,065)	(873)
(4,684)	(4,190)
104,858	91,453

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

12. Deferred revenue and deposits

	2023	2022
	\$	\$
Franchise fee deposits	28,948	25,246
Unearned rent	1,794	1,880
Supplier contributions and other allowances	1,786	3,733
	32,528	30,859
Current portion	(7,775)	(7,571)
	24,753	23,288

Deferred revenues consist mostly of initial, transfer and renewal franchise fees paid by franchisees, as well as upfront fees paid by master franchisees, which are generally recognized on a straight-line basis over the term of the related agreement. Deferred revenues also include amounts paid in upfront fees received from agreements with suppliers, which are amortized over the term of the related agreement.

\$8,055 (2022 - \$7,598) of revenue recognized in the current year was included in the deferred revenue balance at the beginning of the year.

The following table provides estimated revenues expected to be recognized in future years related to performance obligations that are unsatisfied as at November 30, 2023:

Estimate for fiscal year:	\$
2024	7,775
2025	3,416
2026	3,066
2027	2,614
2028	2,026
Thereafter	13,631
	32,528

13. Receivables and advances from ultimate parent, parent company and companies under common control

The receivable from ultimate parent is primarily comprised of a renewable one-year term loan facility of \$148,670 (2022 – \$126,128). The term loan is unsecured and bears interest at the rate set at the Applicable Federal Rates as issued by the US Internal Revenue Service for short-term instruments. The term loan may be repaid, in whole or in part, at any time, without premium or penalty. MTY USA may also request repayment, in whole or in part, at any time, without penalty.

The advance from ultimate parent was repaid in conjunction with the issuance of another loan that makes up a portion of the interest bearing loans totaling \$405,833 seen in note 14 (2022 – \$198,785). Those loans are due December 8, 2029, are unsecured and bear interest at 9.26%. The term loans may be repaid, in whole or in part, at any time, without premium or penalty.

The receivable from company under common control and advances from parent company and companies under common control are non-interest bearing and receivable/due on demand with no specified collection/repayment terms.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

14. Long-term loan from company under common control

Interest-bearing loan at 5.4%, repayable by November 27, 2026 ⁽¹⁾
Two interest-bearing loans at 9.26%, repayable by December 8, 2029

2023	2022
\$	\$
299,850	299,850
405,833	-
705,683	299,850

⁽¹⁾ This loan is subject to a maximum debt-to-EBITDA ratio of 6.00:1.00 starting on November 27, 2019, and a minimum EBITDA interest coverage ratio of 2.00:1.00 to be calculated in conjunction with interest payments based on the past 12 months.

15. Holdback payable

Non-interest-bearing holdback on acquisition of Sauce Pizza and Wine, repayable in December 2023 (Note 3)

Non-interest-bearing holdback on acquisition of the assets of two restaurants located in California, repayable within the next 12 months

Current portion of holdback payable

2023	2022
\$	\$
837	i.—.
75	-
912	-

16. Contingencies

The contingencies for litigation and disputes represent management's best estimate of the outcome of litigations and disputes that are ongoing at the date of the consolidated balance sheet. These contingencies consist of multiple items, a large part of which are insurance claims, including worker's compensation claims, at the Company's corporately owned locations. The timing of the settlement of these contingencies is unknown given their nature, as the Company does not control the litigation timelines.

	2023	2022
	\$	\$
Provision for litigations, disputes and other contingencies, beginning of year	724	966
Reversals	(304)	(421)
Amounts used	(8,797)	(56)
Additions	11,226	235
Provision for litigations, disputes and other contingencies, end of year	2,849	724

17. Common stock

	2023			2022	
	Shares issued	\$	Shares issued	\$	
Balance, beginning and end of year	15	179,154	15	179,154	

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

18. Financial instruments

In the normal course of business, the Company uses various financial instruments which by their nature involve risk, including market risk and the credit risk of non-performance by counterparties. These financial instruments are subject to normal credit standards, financial controls, risk management as well as monitoring procedures.

Fair value of recognized financial instruments

The Company has determined that the fair value of its financial assets and financial liabilities with short-term maturities approximates their carrying value. These financial instruments include cash, restricted cash, accrued liabilities, promotional funds payable, and advances from parent company and from companies under common control.

The table below shows the fair value and the carrying value of other financial instruments as at November 30, 2023 and 2022. Since estimates are used to determine fair value, they must not be interpreted as being realizable in the event of a settlement of the instruments.

	2023		2022
Carrying amount	Fair value	Carrying amount	Fair value
\$	\$	\$	\$
423	423	1,004	1,004

Financial assets
Loans receivable

Determination of fair value

The following methods and assumptions were used to estimate the fair values of each class of financial instrument:

Cash, restricted cash, accounts receivable, receivable from ultimate parent, deposits, accounts payable, holdback payable, accrued liabilities, advance from parent company, advances from companies under common control – The carrying amounts approximate fair values due to the short maturity of these financial instruments.

Risk management policies

The Company, through its financial assets and financial liabilities, is exposed to various risks. The following analysis provides a measurement of risks as at November 30, 2023.

Credit risk

The Company's credit risk is primarily attributable to its trade receivables, loans receivable and deposits. For accounts receivable and loans receivable, the amounts disclosed on the consolidated balance sheets are net of allowances for bad debts, estimated by the Company's management based on past experience and counterparty specific circumstances. The Company believes that the credit risk of accounts receivable is limited for the following reasons:

- The Company's broad client base is spread mostly across the US, which limits the concentration of credit risk.
- The Company accounts for specific bad debt provisions when management considers that the expected recovery is less than the actual account receivable.

The credit risk on cash is limited because the Company invests its excess liquidity in high-quality financial instruments and with credit-worthy counterparties.

The credit risk on deposits is also limited as these are mostly with well-established and credit-worthy companies.

Interest rate risk

Interest rate risk is the Company's exposure to increases and decreases in financial instrument values caused by the fluctuation in interest rates. The Company has limited exposure to interest rate risk as its long-term loans from company under common control and ultimate parent as well as its long-term debt have fixed interest rates

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

19. Revenue

- 1	_				
	-or	tho	vear	and	חם

	November 30, 2023		November 30, 2022			
	Franchising	Corporate	Total	Franchising	Corporate	Total
	\$	\$	\$	\$	\$	\$
Corporate store revenues		337,937	337,937	_	67,015	67,015
Royalties	128,461	<u> </u>	128,461	99,637	_	99,637
Franchise fees, transfer fees and master license fees	4,996		4,996	5,203	_	5,203
Promotional funds	56,734	_	56,734	51,867	-	51,867
Program allowances	32,499	_	32,499	20,619	_	20,619
Breakage income	4,684	_	4,684	4,190	_	4,190
Resale material and retail sales	5,754	_	5,754	8,587	_	8,587
Other	9,215		9,215	6,568	_	6,568
	242,343	337,937	580,280	196,671	67,015	263,686

20. **Operating expenses**

For the year ended

			I OF LITE Y	ai ciided		
	November 30, 2023		November 30, 2022			
	Franchising	Corporate	Total	Franchising	Corporate	Total
	\$	\$	\$	\$	\$	\$
Cost of goods sold	2,284	94,465	96,749	2,994	21,494	24,488
Wages and benefits	54,590	115,976	170,566	36,336	24,271	60,607
Advertising, marketing and promotion	462	5,311	5,773	442	2,130	2,572
Rent	4,369	35,294	39,663	2,964	7,096	10,060
Professional and consulting fees and commission	12,054	4,394	16,448	14,942	197	15,139
Office, travel, meals and entertainment and utilities	13,302	37,713	51,015	9,099	7,273	16,372
Promotional funds	56,734	_	56,734	51,867	-	51,867
Gift card program costs	6,695	_	6,695	6,296	_	6,296
Other (1)	2,342	25,474	27,816	1,573	5,640	7,213
Bad debt expense (recovery)	438	250	688	(70)	120	50
	153,270	318,877	472,147	126,443	68,221	194,664

Other operating expenses are comprised mainly of other office administration expenses.

Franchising operations

The franchising business mainly generates revenues from royalties, supplier contributions, franchise fees and rent.

Corporate store operations

Corporate stores generate revenues from the direct sale of prepared food to customers.

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

21. Interest expense

	2023	2022
	\$	\$
Interest charged by ultimate parent company (Note 24)		1,943
Interest charged by companies under common control (Note 24)	53,977	16,192
Interest expense	53,977	18,135

22. Income taxes

The Company accounts for income taxes in accordance with ASC 740. ASC 740 prescribes the use of the liability method whereby deferred tax asset and liability account balances are determined based on differences between the financial reporting and tax bases of assets and liabilities and are measured using the enacted tax rates. The effects of future changes in tax laws or rates are not anticipated.

Under ASC 740, income taxes are recognized for the following: a) amount of tax payable for the current year and b) deferred tax liabilities and assets for future tax consequences of events that have been recognized differently in the consolidated financial statements than for tax purposes.

Income tax expense (recovery)	2023	2022
	\$	\$
Current tax expense	7,807	9,911
Deferred tax recovery	(6,405)	(2,496)
Total tax expense	1,402	7,415

The provision for income taxes recorded in the consolidated financial statements differs from the amount which would be obtained by applying the statutory federal income tax rate of 21% (2022 – 21%) to the income for the period as follows:

	2023	2022
	\$	\$
Income before income taxes	18,380	26,988
Income tax expense at federal statutory rate	3,860	5,679
State and local income taxes net of federal tax benefit	2,360	2,457
Non-deductible/non-taxable items	(216)	549
Temporary difference for which no deferred tax asset is recognized		(44)
True-up of prior year tax provision	(2,239)	(598)
Rate variation on deferred income tax	(102)	(552)
Credits generated and used in current year	(2,371)	286
Other	110	(362)
Income tax expense	1,402	7,415

Notes to the consolidated financial statements

Years ended November 30, 2023 and 2022

(In thousands of US dollars)

22. Income taxes (continued)

Components of the net deferred tax asset (liability):

	2023	2022
	\$	\$
Inventorias	(004)	40
Inventories	(264)	42
Allowance for credit losses	595	228
Deferred revenue and deposits	4,725	4,482
Gift card liability	2,947	16,835
Accrued liabilities	33,567	6,658
Non-capital losses and other tax credits	4,570	12,832
Other	(1,285)	98
Operating lease liabilities	48,406	42,998
Total deferred tax assets	93,261	84,173
Deferred costs	(948)	(859)
Property, plant and equipment	(11,490)	(12,420)
Operating lease right-of-use assets	(47,497)	(42,260)
Intangible assets	(143,345)	(120,483)
Holdback payable	(233)	(95)
Total deferred tax liabilities	(203,513)	(176,117)
Net deferred tax liability	(110,252)	(91,944)

23. Supplemental cash flow information

During the year ended November 30, 2023, the Company paid \$8,268 (2022 – \$8,894) in income taxes. Furthermore, there are non-cash items included in the proceeds on disposition amounting to a net liability of \$1,574 (2022 – net asset of \$25). The non-cash items were primarily related to commitments made as part of the disposal of a portfolio of corporately-owned locations.

24. Related party transactions

The Company has transactions in the normal course of business with its ultimate parent, parent company and companies under common control. These transactions were in the normal course of business and measured at the exchange amount, which is the amount of consideration established and agreed to by the related parties.



EXHIBIT C

Manchu WOK FRANCHISE AGREEMENT



MTY FRANCHISING USA, INC. EXHIBIT C FRANCHISE AGREEMENT

Franchise #:	
Franchisee:	
Date:	
Traditional Restaurant (YES or NO): _	
Territory:	
-	

TABLE OF CONTENTS

Section	<u>on</u>	<u>Page</u>
1.	DEFINITIONS	2
2.	TERM AND GRANT OF FRANCHISE	5
3.	FEES PAYABLE BY THE FRANCHISEE	10
4.	DEVELOPMENT AND OPENING OF THE PREMISES	15
5.	DUTIES AND RESPONSIBILITIES OF FRANCHISOR	21
6.	FRANCHISEE DUTIES, RESPONSIBILITIES AND OBLIGATIONS	24
7.	INSURANCE	33
8.	RECORDS AND REPORTING	35
9.	LICENSED ITEMS AND TRADEMARKS	38
10.	DEFAULT AND TERMINATION OF FRANCHISE	40
11.	RESTRICTIVE COVENANTS	47
12.	ASSIGNMENT AND TRANSFER	50
13.	GENERAL CONTRACT PROVISIONS	55
14.	DISPUTE RESOLUTION	60
15.	GENERAL PROVISIONS	62
16.	REPRESENTATIONS AND WARRANTIES	63
17.	GENERAL PROVISIONS	65
Sched	<u>lules</u>	
Scheo Scheo Scheo Scheo	dule B Premises and Lease Description dule C Trade Area dule D Lease Addendum dule E Development Obligations	

FRANCHISE AGREEMENT

This Franchise Agreement is dated the	nis the " Effective
Date") between MTY FRANCHISING USA, INC.	, a Tennessee corporation, d.b.a. Manchu Wok
("Franchisor"), and	, a(n)
("Franchisee") and	an individual(s) (collectively
and individually "Guarantor").	

WHEREAS through the expenditure of time, effort and money by the Franchisor and its associated entities, the Franchisor has acquired experience and skill in the development, opening and operating of fast food restaurants utilizing the Manchu WOK Restaurant System (as hereinafter defined) specializing in the preparation, sale and merchandising of Chinese and Asian style food and other related products. These restaurants generally comprise two separate and distinct groups: (i) shopping mall food courts, and non-traditional locations; and (ii) stand-alone street-front locations:

WHEREAS the Manchu WOK Restaurant System, which the Franchisor has developed and continues to develop, includes, without limitation, standards and specifications for buildings, decor, equipment, equipment layouts, supplies and menus, quality and quantity standards, operating procedures for sanitation, maintenance, food and beverage storage, preparation and service, methods and techniques for inventory and cost controls, record keeping and reporting, personnel management, purchasing, sales, promotions and advertising;

WHEREAS The Franchisor is in the business, among other things, of operating restaurants and franchising restaurants that are opened or are to be opened and operated pursuant to franchise agreements and using the Manchu WOK Restaurant System;

WHEREAS the Manchu WOK Restaurant System uses the federally registered Trademarks and the unregistered Trademarks as listed in Schedule A annexed hereto;

WHEREAS the Manchu WOK Restaurant System also uses the trade name "Manchu WOK" and other trade names that include the words "Manchu WOK," all of which have been adopted and used to identify restaurants using the Manchu WOK Restaurant System and the services, food, beverages and other products sold or used in connection therewith; and

WHEREAS the Franchisee recognizes the benefits to be derived from a franchise issued by The Franchisor to be identified with the name "Manchu WOK" and recognizes the value of the Manchu WOK Restaurant System and the necessity of opening and operating the Business (as hereinafter defined) at the Premises in accordance with the Manchu WOK Restaurant System and pursuant to the covenants, terms and conditions set out in this Agreement; and

WHEREAS (If this Franchise Agreement is entered into under an Area Development Agreement) The Franchisee and the Guarantor have represented to the Franchisor that they are successful multi-unit franchisees of other food-based franchise systems, which representations to the knowledge of the Franchisee and the Guarantor have been relied upon by the Franchisor, and

WHEREAS as an inducement to the Franchisor to grant to the Franchisee the right to operate a Manchu WOK Restaurant at the Premises, the Guarantor has agreed to enter into this

Agreement to, among other things, unconditionally guarantee the performance and observance of the duties and obligations of the Franchisee;

NOW THEREFORE in consideration of the covenants, conditions and provisions herein contained and in consideration of the full and faithful performance by the Franchisee of each and every one of the covenants, terms and conditions herein contained, the parties hereto hereby covenant, warrant and agree as follows:

1. **DEFINITIONS**

1.1. Definitions

In this Agreement the following terms shall have the following meanings:

"Affiliate" means, with respect to a corporation or other business entity, any person controlling, controlled by, or under common control with that corporation or business entity. With respect to an individual, Affiliate means that individual's immediate family members, as well as such immediate family members' spouses, and the corporations or other business entities such individual and his or her family members, directly or indirectly, control.

"Agreement" or "Franchise Agreement" means this franchise agreement and all schedules and attachments appended to it and any amendments made to it.

"Approved Products" means all food and beverage ingredients and products, packaging and serving materials, paper goods, utensils, supplies, and other items of inventory, uniforms, marketing or promotional materials, Fixtures, Equipment, as well as all other items required for or used in any manner in the operation of Franchisee's Manchu WOK Restaurant, both in inventory and as finished goods for sale.

"Area Development Agreement" means the area development agreement, if any, entered by the Franchisee, or an affiliate, and the Franchisor pursuant to which the Franchisee is entering this Agreement.

"Business" means the Manchu WOK Restaurant conducted by Franchisee at the Premises.

"Contracts" means any and all of this Agreement, Lease, Offer to Lease, Sublease, any financing or security agreement, license or other contract, agreement, document or instrument to which Franchisee is party with Franchisor or Franchisor's Affiliates or any of them (with or without other parties) including, unless otherwise specifically excluded, a contract relating to premises other than the Premises, any Area Development Agreement, or to any right, license or franchise other than this Franchise Agreement relating to any Manchu WOK Restaurant. A "Contract" means any of the Contracts.

"Equipment" means all the equipment necessary for the operation of a Manchu WOK Restaurant that is not affixed to the premises.

"Fixtures" means tangible moveable or affixed goods, including equipment, fixtures, furnishings and signs.

"Franchise" is as defined in Section 2.1 below.

"Franchisee" means the person, firm or corporation who or which enters into this Agreement with the Franchisor, its or their permitted successors and assigns.

"Franchisee's Fixtures" and "Franchisee's Equipment" means Fixtures and Equipment used in connection with the Business or usually located on the Premises, including, without limitation, until immediately prior to the time of expiration or termination, with or without cause, of the Agreement (at which time there shall then be excluded) anything that is deemed by the Lease to either become part of the Premises or to be owned by the Landlord on expiration or termination (with or without cause) of the Lease.

"Franchisor" means MTY Franchising USA Inc., its successors and assigns.

"Guarantor" means the person or persons who enter into this Agreement to personally guarantee all of the obligations of the Franchisee and who incur thereby certain personal obligations as set out in this Agreement.

"Gross Sales" means all Franchisee's gross receipts from its Business operations, including, without limitation, sales, charges, fees, orders taken, services, gift certificates, concessions, licenses, business interruption insurance and all other revenues of every kind and nature, in, from, about or by reason of the operation of the Business or the use or occupation of the Premises, whether evidenced by check, cash, credit, charge accounts, exchange or otherwise and whether or not made by means of mechanical or other vending machines or devices in or on the Premises, including, but not limited to, telephones, cigarette or confectionery machines or entertainment game machines but excludes bona fide credits and refunds upon returns of merchandise and any sales or use taxes, goods and services taxes, or other similar taxes imposed by any governmental agency on sales of products or services made or rendered by Franchisee and actually remitted by it to the applicable governmental agency. There shall be no deductions allowed for uncollected or uncollectible credit accounts and no allowances shall be made for bad debts. The revenue derived from the sale of any gift certificate, promotional coupon or other right or option shall be included in Gross Sales at the time it is originally sold and shall not be included again when it is redeemed and the original document evidencing same shall be submitted to Franchisor on request. Payments made upon installment or credit shall be treated as a sale for the full price in the week during which the related charge or sale was made, irrespective of the time when Franchisee shall receive payment (whether full or partial) therefor.

"Initial Franchise Fee" is as defined in Section 3.1 below.

"Landlord" means the landlord of the Premises as described in Schedule B hereto.

"**Lease**" means the documents described in **Schedule B** hereto pursuant to which the Franchisee is the tenant of the Premises and includes an Offer to Lease.

"Licensed Items" means: (a) the Trademarks; (b) trademarks, trade names, logo types, trade symbols, emblems, slogans, insignia, designs, logos and other commercial symbols (including, to the extent so used, telephone numbers); (c) Proprietary Products; (d) any and all works in any medium in which Franchisor has right, title or interest by way of copyright, patent, inventor's rights or other form of intellectual or industrial property, whether registered or not, that are now or in the future designated or provided by Franchisor for use in connection with the operation of the Business; (e) the Confidential Manual; (f) all building plans; (g) advertising and

marketing materials; and (h) all confidential or proprietary information or materials provided by Franchisor for use in connection with operation of the Business.

"Manager" means a person fully trained by or otherwise approved and accepted by the Franchisor and duly appointed by the Franchisee on Notice to the Franchisor to be fully responsible for the day-to-day operations of the Franchisee's Business.

"Manchu WOK Restaurant System" means a Chinese and Asian, fast-food-restaurant-business system, including, without limitation, the uniform standards, methods, procedures and specifications designated and approved by Franchisor regarding the construction, operation and management of a Manchu WOK restaurant and may be a modification or derivation in whole or in part of or from the Manchu WOK Restaurant System as established by the Franchisor for other Manchu WOK restaurants (such as those operated in street-front locations), including, without limitation, building plans, Fixtures and Equipment, identification schemes and uniforms; menu, products and services; techniques of product preparation, proprietary products and provision of services; business, accounting, supply, management, operations and marketing programs, techniques and systems and Licensed Items. When used in the plural, the term "Manchu WOK Restaurant Systems" shall be deemed to include both the Manchu WOK Restaurant System and the Manchu WOK Street-Front Restaurant System.

"Manchu WOK Street-Front Restaurant" means a fast-food-restaurant business carried on at a street-front location in accordance with the Manchu WOK Restaurant System.

"Marketing Fee" is as defined in Section 3.5 below.

"Offer to Lease" means any offer accepted by or any agreement with, a landlord for the use of Premises as a Manchu WOK Restaurant.

"Confidential Manual" means all operations, administration and managers' manuals, including but not limited to the operating manual ("Operations Manual"), and all books, computer programs, pamphlets, memoranda, recipes and other publications prepared by or on behalf of Franchisor or its Affiliates and as may be added to, changed, modified, withdrawn or otherwise revised from time to time by Franchisor or its Affiliates setting out the standards, methods, procedures and specifications of the Manchu WOK Restaurant System. All references in the Confidential Manual to a "Manchu WOK Restaurant" or to a "Manchu WOK Outlet" shall be taken to apply, unless specifically directed by the Franchisor or unless the context otherwise requires, to a Manchu WOK Restaurant.

"Payments" all payments to the Franchisor pursuant to this Agreement shall be paid in US Funds.

"POS System" means the point-of-sale system approved, from time to time, for use by the Franchisor and includes, without limitation, the computer, electronic cash register and computer terminals referred to in Section 6.9 below.

"Premises" means the premises more particularly described in **Schedule** "B" annexed hereto.

"Proprietary Products" means products (including, without limitation, specially formulated sauces and menu items) developed by the Franchisor for the Manchu WOK Restaurant System.

"Royalty" is as defined in Section 3.4 below.

"Restaurant" or "Manchu WOK® Restaurant" means the Manchu WOK® restaurant which may be located in a street front location; Shopping Mall Food Court or at a Non-Traditional Location; Non Traditional Locations include highway facilities, airport concourses, professional sports facilities, recreational and entertainment facilities, college campuses, military installations, governmental or institutional locations, supermarkets, and department stores.

"Term" has the meaning set out in Section 2.1 below.

"Trademarks" or "Marks" means the registered and unregistered trademarks described in *Schedule "A"* annexed hereto and otherwise and referred to in Section 9 hereof.

"Trade Area" (if applicable) means the area described in Schedule "C" annexed hereto.

1.2. Table of Contents and Headings

The table of contents, the division of this Agreement into articles and sections and the insertion of articles and section headings are for convenience of reference only and shall not affect the interpretation or construction of this Agreement.

1.3. Number and Gender

In this Agreement the singular number includes the plural and vice versa, the use of any gender includes all genders, and the word "person" includes an individual, a trust, a corporation, a limited liability company, a partnership, an association and any other incorporated or unincorporated organization or entity. Words commencing with "here" (including without limitation "herein" and "herewith" and "hereof" shall, unless the context clearly indicates to the contrary, refer to the whole of this Agreement (including its schedules) and not to any particular article or section. "Including " means "including, without limitation." All forms of the verb "include" shall bear that meaning as well.

1.4. Time

Time is of the essence of this Agreement. When calculating the period of time within which or following which any act is to be done or step taken pursuant to this agreement, the day which is the reference date for calculating the period shall be excluded. If the last day of the period is a non-business day, then the period shall end on the next business day.

2. TERM AND GRANT OF FRANCHISE

- 2.1. This Agreement will commence on the Commencement Date and continue as set forth in either Section 2.1a. or 2.1b. below ("Term"):
 - (a) If you are purchasing a new or existing non-operating Business, the Term will expire on either: (1) the ten (10) year anniversary of the date you open this Business

to the public if you own the property where this Business is located or if you enter into a lease directly with the landlord or other third-party for the property where this Business is located; or (2) if you have entered into a sublease with one of our affiliates, the expiration of the term of the sublease for the property where this Business is located excluding any extensions or renewal options, unless terminated earlier in accordance with Section 9 or any other provisions of this Agreement; a successor franchise agreement is executed in accordance with Section 2.2, or transferred in accordance with Section 12: or

- (b) If you are purchasing an existing and operating Business, the Term will expire on either: (1) the ten (10) year anniversary of the Effective Date if you own the property where this Business is located or if you enter into a lease directly with the landlord or other third-party for the property where this Business is located; or (2) if you have entered into a sublease with one of our affiliates, the expiration of the term of the sublease for the property where this Business is located excluding any extensions or renewal options, unless: terminated earlier in accordance with Section 9 or any other provisions of this Agreement; a successor franchise agreement is executed in accordance with Section 2.2; or transferred in accordance with Section 12.
- 2.2. No Additional Rights Beyond Term. Franchisee does not have any rights to operate the Business beyond the Term, subject to any state laws to the contrary. Notwithstanding the foregoing, Franchisor reserves the right to offer to Franchisee the opportunity to extend its rights to operate the Business by entering into a successor franchise agreement ("Successor Franchise Agreement") for a five-year term and upon such other terms as Franchisor is offering new franchisees when Franchisee's Term is expired. Franchisee acknowledges and agrees that the terms and conditions of any such Successor Franchise Agreement may differ substantially from those in this Agreement. Franchisee also acknowledges that, should Franchisee be offered the opportunity to sign a Successor Franchisee Agreement, Franchise will be required to pay a Successor Franchise Fee of \$15,000.
- 2.3. Interim Period. After the expiration of this Agreement, if a period of time transpires before (a) Franchisee's rights under this Agreement have been fully and finally terminated or (b) Franchisor has offered a Successor Franchise Agreement to Franchisee and Franchisee has not yet signed such Successor Franchise Agreement, and Franchisee continues to accept the benefits of this Agreement, then at the option of Franchisor, this Agreement may be treated either as (a) expired as of the date of expiration with Franchisee then operating without a franchise to do so and in violation of Franchisor's rights; or (b) continued on a month-to-month basis ("Interim Period") until one party provides the other with written notice of such party's intent to terminate the Interim Period, in which case the Interim Period will terminate thirty (30) days after receipt of the notice to terminate the Interim Period. In the latter case, all obligations of Franchisee shall remain in full force and effect during the Interim Period as if this Agreement had not expired, and all obligations and restrictions imposed on Franchisee upon expiration of this Agreement shall be deemed to take effect upon termination of the Interim Period.
- 2.4. Grant of License. On and subject to the terms and conditions contained in this Agreement, the Franchisor grants to the Franchisee the following (collectively, the "License"):

- (a) an exclusive license to carry on the Business only at the Manchu WOK Restaurant under the name MANCHU WOK® or such other Marks as may be specified from time to time by the Franchisor during the Term or any Interim Period;
- (b) a non-exclusive license to use and display the Marks during the Term and any Interim Period in connection with the operation, advertising and promotion of the Business; and
- (c) a non-exclusive license to use, in connection with the Manchu WOK Restaurant, the System.

The License shall only remain in effect during the Term and so long as the Franchisee continues to operate the Business and does not commit an Event of Default under this Agreement. Upon Termination, such license shall immediately be at an end and the Franchisee shall immediately cease the use of the Marks and the System and shall not thereafter in any way hold itself out as being associated with the Franchisor or the restaurant operations owned or licensed by the Franchisor.

Notwithstanding the foregoing, the Franchisee expressly acknowledges that the Franchisor expressly reserves the right to:

- (i) identify the Territory around the Manchu WOK Restaurant for which Franchisee's franchise rights will be granted;
- (ii) establish, operate or license to any other person or entity the right to establish or operate, a restaurant owned or licensed by the Franchisor at any location outside of the Territory;
- (iii) develop, market, own, operate or participate in any other business under the Marks or any other trademarks;
- (iv) develop, lease and license the use of, at any location inside or outside the Territory, trademarks other than the Marks licensed by Franchisor under this Agreement, in connection with the operation of a system which offers products or services which are the same as or similar to those offered under the System on any terms or conditions which the Franchisor deems advisable;
- (v) merge with or be acquired by any other business, including a business that competes with Franchisee's Business, or to acquire and convert to the System operated by the Franchisor any retail stores, including retail stores operated by competitors located inside or outside the Territory or otherwise operated independently or as part of, or in association with, any other system or chain, whether franchised or corporately owned;
- (vi) distribute, sell or license other persons to distribute or sell non-System products and System products whether inside or outside the Territory through all other channels. "Other Channels" means locations other than traditional restaurants owned or franchised or licensed by the Franchisor and includes sale by or through other channels of trade including, without limitation, kiosks, carts, grocery stores, convenience stores, food chains, electronic mail,

Internet sales, malls, universities, schools, hospitals, military bases, casinos, convention centers, arenas, stadiums, health and fitness facilities, office buildings, theme parks, movie theatres, and amusement facilities; and

(vii) implement multi-area marketing programs (including, without limitation, mail drops and other flyer distribution methods) which may allow Franchisor or others to solicit or sell to customers anywhere, including without limitation inside or outside of the Territory. Franchisor also reserves the right to issue mandatory policies to coordinate such multi-area marketing programs.

Each of the Franchisee and the Guarantor further agree to execute and provide the Franchisor with the acknowledgment in substantially the form attached hereto as **Schedule H**.

Provided that the Franchisee does not commit an Event of Default under this Agreement including, without limitation, the Confidential Manual, the Lease or any other agreement between the Franchisee and the Franchisor, during the Term or any Interim Period, the Franchisor shall not operate a restaurant identified by any of the Marks within the Territory, nor shall the Franchiser authorize any other franchisee to operate a restaurant identified by any of the Marks within the Territory during the Term or any Interim Period.

Franchisee expressly agrees that all right, title and interest in and to the Marks, the System, the goodwill associated with the System and confidential trade secrets are owned by the Franchisor or its affiliates, and shall remain solely owned by Franchisor or its affiliates, and are being revealed to Franchisee solely to enable Franchisee to establish and operate a Manchu WOK franchise only in accordance with the terms and conditions of this Agreement.

2.5. Use of Premises; Trade Area

Franchisee is only permitted to conduct and operate a Manchu WOK Restaurant at the Premises and, except as otherwise may be agreed to in writing by the Franchisor, Franchisee shall not be entitled to operate a Manchu WOK Restaurant at any other location. Franchisee acknowledges that the Franchise is non-exclusive and that Franchisor shall have the absolute and unconditional right in Franchisor's absolute discretion to construct and operate and to permit others to construct and operate, other Manchu WOK Restaurants and Street-Front restaurants outside the Trade Area (or other restaurants or outlets using the Trademarks or the Manchu WOK Restaurant System which are not Restaurants wherever situated both within and outside of the Trade Area) at any other location and to use and permit others to use Licensed Items at any other location, in each case on such terms and conditions as Franchisor deems acceptable. Franchisor retains the unrestricted right on behalf of itself and its affiliates to distribute any and all food products, including, without limitation, Proprietary Products and services, to the public or others using either the Trademarks or trademarks other than the Trademarks. Franchisor shall not be accountable in any way to Franchisee, regardless of the proximity of any other Manchu WOK Restaurant outside the Trade Area or other restaurant using the Trademarks or the Manchu WOK Restaurant System wherever situate (whether operated by Franchisor or by a franchisee or licensee of Franchisor) to the Premises. Notwithstanding anything in this Agreement to the contrary, Franchisor reserves the absolute, unconditional right to operate, or grant third-parties the right to operate, Manchu WOK Street-Front restaurants either within or without the Trade Area. The Franchisee shall not solicit customers residing outside its Trade Area nor provide off-premises services including without limitation home delivery to customers.

Without limiting the generality of the foregoing paragraph, during the Term, for so long as the Franchisee is not in default of any Contract, the Franchisor shall not operate, or grant a third party the right to operate, another Manchu WOK Restaurant within the Trade Area (if applicable) .

The Franchise (including, the license of the Licensed Items) granted to the Franchisee has limited exclusivity. In addition to the Franchisor's right to use and grant others the right to use the Licensed Items outside the Trade Area, all rights not expressly granted in this Agreement to the Franchisee concerning the Licensed Items or any other matters are reserved for the Franchisor, including the right to:

- (a) establish, develop, and license or franchise other systems different from the System licensed by this Agreement, within or outside the Trade Area, without offering or providing the Franchisee any rights in, to, or under such other systems;
- (b) sell, within or outside the Trade Area, through dissimilar channels of distribution, such as retail grocery stores, under any terms that the Franchisor deems appropriate, Manchu WOK products and services (including Proprietary Products) similar or identical to those authorized for the Business, using the Trademarks; and
- (c) the unconditional right to operate, or grant third-parties the right to operate Manchu WOK® Street-Front restaurants either within or without the Trade Area.

2.6. System Changes and Modification

Franchisee acknowledges and agrees that the Manchu WOK Restaurant System may continue to evolve in order to effect improvements therein and also to reflect changing markets and accommodate new or changing consumer demands. Franchisee therefore further agrees that:

- (a) Franchisee will, immediately upon receipt of any and all updates and changes to the Confidential Manual, implement the modifications or alterations to the Manchu WOK Restaurant System that are required by such updates and changes;
- (b) upon receipt of written notice from Franchisor, Franchisee will promptly proceed to implement, provide for and use, and to fully complete within thirty (30) days following receipt of Franchisor's notice (or such longer period of time as may be specified in such notice) the implementation of any and all other modifications or alterations to the Manchu WOK Restaurant System as Franchisor, in the good faith exercise of its judgment from time to time, designates as necessary or advisable, including, without limitation, the adoption and use of new Licensed Items, products, techniques and Fixtures; and
- (c) changes and modifications to and evolutionary developments of the Manchu WOK Restaurant System may be introduced by the Franchisor for testing purposes or otherwise in Manchu WOK Restaurants other than that of the Franchisee. Franchisee shall not be entitled to introduce any such change(s) or modification(s) or evolutionary development(s) in its Manchu WOK Restaurant unless and until Franchisor has directed that it is appropriate for Franchisee so to do; in which event Franchisee

shall introduce any such change or modification to its Manchu WOK Restaurant with thirty (30) days of such notification by the Franchisor.

2.7. Acknowledgement and Importance of Compliance

Franchisee acknowledges that the Manchu WOK Restaurant System is an integrated food service system for the retailing of a limited menu of uniform and quality food products, emphasizing prompt and courteous service in a clean, wholesome atmosphere that is intended to be attractive to persons of all ages. The basis for the valuable goodwill and public acceptance of all restaurants utilizing the Manchu WOK Restaurant System, including Franchisee's Manchu WOK Restaurant, is Franchisee's compliance with the Manchu WOK Restaurant System and the Contracts in conjunction with the use of Licensed Items. The essence of this Agreement is Franchisee's accountability for performance of the obligations contained in this Agreement and Franchisee's adherence to the Manchu WOK Restaurant System, including, but not limited to, serving only designated food and beverage products; strict adherence to designated food and beverage specifications and to Franchisor's prescribed standards of quality, preparation, service and cleanliness; the use of prescribed equipment; and methods of use of equipment. The provisions of this Agreement shall be interpreted to give effect to the intent of the parties stated in this Section 2.5 and so as to ensure that the Business is operated in strict conformity with the Manchu WOK Restaurant System as it now exists or may later be modified as set out in Section 2.4 or otherwise.

2.8. Atypical Arrangements

Franchisee acknowledges and agrees that Franchisor may modify the offer of its franchises to other franchisees or its ongoing relationship with other franchisees in any manner and at any time, which offers, agreements and/or modifications have or may have terms, conditions, and obligations which may differ from the terms, conditions, and obligations in this Agreement. The existence of different forms of agreement and the fact that existing or future franchisees may have different rights and obligations shall not in any manner eliminate, modify, or affect the duties of the parties to the Agreement to comply with the terms of this Agreement.

3. FEES PAYABLE BY THE FRANCHISEE

As consideration for Franchisor entering into this Franchise Agreement and the granting to the Franchisee a franchise for a Manchu WOK Restaurant, the Franchisee agrees to pay to the Franchisor, without notice or demand, the amounts set out in this Section together with any and all other amounts as may be payable by Franchisee from time to time pursuant to this Franchise Agreement. Receipt of any check, draft or other commercial paper shall not constitute payment until Franchisor shall have collected all the funds therefrom.

3.1. Initial Franchise Fee

Franchisee shall pay to Franchisor by certified check or bank draft, an Initial Franchise Fee in the amount of Thirty Thousand (\$30,000) Dollars ("Initial Franchise Fee"); provided that in the event that the initial term of the Lease for the Premises is greater than ten (10) years, the Term of this Agreement shall be extended so as to correspond with the initial term of the Lease (less one day) and the Initial Franchise Fee shall be increased by an amount equal to Three Thousand (\$3,000) Dollars for each year or part year of the Term in excess of ten (10) years. The Initial Franchise Fee shall be fully paid not later than the date of execution of this Franchise

Agreement, at which time such payment shall be deemed to have been fully earned by Franchisor. Except as otherwise expressly provided in this Franchise Agreement, Franchisee shall not be entitled to a refund of the whole or any part of the Initial Franchise Fee.

3.2. Development Construction Fees

- (a) In connection with the development of the Premises for the Term, if Franchisor supervises the construction and improvement of the Premises, Franchisee shall pay to Franchisor a fee equal to the greater of: (i) Two Thousand Five Hundred Dollars (\$2,500); and (ii) three percent (3%) of the total cost of the development of the Premises for construction projects for "Manchu WOK" store construction projects in North America (but not to exceed \$15,000); this fee is payable when construction is completed.
- (b) In connection with any remodeling or general refurbishing or upgrading of the Premises for any reason, including, without limitation, any remodeling carried out as part of the conditions applicable to any Successor Franchise Agreement, if Franchisor supervises such remodeling, Franchisee shall pay to Franchisor a fee equal to the greater of: (i) Two Thousand Five Hundred Dollars (\$2,500); and (ii) three percent (3%) of the total cost of the development of the Premises for construction projects for "Manchu WOK" store construction projects in North America (but not to exceed \$15,000); this fee is payable when renovation is completed.
- (c) In connection with the development of the Premises, you will pay us an architect fee ("Architect Fee") ranging from \$18,000 to \$20,000 depending upon the size and scope of the project. The Architect Fee will be due to us before we commence architectural drawings for your Restaurant. If you would like to hire your own architect, a written approval must be obtained from us and all drawings must be prepared in conformance with our specifications and submitted to us for review and approval. You will pay us a \$2,000 Plan Review Fee for the reviewing process.
- (d) In connection with the construction of the Premises, you will pay us a construction deposit ("Construction Deposit") equal to \$30,000 to cover our costs in association with assisting you in the construction of your Restaurant. We reserve the right to reduce or waive the Construction Deposit for any upgrade, refurbishment or renovation at an existing Restaurant.

3.3. Training Fee

Franchisee shall pay to Franchisor the training fees described in Section 5.2.

3.4. Royalty Fee

Franchisee shall pay to Franchisor (in such manner as Franchisor may, from time to time, prescribe in the Confidential Manual or otherwise in writing), on or before Thursday of each week during the Term of the Franchise Agreement, as a continuing fee for Franchisee's right to operate a Traditional Manchu WOK Restaurant at the Premises an amount equal to seven (7%) of the Gross Sales of Franchisee's Manchu WOK Restaurant for the seven (7) day period ending at the close of business on the Sunday preceding such Thursday or for a Non-Traditional Manchu WOK Restaurant at the Premises an amount equal to six (6%) of the Gross

Sales of Franchisee's Manchu WOK Restaurant for the seven (7) day period ending at the close of business on the Sunday preceding such Thursday ("Royalty"); provided that during such period as Franchisee is an Area Developer and continues to operate this Manchu WOK Restaurant pursuant to an Area Development Agreement, the Royalty rate referred to herein shall be reduced to five (5%) percent.

3.5. Marketing Fees

- (a) Franchisee shall pay to Franchisor, in the same manner and at the same time as the Royalty is payable, a marketing contribution ("**Marketing Fee**") to Franchisor's marketing and sales promotion fund ("**Fund**") in an amount equal to one per cent (1 %) of Gross Sales for the seven (7) day period ending at the close of business on the Sunday preceding such Thursday:
 - (b) Franchisee acknowledges and agrees that:
 - (i) the Fund shall be used by Franchisor, as determined in its sole and absolute discretion, for any one or more of the purposes of: formulating, developing and conducting programs and policies for marketing, sales promotional matters, marketing research and communication and development; purchasing advertising or marketing rights or services in any media; administration of any media-buying co-operative established by Franchisor; and development and administration of the Manchu WOK Restaurant System;
 - (ii) Franchisor shall administratively aggregate the monies in the Fund;
 - (iii) Franchisor shall have the right to pay itself out of the Fund (in addition to the payments referred to in Section b.1 above) an annual administration fee of fifteen per cent (15%) of the total amounts paid to the Fund; and
 - (iv) Franchisor shall be entitled, in its sole and absolute discretion, to retain and use the Fund monies in such manner as Franchisor determines; and
 - (v) Franchisee acknowledges that Franchisor shall have the right to commingle contributions to the Fund made by Franchisees of Manchu WOK Restaurants with those of Franchisees of other restaurants using the Manchu Wok Restaurant System under Franchise Agreements entered into with Franchisor and/or its affiliates and apply such funds for such marketing activities as the Franchisor shall deem appropriate.
 - (vi) Franchisee acknowledges the Fund is not a trust fund and Franchisor shall not have any fiduciary duty to Franchisee in connection with the collection or expenditures of Fund monies or any other aspect of their operations.
- (c) Franchisee shall at its expense pay all costs for and participate in any and all marketing and/or promotional programs required by the Lease (if any). Franchisee shall use in the Business and display at the Premises all such marketing and promotional materials (including, but not limited to, banners, signs and point-of-purchase

materials) as Franchisor may, in its absolute discretion, provide. All Marketing other than that specified by Franchisor shall be subject to the prior written approval of Franchisor, such approval not to be unreasonably withheld.

- (d) Franchisee shall pay to the Franchisor at the same time as the Initial Franchise Fee is payable and in addition thereto, the sum of Ten Thousand Dollars (\$10,000) ("Grand Opening and Promotion Fee") to be applied towards the cost of the Grand Opening Marketing and Promotion for the Franchisee's Manchu WOK Restaurant.
- (e) Save for the Grand Opening Advertising and Promotion Fee referred to in Section 3.5(d) above, Franchisee shall not be obligated to spend an amount in excess of five per cent (5%) of Franchisee's Gross Sales in any twelve (12) month period for the aggregate costs incurred for the Marketing Fee, the Regional Cooperative Marketing Fee (defined below), and the advertising and promotional requirements under the Lease, and Franchisor shall be entitled to adjust or apportion any or all the fees and contributions referred to in this Section 3.5(e) as it deems fit within such limit.
- (f) Notwithstanding the provisions of this Section 3.5 Franchisor shall have the right to direct at all or part of the Marketing Fee be expended by the Franchisee on additional Local Marketing which additional Local Marketing (if directed by the Franchisor) shall be carried out pursuant to the provisions of Section 3.5 (c) above.
- (g) The Franchisor reserves the right to create a regional marketing cooperative ("Cooperative") in any DMA, and establish the rules and regulations therefor. ("DMA" means Designated Market Area, local television market areas designated by the "Nielsen" division of the Television Bureau of Advertising, Inc., or a reasonable substitute that we select if Television Bureau of Advertising, Inc. no longer publishes DMA data.) Immediately upon the Franchisor's request, the Franchisee must become a member of the Cooperative for the DMA in which some or all of the Trade Area is located. In no event shall the Franchisee's Manchu WOK Restaurant be required to be a member of more than one Cooperative. The cooperative must be organized, governed, and operated in the manner the Franchisor prescribes. The Cooperative may require each of its members to make contributions thereto ("Regional Cooperative Marketing Contribution") not to exceed the amounts set forth in Section 3.5(e).
- (h) At Franchisor's request and at Franchisee's expense, Franchisee shall be required to participate in all of Franchisor's System marketing and Restaurant monitoring programs including Internet-based programs, including, but not limited to, various e-mail programs (such as e-mail list building), messaging, newsletters, customer retention and acquisition programs, social media campaigns and tracking surveys, campaign tracking, mystery shopper programs, and health and safety monitoring services. Franchisor reserves the right from time to time and at any time to add, delete or modify System marketing and Restaurant programs.
- 3.6. Lease Administration Fee. Upon the execution of any Lease or upon any amendment to or alteration of the Lease (including, without limitation, entering a renewal of or a new Lease in connection with a Successor Franchise Agreement) Franchisee shall

pay to Franchisor a lease administration fee ("Lease Administration Fee") of Eight Thousand Dollars (\$8,000).

- 3.7. Document Administration Fee. A document administration fee of Five Hundred Dollars (\$500) ("**Document Administration Fee**") is payable to us when we must prepare an amendment to your franchise or other related documents.
- 3.8. Non-participation Fee. You must offer and sell at the Premises of the Business all products designated by us, consistent with our System standards. In addition, you must immediately incorporate into the Business all new products and services designated by us and must fully participate in all local, regional, seasonal, promotional and other programs, initiatives and campaigns adopted by us in which we require you to participate. If you fail or refuse to fully participate in any such program, initiative or campaign, you may, in our sole discretion, be required to pay a non-participation fee of One Hundred Dollars (\$100) per day that you are not in compliance ("Non-participation Fee");
- 3.9. Late Report; Interest; Late Charge; Non-Sufficient Funds Fee; Breaching Royalties; Draft Draw Charge. If you fail to submit to us any financial statements, forms, reports or records required to be provided under this Agreement by its due date, including your weekly Gross Sales report for calculating your Royalty and Marketing Fee, you must pay to us a late report charge of One Hundred Dollars (\$100) per week or part thereof.

If any fees or assessments due under this Agreement, including the Royalty Fee and Advertising Fee, are not paid when due, interest shall accrue on the late payment (from the date payment is due until the date payment is made) at the Default Rate, which amount, plus a late charge of five percent (5%) of the unpaid amount, or One Hundred Dollars (\$100), whichever is greater, shall be added to each late payment. For any payments made by you to us under this Agreement which are returned for non-sufficient funds of a processed check, you shall be charged a non-sufficient funds fee of Twenty-Five Dollars (\$25) per occurrence.

We have the absolute right to charge you the greater of: three (3) times the fixed Royalty; or, if on a percentage Royalty the Royalty will be increased up to eighteen percent (18%) of Gross Sales, with respect to any period during which you are in breach or default of your obligations under this Agreement. The Royalty paid or owing to us with respect to the period during which you are in breach or default are referred to as "Breaching Royalties." Breaching Royalties will be charged for a minimum fourteen (14) day period, regardless of the length of the actual breach or default.

If you fail to provide us any necessary information or documentation with respect to our practice of drawing drafts against your bank accounts, you must pay us a draft draw charge in the amount of One Hundred Dollars (\$100) per day that that failure continues.

- If, as a result of your failure to remit payments required under any provision of this Agreement, we retain an attorney or a collection agency to collect such payments, you must pay all collection costs and expenses, including reasonable attorneys' fees and expenses, whether or not legal proceedings are initiated. Our rights under this Section are in addition to any other rights or remedies that we may have as a result of your default under this Agreement.
- 3.10 Early Termination Fee. If you stop operating your Business before this Agreement expires, with or without obtaining our prior written consent, or in the event of a termination

arising from or related to your default under this Agreement, you must pay us an early termination fee ("Early Termination Fee") for breaching your Agreement. The payment of the Early Termination Fee is not in lieu of other damages and does not constitute a release of the obligations to us. The amount of the Early Termination Fee is calculated as follows:

- (a) Compute the average monthly Royalty and Marketing Fee due for any consecutive twelve (12) month period within the forty-eight (48) months immediately preceding the date we receive notification of the closure, or if you failed to timely notify us of the closure then the date the Business closed, or, if the Business has been open for less than twelve (12) months, the average monthly Royalty and Marketing Fee due since the opening of the Business ("Monthly Average");
- (b) Multiply the Monthly Average by the number of months remaining in the Term; and
 - (c) Divide the resulting total computed in b. above by two (2).

For example purposes only: If the average monthly Royalty and Marketing Fee were collectively \$1,000 and there were five years (60 months) remaining in the Term, the Early Termination Fee would be \$30,000, calculated as follows: $$1,000 \times 60 = $60,000 \div 2 = $30,000$.

If you unilaterally terminate this Agreement prior to the end of the Term, you must give us at least ninety (90) days prior notice of the early termination ("Early Termination Notice"). Within ten (10) days after our receipt of your Early Termination Notice, we will calculate the Early Termination Fee, which will be due and payable thirty (30) days prior to the closure of your Business. In the event of a termination arising from or related to your default under this Agreement, or if you do not: (i) provide us with the Early Termination Notice at least ninety (90) days prior to the early termination of your Business and this Agreement; (ii) remain open for at least ninety (90) days after providing us with the Early Termination Notice; and (iii) pay the Early Termination Fee in full at least thirty (30) days prior to closing of the Business, the Early Termination Fee due may, in our sole discretion, be increased as follows: it will be calculated by multiplying the Monthly Average by the number of months remaining in the Term, and will not be divided by two (2).

If you have not paid your Royalty and/or Marketing Fee for any period(s) within the forty-eight (48) months prior to notifying us of your early termination, or if you have not reported your Gross Sales for any period(s) within the forty-eight (48) months prior to notifying us of your intended early termination, we will estimate the Royalty and Marketing Fee based upon prior reports to calculate the Monthly Average.

4. DEVELOPMENT AND OPENING OF THE PREMISES

4.1. Site Location Period. If the Manchu Wok Restaurant or the Territory is not identified and agreed to by the Franchisor prior to the execution of this Agreement, then the Franchisee will have eighteen (18) months following the Commencement Date of this Agreement ("Site Location Period") to use its best efforts to find a suitable location for the Manchu Wok Restaurant acceptable to the Franchisee and the Franchisor in all reasonable respects and to sign a Lease for the Manchu WOK Restaurant. As set forth in

Section 2.1 of this Agreement, the Term of this Agreement will expire if Franchisee is unable to find a suitable location within the Site Location Period, and in such event, Franchisor may retain any and all monies paid to Franchisor, including the Initial Franchise Fee paid pursuant to Section 3.1, as liquidated damages and not as a penalty. Franchisee's failure to sign a Lease for the Manchu Wok Restaurant prior to the expiration of the Site Location Period will constitute abandonment of the Business by Franchisee and entitle the Franchisor to terminate this Agreement without providing Franchisee with notice or the opportunity to cure.

4.2. Site Selection

The location of the Premises must be approved in writing by the Franchisor. The Franchisee understands and acknowledges that the Franchisor's recommendation or acceptance of a location or site for the Premises, or any assistance it may render with respect thereto, does not constitute any assurance that Franchisee's Manchu WOK Restaurant will be profitable at such location or site or more profitable at such location or site in comparison to other locations or sites. Location or site recommendation, acceptance, or assistance merely indicates that such location or site meets the Franchisor's minimum criteria.

4.3. Leasing of Premises

Franchisor shall have the right to require that Franchisee lease the space for the Premises directly and sign a lease (as well as a Lease Addendum, to be signed by Franchisee and the landlord for the Premises, as well as Franchisor) pursuant to Section 4.2(a) below.

(a) Lease of Premises by Franchisee

Within thirty (30) days after (i) execution by all parties of this Franchise Agreement, or (ii) notification of location or site acceptance by the Franchisor, whichever shall first occur, the Franchisee shall enter into a Lease of the Premises with the owner or landlord thereof, in form and terms acceptable to the Franchisor. Such Lease shall contain, among other terms, clauses providing:

- (i) that Franchisee and the Landlord for the Premises shall sign a "Lease Addendum" (in the form that is attached to this Agreement as Schedule D) as a condition to giving Franchisor's approval to Franchisee's lease or sublease;
- (ii) that no action shall be taken to enforce any term of the Lease without first giving no less than ten (10) days' notice to the Franchisor and permitting the Franchisor a reasonable time to correct or remedy any default;
- (iii) for reasonable prior written notice from the Landlord to the Franchisor in the event that the Landlord should elect to cancel or terminate the Lease by reason of any breach or default of the Franchisee in compliance with the terms of the Lease. In the event that the Lease is terminated by the Landlord, the Franchisor may elect to immediately terminate this Franchise Agreement and require the Franchisee to immediately transfer possession of the Premises to the Franchisor or otherwise vacate the Premises in accordance with the Franchisor's written instructions. The Franchisee shall be deemed to have

released the Franchisor and the Landlord from any claims relating the taking of possession of the Premises or the granting of a lease to the Franchisor by the Landlord in such event(s).

In the event of such cancellation or termination and provided that the Franchisor has corrected or remedied or agrees to correct or remedy any default or breach of the Franchisee within thirty (30) days following such notice, cancellation or termination, the Franchisor shall be entitled to take possession of the Premises and to have a lease in its name for the unexpired balance of the initial term of the Lease, plus renewals, on the same terms and conditions as set forth in the Lease.

- for the Franchisor having the right to call for an assignment of the Lease to the Franchisor or, as directed by the Franchisor to an Affiliate or to a franchisee of the Franchisor, or, at the Landlord's option, for a new lease on the same terms and conditions as contained in the Lease, for the unexpired portion of the initial term, plus renewals/option terms, upon producing to the Landlord a certificate from the Franchisor to the effect that that this Franchise Agreement has been terminated and that the Franchisor is entitled to call for such assignment or new lease; for the Franchisee and the Landlord agreeing to transfer possession of the Premises to the Franchisor in such instance and the Franchisee releasing the Franchisor and the Landlord from any claims relating the taking of possession of the premises or the granting of a lease to the Franchisor; for the Landlord agreeing to consent to such assignment or to issue a new lease without regard to any objection by the Franchisee and any dispute between the Franchisee and the Franchisor being dealt with as between themselves; in the event of an assignment of the Lease to the Franchisor, the Franchisor shall covenant to be bound by all the terms and conditions of the Lease in the same manner as if the Franchisor were the original tenant thereunder; provided if Franchisor shall elect not to become an assignee of the Lease as provided for herein, Franchisor shall have the right, at its expense, to enter upon the Premises and remove all Manchu WOK Restaurant signage and identification therefrom;
- (v) the Lease not being altered, amended, or modified in any particular, or renewed or extended without the express prior written consent of the Franchisor which consent shall not be unreasonably withheld; and
- (vi) the Franchisee effecting and maintaining adequate insurance coverage on the building in which the Premises are located in accordance with the Franchisor's standard provisions in that regard.

All other terms and conditions of the Lease shall be otherwise satisfactory to the Franchisor, acting reasonably.

The Franchisee and the Guarantor covenant and agree not to execute any Lease, or renewed or extended, without the prior written consent of the Franchisor. The Franchisee shall deliver at least one fully executed copy of any such Lease to the Franchisor within fifteen (15) days after its execution.

(b) Deposit and Fee

In the event that the Franchisor is required by the terms of any Lease entered into by the Franchisee to guarantee in whole or in part any of the obligations of the Franchisee in the Lease entered into by the Franchisee, the Franchisee shall pay to the Franchisor:

- (i) provided that it is required by the terms of the Lease, a deposit ("Deposit") equal to three months' gross rent for the Premises (or such greater amount that may be required as security and/or a damage deposit under the Lease) to be held by the Franchisor without interest and, to the extent not applied to amounts due under the Lease, or to offset any other monies owing by the Franchisee to the Franchisor or any Affiliate of the Franchisor under any Contract, returned to the Franchisee; and
- (ii) Franchisee shall pay to the Franchisor at such time or times as rent becomes due and payable pursuant to the Lease and in addition thereto, an amount equal to ten (10%) of the base or minimum rent otherwise payable thereunder, for the Franchisor's own account.

4.4. Development Obligations

The Franchisee's obligations regarding the construction, renovation or other preparation of the Premises for operation as a Manchu WOK Restaurant including, without limitation, the purchase of opening inventory and supplies and the purchase and installation of Fixtures and Equipment are as set out in Schedule "E" to the Agreement (collectively called the "Development Obligations"). Except as otherwise specifically set out in the Agreement, the performance of and payment for all Development Obligations including, without limitation, all training costs as provided in this Agreement, including, without limitation, Section 5.2 hereof, shall be promptly paid for or reimbursed to the Franchisor by Franchisee, whether or not performed in whole or in part by Franchisee or Franchisor or the agents or contractors of either or both of them. Notwithstanding anything in this Agreement to the contrary, if Franchisor requests, it must designate, or review and approve, the contractor who constructs and improves the Premises.

4.5. Opening for Business

The Franchisee shall not open for nor start operating the Business until:

- (a) the training described in Section 5.2 below has been completed to the Franchisor's satisfaction.
 - (b) the Franchisee's Manchu WOK Restaurant has:
 - (i) been completed in all respects to the satisfaction of the Franchisor and
 - (ii) is in the opinion of the Franchisor (acting reasonably) ready to be opened to the public;

- (c) the Franchisee has provided the Franchisor with certificates of all insurance policies required by Section 7 below (and such other evidence of insurance coverage as Franchisor reasonably requests); and
- (d) the Franchisee has complied with any other reasonable terms and conditions imposed by the Franchisor.

Notwithstanding anything in this Agreement to the contrary, the Business must open and begin operating no later than 180 days from the date of this Agreement or the required opening date under any Area Development Agreement under which this Agreement has been entered into, whichever first occurs.

4.6. Revocation

Franchisee expressly acknowledges and agrees that Franchisor has the right to revoke the Franchise and terminate the Agreement and all other Contracts relating to the Premises or the Franchise if Franchisee fails for any reason either to fulfill the Development Obligations for the Manchu WOK Restaurant subject of this Franchise Agreement or for other Manchu WOK Restaurant(s) referred to in any Development Agreement entered into with the Franchisor or its affiliates and Franchisee within the time required pursuant to attached Schedule "E" or in Development Schedule attached to any Development Agreement, or if none, within a reasonable time as determined by Franchisor, or to start operating a Manchu WOK Restaurant at the Premises within five (5) days after completion of the Development Obligations (but in no event later than the date set forth in the last sentence of Section 4.4), with the exception that, if local government approval of anything required for the opening of the Business (or other Business referred to in any Development Agreement) is withheld during such period for reasons beyond Franchisee's control, Franchisor shall grant Franchisee such reasonable additional time, up to sixty (60) days, as is necessary to obtain such approval. In the event of such revocation and termination by Franchisor, the parties shall be released and discharged from any and all further obligations under Contracts relating to the Premises or the Franchise or to other Manchu WOK Restaurants to be developed (except for those applicable to, or stated to survive, their termination) and Franchisor shall be entitled to retain for its absolute use, and not be obligated to refund to Franchisee, any and all monies paid under such Contracts.

4.7. Location of Franchisee's Manchu WOK Restaurant

The Franchisee shall operate its Manchu WOK Restaurant only at and from the Premises and may not relocate except with the Franchisor's prior written consent. If, prior to the expiration or sooner termination of this Agreement, the Lease for the Premises expires or terminates without fault of the Franchisee, or if the Premises are destroyed, condemned or otherwise rendered unusable, the Franchisor will grant its consent to the relocation of the Franchised Business to a site acceptable to the Franchisor within the Trade Area, and in accordance with the provisions of Section 4.10 below.

4.8. Equipment and Furnishings

The Franchisee shall use in and about its Manchu WOK Restaurant only such Equipment and Fixtures, which are required by the Franchisor and which strictly conform to the appearance, uniform standards and specifications prescribed by the Franchisor for the Manchu WOK Restaurant System. All Equipment and Fixtures shall be purchased or leased from the

Franchisor or from suppliers designated by it. The Franchisor agrees that the prices charged by it to the Franchisee for items of Equipment and Fixtures purchased from the Franchisor will be competitive with the prices being generally charged in the Franchisee's market area for equivalent items of comparable quality and in similar quantities. If at any time the Franchisor and its designated suppliers are unable to furnish to the Franchisee any item of Equipment and Fixtures then the Franchisee may purchase such item from another supplier approved by the Franchisor in writing, which approval will not be unreasonably withheld.

4.9. Certification of Performance

After it performs its pre-opening obligations under this Agreement, the Franchisor may request the Franchisee to execute a certification (the "Certification of Performance"), in a form the Franchisor reasonably requests, confirming such performance. If the Franchisor does so request, the Franchisee must execute and deliver the Certification of Performance to the Franchisor within three-business days of the request therefore. If, however, the Franchisee does not reasonably believe that the Franchisor has performed all its pre-opening obligations hereunder, it must, within said three-day period, provide the Franchisor with written notice specifically describing the obligations that the Franchisor has not performed. Not later than three-business days after the Franchisor completes all the obligations specifically described in such notice, the Franchisee must execute and deliver the Certification of Performance to the Franchisor. It must do so even if the Franchisor performed such obligations after the time performance was due under this Agreement. The term "preopening obligations" means such of the Franchisor's obligations to the Franchisee under this Agreement that must be performed before the Opening Date for the Business.

4.10 Relocation.

- (a) If you desire to relocate the Business, you may request our consent upon the following conditions:
- (i) Not less than sixty (60) days prior to the desired date of relocation (unless prior notice is impractical because of a required relocation, due to a third-party or our request, in which event notice shall be made as soon as possible), you must make a written request for consent to relocate, describing the reasons for the relocation and providing details respecting any proposed new location.
- (ii) Within twenty-one (21) days after receiving your written request, we shall advise you in writing if the proposed new location meets our minimum real estate site requirements and if you have our authorization to proceed with the relocation. In the event of our denial to proceed with the relocation, you may request an alternative proposed new location pursuant to the provisions of this Section 4.10.
- (iii) The Term will not be extended in connection with the requested relocation.
- (b) At the time you request to relocate the Business, you must also meet each of the following requirements:
- (i) You must not be in default under this Agreement or any other agreement or note then in effect between us or you and any affiliate of ours, and no event shall have

occurred that, with the giving of notice, the passage of time, or both, would constitute a default under this Agreement;

- (ii) You must neither have received more than three (3) notices of default or breach during the Term; nor more than two (2) notices of default or breach during the five (5) years immediately preceding the effective date of the proposed relocation;
- (iii) The equipment, fixtures and signage used in connection with the operation of the Business must either meet our then-existing System specifications and System standards, or you must agree, within a timeframe required by us, to replace or refurbish such items, and otherwise modify the methods of operation of the Business at your cost and expense, in order to comply with our System standards then applicable to new franchise owners.
- (iv) You shall have paid to us a relocation fee in the amount of Five Hundred Dollars (\$500.00).
- (c) If we approve the relocation of your Business, (i) Franchisee and Franchisor must execute an amendment to this Agreement indicating the address for your relocated Business, and (ii) you must open your Business at the new location within ninety (90) days after you close your Business at the current property where the Business is located. During the period of time between the closure of your Business at the current location, and the opening of the Business at the approved relocation address, you will not owe the Royalty (as defined in Section 3.4).
- (d) Any such relocation shall be at the Franchisee's sole expense, including payment to the Franchisor and its Affiliates of all reasonable costs and expenses to which the Franchisor or such Affiliate is put or incurs, and the Franchisee agrees to indemnify the Franchisor and its Affiliates against, and reimburse each of them for, all loss, liability, costs and expenses incurred by the Franchisor in connection therewith.
- (e) If the Franchisee has not found an acceptable new site for its Manchu WOK Restaurant within the time requirements as set forth above, then the Franchisor may thereafter at its sole option immediately terminate this Agreement, as provided in Section 10.1(a) hereof, by written notice to the Franchisee. Except as specifically referred to in this Section, the Franchisee shall have no right to relocate its Manchu WOK Restaurant to any other location without a) the written consent of the Franchisor which consent the Franchisor shall have the absolute right to withhold, and b) payment to the Franchisor of all fees referable to a new store, including, without limitation, a franchise fee in an amount then charged by the Franchisor to new Manchu WOK Restaurant Franchisees.

5. DUTIES AND RESPONSIBILITIES OF FRANCHISOR

Franchisor shall provide Franchisee with the following assistance during the Term of this Agreement so long as Franchisee has not defaulted under any Contracts relating to the Premises or the Franchise or under this Agreement:

5.1. Operating Manuals

- The Operations Manual is available to Franchisee via the Franchisee portal at: https://portal.kahalamgmt.com. In the event Franchisee desires to receive a hard copy of the Operations Manual, then Franchisee shall submit such request in writing to training@kahalamgmt.com. Upon such request, a hard copy of the Operations Manual will be mailed to Franchisee via regular mail within approximately forty-five (45) days thereafter. Franchisor may periodically revise and change the contents of the Confidential Manual. Beginning on the 30th day (or such longer time as specified by Franchisor) after delivery of written notice from Franchisor, Franchisee must comply with each new or changed provision. Revisions to the Confidential Manual may be based on what Franchisor, in its sole discretion, deems in the best interests of the System, including, among other things, promoting quality, enhancing good will, increasing efficiency, decreasing administrative burdens, or improving profitability of Franchisor or the Franchisee or Manchu WOK franchisees generally. Because complete and detailed uniformity under many varying conditions may not be possible or practical, Franchisor may, in its sole discretion and as it may deem in the best interests of all concerned in any specific instance, vary standards for any franchisee based upon the peculiarities of the particular site or circumstances, density of population, business potential, population of trade area, existing business practices, or any condition that Franchisor deems important to the successful operation of such franchisee's Franchised Business. In such event(s) Franchisee is not entitled to require Franchisor to grant to Franchisee a like or similar variation under this Agreement. Franchisee must at all times ensure that its copy of the Confidential Manual contains all updates Franchisor delivers. In the event of any dispute as to the contents of the Confidential Manual, the terms contained in the Master Copy of each of the Confidential Manual that Franchisor maintains at Franchisor's home office are controlling. Any required standards exist to protect Franchisor's interests in the System and the Marks and not for the purpose of establishing any control or duty to take control over those matters that are reserved to Franchisee. The Confidential Manual also will include guidelines or recommendations in addition to required standards. In some instances, the required standards will include recommendations or guidelines to meet the required standards. Franchisee may follow the recommendations or guidelines or some other suitable alternative, provided Franchisee meets and complies with the required standards. In other instances, no suitable alternative may exist. In order to protect Franchisor's interests in the System and Marks, Franchisor reserves the right to determine if Franchisee is meeting a required standard and whether an alternative is suitable to any recommendations or guidelines.
- (b) Franchisor shall have the right at its discretion to establish an internet based "extranet" system for use by Manchu WOK Franchisees for dissemination of all information which Franchisor provides to Franchisees. Franchisee shall at all times maintain at its expense such computer hardware and software as may be necessary from time to time inclusive of high speed internet connection, to enable Franchisee to access the Manchu WOK Extranet. When established, the Franchisor shall have the right on appropriate notice to the Franchisee, to use the Manchu WOK Extranet for delivery to Franchisee of all notices and changes to the Manchu WOK System including without limitation, changes or amendments to the Confidential Manual.

5.2. Training

Franchisor shall provide an initial training course regarding the Manchu WOK Restaurant System ("Initial Training") for up to two (2) persons - Franchisee and one of the employees as

designated by Franchisee. Initial Training is made up of the "In-Store Training," which is approximately one hundred twenty (120) hours, and "New Owner Training," which is approximately forty (40) hours. One of the trainees shall be Franchisee if Franchisee is an individual or one of Franchisee's designated shareholders (as provided in Section 6.2 hereof) if Franchisee is not an individual and one of the trainees must be the store Manager who will be responsible for the operation and management of your Manchu WOK® Restaurant. Franchisee has previously operated a Manchu WOK Restaurant then, prior to the opening of the Business, Franchisor shall provide refresher training in food preparation techniques and other aspects of the physical operation of a Manchu WOK Restaurant ("Refresher Training") to two (2) persons designated by Franchisee (one of whom shall be Franchisee as Manager if Franchisee is an individual or one of Franchisee's designated shareholders (as provided in Section 6.2 hereof) as Manager if Franchisee is not an individual). Franchisor may from time to time thereafter provide such further training assistance as Franchisor deems necessary ("Supplemental Training"), including assistance with any new developments in the Manchu WOK Restaurant System or matters that may be described in revisions made to the Confidential Manual. All training described in this paragraph shall be provided at a place chosen by Franchisor and may include both in-store and classroom training. All persons attending such training shall complete the same to the satisfaction of Franchisor. One Initial Training, including all requisite materials, is included in the Initial Franchise Fee. Any subsequent Initial Training, Refresher Training, Supplemental Training or training required on a Transfer, the Franchisee shall pay to the Franchisor the non-refundable amount of Three Thousand Dollars (\$3,000) for each person attending such training. In addition, in connection with every type of training, Franchisee shall be responsible for all personal expenses incurred by the persons attending such training (including, but not limited to, accommodation and travel expenses) and for any and all injury or damage to person(s) or property occurring to, incurred by or caused by persons attending such courses. No compensation shall be payable to Franchisee or its employees for any services rendered by them during the training courses. Franchisee or at least one of the designated shareholders (as provided in Section 6.2 hereof) of a corporate Franchisee, must attend and satisfactorily complete all training. If the Franchisor designates any Supplemental Training to be mandatory, the Franchisee must attend such training; if it does not, such failure is an Event of Default.

5.3. Advice and Guidance

- (a) The Franchisor may make available to the Franchisee such assistance as the Franchisor considers necessary based upon the experience and judgment of the Franchisor in connection with the pre-opening, opening and initial operation of the Business and conforming to the requirements of the Manchu WOK Restaurant System; and
- (b) Franchisor, to the extent it deems it appropriate or necessary, may provide continuing advice and guidance to Franchisee concerning the use of the Manchu WOK Restaurant System and the operation of the Business, including advice and guidance regarding formulation and implementation of local advertising and promotional programs, purchasing of goods and other supplies, establishment and maintenance of any administrative, bookkeeping, accounting and general operating procedures required by Franchisor, hiring and training of employees and the provision from time to time of results of research on market trends, where practicable.

(c) Franchisee acknowledges and agrees that any training, support, guidance or tools provided to Franchisor as part of the Franchised Business under this Agreement are for the purpose of protecting the System's brand and Marks and to assist Franchisee in the operation of the Franchised Business and not for the purpose of controlling or in any way intended to exercise or exert control over Franchisee's decisions or day-to-day operations of the Franchised Business. Franchisee further acknowledges and agrees that no employee of Franchisee will be deemed to be an employee of Franchisor for any purpose whatsoever, and nothing in any aspect of the System or the Marks in any way shifts any employee or employment related responsibility from Franchisee to Franchisor.

5.4. Special Assistance

In response to any written request by the Franchisee and to the extent the Franchisor deems it appropriate, necessary or advisable and has personnel available for such purpose, the Franchisor may furnish additional or special assistance to the Franchisee (in addition to the assistance referred to in Section 5.3 above) to rectify any specific problems encountered by Franchisee in the operation of the Franchisee's Manchu WOK Restaurant. In such event, the Franchisee shall promptly pay the Franchisor for each person designated by the Franchisor to provide additional or special services pursuant to the provisions of this Section, the amount of Five Hundred Dollars (\$500) per day or part thereof during which any additional assistance is provided, together with for all reasonable costs incurred by the Franchisor on account of travel, lodging and meals for all personnel so designated by the Franchisor.

5.5. Opening Marketing

The Franchisor, using the amount the Franchisee paid under Section 3.5(d) shall prepare, coordinate, and conduct, on and in the Franchisee's behalf, the Grand Opening Marketing and Promotion program referred to in Section 3.5(d), which is a marketing program to promote the opening of the Business. The nature and scope of such opening promotional marketing program shall be determined by the Franchisor, in its sole and absolute judgment, and the cost of same shall be payable by the Franchisee.

5.6. Promotional Materials

Franchisor shall review and approve prior to any use by Franchisee, all marketing and promotional materials which Franchisee proposes to use in connection with the Business and may, at its option and in its sole and absolute discretion, provide promotional and marketing materials to Franchisee for use in the Business whether pursuant to payments to the Fund or otherwise.

5.7. Conventions and Seminars

Franchisor may invite Franchisee to attend all appropriate conventions, seminars and other franchisee-oriented functions sponsored by Franchisor.

6. FRANCHISEE DUTIES, RESPONSIBILITIES AND OBLIGATIONS

The Franchisee acknowledges that the Franchisor maintains and demands the maintenance of high standards of quality of product and service in restaurants utilizing the

Manchu WOK Restaurant System and that the Franchisor may, in the interest of improving thereon, and enhancing public acceptance of Manchu WOK Restaurants, upgrades its standards from time to time. Therefore, in order to maintain a uniform standard of operation and quality for all Franchisees operating under the name Manchu WOK and to further protect the goodwill of the Franchisor, the Franchisee covenants and agrees to comply with all of the following:

6.1. Conform to Manchu WOK System

Franchisee agrees that it shall operate its Manchu WOK Restaurant in conformity with such uniform methods, standards and specifications as the Franchisor may from time to time prescribe for general application in the Confidential Manual or otherwise in writing to ensure that the highest degree of quality and service is uniformly maintained, refrain from any deviation therefrom and from otherwise operating in any manner that reflects adversely on the Franchisor's name and goodwill, or on the name "Manchu WOK" as associated with the Business.

6.2. Management of Business

Franchisee agrees that it shall manage and operate the Business using at all times at least the two (2) employees (one of whom may be the manager of the Business if such Manager is actually involved in the day to day operations of the Business) who have been trained in the operation of a Manchu WOK Restaurant (as set out in Section 5.2 hereof), who have signed a Confidentiality and Non-Competition Agreement in terms similar to the confidentiality and noncompetition provisions of this Agreement and who carry out their duties courteously and with a view to complete customer satisfaction. Franchisee (if an individual) shall be the Manager of the Business. If Franchisee is a corporation, or other entity, one shareholder, or owner, shall be designated by Franchisee to be the Manager of the Business and that Manager shall have the authority to receive and implement all directions received from time to time from the Franchisor and to represent the Franchisee in all dealings with the Franchisor. Before his or her appointment as Manager, the Franchisee must submit in writing the name and qualifications of any proposed Manager, or replacement Manager, together with any other information relating thereto the Franchisor requests, to Franchisor for its review and written approval of the proposed Manager. Without such written approval, the Franchisee may not appoint any person a Manager. The Manager shall have, and at all times maintain, at minimum, a ten percent (10%) equity interest in the ownership of the Business. The Manager must devote his or her full-time efforts to on-premises supervision of the Manchu WOK Restaurant. Franchisee shall reimburse Franchisor's reasonable costs and expenses of providing any and all training following the initial training of the Manager or any employees and retraining or training of all Business personnel in new methods, all as deemed necessary or advisable from time to time by the Franchisor.

6.3. Conform to Laws

Franchisee agrees that it shall operate in accordance with, and immediately correct any failure to comply with, all governmental requirements, applicable laws, regulations and orders ("Laws and Regulations") of competent authorities (whether federal, state, municipal or otherwise), including, without limitation, Laws and Regulations relating to health, sanitation and safety, employment, human rights, wages, child labor, misleading advertising, and including obtaining all necessary licenses, permits and approvals ("Licenses") as may be required to

permit the operation of a Manchu WOK Restaurant at the Premises, and Franchisee shall deliver to Franchisor on a timely basis (i) a copy of each License and (ii) a copy of any and all notices of violation of any Law or Regulation as may be received by Franchisee from time to time (whether or not such violation has been corrected by Franchisee).

6.4. Purchasing and Supply of Approved Products

- Approved Products and Supplies. Franchisee agrees that it shall buy all Approved Products, ingredients, supplies, materials (such as packaging), and other products used or offered for sale at Franchisee's Manchu WOK Restaurant only from suppliers that Franchisor has approved in writing (and whom Franchisor has not In determining whether Franchisor will approve any subsequently disapproved). particular supplier, Franchisor will consider various factors, including but not limited to: (1) whether the supplier can demonstrate, to Franchisor's continuing reasonable satisfaction, the ability to meet Franchisor's then-current standards and specifications for such items; (2) whether the supplier has adequate quality controls and capacity to supply Franchisee's needs promptly and reliably; (3) whether approval of the supplier would enable the Manchu WOK Restaurant System, in Franchisor's sole opinion, to take advantage of marketplace efficiencies; and (4) whether the supplier will sign a confidentiality agreement and a license agreement in the form that Franchisor may require (which may include a royalty fee for the right to use Franchisor's Proprietary Marks and any other proprietary rights, recipes, and/or formulae). For the purpose of this Agreement, the term "supplier" is defined to include, without limitation, manufacturers, distributors, resellers, and other vendors. Franchisee also recognizes and agrees that Franchisor has the right to appoint only one supplier for any particular Approved Product or item, which may be the Franchisor or one of its affiliates.
 - (i) Franchisee agrees to sell only Approved Products (whether edible or non-edible) at Franchisee's Manchu WOK Restaurant. Franchisee shall not sell anything at Franchisee's Manchu WOK Restaurant that is not an Approved Product.
 - (ii) Notwithstanding anything to the contrary in this Agreement, Franchisee agrees to buy all of Franchisee's requirements for any Proprietary Products only from Franchisor or from its designees, as provided in Section 6.4(b) below (possibly through one or more suppliers that Franchisor designates in writing). Franchisor has the right, but not the obligation, to introduce additional Approved Products.
 - (iii) If Franchisee wants to buy any Approved Products or any items (other than Proprietary Products) from an unapproved supplier, Franchisee must first submit a written request to Franchisor asking for its prior written approval. Franchisee agrees not to buy from any such supplier unless and until Franchisor has given Franchisee the Franchisor's prior written consent to do so. Franchisor has the right to require that its representatives be permitted to inspect the supplier's facilities, and that samples from the supplier be delivered, either to Franchisor or to an independent laboratory that Franchisor has designated for testing. The supplier may be required to pay a charge, not to exceed the reasonable cost of the inspection, as well as the actual cost of the test. Franchisor has the right to also require that the supplier comply with such other

requirements that Franchisor has the right to designate, including but not limited to payment of reasonable continuing inspection fees and administrative costs and/or other payment to Franchisor by the supplier on account of their dealings with Franchisee or other franchisees, for use of Franchisor's trademarks, and for services that Franchisor may render to such suppliers. Franchisor also reserves the right, at Franchisor's option, to periodically re-inspect the facilities and products of any such approved supplier and to revoke Franchisor's approval if the supplier does not continue to meet any of Franchisor's then-current criteria. Franchisor is not required to approve any particular supplier, or to make available Franchisor's standards, specifications, or formulas to prospective suppliers, which Franchisor has the right to deem confidential.

- Franchisee acknowledges and agrees that Franchisor has the right to establish one or more strategic alliances or preferred vendor programs with one or more nationally or regionally-known suppliers that are willing to supply all or some Manchu WOK Restaurants with some or all of the products and/or services that Franchisor requires for use and/or sale in the development and/or operation of Manchu WOK Restaurants, notwithstanding anything to the contrary contained in this Agreement. In this event, Franchisor may limit the number of approved suppliers with whom Franchisee may deal, designate sources that Franchisee must use for some or all Approved Products and other products and services, and/or refuse any of Franchisee's requests if Franchisor believes that this action is in the best interests of the Manchu WOK Restaurant System or the network of Manchu WOK Restaurants. Franchisor has the right to approve or disapprove of the suppliers who may be permitted to sell Approved Products to Franchisee. Any of Franchisor's affiliates that sell products to Franchisee will do so at Franchisor's direction. If Franchisee is in default of this Agreement, Franchisor reserves the right to direct its affiliates not to sell products to Franchisee, or to withhold certain discounts that may otherwise be available to Franchisee.
- (v) Franchisee acknowledges and agrees that Franchisor has the right to collect and retain all manufacturing allowances, marketing allowances, rebates, credits, monies, payments or benefits (collectively, "Allowances") offered by suppliers to Franchisee or to Franchisor (or our affiliates) based upon Franchisee's purchases of Approved Products and other goods and services. These Allowances include those based on purchases of Approved Products, ingredients, paper goods, beverages, and other items (such as packaging). Franchisee assigns to Franchisor or its designee all of Franchisee's right, title and interest in and to any and all such Allowances and authorizes Franchisor (or its designee) to collect and retain any or all such Allowances without restriction.
- (vi) If Franchisor requires Franchisee to use items that bear Franchisor's Marks (for example, plates, cups, napkins, take-out bags, other paperware, etc.), then Franchisee must buy, use, and sell only the items that Franchisor requires, and Franchisee must buy those items only from Franchisor's approved suppliers (or as otherwise required under Section 6.4(b) below).
- (b) <u>Proprietary Products</u>. Franchisee acknowledges and agrees that certain menu items offered and sold at Franchisee's Manchu WOK Restaurant are

manufactured in accordance with Franchisor's secret recipes, standards, and specifications and/or those of Franchisor's affiliates, and that all of those items are Franchisor's Proprietary Products. In order to maintain the high standards of quality, taste, and uniformity associated with such menu items, and other Approved Products and packaging bearing the Marks, Franchisee agrees to buy all Proprietary Products, and Approved Products and packaging bearing the Proprietary Marks (and any other products Franchisor may now or in the future designate), only from Franchisor, its affiliates, and/or Franchisor's approved suppliers, and not to offer or sell any other such products at or from Franchisee's Manchu WOK Restaurant.

- (c) <u>Quality.</u> Franchisee is solely responsible for the quality of products that Franchisee sells.
- (d) <u>Use of the Marks</u>. Franchisee must require all marketing materials, signs, decorations, paper goods (including, without limitation, disposable food and beverage containers, bags, napkins, menus, and all forms and stationery used in Franchisee's Manchu WOK Restaurant), and other items which Franchisor may designate to bear the Marks in the form, color, location, and manner Franchisor prescribes (and subject to Franchisor's prior written approval, for example as provided in Section 3.5(c.) above).
- (e) <u>Innovations.</u> All products, services, ingredients, concepts, methods, techniques, and/or new information relevant to Franchisee's operation of Franchisee's Manchu WOK Restaurant (collectively, the, "Innovations"), whether or not constituting protectable intellectual property, that Franchisee or its employees create, or that are created on Franchisee's behalf, must be promptly disclosed to Franchisor. All such Innovations will be deemed to be Franchisor's sole and exclusive property and works made-for-hire for Franchisor. Franchisee and each of its Principals and owners agree to: (1) sign the assignment and/or other documents Franchisor requests in order to implement this clause in order to evidence Franchisor's ownership; (2) cause Franchisee's employees and contractors to sign such assignment documents as Franchisor may request for this purpose; and (3) assist Franchisor in securing intellectual property rights in such Innovations.

6.5. Approved Design

Franchisee agrees that it shall use in the operation of the Franchisee's Manchu WOK Restaurant all, but only those, methods of exterior and interior design and construction, and signage as may be prescribed by the Franchisor or which conform to the requirements and specifications of the Manchu WOK Restaurant System, and no others and to make no material alteration or addition to, or replacement of or improvement or change to, the interior or exterior of the Premises or the Fixtures or Equipment without the prior written consent of Franchisor;

6.6. Maintenance and Renovations

Franchisee agrees that it shall maintain the condition and appearance of the Premises, the Fixtures and the Equipment in a manner consistent with the high quality image of the Manchu WOK Restaurant System as an attractive, modern, clean, convenient and efficiently operated Manchu WOK Restaurant offering high quality products and efficient and courteous service. The Franchisee will effect such maintenance of the Premises, the Fixtures and the Equipment as is required by the Lease and as reasonably required from time to time by the

Franchisor and the Landlord and as required by Laws and Regulations to maintain such condition, appearance and efficient operation, including, without limitation, replacement of wornout or obsolete Fixtures and Equipment, repair of the interior and exterior of the Premises, and periodic cleaning, decorating and redecorating. If, at any time in the Franchisor's reasonable opinion, the general state of repair, appearance or cleanliness of the Premises, the Fixtures or the Equipment or the general state of the Premises does not meet Manchu WOK Restaurant System standards or the requirements of the Lease, the Franchisor may give notice to the Franchisee specifying the action to be taken by the Franchisee to correct such deficiency, and the Franchisee shall initiate such action forthwith upon receipt of such notice. If the Franchisee fails to take such action within three (3) days after receipt of such notice from the Franchisor, the Franchisor may, without prejudice to any other rights or remedies hereunder, cause such repairs or maintenance to be done at the sole cost and expense of the Franchisee. In such event, the Franchisee will reimburse the Franchisor for the amount of such costs and expenses upon demand plus an administrative fee of fifteen percent (15%) of such costs and expenses. Franchisor shall have the right at all times to enter upon the Premises for purposes of enforcing the terms of this paragraph as set out in Section 6.7 below.

In addition, the Franchisee shall, on the fifth (5th) anniversary of the Opening Date make at Franchisee's expense, such renovations, alterations and upgrades to the Premises, Fixtures and Equipment as may be required by Landlord and such renovations, alterations and upgrades to the Premises, Fixtures and Equipment as may be required by Franchisor in order to renovate, upgrade and modernize the Franchisee's Manchu WOK Restaurant so as to reflect the thencurrent image of the Manchu WOK Restaurant System for Manchu WOK Restaurants. All such work shall be completed (unless otherwise agreed to by Franchisor in writing) within 90 days. Required renovations and alterations may include complete remodeling and upgrading of the Premises and the Fixtures and Equipment. In connection with any renovation, alteration, upgrading, etc., required under this Section 6.6, the Franchisee shall pay the applicable fees specified in Section 3.2(b) hereof.

6.7. Inspection

Franchisee agrees that it shall permit the Franchisor or its authorized representatives to enter and inspect the Premises at reasonable times and to examine and test the Fixtures and Equipment (including, without limitation, the POS System) and the goods and services offered by the Business, for the purpose of ascertaining whether the Franchisee is operating its Manchu WOK Restaurant in accordance with the terms of this Agreement. Inspections may be conducted at any time at the discretion of the Franchisor and without prior notice to the Franchisee. The Franchisor agrees to give notice to the Franchisee of any deficiencies detected during any inspection. The Franchisee will diligently and promptly correct all such deficiencies. If the Franchisee receives notice from the Franchisor that any Fixtures or Equipment or goods or services do not meet the specifications, standards and requirements established by Franchisor for Manchu WOK Restaurants, the Franchisee will immediately refrain from the further use thereof until the same are repaired or replaced to the reasonable satisfaction of the Franchisor.

6.8. Use POS System

Franchisee agrees that it shall give to each customer of the Business a receipt detailing such customer's purchases from the Business and record all sales on an electronic cash register, computer and computer terminals ("POS System") approved by Franchisor

containing: (i) devices that will record cumulative sales and provide sequential customer count recording on a daily and cumulative basis and that cannot be turned back or reset; and (ii) such devices permitting the electronic polling of cash register information via high speed Internet connection as may be required by Franchisor. Franchisor shall have the unlimited right to use any information obtained from Franchisee's POS System for any business purpose whatsoever. The Franchisor may, from time to time, require the Franchise to upgrade, supplement, or replace any or all the components of the POS System. The Franchisor may poll the POS System at any time it elects; to permit such polling, the Franchisee must configure and constitute its POS System so that the Franchisor has complete, continuous, and unfettered access to all information the POS System generates and collects. At the Franchisor's request, the Franchisee must provide any assistance the Franchisor requires to bring the POS System "on-line" with the Franchisor's computer at the Franchisor's headquarters. The Franchisee is exclusively responsible for the telephonic cost of such polling.

You are required to accept debit and credit cards and gift/loyalty cards from consumers at the Business. Prior to the opening of your restaurant, you will be required to acquire an approved debit, credit and gift/loyalty card processing system to use during the operation of the Business. Additionally, you must utilize our approved third-party payment card processor for processing all such debit, credit, rewards, and gift/loyalty card transactions, including entering into any such approved vendor agreements for the gift/loyalty card processing services. The Payment Card Industry ("PCI") requires all companies that process, store, or transmit credit card information to protect the cardholders' information by complying with the PCI Data Security Standard ("PCI DSS"). Therefore, as a franchisee who accepts credit cards, you are required to be PCI compliant by following and adhering to PCI DSS, which includes ensuring that your point-of-sale system, back office computer (if supplied), and any other device that is plugged into the network is only used for business purposes. You are also required to complete an annual questionnaire and quarterly network PCI scans and install a network firewall appliance for logging, tracking, reporting, and security assessment. We require your Business' point-ofsale system, including terminals, computers, and software to be in compliance with the PCI DSS at all times. The PCI DSS is often updated, and you are required to obtain and comply with all updated standards. You must also be PCI compliant in order to obtain cyber liability/data breach insurance coverage.

6.9. Hours of Operation

Franchisee agrees that it shall keep the Business open and in normal operation during such hours of the day and during such days of the year as are directed by the Franchisor and/or as required by the Lease.

6.10. Pay Suppliers

Franchisee agrees that it shall promptly pay when due all trade creditors and vendors (including but not limited to any that are affiliated with us) that supply goods or services to Franchisee and/or the Manchu WOK Restaurant (including but not limited to food and beverage products, packaging and serving materials, paper goods, utensils, supplies and other items of inventory, uniforms, marketing or promotional materials, furniture, Fixtures, Equipment and all other items used in the Business). Franchisee also agrees to promptly pay Franchisor for any expenditures or payments that Franchisor chooses (or is required) to make to trade creditors or vendors on Franchisee's behalf to pay for obligations that Franchisee incurred in connection

with the Manchu WOK Restaurant for example, products and/or services that Franchisee ordered but for which Franchisee did not pay.

6.11. Signage, Notices and Menus

Franchisee agrees that it shall post on the Premises such signs, in size, form, color and location as the Franchisor may require, notifying customers of the Business and others of: (i) Franchisee's identity and status as an independent contractor and operator of the Business as a Franchisee of the Franchisor; (ii) the manner in which to contact Franchisor; (iii) each customer's right to a receipt in respect of goods or services purchased from the Business; (iv) such other matters as Franchisor may, in its absolute discretion, determine necessary or desirable in connection with the operation of the Business and to maintain and display signs (including all or any part of a menu-board, digital menu-board, or store sign) reflecting the thencurrent image of the Manchu WOK Restaurant System, which are of the color(s), size, design and location as determined and specified by the Franchisor.

6.12. Pay Franchisor

Franchisee agrees that it shall pay to the Franchisor as and when due all monies due the Franchisor pursuant to this Franchise Agreement from time to time, together with the amount, calculated in accordance with the applicable laws, equal to any and all, sales taxes, value added taxes, business transfer taxes or any other taxes imposed on or collectable from Franchisee in respect of any and all amounts payable by Franchisee to Franchisor under the Agreement.

6.13. Marketing Obligations

Franchisee agrees that it shall carry out the marketing obligations set out in Section 3.5 above.

6.14. Banking

Franchisee agrees that it shall open and use one (1) bank account only and deposit all revenues received from the Business in such account and notify Franchisor of such bank and account number and authorize Franchisee's banker to provide to Franchisor such information as may be required by Franchisor. Franchisee shall also, at its own expense and, at the request of Franchisor (whether or not Franchisee is or ever has been in default in any payment to Franchisor), implement an automatic bank transfer or other electronic funds transfer from the said account in connection with the payment of any and all amounts payable to the Franchisor pursuant to this Franchise Agreement or to Franchisor or any Landlord or sub-landlord payable pursuant to any Lease.

6.15. Financial Reporting

Franchisee agrees that it shall provide to Franchisor, without notice or demand, the reports, statements and financial information at the times and in the forms as provided in Section 8.2 hereof.

6.16. Advise Suppliers

Franchisee agrees that it shall advise each supplier and all others with whom the Franchisee deals that the Franchisee is an independent contractor/business person and that all debts incurred by it are for its account only and are not debts of the Franchisor.

6.17. Telephone Listings

Franchisee agrees that it shall obtain and maintain, at its sole cost and expense, a listing for its Manchu WOK Restaurant in the principal regular (white pages) and classified (yellow pages) telephone directories serving the location of the Business, in such directory categories as are specified by the Franchisor from time to time. All such listings will be of such type and size as are specified by the Franchisor from time to time. If the Franchisee does not obtain such listings, as aforesaid, the Franchisor may obtain such listings on behalf of and at the expense of the Franchisee, and the Franchisee shall forthwith reimburse the Franchisor for all the Franchisor's costs and expenses for so doing, upon demand

6.18. Promote Franchise

Franchisee agrees that it shall obtain and maintain an adequate supply of brochures, pamphlets, videotapes and other special promotional materials prescribed by the Franchisor (including, without limitation, materials directed to prospective franchisees of the Manchu WOK Restaurant Systems) at such locations in and about the Premises as the Franchisor may require from time to time for its franchisees generally. All such brochures, pamphlets, videotapes and other special promotional materials (save for those directed to prospective franchisees of the Manchu WOK Restaurant System which shall be provided to the Franchisee without charge) shall be purchased from the Franchisor or from its approved suppliers.

6.19. Administration Programs

Franchisor shall have the right, in its sole and absolute discretion, to inaugurate and administer for the benefit of the franchisees of the Manchu WOK Restaurant System various third-party programs including, without restriction, lease and rental payment administration, insurance plan(s), realty tax payment administration, and the like. In such event, the Franchisor shall be entitled to impose a reasonable administration fee in respect of the conduct of each such program, not to exceed fifteen percent (15%) of the gross amount of the payments received by the Franchisor from time to time in respect thereof, and the Franchisee agrees to participate in and abide by the terms of each such program and to execute all such documents and authorizations as may be required by the Franchisor in connection therewith.

6.20. Seminars, Conventions, Training

As and when required by the Franchisor, the Franchisee, the Manager, and such other employees of the Franchisee's Business as may be from time to time directed by the Franchisor (acting reasonably) shall attend and participate in (at Franchisee's sole cost and expense) management seminars and training and refresher courses and Franchisor and/or industry conventions, including, without limitation, the Alternate Year Convention, and participate in ongoing aptitude testing conducted by or on behalf of the Franchisor from time to time. The Franchisor agrees that the Franchisee, the Manager, and such employees shall not be required to attend more than two (2) such seminars, conventions or other courses in each calendar quarter. The foregoing restriction does not apply to Initial Training or any other training that the Franchisor may require the Franchisee's Manager or any replacement Manager to attend.

6.21. Franchisee Organizations

Franchisee shall actively participate in any and all franchisee and franchisee/Franchisor committees and organizations (Bodies) (including without limitation a Franchisee Council) as may be established from time to time by the Franchisor and which may at Franchisor's discretion become Bodies elected by the franchisees, all for the benefit of the Franchisee, other franchisees of the Manchu WOK Restaurant System and the Franchisor.

6.22. Employee Attire and Personal Appearance

Franchisee's employees must comply with such dress code or standards as Franchisor may require, which may include use of branded (or other "uniform") apparel, and otherwise identify themselves with the Marks at all times in the manner Franchisor specifies (whether in the Confidential Manual or otherwise in writing) while on a job for Franchisee's Manchu WOK Restaurant. Franchisor may also require that Franchisee and its employees comply with personal appearance standards (including but not limited to dress code, shoes, hair color, body art, piercing, sanitation and personal hygiene, foundation garments, personal displays at work stations, etc.).

6.23. Acceptance of Alternative Payment Methods

Franchisee shall, at Franchisor's request, acquire and maintain Internet connectivity and accept debt cards, credit cards, stored value gift cards, or other non-cash systems specified by Franchisor to enable customers to purchase the products offered by the Restaurant. Franchisee shall acquire, at its expense, all necessary Internet connectivity, hardware and/or software used in connection with these non-cash systems.

7. INSURANCE

7.1. Mandatory Insurance Coverage

At all times during the Term, you must maintain in full force and effect at least the minimum types and amounts of insurance coverage that we require, under one (1) or more policies of insurance (each of which shall be primary coverage and shall not be contributory or secondary to any other coverage maintained by us), insured under the particular name of the Franchisee and for the particular address of the Franchised Business. You understand and acknowledge that you remain fully liable for any and all claims in connection with the operation of the Franchised Business arising during the Term of this Agreement, including without limitation, for any time period during which you may be in the process of obtaining a valid and complete certificate of insurance or is otherwise out of compliance. We do not waive and hereby reserve your obligations under this Agreement, including without limitation your insurance obligations.

Such insurance policies must be issued by insurers acceptable to us having an A.M. Best's financial strength rating of at minimum "A-VIII," and grant us authority to obtain copies of your certificate of insurance directly from the carrier or your agent/broker. The particular requirements of our minimum insurance coverage will be made available to you throughout your term. Currently, the general liability insurance required by this Agreement must: (i) name MTY Franchising USA, Inc. as the certificate holder; (ii) name Kahala Franchising, L.L.C. and MTY Franchising USA, Inc. and their parents, subsidiaries, affiliates, officers, directors, and

employees as additional insureds; (iii) contain a waiver by the insurance carrier of all subrogation rights against us and our affiliates and our affiliates' respective officers, directors and employees for casualty losses; (iv) indicate the address of the Franchised Business being insured; and (v) provide that we will receive a copy, via an endorsement, thirty (30) days' prior notice of cancellation of any such policy. Additional minimum insurance coverage requirements (subject to increase or otherwise change in our sole discretion) are as follows:

TYPE OF COVERAGE	LIMITS/SPECIFICATIONS
General Liability	\$1,000,000 Bodily Injury/Property Damage Per Occurrence / \$2,000,000 Aggregate
Building Improvements and Betterments	100% of Full Replacement Cost – No Coinsurance (minimum of \$100,000)
Business Personal Property	100% of Full Replacement Cost – No Coinsurance – Special Form or equivalent (minimum of \$100,000)
Spoilage	\$5,000
Flood, Earthquake and Volcanic Eruption	Subject to Territory Limitations – required if in a designated Flood Zone
Workers' Compensation and Employer's Liability Insurance	As required by law
Hired and Non-Owned Automobile Liability	\$1,000,000 Combined Single Limit per accident

You need to evaluate if your particular business will require greater coverage or other types of insurance. For example, we strongly recommend that you consult with an insurance broker to discuss whether your particular lease/situation requires and/or should obtain additional common types of insurance (including without limitation, employment practices liability, umbrella insurance, and cyber liability/data breach insurance coverage). Such insurance may significantly increase your premiums, but may also save you money in the long run.

You are responsible for maintaining insurance coverage and limits as required by us, at minimum, pursuant to this *Section 7*, provided, if your landlord requires additional coverage, higher limits, or any other requirements not required by us, then you are responsible for maintaining such additional items as well. You must always keep the required insurance coverage in force at all times during the operation of the Franchised Business, and you must comply with any changes we make periodically to our insurance requirements. Upon 30 days' notice to you, we may require you to increase and/or otherwise change the minimum coverage of the insurance referred to above, including to reflect identification of special risks, changes in law or standards of liability, higher damage awards or other relevant changes in circumstances.

Subject to Section 4, before you may open your restaurant, annually thereafter at least ten (10) days prior to renewal of your insurance coverage, and at any other time on our request,

you must provide us with certificates of insurance or copies of insurance policies showing that you are in compliance with our insurance requirements, as well as proof that you have paid the premiums you owe for the insurance we require. You will pay your insurance premiums to your insurance broker or to the insurance company issuing the policy. We or our affiliate may, at our option and in addition to our other rights and remedies under this Agreement, obtain such insurance coverage on your behalf, and you must promptly execute any applications or other forms or instruments required to obtain any such insurance and pay to us, on demand, any costs, expenses and premiums (in whole or part) incurred by us. Your obligation to obtain and maintain the insurance described above will not be limited in any way by reason of any insurance maintained by us, nor will your performance of such obligations relieve you of any obligations under *Section 13.2*.

8. RECORDS AND REPORTING

8.1. Records

The Franchisee shall at its cost establish a bookkeeping, accounting and record-keeping system conforming to the requirements prescribed from time to time by the Franchisor (including, without, limitation the use of an approved POS System in accordance with Section 6.8 hereof and retention of cash register tapes, invoices, purchase orders, daily inventory records, payroll records, check stubs, bank deposit receipts, sales tax records and returns, cash disbursement journals and general ledgers, together with such further and other records and documents as may from time to time be required by the Franchisor to accommodate any changes in such systems) and shall on request of the Franchisor install and maintain at the Franchisee's expense such microcomputer hardware, bookkeeping and accounting software systems, and telecommunications systems, and to upgrade, supplement, and/or replace any or all of said items as the Franchisor may from time to time require generally for the Manchu WOK Restaurant System. All personnel employed by the Franchisee shall record at the time of sale, in the presence of customers, all receipts from sales or other transactions, on the POS System as designated by the Franchisor and as provided in Section 6.8 hereof. Such POS System shall be sealed in a manner approved by the Franchisor and shall have such other features as may be reasonably required by the Franchisor from time to time.

8.2. Reports and Financial Information

The Franchisee shall furnish to the Franchisor throughout the Term of the Agreement, in respect of the Business, in a form and method from time to time prescribed by the Franchisor (including through a third-party vendor) but at the sole cost and expense of the Franchisee:

- (a) not later than thirty (30) days preceding the scheduled Opening Date (in accordance with Section 2.1 hereof), and not later than thirty (30) days preceding the anniversary of the Opening Date in each year thereafter, a forecast signed and verified by the Franchisee if an individual or if a corporation, by the Franchisee's Chief Financial Officer containing (among other details specified by the Franchisor) a budget of anticipated revenue and expenditure for the ensuing fiscal year of the Business ("Forecast");
- (b) on or before Thursday of each week, a written statement in the form approved by Franchisor and signed by Franchisee, of the Gross Sales for the seven (7)

day period ending at the close of business on the preceding Sunday, together with the Royalty and Marketing Fees;

- (c) on or before the twentieth (20th) day following the end of each calendar month, an unaudited balance sheet for the Business as at the close of such period together with an unaudited statement of profit and loss of the Business for such period, in each case signed and verified by the Franchisee if an individual and, if a corporation, by the Franchisee's Chief Financial Officer, together with a report of variance (if any) from the Forecast and accompanied by such other information and supporting records as the Franchisor may from time to time require;
- as soon as practicable, and in any event within sixty (60) days after the end of each of Franchisee's fiscal years, audited of the Business prepared in accordance with generally accepted accounting principles consistently applied ("GAAP"), as well as generally accepted auditing standards ("GAAS"), and consisting at a minimum of (i) a balance sheet as at the close of such fiscal year, and (ii) a statement of income and a statement of cash flows for such fiscal year, setting forth in each case in comparative form the corresponding figures for the previous fiscal year, in each case signed and verified by the Franchisee if an individual or, if a corporation, by the Franchisee's Chief Financial Officer, certified and prepared by an certified public accountant approved by the Franchisor and accompanied by accountant's comments and a report of variance (if any) from forecast. The audit required by this Section shall be carried out at the expense of the Franchisee by a firm of certified public accountants acceptable to the Franchisor and to be accompanied by an unqualified opinion thereon of such firm, all at the Franchisee's sole cost and expense; and when it files such returns with the appropriate tax authorities, the Franchisee must also provide the Franchisor with copies of its federal and state income tax returns; all financial statements and other periodic reports must segregate the Business's income and related expenses from those of any other business that the Franchisee may conduct; and
- (e) such other periodic forms and reports (including, without limitation, reports of purchases and sales) as may be required by Franchisor.

The Franchisee shall at its expense provide a modem or other electronic communications device as shall be determined from time to time by the Franchisor and a dedicated high speed communications line so as to enable the Franchisor to gather information from the Franchisee's POS System ("Franchisee's System"); this dedicated communications line, which shall be installed and maintained at the expense of the Franchisee, shall be dedicated exclusively to providing the Franchisor with complete and unfettered access to the Franchisee's System so the Franchisor may poll same at such times as it deems appropriate. The Franchisee shall at all times ensure that the Franchisor shall have access to Franchisee's System at Franchisee's ongoing cost and expense, including all polling and maintenance costs. Failure to comply with the provisions of this paragraph shall constitute an Event of Default, entitling the Franchisor to immediately terminate this Agreement, and which may be otherwise dealt with as set out in Section 10.1 and Section 10.2 of this Agreement.

The Franchisee acknowledges that all financial information will be furnished by it solely for the information of the Franchisor, and the Franchisor does not assume nor undertakes any obligation to review such financial information or to advise the Franchisee in connection therewith.

Franchisee acknowledges and agrees that it is Franchisee's responsibility to ensure that Franchisee is in compliance with all laws that are applicable to the proprietary software or other technology used in the operation of the Franchised Business, including all data protection or security laws as well as payment card industry compliance.

8.3. Franchisor's Right to Audit

The Franchisor shall have the right at reasonable times without prior notice to the Franchisee to inspect and audit the business records, bookkeeping and accounting records, cash register tapes, invoices, purchase orders, payroll records, check stubs and bank deposit receipts of the Business, the periodic reports, financial statements, tax returns and other forms, information and supporting records which the Franchisee is required from time to time to submit to the Franchisor and the books and records of the Franchisee, and the books and records of the Guarantors as the latter relate to the Business. The Franchisee shall fully co-operate with representatives of the Franchisor and independent accountants hired by the Franchisor to If any such inspection or audit discloses an conduct any such inspection or audit. understatement of Gross Sales for any period, the Franchisee shall pay to the Franchisor, within ten (10) days after receipt of the inspection or audit report, the Royalty and other sums due on account of such understatement. Further, if such inspection or audit is made necessary by the failure of the Franchisee to furnish reports, financial statements, tax returns or schedules or any other documentation as herein required, or if an understatement of Gross Sales for any period is determined by any such inspection or audit to be greater than two percent (2%), the Franchisee shall on demand (in addition to payment of amounts due by reason of understatement plus interest on such amount at the rate of Eighteen Percent (18%) per year [or the maximum permitted by law, whichever is the less] computed from the date payment was due to the date payment is received) reimburse the Franchisor for the full cost of such inspection or audit, including, without limitation, the charges of any independent accountant or other professionals retained by the Franchisor in connection with such audit or inspection and the travel expenses. room, board and compensation of employees of the Franchisor and/or such independent accountant. Any evaluation or inspection conducted by Franchisor is not intended to exercise, and does not constitute, control over Franchisee's day-to-day operation of the Franchised Business or to assume any responsibility for Franchisee's obligations under this Agreement.

8.4. Inquiry by Franchisor

By its execution hereof, the Franchisee specifically authorizes the Franchisor to make reasonable inquiry of the Franchisee's bankers, suppliers, landlord and other trade creditors as to their dealings with the Franchisee in relation to the Business, and to discuss the affairs, finances and accounts of the Business with the Franchisee's bankers (and the Franchisee hereby authorizes and directs its bankers and other suppliers to discuss with the Franchisor the affairs, finances and accounts of the Business). In order for the Franchisor to obtain information and copies of invoices relating to sales or other dealings between all such persons and the Franchisee in any way referable to the Business, the provision of a copy of this paragraph duly certified by the Franchisor to the Franchisee's banker or other supplier shall constitute the express and irrevocable direction of the Franchisee to such banker or other supplier to supply the requested information to the Franchisor. The Franchisee shall on request of the Franchisor, promptly execute such other directions and other documents as the Franchisor may require, in furtherance of this Section. Neither the Franchisor nor any person, firm. or corporation to whom Franchisor may make inquiry shall be deemed in breach of any privacy laws from time to time enacted in any jurisdiction with authority over this Agreement or the parties by reason only

of making or answering inquiries as provided for in this Section or otherwise in this Franchise Agreement.

9. LICENSED ITEMS AND TRADEMARKS

9.1. Marks

The Franchisee acknowledges and agrees that "Manchu WOK" is a registered trademark, that the Trademarks and Licensed Items (both registered and unregistered and as presently developed or as may be developed in the future) constitute part of the Manchu WOK Restaurant Systems and further acknowledges the existence and validity of the Licensed Items, including, without limitation, the Trademarks and that the Franchisor has the exclusive right to use the Licensed Items and Marks and to license others to use the Licensed Items and Marks. The Franchisee shall not, in any way, do anything to infringe upon, harm or contest the rights of the Franchisor (or of the owner of the Licensed Items and Marks if not the Franchisor) in and to the Licensed Items and Marks. The Franchisee shall not use any Licensed Items or Mark other than as herein permitted in connection with the business of the Franchisee's Manchu WOK Restaurant. Any materials or products which Franchisee develops in whole or in part for use in its Manchu WOK Restaurant shall become the exclusive property of the Franchisor and the Franchisee does hereby assign and transfer all right, title and interest in any such materials or products to the Franchisor. The Franchisee shall not place any Licensed Items or Mark, other than the Licensed Items or Marks originally appearing thereon, on any products, packages or other materials which the Franchisee obtains from the Franchisor or an approved supplier or with the Franchisor's written permission from third parties. The Franchisee agrees not to add any other names or words to the Licensed Items or Marks, or to make any variations in the use of the Licensed Items or Marks, unless the Franchisee has received prior written approval from the Franchisor.

9.2. Corporate Name

The Franchisee shall not use the name "Manchu WOK" or any derivation or contraction thereof in its corporate name, nor as part of any e-mail address, domain name, and/or other identification of Franchisee in any electronic medium (except with our prior written consent).

9.3. Goodwill

The Franchisee recognizes that the use of the Licensed Items and Marks, or of any mark or name that incorporates the mark "Manchu WOK" and all other Marks or other trademarks which may be developed by the Franchisor inures solely to the benefit of the Franchisor and that all goodwill arising from such use by the Franchisee shall belong and revert to the Franchisor upon the termination or expiration of this Agreement.

9.4. e-Commerce and the Web

The Franchisor retains unto itself the exclusive right to utilize the Licensed Items and Marks for e-commerce purposes and for use on the "world wide web" or "Internet". The Franchisee shall not use the Licensed Items or Marks on the world wide web (or in any other electronic medium, including but not limited to social media and social networking sites) without the express written consent of the Franchisor, which consent the Franchisor shall have the absolute right to withhold. Without limiting the generality of the foregoing, without the Franchisor's prior

written consent, Franchisee shall not operate or advertise, market, or otherwise promote the Business on the "Internet" nor shall Franchisee register any domain name containing the words **MANCHU WOK**, or any variation thereof, or establish, operate, or participate in a Web site on which such words appear. With respect to any aspect of the Manchu WOK Restaurant Systems, the Manchu WOK chain, or the Business (including, without limitation, the use of the Marks), Franchisor retains the sole right to advertise on the Internet, create or operate a Web site or sites, and use **MANCHU WOK** as part of any domain name. Franchisee acknowledges and agrees that the Franchisor exclusively owns all rights in such domain names and such other domain names as Franchisor shall designate, acquire or from time to time establish, to the absolute exclusion of the Franchisee. The parties agree that as used in this Agreement, the term "Internet" includes, but is not limited to, the world wide web, Social Media Sites (as defined in Section 9.5 below), social networking sites, mobile applications, and all other electronic media, whether created or invented now or in the future.

9.5. Social Media. Franchisee acknowledges that the use of any social networking websites or webpages, including but not limited to Facebook, YouTube, LinkedIn, and Twitter, which exploits, utilizes, displays, or otherwise makes use of any of the Marks or Franchisor's intellectual property ("Social Media Sites") is Franchisor's sole property, and Franchisee shall promptly submit to Franchisor all passwords for such Social Media Site(s) and any changes to a password shall be submitted to Franchisor within three (3) days of the change. Franchisor shall be granted full access to any Social Media Sites. Franchisee shall have no right, title or interest to any webpage on any of Franchisee's Social Media Sites including, but not limited to, all "fans", "followers", "friends" and "contacts" associated therewith which mentions, uses or refers in any way to the Proprietary Marks or Franchisor's intellectual property even if such webpage is established by Franchisee or otherwise held in Franchisee's name or by any guarantor. Franchisee acknowledges it is responsible for paying all expenses associated with such Social Media Sites.

Upon expiration, transfer or termination of this Agreement, Franchisee shall immediately take whatever steps are necessary to cancel or dismantle any such social networking account or webpage or transfer the account or webpage and all related information, including all "fans", "followers", "friends" and "contacts" associated with such accounts or webpages, to Franchisor or as Franchisor may otherwise direct. Franchisor reserves the right to hyperlink any Social Media Sites used by Franchisee to any other Internet site related to the System that Franchisor designates. Franchisor also reserves the rights to modify or use any of the content on such Social Media Sites and to authorize third parties to modify or use any of the content on such Social Media Sites, subject to the requirements of the social network service that hosts the applicable Social Media Sites, without payment of any compensation to Franchisee. Franchisee must keep Franchisor advised at all times of the addresses and domain names of any Social Media Sites used by Franchisee. Franchisee acknowledges and agrees that Franchisor has sole ownership of all domain names related to the Social Media Sites used by Franchisee, subject to the requirements of the social network service that hosts the applicable Social Media Sites.

9.6. Infringement; Discontinuing Use of the Marks

(a) The Franchisor retains the sole and discretionary right to assert infringement of the Marks against any person or concern who may use such name or marks without permission, license or authority of the Franchisor. The Franchisee shall

promptly notify the Franchisor of any infringement, actual or suspected, of which the Franchisee becomes aware, and shall co-operate at the Franchisor's expense (unless caused or contributed to by the Franchisee's negligence or misconduct) in doing such acts and things as may be asked of the Franchisee by the Franchisor or the Franchisor's counsel in the furtherance of any such proceedings.

- (b) The Franchisor shall defend the Franchisee from any and all claims of trademark infringement for use of the Marks and hold the Franchisee harmless from any such claims, provided that the Franchisee has at all times complied with the terms of this Agreement, promptly gives written notice to the Franchisor of any such claim, action or suit and grants the Franchisor complete authority to defend against same, and otherwise cooperates with the Franchisor in accordance with its instructions in the conduct thereof. The Franchisor shall conduct such legal action (or settle same) on such terms as it considers appropriate, in its sole discretion, shall bear all costs of defense of any such claim or suit, provided the Franchisee has satisfied the conditions set forth above.
- (c) Notwithstanding anything in this Agreement to the contrary, if, at any time, in the Franchisor's sole and absolute discretion, it becomes advisable to modify or discontinue the use of any of the Marks and/or use one or more additional or substitute names or marks for reasons including, but not limited to, the rejection of any pending registration or revocation of any existing registration of any of the Marks, or the superior rights of senior users, the Franchisee shall do so at its sole expense within 30 days of the Franchisor's request. In such event, the Franchisor is liable solely to reimburse the Franchisee for its reasonable direct printing and signage expenses incurred to modify or discontinue the use of the Mark(s) and substitute a different Mark(s). Such reimbursable expenses shall not include any expenditures the Franchisee makes to promote or market a modified or substitute Mark.

10. DEFAULT AND TERMINATION OF FRANCHISE

- 10.1. Events of Default <u>With Notice and No Opportunity to Cure</u>. Except as provided in Section 13.8, this Agreement and the rights conferred upon the Franchisee hereunder shall, at the option of the Franchisor, be terminated upon the occurrence of any of the following events or conditions, each of which shall be an "**Event of Default**" and which cumulatively may be referred to as "**Events of Default**," such termination is effective immediately upon the receipt by the Franchisee of notice to that effect:
 - (a) if the Franchisee loses the right to possession of the Premises and does not relocate and open for business as contemplated in Section 4.10 within the time periods specified therein, or if the Franchisee otherwise forfeits the right to conduct the Business at the Premises;
 - (b) if the Franchisee understates Gross Sales for any period by more than two percent (2%) or knowingly maintains false books and records or submits any false reports or statements to the Franchisor;
 - (c) if the Franchisee or, if it is a corporation or other entity, any shareholder or other owner, assigns or encumbers any interest in this Agreement, any interest in the

assets of the Business, or any interest in the Franchisee, without fully complying with the requirements of Section 12;

- (d) if the Franchisee or, if it is a corporation or other entity, a shareholder or other owner, breaches the provisions of Sections 11.1, 11.2, 11.3 or if section 11.4 is violated, whether by Franchisee or by Guarantor or any individual associated with either of them, or otherwise as set out in section 11.4;
- (e) if the Franchisor during any consecutive twelve (12) month period gives to the Franchisee or Guarantor three (3) or more notices of any Event of Default (whether such notices relate to the same or different Events of Default and whether or not such Events of Default have been remedied by the Franchisee);
- (f) if there has been any intentional misrepresentation of any information supplied to the Franchisor by the Franchisee or the Guarantor in connection with the Franchisee's application for this franchise;
- (g) if the Franchisee or a Guarantor commits or suffers any act of bankruptcy or is adjudicated a bankrupt, or becomes an insolvent person, or if a petition in bankruptcy is filed against the Franchisee or a Guarantor, or if the Franchisee or a Guarantor takes or attempts to take advantage of any federal or state law now or hereafter in effect for the relief of debtors;
- (h) if any proceedings are brought with respect to the Franchisee seeking any reorganization, arrangement, composition, liquidation or dissolution or similar relief under any relevant statute or law;
- (i) if a receiver or other custodian (permanent or temporary) of any assets of the Business or of any shares in the capital of the Franchisee is appointed or a petition seeking same is filed, or if execution, attachment or similar process is levied against any such assets or shares, or if distress or other analogous process is made against any of the assets of the Business;
- (j) if the Franchisee ceases the operation of the Business or otherwise abandons it; if the Franchisee does not, for any reason, open the Business to the public for three consecutive days, or 14 days in any 12-consecutive-month period, it is deemed to have abandoned the Business;
- (k) if the Franchisee or any affiliated entity or any other entity to which the Guarantor is a party in favor of the Franchisor or an Affiliate is in default under any other Agreement with the Franchisor including without limitation any Franchise Agreement and/or Area Development Agreement and such default is not fully remedied within the time therein set out:
- (I) If an imminent threat or danger to public health or safety results from the operation of the Restaurant;
- (m) If Franchisee takes, withholds, misdirects or appropriates for Franchisee's own use any funds withheld from Franchisee's employees' wages for employee taxes,

- FICA, insurance or benefits or generally fails to deal fairly and honestly with Franchisee's employees and/or customers; or
- (n) If Franchisee's assets, property or interests are "blocked" under any Anti-Terrorism Law or if Franchisee is otherwise in violation of such law.
- 10.2. Events of Default <u>With Notice and With Opportunity to Cure.</u> The Franchisor may, at its sole option, terminate this Agreement and all rights hereunder, after affording Franchisee an opportunity to cure the default as set forth below, effective immediately, unless otherwise specified herein or in the notice of termination, upon delivery of notice of termination if the Franchisee and/or Guarantor or any corporation to which the Franchisee and/or Guarantor have assigned this Agreement:
 - (a) if the Franchisee or the Manager or any replacement Manager does not after two (2) attempts successfully complete his/her Initial Training within ten (10) days after the Franchisor delivers notice of such Event of Default:
 - (b) if the Franchisee fails to pursue or to successfully complete the Development Obligations in this Agreement or any other franchise agreement or Area Development Agreement to which Franchisee is a party, on a timely basis or does not commence the operation of the Business as required by Sections 4.3 and 4.4 and fails to remedy such default within ten (10) days after the Franchisor delivers notice of such Event of Default;
 - (c) if the Franchisee fails to pay any amount owing to the Franchisor (whether or not under this Agreement) or any Affiliate of the Franchisor and fails to remedy such default within five (5) days after receiving notice of such Event of Default from the Franchisor (or such shorter period of time as may be provided to cure such default, if any);
 - (d) if a default occurs under the Lease or sublease of the Premises and Franchisee fails to cure such default as provided in the Lease or sublease;
 - (e) if the Franchisee fails to furnish reports, financial statements, tax returns or schedules or any other documentation required by the provisions of this Agreement or to provide electronic access to its records and its system, or fails to furnish the same at the times herein specified and Franchisee does not correct such failure within five (5) days, (or such longer period of time as the Franchisor in its sole discretion deems appropriate,) after written notice from the Franchisor (which shall describe the action that the Franchisee must take) is delivered to the Franchisee;
 - (f) if the Franchisee or the Guarantor or if either is a corporation or other entity, a shareholder or other owner, engages in any conduct that, in the reasonable opinion of the Franchisor, reflects unfavorably upon or is detrimental to the Licensed Items or Marks, to the good name, goodwill or reputation of the Franchisor or any Affiliate of the Franchisor, or the Manchu WOK Restaurant System, or to the business, reputation or goodwill of its franchisees (such conduct to include by way of example only and without limitation the failure to pay on a timely basis any amount owing to a trade creditor of the Franchisee or the misuse of any of the Licensed Items or Marks) and the Franchisee or the shareholder, as applicable, fails to remedy or cease such

conduct within ten (10) days after the Franchisor delivers notice of such Event of Default; or

(g) except as otherwise provided in this Section 10.2, if the Franchisee breaches any of its other obligations from time to time to the Franchisor (whether under this Agreement or any other Contract) or any Affiliate of the Franchisor, including, without limitation, Franchisee's obligations as set out in Article 6 and fails to cure such breach within five (5) days after the Franchisor delivers notice of same (or such shorter period of time as may be provided to cure such breach, if any); or if the Franchisor, or any of its Affiliates, terminates any contract.

Franchisor's waiver of any Franchisee's defaults will not constitute a waiver of any other default and will not prevent Franchisor from requiring Franchisee to strictly comply with this Agreement and/or any other agreement with the Franchisor and/or its affiliates.

10.3. Termination Pursuant to Section 10.2(a)

If the Franchisor terminates this Agreement pursuant to Subsection 10.2(a), then the Franchisor, the Franchisee and the Guarantor shall each execute such documents as may be required to fully rescind all Contracts between them in respect of the franchise herein granted, except for the provisions of Section 11 of this Agreement, and the Franchisee shall promptly return to the Franchisor the Confidential Manual and other documents, including any copies thereof, furnished to or otherwise obtained by the Franchisee in connection with the Franchise herein granted. In the event that the Franchisee shall fail to execute such documents as and when required by this Section, Franchisor shall have the right so to do in Franchisee's name on behalf of the Franchisee and this shall constitute Franchisors complete authority so to do. Upon compliance with the foregoing, the Franchisor shall refund to the Franchisee all amounts received by the Franchisor in respect of the Initial Franchise Fee referred to in Section 3.1, less i) the sum of Ten Thousand Dollars (\$10,000) and ii) any out of pocket costs or expenses as Franchisor may have incurred pursuant to this Agreement, including, without limitation, any and all costs associated with training or the leasing and/or development of the Premises, which amounts are acknowledged by the Franchisee and the Guarantor to be Franchisor's reasonable costs and expenses incurred to date of termination in connection with this Agreement (and not a penalty).

10.4. Cease Use of Marks and Licensed Items

Upon the expiration or sooner termination of this Agreement for any reason, all rights of the Franchisee hereunder shall be at an end and the Franchisee shall immediately cease to use the Licensed Items and Marks, by advertising or otherwise, and shall remove the Licensed Items and Marks from all signs, slogans, symbols, letterheads, stationery and from all documents, instruments or forms of whatever character which use of any of the Licensed Items and Marks. The Franchisee shall further execute such documents and take such action as the Franchisor may, in its sole and absolute discretion, deem necessary or advisable to evidence the fact that the Franchisee has ceased such use and has no further interest or right therein. Without limiting the generality of the foregoing, the Franchisee shall, at the request of the Franchisor, take all such action as may be necessary to cancel any trade name or assumed name registration with respect to any of the Marks. The Franchisee shall furnish to the Franchisor evidence satisfactory to the Franchisor of compliance with its obligations hereunder within thirty (30) days after expiration or sooner termination of this Agreement. Unless the

Franchisor has entered into possession of the Premises pursuant to a conditional lease assignment clause in the Franchisee's lease of the Premises, the Franchisee shall immediately upon the expiration or sooner termination of this Agreement for any reason, make such removals or changes in signs and decor of the Premises as the Franchisor shall request, so as to effectively distinguish the Premises from their former appearance as a Manchu WOK Restaurant. If the Franchisor relinquishes possession of the Premises to the Franchisee following a re-entry pursuant to Section 10.8 hereof, then unless otherwise agreed by the Franchisor, the Franchisee shall immediately de-identify the Premises as aforesaid. Neither the Franchisee nor any Guarantor shall at any time following expiration or sooner termination of this Agreement identify itself or himself in any manner as having been at any time involved in a business relationship with the Franchisor; nor may the Franchisor nor any of its owners or any Guarantor disparage the Franchisor or the Manchu WOK Restaurant System.

10.5. Return of Manuals

Immediately following the expiration or sooner termination of this Agreement, the Franchisee shall return to the Franchisor all copies of the Confidential Manual (which are provided to the Franchisee on a "loan basis" only and shall not at any time become the property of the Franchisee in any event) and all bulletins and advertising matter used by or in the possession of the Franchisee or the Guarantor relating to the Business, together with all materials bearing any of the Marks.

10.6. Option to Purchase

The Franchisor shall have the option, for a period of thirty (30) days following the expiration or sooner termination of this Agreement, to purchase any or all Equipment and Fixtures, leasehold improvements that are not the property of the Landlord, signage, inventory and supplies other than items of a personal nature (the "Franchise Assets") owned and used by the Franchisee in connection with the Business, for a purchase price equal to their then current value determined on an asset-by-asset liquidation-sale basis (rather than on a going-concern basis). Without limiting the generality of the foregoing, the then current value of the Franchise Assets shall not contain any factor or increment for goodwill and other intangibles, and the Franchisor may exclude from the Franchise Assets purchased hereunder any items which, in the reasonable opinion of the Franchisor, are obsolete, shop-worn, damaged or otherwise not in marketable or useable condition. If the parties cannot agree upon the value of the Franchise Assets within a period of ten (10) business days following the exercise of such option by the Franchisor, an independent appraiser shall be designated by the Franchisor to do so, and his determination shall be final and binding on all parties, and no appeal shall lie therefrom. If the Franchisor elects to exercise its option to purchase, it shall be entitled to set off against the purchase price for the assets all amounts owing to it from the Franchisee (whether or not under this Agreement) and the cost of the appraisal, if any. If the Franchisor exercises its purchase option, the transaction of purchase and sale shall be closed on a date to be determined by the Franchisor, within five (5) days of the date that the purchase price is finally determined. The transaction of purchase and sale shall be completed in accordance with all applicable legislation, and without limiting the generality of the foregoing, the Franchisee shall comply with applicable bulk sales legislation in such manner as will permit the Franchisor to pay the entire purchase price to the Franchisee at closing. The Franchisee shall deliver against payment for the Franchise Assets being purchased a bill of sale with full covenants and warranties as to title (including, without limitation, that all items are being sold free and clear of all liens, claims, and encumbrances), together with such other documents as may be necessary or desirable to

complete the transaction of purchase and sale. If the Franchisor exercises its purchase option, it may, in addition to the right of set-off provided above, pay from monies otherwise payable to the Franchisee any debts which the Franchisee owes to any Affiliate of the Franchisor or to any trade creditor (including without limitation any designated or approved supplier) of the Business and remit the balance of the purchase price then remaining, if any, to the Franchisee.

10.7. Removal of Equipment

If following the expiration or sooner termination of this Agreement the Franchisor enters into possession of the Premises and elects not to exercise any of its purchase options under Section 10.6 then the Franchisee shall have five (5) days within which to remove its Franchise Assets from the Premises, failing which the Franchisor may cause the same to be removed at the cost of the Franchisee. In such event, the Franchisor may dispose of the Franchise Assets in whole or in part, in such manner as it, in its sole and absolute discretion, deems appropriate. The proceeds of such disposition shall be applied first in payment of all expenses incurred by the Franchisor in connection with the removal and disposition of such property, then in payment of any amounts owing by the Franchisee to the Franchisor or its Affiliates (whether under this agreement or otherwise) and the balance, if any, shall be paid to the Franchisee. Ownership of any Franchise Assets not so removed by the Franchisee shall be deemed to have passed to the Franchisor and may be used by the Franchisor free of charge for the purpose of carrying on the business of a Manchu WOK Restaurant or otherwise at the Premises or elsewhere as the Franchisor may from time to time determine.

10.8. Right of Re-entry

Forthwith upon expiration or sooner termination of this Agreement or upon the occurrence of an Event of Default under this Agreement, and whether or not this Agreement shall have been terminated as a result thereof, the Franchisor may enter upon, occupy and use all or any part of the Premises, and any Equipment and Fixtures, leasehold improvements and other assets located in, on or about the Premises and used in connection with the Business (including, without limitation, all telephone numbers and telephone listings used in connection with the Business). The Franchisor shall pay a reasonable use charge to the Franchisee for its use of such assets if it does not elect to purchase same pursuant to the terms of this Agreement. The Franchisor shall not be liable for any trespass, tort, or neglect in so doing or in respect of any depreciation or damages in connection with such action. All revenues, monies, profits, benefits and advantages derived from the operation of the Premises throughout such period of occupation shall be for the exclusive account of the Franchisor, and the Franchisor shall pay and discharge all debts and liabilities incurred by it in connection with its operation of the Premises. In addition, the Franchisor shall have the option, but not the obligation, to pay all amounts owing by the Franchisee to any creditor of the Business and any amount so paid shall be charged to and repaid by the Franchisee to the Franchisor upon demand (or deducted by Franchisor from any monies otherwise due Franchisee) and until so repaid shall bear interest as provided in this Agreement for overdue amounts. The Franchisor shall have no obligation to retain any employees of the Business nor to honor any contractual commitments previously made by the Franchisee in connection therewith. Any liability with respect thereto shall be exclusively borne and paid for by the Franchisee. If the Franchisor elects to retain any such employee, such employment shall be pursuant to a new employment agreement between the Franchisor and such employee and shall commence on the first business day on which the Franchisor commences operating the Premises. Any claim of such employee for unpaid salary, vacation or other benefits arising from his employment with the Franchisee shall be the

exclusive responsibility of and be paid solely by the Franchisee. The Franchisee and the Guarantor specifically indemnify and save the Franchisor harmless from any and all claims by employees or by government agencies pursuant to any federal or state employment or labor law in respect of past or continuing employees. The right of the Franchisor to use the Premises and the assets of the Business as aforesaid shall continue until the expiration of the option period in Section 10.6, but if the Franchisor elects to exercise such option, such right shall continue throughout the period preceding and including the closing of the transaction of purchase and sale.

10.9. Transfer of Telephone Numbers

The Franchisee agrees that upon the expiration or sooner termination of this Agreement all interest in and right to use all telephone numbers and all listings applicable to the Business (other than listings of a personal nature placed by the Franchisee) in use at the time of such expiration or sooner termination shall, upon request by the Franchisor, be transferred to and vested in the Franchisor. The Franchisee acknowledges that its execution of this Agreement shall, upon the expiration or sooner termination of this Agreement constitute its assignment of all its rights in all telephone numbers and listings applicable to the Business; notwithstanding the foregoing, Franchisee shall execute such additional documentation as may be reasonably required by the Franchisor from time to time to enable the Franchisor to obtain an assignment of the telephone number and listings applicable to the Franchise business. The Franchisor shall thereupon have the exclusive right to use such numbers and listings and to authorize the use thereof by any other person. To facilitate the completion of any such transfer, the Franchisee further agrees that upon the request of the Franchisor, the Franchisee shall pay to the telephone company all charges and other amounts which may be payable in connection with the telephone numbers being so transferred up to and including the date of transfer. The Franchisee agrees to reimburse the Franchisor upon demand for all amounts paid by the Franchisor on account of any such charges as a result of the Franchisee's failure to pay same. together with interest thereon as provided in this Agreement for overdue amounts or to deduct such amounts from any monies otherwise due Franchisor by Franchisee.

10.10. Payment of Monies Due

Franchisee shall promptly pay all sums owing to Franchisor and its affiliates. In the event of termination for any Event of Default of Franchisee, such sums shall include all damages, (including future royalty fees that otherwise likely would have been paid from the termination date through the end of the initial term but for the early termination of the Agreement), costs, and expenses, including reasonable attorneys' fees, incurred by Franchisor as a result of the Event of Default, which obligation shall give rise to and remain, until paid in full, a lien in favor of Franchisor against any and all of the personal property, furnishings, equipment, signs, fixtures, and inventory owned by Franchisee and on any and all premises operated hereunder at the time of default.

10.11. Settlement of Accounts

Upon the earlier of sixty (60) days after the expiration or sooner termination of this Agreement and the closing of the transaction of purchase and sale contemplated in Section 10.6, there shall be an accounting among the parties with respect to the monies due by each to the others. Each of the parties agrees promptly to pay to the others, by certified check, whatever money shall be found to be owing to the others pursuant to such accounting.

10.12. Power of Attorney

If at any time the Franchisee does not execute any document reasonably requested by the Franchisor in connection with any of the Franchisor's rights under this Section 10.12 or elsewhere in this Agreement, the Franchisor may do so on behalf of the Franchisee and the Franchisee does hereby constitute the Franchisor as its true and lawful attorney to execute any such document(s) on its behalf.

10.13. Assignment of Lease

Notwithstanding anything in this Agreement to the contrary, upon the expiration or sooner termination of this Agreement, if the Franchisee has directly entered a Lease with the Landlord, the Franchisee shall assign the Lease to the Franchisor if the Franchisor so requests (irrespective of whether the Franchisor exercises any of its purchase options under Section 10.6).

11. RESTRICTIVE COVENANTS

11.1. Secrecy and Confidentiality

The Franchisee and the Guarantor for itself/himself/themselves and its/their officers directors, shareholders, employees, executors, administrators and permitted assigns acknowledge that in connection with the operation of the Business the Franchisor will be disclosing to them confidential information and trade secrets of the Franchisor, including, without limitation, information in the Confidential Manual. For the purposes of this Agreement, confidential information includes all information that the Franchisor specifically designates as confidential, as well as all information of a non-public nature, whether or not so designated. The Franchisee and the Guarantor each jointly and severally agree that none of them will use or divulge to any person any such trade secrets or confidential information, except that during the Term of this Agreement the Franchisee may use such trade secrets and confidential information solely for the purpose of conducting the Business in accordance with the provisions of this Agreement and may divulge such trade secrets and confidential information to its employees, but only on a "need-to-know" basis and in circumstances that will maintain its confidentiality. The Franchisee must cause its employees and agents to maintain the confidentiality of such confidential information and trade secrets, and shall obtain from the foregoing persons, and deliver copies thereof to the Franchisor, written covenants to maintain such confidentiality, in form prescribed by the Franchisor. A breach of any of the provisions of this Section 11.1 by any of the Franchisee's employees or agents is considered a direct breach by the Franchisee thereof.

11.2. In Term Competition: Franchisee

The Franchisee agrees that it will not directly or indirectly (without the prior written consent of the Franchisor, which consent may be unreasonably withheld) during the Term, as owner, officer, director, employee, agent, lender, broker, consultant, franchisee, or in any other similar capacity whatsoever, be connected in any manner with the ownership, management, operation or control, or conduct of any restaurant business or retail food outlet that specializes in the sale of Chinese and Asian-style food or that is reasonably deemed by the Franchisor to be similar in nature, style or otherwise to any Manchu WOK Restaurant or Street-Front restaurant or using the Manchu WOK Restaurant Systems or any part or parts thereof.

11.3. In Term Competition: Guarantor

The Guarantor agrees that he or she will not directly or indirectly (without the prior written consent of the Franchisor, which consent may be arbitrarily withheld) during the Term, as owner, officer, director, employee, agent, lender, broker, consultant, franchisee, or in any other similar capacity whatsoever, be connected in any manner with the ownership, management, operation or control, or conduct of any restaurant business or retail food outlet that specializes in the sale of Chinese and Asian-style food or that is reasonably deemed by the Franchisor to be similar in nature, style or otherwise to a Manchu WOK Restaurant or Street-Front restaurant or using the Manchu WOK Restaurant System or any part or parts thereof.

11.4. Post Term Competition

The Franchisee and the Guarantor acknowledge that the Licensed Items, Marks, the business reputation and goodwill associated therewith, the methods and techniques employed by the Franchisor, the training and instruction and confidential information to be provided hereunder, the knowledge of the services and the operational, sales, promotional and marketing methods and techniques of the Franchisor and the opportunities, associations and experiences established and acquired by them hereunder are of considerable value, as well as is the location of the Business and Trade Area pertinent thereto, and constitute legitimate business interests of the Franchisor and as a result the Franchisee and Guarantor agree to reasonably protect such legitimate business interests as follows:

The Franchisee and the Guarantor agree that he/she/it/they will not directly or indirectly (without the prior written consent of the Franchisor which consent may be arbitrarily withheld) at any time during the period of eighteen (18) months from the latest date of termination of this Agreement, whether as owner, shareholder, officer, director, employee, agent, lender, broker, consultant, franchisee, or in any other similar capacity whatsoever, be connected in any manner with the ownership, management, operation or control, or conduct of any restaurant business or retail food outlet located at the Premises, or the premises at which any restaurant or outlet utilizing the Manchu WOK Restaurant Systems (whether franchised or company or affiliate owned) within three (3) miles of the exterior boundaries of any building in which (i) the Premises are located or (ii) any premises at which any restaurant utilizing the Manchu WOK Restaurant Systems is then being operated, or is then being constructed or then under contract to be constructed.

11.5. Contravention by Related Persons

Franchisee and Guarantor acknowledge and agree that they or any of them will be deemed to be in contravention of any of the provisions of Section 11 if any restaurant business or retail food outlet be opened or operated (or intended to be opened or operated) in any manner that if carried out by the Franchisee or Guarantor, would violate any of the provisions of this Section by or on behalf of a spouse, child, shareholder or other owner, director, or officer of Franchisee or Guarantor or by a person, firm, corporation, association, or other business or legal entity in which such spouse, child, shareholder or other owner, etc., has any interest (legal, beneficial, economic, or otherwise) whatsoever, whether or not Franchisee or Guarantor has any legal or beneficial interest in such restaurant business. All such parties must enter a written agreement, in form and substance satisfactory to the Franchisor, under which they agree, jointly and severally, to be bound by the provisions of this Section, as well as the confidentiality provisions contained in this Agreement.

11.6. Solicitation

- (a) The Franchisee and Guarantor jointly and severally covenant agree that during the Term and for a period of eighteen (18) months thereafter:
 - (i) Not applicable;
 - (ii) they nor either of them will not divert or attempt to divert any business or any customers of the Business to any other competitive establishment, by direct or indirect inducement or otherwise; and
 - (iii) they nor either of them will not counsel or induce any franchisee of the Franchisor to leave the Manchu WOK Restaurant System.
- (b) The parties agree that each of the covenants contained in Subsections (i), (ii) and (iii) of this Section 11.6(a) are independent of each other and are severable.

11.7. Acknowledgement

- (a) The Franchisee and the Guarantor acknowledge, confirm, covenant and agree that the restrictions contained in this Section are reasonable in order to protect the legitimate business interests of the Franchisor, its Affiliates, and the franchisees of the Franchisor, and all defenses to the strict enforcement of such restrictions by the Franchisor are hereby waived.
- (b) The Franchisee and Guarantor acknowledge, confirm, covenant and agree that the scope of activities prohibited in this Section, as well as the length of the term and geographical restrictions contained therein, are necessary to protect the Franchisor's legitimate business interests and are fair and reasonable and not the result of overreaching, duress, or coercion of any kind. The Franchisee's and Guarantor's full, uninhibited, and faithful observance of each of the covenants contained in this Section will not cause any undue hardship, financial or otherwise. Enforcement of each of the covenants contained in this Section will not impair the Franchisee's or the Guarantor's ability to obtain employment commensurate with its abilities and on terms fully acceptable to it or otherwise to obtain income required for the comfortable support of itself and its family, and the satisfaction of the needs of its creditors. The Franchisee's and Guarantor's special knowledge of the business of a Manchu WOK Restaurant (and anyone acquiring such knowledge through the Franchisee or Guarantor) is such as would cause the Franchisor and its franchisees serious injury and loss if it (or anyone acquiring such knowledge through the Franchisee or Guarantor) were to use such

knowledge to the benefit of a competitor or were to compete with the Franchisor or any of its franchisees.

- (c) If any court of final jurisdiction finally holds that the time or territory for or to which this Section applies or the scope of activities prohibited thereunder—or that any provision stated in this Section —constitutes an unreasonable restriction upon the Franchisee or any Guarantor, the provisions of this Agreement are not thereby rendered void, but shall apply as to time and territory or to such other extent as such court finally concludes or indicates is a reasonable restriction under the circumstances. The time periods set forth in this Section are suspended during any period in which the Franchisee or Guarantor is breaching any of its terms or involved in a legal action or proceeding challenging the validity or enforceability thereof.
- (d) Without the Franchisee's or Guarantor's consent, the Franchisor, in its sole discretion, may at any time or times unilaterally reduce or limit the scope of any covenants set forth in this Section. Any such reduction is effective immediately upon the Franchisor's delivery of written notice to either of them. The Franchisee and Guarantor shall comply immediately with any covenant as so modified. Such modified covenant is fully enforceable to the extent permitted by applicable law and shall be subject to the provisions of Section (c) above.
- (e) The covenants in this Section shall be construed as independent of any other covenant or provision of this Agreement. The existence of any claim the Franchisee or Guarantor may have against the Franchisor or any of its affiliates (whether or not arising from this Agreement) shall not be deemed to be a defense to the enforcement of such covenant(s) against the Franchisee or Guarantor.
- (f) The Franchisee and the Guarantor further agree that their failure to fully comply with the restrictions contained in this Section will constitute, among other things, unfair competition with the Franchisor. The Franchisee and the Guarantor acknowledge that the Franchisor would suffer irreparable injuries if the Franchisee or any person referred to in this Section breaches any of these restrictions and, accordingly, agree (in addition to any other remedies available to Franchisor whether pursuant to this Agreement or otherwise) to the granting without prior notice (to the extent that applicable notice requirements may be waived) of temporary and permanent injunctions restraining any such breach, or any other manner of equitable relief. Such injunctive relief may be granted without the need for the Franchisor demonstrating irreparable injury or posting any bond or other security (or if such posting of security may not be waived, by posting the sum of \$1,000 as security). The Franchisee and the Guarantor further jointly and severally agree to indemnify and save the Franchisor harmless in respect of all costs and expenses (including legal fees) incurred by the Franchisor in seeking to obtain any such injunction(s) and in pursuing any such other remedies.

12. ASSIGNMENT AND TRANSFER

12.1. Assignment or Transfer by Franchisor

The Franchisor may assign or delegate all or any part of its rights or duties, arising from this Agreement, provided that any assignee, or delegate, shall agree in writing to assume all obligations undertaken by the Franchisor herein relating to the rights so assigned, or the

obligations delegated. Upon such assignment or delegation and assumption, the Franchisor shall be released from and under no further obligation with respect to any obligations so assumed. The Franchisee and the Guarantor agree that any such assignment or delegation shall not affect any of their responsibilities under this Agreement to and under which they remain jointly and severally subject and bound.

12.2. Assignment or Transfer by Franchisee

During the Term of this Agreement, the Franchisee and the Guarantor shall not sell, assign, charge, grant a security interest in or otherwise transfer or encumber in whole or in part in any manner, including, without limitation, by operation of law, any interest in this Agreement, the lease for the Premises, the property and assets of the Business or any ownership interest in the Franchisee, nor may the Franchisee issue any further shares in its capital or redeem or purchase any of its issued shares, nor may the Franchisee merge, amalgamate, re-organize or engage in any other similar proceeding, (collectively, "Transfer") without in each case obtaining the prior written consent of the Franchisor which consent shall not be unreasonably or arbitrarily withheld. Any proposed or actual transfer of a direct or indirect ownership interest in Franchisee shall also be considered as a "Transfer" for all purposes under this Agreement. Any actual or purported Transfer without the Franchisor's prior written consent shall constitute a material default of this Agreement and shall be null and void. In considering any request for its consent to any such Transfer the Franchisor may consider, among other things, the qualifications, background, apparent ability and credit standing of the proposed transferee ("Transferee"), as well as the probable post-closing operational viability of the Manchu WOK Restaurant should the intended Transfer be completed in accordance with the proposed terms of the Transfer. In addition, it is required as a condition precedent to the granting of the Franchisor's consent that:

- (a) there shall be no existing default in the performance or observance of any of the Franchisee's or the Guarantor' obligations pursuant to any Contract or with any supplier or other trade creditor of the Business, the Franchisor or its Affiliates;
- (b) the Franchisee shall have settled all outstanding accounts with the Franchisor, its Affiliates and all other trade creditors of the Business;
- (c) the Franchisee (and all of its owners, including the Guarantor) shall have delivered to the Franchisor a general release of its/their claims against the Franchisor, its Affiliates and their respective directors and officers, in form and substance satisfactory to the Franchisor;
- (d) the Transferee and, if it is a corporation, limited liability company, or partnership, its shareholders, members or partners as the case may be, shall have entered into a written assignment and guarantee in form and substance prescribed by the Franchisor, assuming and agreeing to discharge all of the Franchisee's obligations under this Agreement or, at the option of the Franchisor, shall have executed a new franchise agreement, for no less than the unexpired portion of the Term of the Agreement, in the standard form then being used by the Franchisor for the grant of franchises for Manchu WOK Restaurants for new single-unit franchisees. Such franchise agreement shall not provide for payment of any Initial Franchise Fee, but may otherwise contain financial, legal and other terms and conditions substantially different from, and more onerous than, those in this Agreement, including, without limitation, different terms and conditions regarding: royalties, advertising obligations and contributions, rent or

other payments or expenditures (any of which may be higher) and the methods of computing same. In addition, the Transferee and its shareholders or partners, as the case may be, shall have executed such other documents and agreements as are then customarily used by the Franchisor in the granting of franchises for Manchu WOK Restaurants;

- (e) the Transferee and/or its proposed Managers (one or both of whom will be responsible for the ongoing operations of the Business and any other persons required by the Franchisor shall have satisfactorily completed at his/her own expense the Franchisor's training program then in effect for all new franchisees of Manchu WOK Restaurants and shall have paid the Training Fee applicable thereto; If Transferee or any of its employees or representatives cancel or reschedule participation in any training course or program which Transferee schedules with Franchisor, Transferee or Franchisee must reimburse Franchisor for all of its employee's travel expenses, wages and other expenses incurred as a result of such cancellation or rescheduling;
- (f) unless the Franchisor otherwise agrees in writing, any transfer of assets shall be effected in compliance with all applicable bulk sales legislation;
- (g) a transfer fee shall, upon application for the consent of the Franchisor to the Transfer, have been paid to the Franchisor by the Franchisee in the amount of Ten Thousand Dollars (\$10,000), which transfer fee shall be refundable, less the Franchisor's reasonable costs and expenses, together with any professional and related expenses incurred in connection therewith, including, without limitation, those expenses relating to investigations of the proposed transferee, if the proposed transfer is not approved by the Franchisor;
- (h) if the Franchisee leased the Premises pursuant to Section 4.2(a), then the Transferee and Landlord must sign and deliver to Franchisor a new Lease Addendum; and
- (i) if required by the Franchisor, the Franchisee must use and pay for the services of a third-party escrow agent, approved in advance by the Franchisor, to facilitate the transfer process and ensure all debts and other payments are properly made.
- (j) the assignee, at its own cost, agrees to make all capital expenditures requested by the Franchisor to renovate, refurbish, remodel and modernize the Store so as to reflect the then-current image for the Franchisor prior to the transfer or within a timeframe specified by Franchisor;

12.3. Additional Documentation

(a) The Franchisee and Guarantor shall cause each person who, from time, to time becomes a shareholder, member, manager or other owner, , director, and/or officer of the Franchisee, before they obtain such status, to execute a written agreement with the Franchisor:

- (i) in the case of a director or officer who is not also a shareholder of the Franchisee, undertaking to be bound by all of the restrictions of Section 11, and all the confidentiality provisions in this Agreement, as though he was a Guarantor: and
- (ii) in the case of a shareholder, member, manager or other owner of the Franchisee, undertaking to be bound by all obligations of the Guarantor under this Agreement (including the guarantee in Section 15.2) as though he was a Guarantor.
- (b) The Franchisee shall furnish to the Franchisor contemporaneously with signing this Agreement, and thereafter annually with the financial statements referred to in Section 8.2 (c) and as well contemporaneously with all transfers or issuances of shares in its capital, a list of all shareholders having an interest in the Franchisee as currently set out in Schedule 'F' hereto, reflecting the percentage interest of each shareholder and the number and class of shares directly or indirectly owned or controlled by him. In addition, the Franchisee shall furnish to the Franchisor on signing this Agreement and thereafter within five (5) days after any transfer or issue of any of its shares the other documents referred to above in this Section, duly executed by all required parties other than the Franchisor. The Articles of Incorporation and by-laws of the Franchisee shall recite that the issue or transfer of any shares in its capital is restricted by the terms of this Agreement, and copies of such articles of incorporation and by-laws shall be furnished to the Franchisor forthwith upon request. The Franchisee shall maintain stop transfer instructions against the transfer of shares on its records subject to the foregoing restrictions of this Section and shall have all outstanding share certificates endorsed with the following legend printed conspicuously upon the face of each such certificate:

"The transfer of shares represented by this certificate is subject to the terms and conditions of a Franchise Agreement made with MANCHU WOK FRANCHISING USA INC. A copy of that agreement is available for inspection at the Franchisor's principal place of business."

The Franchisor shall have access to the corporate records and minute books of the Franchisee at reasonable times and shall be entitled to make copies thereof for its own purposes.

The Franchisee and the Guarantor jointly and severally represent and warrant to the Franchisor that Schedule "F" is an accurate list of all of the registered and beneficial shareholders of the Franchisee, showing the number and class of shares held by them. The Franchisee will advise the Franchisor of any changes to Schedule "F" which must be in conformity with the provisions of this Section and of this Agreement.

- 12.4. Death or Permanent Disability
 - (a) If any Franchise Owner (defined below):
 - (i) dies; or
 - (ii) becomes permanently disabled as defined in Section 12.5;

then within thirty (30) days after either event, the Franchisee (or its legal representative) must hire and maintain a replacement satisfactory to the Franchisor to perform such obligations.

- (b) Within six months of the permanent disability or death of any of the following individuals, such individual's estate must transfer his or her interests in the Franchisee (or in any of the Franchisee's owners) or in this Agreement in accordance with the terms of this Section 12:
 - (i) any individual who holds a 25% or greater voting or ownership interest in the Franchisee (or in any of the Franchisee's owners); or
 - (ii) any individual who is the Franchisee.

For the purposes of this Section 12.4, "Franchise Owner" means the Franchisee if the Franchisee is an individual or the shareholder designated as Manager under Section 6.2

12.5. Deemed Disability

For the purposes of this Section 12, the Franchise Owner shall be deemed to be permanently disabled if his usual day to day participation in the Business (as required by the terms of this Agreement) is for any reason impaired or he is otherwise unable to perform his or her obligations under this Agreement by reason of mental or physical disability for a cumulative period of forty-five (45) days in any twelve (12) month period.

12.6. Right to Manage

Upon the death or permanent disability of the Franchisee or of the Manager, if the Franchisee does not timely comply with Section 12.4(a), the Franchisor shall have the right (but without obligation) to operate and manage the Business for the account of the Franchisee. If the Franchisor so operates and manages the Business, (i) the Franchisor shall make a complete accounting to and return the net income from such operation (being revenues less all applicable expenses and charges incurred in the operation of the Business inclusive of amounts sufficient to reimburse the Franchisor for its reasonable expenses in connection therewith and a reasonable management fee) to the Franchisee; and (ii) in no event shall the Franchisor be liable for other than gross negligence in the operation and management of the Business. Notwithstanding anything in this Section 12.6 to the contrary, the primary obligation to operate and manage the Business shall, at all times be upon the Franchisee, and the Franchisor makes no representation or warranty whatever that the operation or management or any part thereof by or on behalf of the Franchisor shall succeed in returning or maintaining a profit.

12.7. Franchisor's Right to Terminate

If any disposition of shares or assets required under Section 12.4 has not taken place within six (6) months as required by Section 12.4(b) or if the Manager shall not have been replaced as required by section 12.4(a), the Franchisor shall thereafter have the continuing right, exercisable upon ten (10) days' notice to the Franchisee, to terminate this Agreement and all the Franchisee's rights hereunder, including the franchise hereby granted. In such event and if the Franchisor so elects, then the provisions of Sections 10.4 to 10.13 inclusive shall thereupon become applicable.

12.8. Option to Purchase

Without in any way derogating from the Franchisor's right to reject a Transfer pursuant to Section 12.2, if at any time during the Term of this Agreement the Franchisee or any shareholder or partner, as the case may be, obtains a bona fide offer ("Offer") to acquire the whole or any part of its or his interest in this Agreement, in the Lease for the Premises, in any assets of the Business (other than in the ordinary course of business) or in any ownership interest in the Franchisee which the Franchisee or the shareholder or partner, as the case may be, wishes to accept, then the Franchisee or shareholder or partner, as the case may be, shall promptly deliver to the Franchisor written notice of and a true copy of such Offer. Upon receipt of the notice, the Franchisor shall have the option of purchasing the property forming the subject matter thereof upon the same terms and conditions as those set out in the Offer, except (i) that there shall be deducted from the purchase price the amount of any commission or fee that would otherwise have been payable to any broker, agent or other intermediary in connection with the sale of such property to the (other) offeree, (ii) the Franchisor shall have the right to substitute cash for any other form of consideration specified in the Offer and if the cash amount cannot be agreed upon within ten (10) days, the question shall be referred to an arbitrator who shall be chosen in accordance with the provisions of Section 14 hereof, (iii) the Franchisor shall have the further right to pay in full the entire purchase price at the time of closing (with an appropriate reduction of the purchase price applicable to the present value of paying in cash should the Franchisor elect so to do) and (iv) any and all monies due the Franchisor or any supplier to the Business shall be deducted from the purchase price and applied or paid by the Franchisor. In the case of a sale of assets to which bulk sales laws apply, the Franchisee shall comply with such laws in a manner which will permit the Franchisor to pay the entire purchase price to the Franchisee at closing. The Franchisor may exercise its option at any time within thirty (30) days after receipt of the notice and copy of the Offer by giving notice of acceptance to the Franchisee or the shareholder or partner, as the case may be. If the Franchisor declines to exercise such option and if such transfer is otherwise approved by the Franchisor, the Franchisee or the shareholder or partner, as the case may be, shall be at liberty to complete the transfer to the third party purchaser in accordance with the terms of the Offer, but always pursuant to the provisions of this Agreement, except that notwithstanding the terms of the Offer such transaction must be completed within thirty (30) days of the date on which the Franchisor notifies the Franchisee or the shareholder or partner, as the case may be, of its approval of such transaction. If the transaction is not completed within such thirty (30) day period then the foregoing provisions of this Section shall apply again in respect of the Transfer and so on from time to time.

12.9. No Security

In the event of a sale of the Business, no interest (including any security interest or lien) in this Agreement, in the Lease of the Premises, in any assets of the Business or in any shares in the capital of the Franchisee may be retained or reserved by the Franchisee or by any Guarantor as security for the payment of any obligations that may arise by reason of any Transfer pursuant to the provisions of this Section.

13. GENERAL CONTRACT PROVISIONS

13.1. Security

As security for all the Franchisee's and its Affiliates' monetary and other obligations to the Franchisor or its Affiliates arising under or relating to this Agreement or any other Contract, the Franchisee grants to the Franchisor a first-priority security interest in all the Franchisee's assets, including all personal and fixture property of every kind and nature wherever located, whether now owned or hereafter acquired or arising, and all proceeds and products thereof, including all goods (including inventory, equipment, and accessions thereto), instruments (including promissory notes), documents, accounts, chattel paper, deposit accounts, letter-ofcredit rights, commercial tort claims, securities and all other investment property, supporting obligations, any other contract rights or rights to the payment of money, insurance claims and proceeds, and all general intangibles (including all payment intangibles). The Franchisee shall execute such financing statements, continuation statements, notices of lien, assignments or other documents as may be required in order to perfect and maintain the Franchisor's security interest, including a UCC-1 Financing Statement in substantially the same form attached as Schedule G. The Franchisee authorizes the Franchisor to execute, on behalf of either or both of the parties, any of the documents described in the prior sentence. The Franchisee shall pay all filing fees and costs for perfecting the Franchisor's security interest. The provisions of this paragraph constitute a security agreement under the Uniform Commercial Code; the parties need not enter any further agreement for the Franchisor's security interest to attach to the collateral described above. Notwithstanding the foregoing sentence, if the Franchisor requests the Franchisee to enter its then standard-form security agreement to more fully document the terms of the security interest granted above, the Franchisee shall immediately do so.

13.2. Indemnity

The Franchisee and the Guarantor agree, during and after the term of this Agreement, to, jointly and severally, indemnify and save the Franchisor and its Affiliates harmless against all liability, losses, damages, costs and expenses (including legal fees whether relating to pre-trial, trial, post-trial, appellate, or bankruptcy proceedings) that may be incurred, suffered or sustained by the Franchisor or any of its Affiliates or any of their respective officers, directors, shareholders or owners, employees, representatives, or agents, as a result of the Franchisee's ownership or operation of the Business, any violation of this Agreement by the Franchisee or the Guarantor, or as a result of any act or omission of the Franchisee or the Guarantor or any of its or their Affiliates or their respective officers, directors, shareholders or owners, employees, representatives, or agents or other persons for whom they are at law or pursuant to this Agreement responsible. The within indemnity obligation shall survive any act of bankruptcy or insolvency of any party and shall be enforceable at the instance of the Franchisor against any party.

13.3. Independent Contractor

The Franchisee acknowledges that it is not an agent or employee of the Franchisor but is an independent contractor completely separate from the Franchisor and any of Franchisor's Affiliates, and that it has no authority to bind or attempt to bind the Franchisor or any of Franchisor's Affiliates in any manner or form whatsoever or to assume or to incur any obligation or responsibility, express or implied, on behalf of or in the name of the Franchisor. This Agreement shall not be construed so as to constitute the Franchisee a partner, joint venturer, agent, employee or representative of the Franchisor for any purpose whatever. The Franchisee shall use its own name in obtaining or executing contracts and making purchases, so that the transaction indicates that the Franchisee is acting on its own behalf and not for the Franchisor. The Franchisee shall prominently display on the Premises, and on any other item the Franchisor

may from time to time require (including menus and stationary), a notice indicating that it is an independent business operating as a franchisee of the Manchu WOK Restaurant System.

13.4. Interest

Except a provided in Section 8.3 hereof, all monies due and owing to the Franchisor or its Affiliates from time to time by the Franchisee pursuant to this Agreement shall bear interest after the due date until paid in full at a rate equal to the prime rate of interest charged by the Franchisor's banker to its best commercial customers ("Bank Prime") plus five (5%) percent (but in no event higher than the maximum interest rate permitted by law to be charged to Franchisee). The acceptance of any principal or interest payment shall not be construed as a waiver by the Franchisor of its rights in respect of the default giving rise to such payment and shall be without prejudice to the Franchisor's right to terminate this agreement in respect of such default. The failure of the Franchisor to charge interest on any amount due at the time such payment shall first become due, shall not constitute a waiver of its right to charge interest at the rate set out in this Agreement on any or all outstanding amounts, from the date such amounts first become due and payable to the Franchisor pursuant to this Agreement.

13.5. Dishonored Payment

Should any payment be made to the Franchisor by any check or other financial instrument including electronic funds transfer and such payment be dishonored for any reason, the Franchisee shall pay to the Franchisor, together with the replacement of such dishonored payment, a fee ("Dishonored Check Fee") in the amount of One Hundred Dollars (\$100) or the amount charged to the Franchisor by its banker on account of such dishonored check or financial instrument, whichever shall be the greater amount.

13.6. Taxes

The Franchisee agrees to promptly pay when due all taxes levied or assessed by reason of its performance hereunder or by reason of the operation or ownership of the Franchised Business, including any sales or use tax on Equipment and Furnishings purchased or leased, and including, without limitation, its municipal taxes and/or business taxes and any taxes imposed on Franchisor based upon any payments due from Franchisee to the Franchisor whether levied directly or indirectly. If there is a bona fide dispute as to the liability for any tax assessed against the Franchisee, the Franchisee shall have the right to contest the validity of the assessment or the amount of tax owing in accordance with the procedures of the relevant taxing authority, provided there is no cost or expense to the Franchisor or Landlord as a result of such contesting and provided that the Franchisee shall first have paid to the Franchisor the full amount of any sums in dispute, which monies shall be disbursed by the Franchisor when the dispute is resolved in accordance with the decision of the taxing authority or agreement of all relevant parties. Notwithstanding the foregoing, the Franchisee shall in no event permit a tax sale procedure by levy of execution or otherwise to occur against the Premises, the Business or any of the assets thereof.

13.7. Right of Set-Off

Notwithstanding any other provision of this Agreement, upon the failure of the Franchisee to pay to the Franchisor or any of its Affiliates or any approved supplier as and when due any amount owing to them, the Franchisor may, at its election, deduct such unpaid amount

from any monies or credit held by the Franchisor or any of its Affiliates for the account of the Franchisee and pay such amounts to the payee thereof.

13.8. Survival

All covenants, agreements, representations and warranties made in this Agreement by the Franchisee and the Guarantor (including those set forth in Section 13.2 and Sections 10, 11, 14, and 15) shall continue in full force and effect subsequent to and notwithstanding the Agreement's expiration or sooner termination and until they are satisfied or by their nature expire.

13.9. Waiver of Obligations

The Franchisor's failure or delay at any time to require performance by the Franchisee or the Guarantor of any provision of this Agreement, even if known, or of any other franchisee or guarantor, or under any custom or practice of the parties at variance with the terms of this Agreement shall not affect the Franchisor's right to require performance of that provision or to exercise any right, power or remedy hereunder, including without limitation, the Guarantor's personal guarantee of the Franchisee's obligations to the Franchisor. The Franchisor's waiver of any breach of any provision of this Agreement should not be construed as a waiver of any continuing or succeeding breach of such provision, a waiver of the provision itself, or a waiver of any right, power or remedy under this Agreement. No notice to or demand on any party in any case shall, of itself, entitle such party to any other or further notice or demand in similar or other circumstances. The Franchisor, however, may by written instrument unilaterally waive any obligation of or restriction upon the Franchisee under this Agreement.

13.10. No Withholding

The Franchisee and the Guarantor agree that they will not, on grounds of the alleged non-performance by the Franchisor of any of its obligations hereunder or for any other reason whatsoever, withhold payment of any administration fee or any other fee or amounts due hereunder (including without limitation Royalty or Marketing fee or contribution) or otherwise to the Franchisor or any of its Affiliates.

13.11. Rights Cumulative

The rights of the Franchisor hereunder are cumulative and no exercise or enforcement by the Franchisor of any right or remedy shall preclude the exercise or enforcement by the Franchisor of any other right or remedy, either hereunder or which the Franchisor is otherwise entitled by law to enforce.

13.12. Force Majeure

No party shall be responsible to another for non-performance or delay in performance occasioned by any causes beyond its reasonable control, including, without limitation, acts or omissions of the other party, acts of civil or military authority, labor disputes, strikes, lockouts, unusual delay by common carrier, unavailability of supplies, embargoes, insurrections or acts of God, provided that inability of a party to obtain funds shall be deemed to be a cause within the control of such party. If any such delay occurs, any applicable time period shall be automatically extended for a period equal to the time lost, as long as the party affected makes

reasonable efforts to correct the reason for such delay and gives to the other party prompt notice of any such delay. Notwithstanding the foregoing, nothing shall excuse late payments, or extend the due date, of amounts owed to the Franchisor.

13.13. Notices

All notices, requests, demands, consents and other communications required or permitted under this Agreement to any party shall be effective only when delivered in writing (including facsimile, telex and telegraphic communication) and must be (as elected by the person giving such notice) hand delivered by messenger or courier service, telecommunicated, faxed or mailed (airmail if other than local) by registered or certified mail (postage prepaid), return receipt requested, addressed to the appropriate party or parties at their respective addresses set forth above (and, if to the Franchisor, must be addressed to its President and C.E.O.) or to such other address as each such party may designate by notice complying with the terms of this Section. Each such notice is deemed delivered:

- (a) on the date delivered if by personal delivery;
- (b) on the date of transmission, with confirmed answer back if by fax or other telegraphic method; or
- (c) on the date upon which the return receipt is signed or delivery is refused or the notice is designated by the postal authorities or courier service as not deliverable, as the case may be, if mailed or couriered.

The Confidential Manual, any revisions to the Confidential Manual, and/or any written instructions that Franchisor furnishes to Franchisee relating to operational matters shall not be deemed as "Notices" for the purposes of the delivery requirements of this Section 13.13.

- 13.14 Interpretation of Rights and Obligations. The following provisions apply to and govern the interpretation of this Agreement, the parties' rights under this Agreement, and the relationship between the parties.
 - 13.4.1 Franchisor's Rights. Whenever this Agreement provides that Franchisor has a certain right, that right is absolute and the parties intend that Franchisor's exercise of that right will not be subject to any limitation or review. Franchisor has the right to operate, administrate, develop, and change the System in any manner that is not specifically precluded by the provisions of this Agreement, although this right does not modify any express limitations set forth in this Agreement.
 - 13.4.2 Franchisor's Reasonable Business Judgment. Whenever Franchisor reserves discretion in a particular area or where Franchisor agrees to exercise its rights reasonably or in good faith, Franchisor will satisfy its obligations whenever it exercises reasonable business judgment ("**Reasonable Business Judgment**") in making a decision or exercising a right. Franchisor's decisions or actions will be deemed to be the result of Reasonable Business Judgment, even if other reasonable or even arguably preferable alternatives are available, if Franchisor's decisions or actions are intended, in whole or significant part, to promote or benefit the System generally even if the decision or action also promotes Franchisor's financial or other individual interests. Examples of items that will promote or benefit the System include, without limitation, enhancing the

value of the Marks, improving customer service and satisfaction, improving product quality, improving uniformity, enhancing or encouraging modernization and improving the competitive position of the System.

14. DISPUTE RESOLUTION

14.1. Arbitration

- (a) Except as otherwise provided herein, any dispute, claim or controversy arising out of or relating to this Agreement, the breach hereof, the rights and obligations of the parties hereto or the relationship between the parties, or the entry, making, interpretation, or performance of either party under this Agreement ("Dispute"), which cannot be resolved by mediation under Section 14.1d. or is not subject to mediation under the terms of this Agreement, shall be settled by binding arbitration administered by the American Arbitration Association ("AAA") before a single arbitrator in accordance with its Commercial Arbitration Rules as modified below, and pursuant to the Federal Arbitration Act, 9 U.S.C. §1 et seq.
- Except to the extent that the United States Trademark Act of 1946, as amended (15 U.S.C. § 1051 et seq.), the laws of the State where the Franchised Business is located govern all rights and obligations of the parties under this Agreement without regard to conflict of law, and any arbitration shall take place before a sole arbitrator in the County of the State where the Franchised Business is located. The parties agree that the arbitrator shall be an attorney licensed to practice law in the United States and must have a minimum of five (5) years of experience in franchise law. Judgment on the award rendered by the arbitrator may be entered in any court of competent jurisdiction. The arbitrator shall, in the award, allocate all of the costs and expenses of the arbitration, including the fees of the arbitrator and the reasonable attorneys' fees of the prevailing party, against the party who did not prevail. This agreement to arbitrate shall survive termination or expiration of this Agreement. The parties agree that any arbitration shall be solely between them (including any affiliates) and shall not include as a party, by consolidation, joinder, or in any other manner, any other person or entity, unless both parties to this Agreement consent in writing. Both parties shall have the absolute right to refuse such consent. The parties expressly waive any right to bring or participate in any class or other consolidated, joined or multi-party arbitration claim or proceeding, whether or not permissible under the AAA Commercial Arbitration Rules, including any claim brought on their behalf by an association of which it, he or she is a member. At the request of either party, the arbitration shall be conducted in a manner that maintains the confidentiality of the proceedings.
- (c) The arbitrator will issue a reasoned award, with findings of fact and conclusions of law. Actions to enforce an express obligation to pay monies may be brought under the Expedited Procedures of the AAA's Commercial Arbitration Rules. Discovery shall be conducted according to the AAA Commercial Arbitration Rules.
- (d) Prior to the commencement of an arbitration proceeding, a party must notify the other party of its intent to commence arbitration by providing written notice to the other party that describes the facts and subject matter of the Dispute at least sixty (60) days prior to commencing arbitration with the AAA. During the sixty (60) days after a written notice of intent to arbitrate is received by either party, the parties must first

mediate the Dispute with a mutually agreed upon private mediator. At the request of any party, the mediation will be confidential. The mediation shall be conducted in the County of the State where the Franchised Business is located, unless the parties mutually agree to a different location. The parties to the mediation will share equally in its costs and expenses, except those costs and expenses incurred separately by each party, including counsel fees and expenses. The mediation process will be deemed "Completed" when the parties agree that it has been completed, the mediator declares that any impasse exists, or sixty (60) days have elapsed since the date of the initiating party's notice to the other party that it is initiating the mediation process, whichever occurs first.

- Notwithstanding anything contained in this Agreement to the contrary, the provisions of sections 14.1a, 14.1b., 14.1c. and 14.1d. do not apply to a Dispute where: (i) we bring an action for an express obligation to pay monies, declaratory relief, preliminary or permanent equitable relief, any action for damage to our goodwill, the Confidential Information, the Proprietary Marks or for fraudulent conduct by you; or (ii) the delay resulting from the dispute resolution process set forth in Section 14.1 may endanger or adversely affect the public (for example, unhealthy, unsafe or unsanitary conditions would continue to exist), or may cause irreparable harm to Franchisor or its affiliates. For such Disputes, we may bring an action in any federal or state court having jurisdiction, whether for monetary damages, temporary and permanent injunctive relief or specific performance in addition to, and not exclusive of, any other remedies available to us. You hereby consent to and waive any objection or defense and agree not to contest venue, forum non conveniens or jurisdiction of such court. THE PARTIES AGREE THAT ALL DISPUTES SUBMITTED TO THE COURT PURSUANT TO THIS SECTION SHALL BE TRIED TO THE COURT SITTING WITHOUT A JURY, NOTWITHSTANDING ANY STATE OR FEDERAL CONSTITUTIONAL OR STATUTORY RIGHTS OR PROVISIONS.
- (f) Subject to this Section 14.1, the right and duty of the parties to this Agreement to resolve any disputes by arbitration shall be governed exclusively by the *Federal Arbitration Act*, 9 U.S.C. § 1 *et seq.*, as amended, and arbitration shall take place according to the Commercial Arbitration Rules of the American Arbitration Association in effect as of the date the demand for arbitration is filed.
- (g) To the extent Law permits, any Dispute arising out of or relating to this Agreement, whether in arbitration or in court, will be conducted only on an individual basis and not in a class, consolidated or representative action. Notwithstanding any other provision of this Agreement or the AAA Rules, disputes regarding the interpretation, applicability, or enforceability of this class waiver may be resolved only by a court and not by an arbitrator. If this waiver of class or consolidated actions is deemed invalid or unenforceable, neither party is entitled to arbitration.

14.2. Reserved.

14.3. Remedies

The court or arbitrator will have the right to award any relief which it deems proper in the circumstances, including, without limitation, money damages (with interest on unpaid amounts

from the date due), lost profits, specific performance, injunctive relief, and attorney's fees and costs. The parties agree that any claim for lost earnings or profits by Franchisee shall be limited to a maximum amount equal to the net profits of the Business for the prior year as shown on Franchisee's federal income tax return. The parties further agree that, in addition to such other damages awarded by the court, if this Agreement is terminated because of Franchisee's default, Franchisee shall be liable to Franchisor for a lump sum amount equal to the net present value of the royalties and Fund contributions that would have become due following termination of this Agreement for the period this Agreement would have remained in effect but for Franchisee's default. Royalties and Marketing Fund Contributions for purposes of this Section shall be calculated based on the Business' average monthly Gross Sales for the twelve (12) months preceding the termination date.

14.4. Legal Fees

If Franchisee is in default under this Agreement, then Franchisee shall pay Franchisor all damages, costs, and expenses (including legal fees) that Franchisor incurs as a result of Franchisee's default under this Agreement or any other agreement between Franchisee (and its affiliates) and Franchisor (and our affiliates).

14.5. Reduced Time Within Which to Commence Actions

ANY AND ALL CLAIMS AND ACTIONS ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE RELATIONSHIP OF FRANCHISOR AND FRANCHISEE, OR FRANCHISEE'S OPERATION OF THE RESTAURANT, BROUGHT BY ANY PARTY HERETO AGAINST ANY OTHER SHALL BE COMMENCED WITHIN TWO (2) YEARS FROM THE OCCURRENCE OF THE FACTS GIVING RISE TO SUCH CLAIM OR ACTION, OR SUCH CLAIM OR ACTION SHALL BE FOREVER BARRED.

15. GENERAL PROVISIONS

15.1. Time

All date and time measurements under this Agreement shall refer to the date and time at Franchisor's headquarters in Markham, Ontario.

15.2. Dollars

All payments required under this Agreement shall be made in U.S. Dollars.

15.3. Independent Investigation

The Franchisee and the Guarantor acknowledge that they have conducted an independent investigation of the Business franchised hereunder, recognize that the business venture contemplated by this agreement involves business risks and also recognize that its success will be largely dependent upon the ability of the Franchisee and Guarantor as independent business persons. The Franchisee and the Guarantor acknowledge that the Franchisor has not made, save as may have been set out in the appropriate Disclosure Document if any, any representation as to current or future sales, income, costs, or profits or to Franchisee's proposed business. The Franchisee and the Guarantor acknowledge that they have not received from the Franchisor or any other person on behalf of the Franchisor any

representation, warranty or guarantee, express or implied, with respect to the above described matters or as to success of the business venture contemplated by this Agreement.

15.4. Independent Advice

The Franchisee and the Guarantor acknowledge that they have received, have had an ample time to read and have read this Agreement and fully understand its provisions. The Franchisee and the Guarantor further acknowledge that they have had an adequate opportunity to be advised by advisors of their own choosing (including without limitation their own lawyers, accountants, and bankers) regarding all pertinent aspects of this franchise and the franchise relationship created hereby and that they have done so with the full understanding and knowledge of all of the terms hereof.

16. REPRESENTATIONS AND WARRANTIES

16.1. No Reliance

Except as expressly provided to the contrary in this Agreement, the Franchisor makes no representations, warranties, or guarantees upon which the Franchisee may rely. Nor does the Franchisor assume any liability or obligation to the Franchisee by providing any waiver, approval, consent, or suggestion to the Franchisee in connection with this Agreement; or by reason of any neglect, delay, or denial of any request therefore unless such conduct would otherwise constitute a breach of an express obligation of the Franchisor under this Agreement. However, nothing in this Agreement or in any related agreement is intended, however, to disclaim the representations we made in the franchise disclosure document that we furnished to you.

THE FRANCHISOR MAKES NO WARRANTY AS TO THE FRANCHISEE'S ABILITY TO OPERATE THE FRANCHISED BUSINESS IN THE JURISDICTION IN WHICH THE FRANCHISED BUSINESS IS TO BE OPERATED. THE FRANCHISEE ITSELF MUST SEEK OR OBTAIN ADVICE OF COUNSEL SPECIFICALLY WITH RESPECT TO THIS ISSUE. IF LAWS OR REGULATIONS PROMULGATED, BY ANY GOVERNMENTAL BODY PREVENTS THE FRANCHISEE FROM OPERATING THE FRANCHISED BUSINESS, THE FRANCHISOR SHALL NOT BE LIABLE FOR DAMAGES NOR REQUIRED TO INDEMNIFY THE FRANCHISEE IN ANY MANNER WHATSOEVER NOR TO RETURN ANY MONIES RECEIVED FROM THE FRANCHISEE.

16.2. The Franchisee's Representations

The Franchisee and Guarantor(s) represent and warrant to Franchisor, and agree, that:

- (a) The Franchisee is not obtaining this Franchise for speculative or investment purposes. The Franchisee has no present intention to sell or transfer or attempt to sell or transfer this Agreement or the Business in whole or in part.
- (b) Copies of all documents the Franchisee, any of its owners, and the Guarantor must furnish to the Franchisor in connection with obtaining the Franchise, and as required in the future, have been and will be true and complete copies of such documents (including all amendments or modifications thereof) and contain no misleading or incorrect statements or material omissions.

- (c) The Franchisee has received from the Franchisor a Franchise Disclosure Document ("FDD") for the state where the Business will be located and, if different, the state where the Franchisee resides (with all exhibits and supplements thereto), at least 14 calendar days prior to:
 - (i) executing this Agreement and every other agreement imposing a binding obligation on the Franchisee in connection with the sale of the Franchise; and
 - (ii) the Franchisee's paying any consideration in connection with the sale, or proposed sale, of the Franchise.
- (d) No material changes to the Franchise Agreement or any other agreement that Franchisee is signing with the Franchisor or any affiliate of the Franchisor has been made within the seven calendar days immediately prior to the date that Franchisee executes the Franchise Agreement.
- (e) Franchisor expressly disclaims the making of any guarantee, express or implied, and Franchisee acknowledges that it has not received or relied upon any representations of revenue, as to potential volume profits or success of the business venture contemplated by this Agreement, including any statement regarding the profits or existing franchised or company-owned Manchu WOK Restaurant.

16.3. Acknowledgement of Risk

The Franchisee acknowledges and agrees to the following:

- (a) THE FRANCHISEE'S SUCCESS IN OWNING AND OPERATING THE FRANCHISED BUSINESS IS SPECULATIVE AND WILL DEPEND ON MANY FACTORS. SUCH FACTORS INCLUDE, TO A LARGE EXTENT, THE FRANCHISEE'S INDEPENDENT BUSINESS ABILITY. EXCEPT AS SPECIFICALLY INCLUDED IN THIS AGREEMENT, NO REPRESENTATIONS OR PROMISES, EXPRESS OR IMPLIED, HAVE BEEN MADE BY THE FRANCHISOR OR ANY EMPLOYEE, BROKER, OR REPRESENTATIVE OF THE FRANCHISOR, TO INDUCE THE FRANCHISEE TO ENTER INTO THIS AGREEMENT. NO EMPLOYEE, OFFICER, DIRECTOR, BROKER OR REPRESENTATIVE IS AUTHORIZED TO DO OTHERWISE.
- (b) ALL OF THE FRANCHISEE'S DEALINGS WITH THE FRANCHISOR'S OWNERS. OFFICERS, DIRECTORS, EMPLOYEES. AND REPRESENTATIVES, SUCH INDIVIDUALS ACT ONLY IN THEIR REPRESENTATIVE CAPACITY AND NOT IN AN INDIVIDUAL CAPACITY. THE FRANCHISEE FURTHER ACKNOWLEDGES THAT THIS AGREEMENT AND ALL BUSINESS DEALINGS BETWEEN THE FRANCHISEE AND SUCH INDIVIDUALS AS A RESULT OF THIS AGREEMENT ARE SOLELY BETWEEN THE FRANCHISEE AND THE FRANCHISOR. NOTWITHSTANDING THE FOREGOING, IF FRANCHISOR ENGAGES ANY BROKER, SUCH BROKER SHALL BE SOLELY LIABLE FOR ITS CONDUCT IN CONNECTION WITH THE FRANCHISEE EXCEPT THAT FRANCHISOR SHALL REMAIN LIABLE FOR THE BROKER'S CONDUCT SOLELY TO THE EXTENT OF FRANCHISOR'S OWN CRIMINAL, INTENTIONAL OR GROSSLY NEGLIGENT CONDUCT REGARDING ENGAGING SUCH BROKER.

16.4. Business Opportunity Disclaimers

The Franchisor makes no representation, warranty, or guaranty—express or implied—concerning any of the following matters:

- (a) that it will purchase any products made, produced, fabricated, or modified by the Franchisee using supplies or services sold to the Franchisee; or
- (b) that the Franchisee will derive income from the Franchise that exceeds the price paid for it; or that the Franchisor will refund all or part of the purchase price for the Franchise, or repurchase any of the products, equipment, or supplies provided by the Franchisor, if the Franchisee is unsatisfied with the Franchise.

17. GENERAL PROVISIONS

17.1. Release of Prior Claims

By executing this Agreement, the Franchisee (and all of its owners, including the Guarantor) and each of its/his/her successors under this Agreement, forever releases and discharges the Franchisor and its affiliates, its designees, franchise sales brokers, if any, and other agents, and their respective officers, directors, representatives, employees and agents, from any and all claims of any kind, in law or in equity, that may exist as of the date of this Agreement relating to this Agreement or any other agreement between the parties, or relating in any other way to the conduct of the Franchisor, its affiliates, its designees, franchise sales brokers, if any, or other agents, and their respective officers, directors, representatives, employees and agents before the date of this Agreement, including any and all claims, whether presently known or unknown, suspected or unsuspected, arising under the franchise, business opportunity, securities, antitrust or other laws of the United States, any state or locality.

17.2. Amendments

The provisions of this Agreement may be amended, supplemented, waived or changed only by a written document signed by the party as to whom enforcement of any such amendment, supplement, waiver, or modification is sought and making specific reference to this Agreement.

17.3. Binding Effect

All of the terms and provisions of this Agreement, whether so expressed or not, are binding upon, inure to the benefit of, and are enforceable by the parties and their respective personal representatives, legal representatives, heirs, successors, and permitted assigns.

17.4. Severability

(a) If any provision of this Agreement, or any other agreement entered into under or in connection with this Agreement, is contrary to, prohibited by, or deemed invalid under applicable law or regulation, such provision is inapplicable and deemed omitted to the extent so contrary, prohibited, or invalid, but the remainder of this Agreement is not invalidated thereby and must be given full force and effect so far as possible. If any provision of this Agreement may be construed in two or more ways, any

of which would render the provision invalid or otherwise voidable or unenforceable and another of which would render the provision valid and enforceable, such provision has the meaning that renders it valid and enforceable.

(b) If any applicable law of any jurisdiction requires greater prior notice of termination or non-renewal of this Agreement than is required hereunder or the taking of some action not required under this Agreement, the greater prior notice and/or other action required by such law must be substituted for its counterpart under this Agreement. If, under any applicable law of any appropriate jurisdiction, any provision of this Agreement or any requirement prescribed by the Franchisor is invalid or unenforceable, the Franchisor may, in its sole discretion, modify such invalid or unenforceable requirement to the extent required to be valid and enforceable. Unless the Franchisor elects to give them greater applicability, the foregoing substitutions and modifications to this Agreement are effective only in the jurisdiction in which they are required. In all other jurisdictions, this Agreement must be enforced as originally made and entered.

17.5. Effectiveness - Counterparts

This Agreement is not effective or binding or enforceable against the Franchisor until it is accepted by the Franchisor at its home office in Markham, Ontario, Canada. This Agreement may be executed in one or more counterparts, each of which is an original, but all of which together constitute one and the same instrument. Confirmation of execution by telecopy or fax transmission (or other mutually-acceptable electronic exchange of signature pages) of a facsimile signature page is binding upon any party to such confirmation.

17.6. Interpretation

Each of the parties has been or has had the opportunity to have been represented by their own counsel throughout the negotiations, as well as at the execution of this Agreement and all the other documents executed incidental to this Agreement. Therefore, either while this Agreement is effective or after its expiration or sooner termination, none of the parties may claim or assert that any provision of this Agreement or of the other documents should be construed against the drafter thereof.

17.7. Entire Agreement

This Agreement (together with its Exhibits, and all other written agreements related to this Agreement that are expressly referenced herein) represents the entire understanding and agreement between the parties with respect to the subject matter of this Agreement, and supersedes all other negotiations, understandings, and representations, if any, made by and between the parties. No representation, inducement, promise, or agreement, oral or otherwise not embodied in this Agreement, its Exhibits, or other written agreement related to this Agreement and expressly referenced herein shall have any force or effect. However, nothing in this Agreement or in any related agreement is intended, however, to disclaim the representations we made in the franchise disclosure document that we furnished to you.

17.8. Liability of Multiple Franchisees

If the Franchisee consists of more than one person, all such persons are jointly and severally liable for the Franchisee's liabilities and obligations under this Agreement.

17.9. Third Parties

Except as expressly provided in this Agreement to the contrary with respect to any affiliates of the Franchisor, nothing in this Agreement whether express or implied, confers any rights or remedies under or by reason of this Agreement on any persons (including other Manchu WOK franchisees) other than the parties and their respective personal or legal representatives, heirs, successors, and permitted assigns.

17.10. Guarantee

All the Franchisee's legal and beneficial owners must sign the guarantee in the form attached as Schedule H to this Agreement. By doing so, each of such owners agrees that he or she is bound by all the terms of this Agreement as if he or she were the Franchisee thereunder and agrees that he or she is jointly and severally liable with the other owners and the Franchisee for all the Franchisee's obligations under this Agreement.

17.11. Designated Franchisee Spokesperson

Franchisee and Franchisor recognize that more than one individual may have a legal or equitable ownership interest in Franchisee ("Owners"). For this reason, and in order to (i) streamline communications between Franchisor and Franchisee; and (ii) protect and insulate Franchisor from potential claims from or liability to any Owner that may arise as a result of actions or inactions taken by Franchisor after having received conflicting advice and/or instructions from Owners, the parties appoint the individual whose name appears on the signature page of this Agreement as the Designated Spokesperson pursuant to this Section 17.11. The Designated Spokesperson shall have full authority to speak on behalf of, as well as bind and commit, Franchisee with respect to all rights, obligations and performance pursuant to this Agreement. The Designated Spokesperson shall not be changed without the prior written consent of both Franchisor and Franchisee.

17.12. Franchisor's Business Judgment

Whenever Franchisor exercises a right and/or discretion to take or withhold an action, or to grant or decline to grant Franchisee a right to take or withhold an action, except as otherwise expressly and specifically provided in this Agreement, Franchisor may make such decision or exercise its right and/or discretion on the basis of its judgment of what is in its best interests. "Best interests" includes what Franchisor believes to be the best interests of the System at the time the decision is made or the right or discretion is exercised even though (a) there may have been other alternative decisions or actions that could have been taken; (b) Franchisor's decision or action taken promotes Franchisor's financial or other individual interest; or (c) Franchisor's decision or the action it takes may apply differently to different franchisees or any companyowned or affiliate-owned Restaurants. In the absence of an applicable statute, Franchisor will have no liability to Franchisee for any such decision or action. The exercise of the right or discretion will not be subject to limitation or review. If applicable law implies a covenant of good faith and fair dealing in this Agreement, Franchisor and Franchisee agree that such covenant shall not imply any rights or obligations that are inconsistent with a fair construction of the terms of this Agreement and that this Agreement grants Franchisor the right to make decisions or take/refrain from taking actions not inconsistent with Franchisee's rights and obligations hereunder.

17.13. Governing Law

This Agreement shall be governed by and be construed in accordance with the laws of the State of Arizona.

--- Signatures on following page --

IN WITNESS WHEREOF this Agreement has been executed on the date first above written.

FRANCHISOR: MTY FRANCHISING USA, INC. By: [Name, Title] I have authority to bind the Corporation **FRANCHISEE:** [FRANCHISEE] By: [Name, Title] I have authority to bind the legal entity By: [Name, Title] I have authority to bind the legal entity **GUARANTOR:** By: [Name], an individual **GUARANTOR:** [Name], an individual **GUARANTOR:**

[Name], an individual

Schedule "A"

Trademarks

Manchu WOK
Manchu WOK Fast & Fresh Chinese Cuisine

Schedule "B"

Premises and Lease Description

The "Manchu WOK Restaurant" shall only be the following street address:					
The Lease commencement date is					
The Lease Expiration Date is					
The Designated Manager is					

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

Schedule "C"

Trade Area (if applicable)

Schedule "D"

Lease Addendum (Section 4.2(a) (i))

LEASE ADDENDUM TO LEASE AGREEMENT

("Lease Addendum")

Dated	l:, 20
	between
	and
(Landlord Name)	(Tenant Name)
(A	ddress of "Premises")

1. Use of Premises.

During the term of the Lease, the Premises may be used only for the operation of a quick service restaurant under the Manchu WOK, trademarks, trade names, and logos, which specialize in the sale of Chinese and Asian-style food and drinks and any other items sold under the Manchu WOK system. Landlord consents to Tenant's use of such trademarks, trade names, logos, tag lines, signs, décor items, color schemes, and related components of the Manchu WOK franchise system as MTY Franchising USA, Inc., franchisor of the Manchu WOK brand ("Franchisor"), or any of its affiliates, may prescribe for Manchu WOK franchisees and which may be altered or changed by Franchisor from time to time.

2. Assignment and Notices.

a. Notwithstanding anything to the contrary in the Lease, Tenant shall have the right to assign the Lease and all amendments thereto, as applicable, including this Lease Addendum (collectively, "Lease"), and all rights hereunder to Franchisor or any of its affiliates, or to a third-party Manchu WOK franchisee approved by Franchisor upon the expiration or termination of the current franchise agreement by and between Franchisor and Tenant ("Franchise Agreement"), without obtaining Landlord's consent and without the imposition of any assignment fee or similar charge. Landlord shall not accelerate the rent owed hereunder in connection with such assignment(s), so long as Franchisor, its affiliate(s) or its third-party franchisee(s) assumes in writing the obligations of Tenant under the Lease. Nothing in this Section 2.a shall serve to extend the term of the Lease or provide Franchisor any occupancy rights, options to renew or other rights not expressly set forth to Tenant in the Lease.

- b. Landlord agrees to furnish Franchisor with copies of any and all letters and notices to Tenant pertaining to the Lease and the Premises at the same time that such letters and notices are sent to Tenant. Landlord further agrees that, if it intends to terminate the Lease, the Landlord will give Franchisor the same advance written notice of such intent as provided to Tenant, specifying in such notice all defaults that are the cause of the proposed termination. Franchisor shall have the right to cure, at its sole option, any such default within the time periods granted to Tenant under the Lease. If neither Tenant nor Franchisor cures all such defaults within said time periods (or such longer cure periods as may be specifically permitted by the Lease), then the Landlord may terminate the Lease, re-enter the Premises and/or exercise all other rights as set forth in the Lease.
- c. Prior to the expiration or termination of the Lease, Franchisor shall have the right to enter the Premises to make any reasonable modifications or reasonable alterations necessary to protect Franchisor's interest in the Manchu WOK brand and its trademarks, trade names, logos, tag lines, signs, décor items, color schemes, and related components of the Manchu WOK franchise system, or to cure any default under the Lease, and Landlord and Tenant agree that Franchisor shall not be liable for trespass or any other crimes or tort.

3. Notices.

All notices and demands required to be given hereunder shall be in writing and shall be sent by personal delivery, expedited delivery service, certified or registered mail, return receipt requested, first-class postage prepaid, email or facsimile (provided that the sender confirm the facsimile by sending an original confirmation copy by certified transmission), to the respective parties at the following addresses unless and until a different address has been designated by written notice to the other parties.

If directed to Tenant, the noti	ice shall be addressed to:
Attn:	
Facsimile:	
Email:	
If directed to <u>Landlord</u> , the ne	
Attn:	
Facsimile:	
Email:	<u> </u>
If directed to <u>Franchisor</u> , the	notice shall be addressed to
MTY Franchising USA, Inc.	
9311 E. Via De Ventura	

Scottsdale, AZ 85258

Attn: Real Estate Department Facsimile: (480) 362-4792

Email:

leases@kahalamgmt.com

Any notices sent by personal delivery shall be deemed given upon receipt. Any notices given by email or facsimile shall be deemed given on the business day of transmission, provided confirmation is made as provided above. Any notice sent by expedited delivery service or registered or certified mail shall be deemed given three (3) business days after the time of mailing. Any change in the foregoing addresses shall be effected by giving fifteen (15) days written notice of such change to the other parties.

4. Amendments.

Landlord and Tenant will not amend, renew, extend or otherwise modify the Lease in any manner which would materially affect any of the provisions in this Lease Addendum without Franchisor's prior written consent.

5. Right of First Refusal.

Landlord and Tenant agree that should Landlord and Tenant desire to terminate the Lease prior to the scheduled expiration date, Landlord will notify Franchisor of the proposed termination not less than thirty (30) days in advance of the proposed termination date. Franchisor shall have a right of first refusal to assume the Lease whereby Tenant will assign the Lease to Franchisor and Franchisor will assume all rights and obligations of Tenant under the Lease.

6. Miscellaneous.

The terms and conditions of this Lease Addendum will supersede any conflicting terms of the Lease. Any capitalized term not specifically defined in this Lease Addendum shall have the meaning ascribed to such term in the Lease.

IN WITNESS WHEREOF, the parties hereto, by and through their respective representatives authorized to enter into and bind each respective party without further consent or authorization, have duly executed and delivered this Lease Addendum in duplicate as of the date set forth above.

LANDLORD:	, a(n)
	By:
	Name:
	Title:

TENANT:	, a(n) _	
	By:	
	Name:	
	Title:	
	By:	
	Name:	
	Title:	

Schedule "E"

Development Obligations (Section 4.3)

Schedule "F"

List of Persons With a Legal and/or Beneficial Ownership Interest in Franchisee (Section 12.3)

Name of Franchisee:		
Persons:		
<u>Name</u>	Ownership Interest Percentage	
1	%	
2	%	
3	%	
4.	%	

Schedule "G"

UCC-1 and Rider (Section 13.1)

Schedule "H"

Guarantee, Indemnification, and Acknowledgment

	As a	an inducement	to MTY	Franc	hising USA,	Inc. ("Franch	n isor") to e	xecute and de	elive
the	MTY	Franchising	USA,	Inc.	Franchise	Agreement	between	Franchisor	and
				("	Franchisee'	') dated		, 20	(th€
"Ag	reemer	nt"), the unde	rsigned,	jointl	y and seve	ally, hereby	uncondition	ally guarante	e to
Fran	chisor	and Franchis	or's su	ccesso	ors and as	signs that al	I of Franc	hisee's mon	etary
oblig	ations	under the Agre	ement v	vill be	punctually pa	aid and perfor	med.		

Upon demand by Franchisor, the undersigned will immediately make each payment required of Franchisee under the Agreement (including but not limited to payments Franchisee is required to make to Franchisor and its affiliates). The undersigned hereby waive any right to require Franchisor to: (a) proceed against Franchisee for any payment required under the Agreement; (b) proceed against or exhaust any security from Franchisee; or (c) pursue or exhaust any remedy, including any legal or equitable relief, against Franchisee. Without affecting the obligations of the undersigned under this Guarantee, Franchisor may, without notice to the undersigned, extend, modify, or release any indebtedness or obligation of Franchisee, or settle, adjust, or compromise any claims against Franchisee. The undersigned waive notice of amendment of the Agreement and notice of demand for payment by Franchisee, and agree to be bound by any and all such amendments and changes to the Agreement.

The undersigned hereby agree to defend, indemnify and hold Franchisor harmless against any and all losses, damages, liabilities, costs, and expenses (including, but not limited to, reasonable attorney's fees, reasonable costs of investigation, court costs, and fees and expenses) resulting from, consisting of, or arising out of or in connection with any failure by Franchisee to perform any obligation of Franchisee under the Agreement, any amendment thereto, or any other agreement execute and delivered by Franchisee referred to therein.

The undersigned hereby acknowledge and agree to be individually bound by all of the promises contained in the following Sections of the Agreement: Section 9 (with respect to trademarks); Section 10 (with respect to obligations after termination of the Agreement); Section 11 (with respect to confidentiality and covenants against competition); and Section 12 (with respect to transfers of interests in Franchisee).

The undersigned acknowledge and agree that: (a) this Guarantee does not grant the undersigned any right to use any of Franchisor's marks (including but not limited to the "Manchu WOK" marks) or the system licensed to Franchisee under the Agreement; (b) that they have read, in full, and understand, all of the provisions of the Agreement that are referred to above in this paragraph, and that they intend to fully comply with those provisions of the Agreement as if they were printed here; and (c) that they have had the opportunity to consult with a lawyer of their own choosing in deciding whether to sign this Guarantee.

This Guarantee shall terminate upon the termination or expiration of the Agreement, except that all obligations and liabilities of the undersigned which arose from events which occurred on or before the effective date of such termination shall remain in full force and effect until satisfied or discharged by the undersigned, and all covenants which by their terms continue in force after the expiration or termination of the Agreement shall remain in force according to their terms. Upon the death of an individual guarantor, the estate of such guarantor shall be

bound by this Guarantee, but only for defaults and obligations hereunder existing at the time of death; and the obligations of the other guarantors will continue in full force and effect.

Unless specifically stated otherwise, the terms used in this Guarantee shall have the same meaning as in the Agreement, and shall be interpreted and construed in accordance with Section 14 of the Agreement (including but not limited to the waiver of punitive damages, waiver of jury trial, agreement to bring claims within two years, and agreement not to engage in class or common actions). This Guarantee shall be interpreted and construed exclusively under the laws of the State of Arizona. In the event of any conflict of law, the laws of the State of Arizona shall prevail (without regard to, and without giving effect to, the application of Arizona conflict of law rules).

IN WITNESS WHEREOF, each of the undersigned has signed this Guarantee as of the date of the Agreement.

(in his/her personal capacity)	(in his/her personal capacity)	(in his/her personal capacity)	
Printed Name:	Printed Name:	Printed Name:	
Date:	Date:	Date:	
Home Address:	Home Address:	Home Address:	



EXHIBIT D

Manchu WOK SUBLEASE AGREEMENT

SUBLEASE

DATED THIS	DAY OF	, 20 FOR A TERM COMMENCING THE
DAY OF	, 20	, 20 FOR A TERM COMMENCING THE
BETWEEN:		
	MTY FRANC	HISING USA, INC., a corporation incorporated under
		e State of Tennessee (hereinafter referred to as the
		Franchisor"), whose principal offices are located at
	9311 East Via	a de Ventura, Scottsdale, AZ 85258.
AND:		
		Company or Franchisee a corporation incorporated
	under the lav	ws of the State of (hereinafter the "Sublessee"), whose principal offices are located
	at	
		
AND:		
AND.		
	Guarantors	
	*both/an indiv	vidual residing in the State of (hereinafter
		referred to as the "Guarantor"), whose principal
	residence(s)	is/are located at
	-	
	_	
PREAMBLE		
ITLAMBLE		
WHEF	REAS by Lease dated	the day of, 20_, a copy of which has
		tached hereto as Schedule "A" (hereinafter referred ereinafter referred to as the " Lessor ") leased to MTY
		r Franchisor) certain premises situated at address of
restaurant (hereinafte	er referred to as the "F	Premises "), for a term of (#) years, the whole as
more fully set out and	d described in the said	Lease;
and W	HEREAS MTY FRAN	CHISING USA, INC. and the Sublessee have entered
	greement of even of	date for a term commencing the day of
		referred to as the "Franchise Agreement"), a copy of
which each party her	eto acknowledges hav	ing received;
and W	/HEREAS the Lessee	and Sublessee wish to execute this Sublease (the
"Sublease"); in respe	ect of the Premises;	
NOW	THEREFORE IN CO	NSIDERATION OF THE RENTS, COVENANTS AND
		TH AND THE SUM OF TEN DOLLARS (\$10.00) PAID

BY EACH OF THE PARTIES HERETO TO THE OTHER, THE RECEIPT AND SUFFICIENCY WHEREOF IS HEREBY ACKNOWLEDGED, THE PARTIES AGREE AS FOLLOWS:

SECTION I - SUBLEASE AND TERM

1.1	The Lessee hereby subleases the Premises to the Sublessee in accordance with
and subje	ect to the terms and conditions of this Sublease, for a term commencing on the
day of _	, 20, and terminating, without notice or demand, on the date o
expiration	n of the Lease, less one (1) day. The Lessee excepts from the term of this Sublease the
last day o	of the term of the Lease, each party acknowledging and agreeing this Sublease is, and
is intende	ed to be, a Sublease of the Premises and that this Sublease shall not operate as ar
assignme	ent of the Lease to the Sublessee.

SECTION II - NET SUBLEASE

2.1 It is the express intent of the parties that this Sublease is to be an absolutely net Lease to the Lessee during the term hereof, free and clear of all payments, charges, taxes and obligations of any nature whatsoever with respect to the Premises, the contents thereof or the use or occupancy thereof, except as may be expressly set forth in this Sublease.

SECTION III - MINIMUM AND ADDITIONAL RENT

- 3.1 The Sublessee covenants to pay minimum or basic rent in accordance with the minimum or basic rent provisions of the Lease.
- 3.2 The Sublessee covenants to pay all other amounts required to be paid by the Lessee, whether to the Lessor or otherwise, pursuant to the Lease including, if applicable but without limitation, percentage rent, operating expenses, merchant association dues, advertising and promotion fund payments, public utility charges, business taxes in respect of the business carried on in the Premises or the improvements, equipment and facilities thereon, as well as the real estate taxes, rates, duties and assessments as may be levied, charged or assessed against or in respect of the Premises or the building in which they are situated, and any and all taxes, whether of the same nature or not, which may in the future be levied in addition to or in lieu of such taxes, whether any such taxes, rates, duties or assessments are charged by any federal, municipal, state, school or other governmental body; provided that all such amounts payable pursuant to the Lease shall be calculated and payable in accordance with the provisions of the Lease.
- 3.3 Notwithstanding the provisions of section 3.4 below, Sublessee shall pay to the Lessee a one-time fee equal to \$_____ ("Lease Administration Fee") to compensate Lessee for entering into the sublease agreement.
- 3.4 All payments required to be made by the Sublessee hereunder shall, unless otherwise specified by the Lessee herein or otherwise by notice in writing to the Sublessee, be made by the Sublessee directly to the Lessor. Furthermore, all payments shall be made without the necessity of any demand therefor, without any deduction or abatement and regardless of any claim, set off or compensation which may be asserted by the Sublessee.
- 3.5 Except as otherwise expressly provided under this Sublease, all amounts required to be paid by the Sublessee hereunder shall be deemed to be rent and collectable by the Lessee

as rent, with all rights and remedies as reserved to the Lessee as it has under this Sublease and at law in respect of rent in arrears.

- 3.6 The Lessee acknowledges receipt from the Sublessee of advance rent in the amount specified in the Schedule forming part of this Sublease, to be held without interest by the Lessee and applied on account of the rent payable for the first complete calendar month of the term hereof.
- 3.7 If required by the terms of the Lease, the Sublessee shall present to the Lessee at the beginning of each year of the term of this Sublease a series of twelve monthly post-dated checks to cover the payment of minimum or basic rent and all additional rent due and payable hereunder and under the Lease, for each year of the term of the Sublease. The Sublessee agrees to deliver the said post-dated checks to the Lessee not later than ten (10) days prior to the commencement of each year of the term of this Sublease, except with respect to the first year of the term, in which case the Sublessee shall deliver the required post-dated checks on the execution of this Sublease. The Lessee also reserves the right if required by the Lease, to require the Sublessee to make the payments of minimum or basic rent and all additional rent due and payable hereunder by automatic bank transfer on the terms of, and following the procedure provided for in Section 6.15 of the Franchise Agreement.

SECTION IV - SUBLESSEE COVENANTS

- 4.1 The Sublessee covenants and agrees with the Lessee:
- 4.1.1 to perform all of the obligations of the tenant/lessee under the Lease and to be bound by the terms of the Lease, as such obligations and/or terms may be qualified by another provision of this Sublease, and in each case as if named therein as tenant/lessee with each reference therein to the Landlord/lessor being deemed to include both the Lessor and Lessee and with each reference therein to the tenant/lessee being deemed to mean the Sublessee;
- 4.1.2 to maintain and keep in first class order, condition and repair the Premises and all equipment, fixtures, chattels and improvements herein or thereon, and upon notice in writing from the Lessee, to repair the Premises as the Lessee or Lessor may so require;
- 4.1.3 to use the Premises only as a **MANCHU WOK RESTAURANT** ("**Restaurant**") (as defined in the Franchise Agreement) and for no other purpose and, from the commencement of the term of this Sublease and thereafter, continuously operate the Restaurant at the times and in the manner required under the provisions of the Lease and the Franchise Agreement;
- 4.1.4 to take out and maintain at all times during the term hereof in the names of the Sublessee and the Lessee (as an additional insured) the insurance required to be maintained by the tenant/lessee under the Lease and by the Franchisee under the Franchise Agreement, including (notwithstanding the terms of the Lease) business interruption and /or loss of profit insurance in such amounts as Lessee must consent to in writing;
- 4.1.5 not to permit any work, renovation, or alteration to be carried out with respect to the Premises unless any such work, renovation or alteration is in accordance with

the terms of the Lease, all applicable governmental by-laws and other legal requirements, moreover, no such work, renovation or alteration shall be commenced without the prior written consent of the Lessee and, if required, the Lessor. All such work shall be subject to inspection by and reasonable supervision of the Lessee (the Lessee's reasonable costs of which shall be paid by the Sublessee on demand) and shall be performed in accordance with any reasonable conditions or regulations imposed by the Lessee; and

4.1.6 at the expiration or sooner termination of the term hereof, to peaceably surrender and yield unto the Lessee the Premises with all appurtenances thereto in first class order, repair and condition and to surrender all keys for the Premises to the Lessee at the place then fixed for the payment of rent and to inform the Lessee of all combinations and all locks, safes and bolts, if any, in the Premises. The Sublessee's obligation to observe and perform the provisions of this subsection shall survive the expiration or sooner termination of this Sublease.

SECTION V - SECURITY DEPOSIT

The Lessee acknowledges receipt of a security deposit in the amount specified in the attached Schedule A, to be held by the Lessee without interest throughout the term hereof as security for the faithful performance by the Sublessee of all of the terms, covenants and conditions of this Sublease and for any lease adjustments made pursuant to this Sublease or the Head Lease subsequent to the end of the term. Provided that the Sublessee has maintained all its obligations under this Sublease, the aforesaid security deposit, or any balance then remaining shall be returned to the Sublessee within 30 days of the receipt of the final lease adjustments by the Lessee. If at any time, the security deposit or any part thereof is applied by the Lessee to remedy any default pursuant to this Sublease, or to repair or maintain the Premises or applied for any other purpose permitted under this Section 5.1 or elsewhere in this Sublease, then the Sublessee shall, upon written demand of the Lessee, forthwith remit to Lessee a sufficient amount to restore the security deposit to the original sum deposited.

SECTION VI - RECORDS AND STATUS STATEMENT

- 6.1 The Sublessee shall maintain and deliver to the Lessee all statements, reports, financial statements and any other records required to be delivered by: (i) the Franchisee to the Franchisor pursuant to the Franchise Agreement and, (ii) the tenant/Lessee to the Lessor pursuant to the Lease, and shall permit the Lessor and Lessee, their respective officers, agents or auditors to have the same access to the records of the Sublessee as granted to the Lessor by the Lessee in respect of the Lessee's records under the Lease.
- 6.2 Within ten (10) days after the Lessee requests it from the Sublessee (or such shorter period of time as may be required under the Lease for such statements and/or certificates), the Sublessee will deliver to the Lessee, on a form supplied by the Lessee, a status statement or certificate to the Lessee and/or any proposed mortgagee, purchaser or other transferee of part or all of the Premises stating;
- 6.2.1 that this Sublease is in full force and effect, except only for any modifications that are set out in the statement of certificate;
- 6.2.2 the commencement and expiration dates of the Sublease;

- 6.2.3 the date to which rent has been paid under the Sublease and the amount of any prepaid rent or any deposits held by the Lessee or Lessor;
- 6.2.4 that there is not any uncured default on the part of the Sublessee, Lessee or Lessor or if there is a default, a certificate which states the particulars of such default; and that no event has occurred, nor does any condition exist, that with notice or the passage of time, or both, would constitute a default hereunder or under the Lease;
- 6.2.5 whether there are any setoffs, defenses or counterclaims against enforcement of the obligations to be performed by the Sublessee or Lessee under this Sublease;
- 6.2.6 with reasonable particularly, details concerning the Sublessee and any indemnitor's financial standing and corporate organization; and
- 6.2.7 any other information or statement that a proposed mortgagee, purchaser, or transferee may reasonably require.

SECTION VII - ASSIGNMENT OR SUBLET

- 7.1 Notwithstanding anything to the contrary, the Sublessee hereby agrees that it shall not assign or transfer its rights in this Sublease nor sublet the Premises or any part thereof, nor permit the use or occupation of all or any part of the Premises by any other, nor mortgage or encumber this Sublease, the Sublessee's interest herein, the Premises or any equipment, fixtures or furniture therein or thereon, without obtaining, in each instance, the previous written consent of the Lessor (if required under the Lease) as well as the Lessee, and such consent by the Lessee may, notwithstanding any statutory provision to the contrary, be arbitrarily or unreasonably withheld. Without in any way limiting the right of the Lessee to arbitrarily or unreasonably withhold its consent to any assignment or subletting, as aforesaid, the Sublessee acknowledges and agrees that the Lessee, in determining whether to grant its consent, shall be entitled (in addition to any other tests or requirements which the Lessee may consider, at law or otherwise) require the proposed sublessee or assignee to fully comply with all of the requirements and conditions set out in the Franchise Agreement. Notwithstanding any assignment or sublease, the Sublessee and the Guarantor shall remain jointly and severally liable with any assignee or subtenant for all obligation of the Sublessee under this Sublease and shall not be released from performing any of the terms, covenants and conditions of this Sublease or the Franchise Agreement. Further, any consent granted by the Lessee shall be conditional upon and subject to the Sublessee causing any such assignee, sublessee or other occupant of the Premises to enter into an agreement directly with the Lessee in writing whereby such assignee or sublessee or other occupant covenants and agrees (i) to assume and continue to perform and be bound by all of the terms, covenants and conditions contained in this Sublease and the obligations of the Lessee contained in the Lease; and (ii) to assume and continue to perform and be bound by all the terms and conditions contained in the Franchise Agreement. The Sublessee and the Guarantor and any such assignee, sublessee or other occupant shall jointly and severally indemnify the Lessee and save it harmless from any costs including legal and administrative charges incurred by the Lessee in connection with such assignment, sublease or other agreement and in providing consent thereto.
- 7.2 If the Sublessee is a corporation, any transfer or issue by sale, assignment, bequest, inheritance, operation of law or other disposition, or by subscription from time to time, of all or any part of the legal or beneficial ownership of the corporate shares of the Sublessee or of an affiliate, or by any manner of corporate reorganization, which results in any change from the

date hereof in the effective voting control of, or ownership of the economic interests in, the Sublessee shall be deemed to be an assignment hereunder.

<u>SECTION VIII - LESSEE COVENANTS</u>

- 8.1 Subject to the rights of the Lessor and the due performance by the Sublessee of its obligations under this Sublease and the Franchise Agreement, the Lessee covenants and agrees with the Sublessee as follows:
- 8.1.1 that the Sublessee shall peaceably and quietly hold and enjoy the Premises during the term hereof without hindrance or interruption by the Lessee, subject nevertheless to the terms and conditions of this Sublease and the Lease:
- 8.1.2 that if the Lessor shall fail to observe or perform any of the obligations on the part of the Lessor contained in the Lease, then the Lessee shall give such notice to the Lessor and take such reasonable action as may be necessary to enforce such obligations against the Lessor, provided that the Lessee shall not be obligated to give such notice or to take such action unless
 - (a) the same has been requested in writing by the Sublessee;
 - (b) the Sublessee agrees to reimburse the Lessee for all its costs and expenses in giving such notice or taking such action and, if required by the Lessee, pay to the Lessee in advance an amount equal to the Lessee's reasonable estimate of such costs and expenses; and
 - (c) such notice or action shall subject the Lease to risk or forfeiture or termination.

The Lessee's sole obligations under this Sublease are those expressly stated herein. The Lessee shall not be liable for any failure on the part of the Lessor to observe and perform the covenants and agreements contained on the part of the Lessor under the Lease nor shall the Lessee be required to perform any of the Lessor's obligations under the Lease.

SECTION IX - LOSS OR DAMAGE AND INDEMNIFICATION

9.1 The Lessee shall not, for any reason whatsoever, including, without limitation, if same results from the fault, nonfeasance, or negligence of the Lessor, the Lessee, or their respective agents, employees or other persons for whom it may be in law responsible, be responsible for damages suffered by the Sublessee or any other person by reason of the manner of operation of, defect in or want of repair of the Premises or the building in which they are contained or any part thereof. Without restricting the generality of the foregoing, there shall be no abatement or reduction of rent, or any other monies, due hereunder nor shall the Sublessee be entitled to damages or losses from the Lessee on account of partial or total failure or lessening of the supply of air-conditioning, electrical or any other service, nor on account of anything coming through or leaking from the roof, windows or otherwise whereby steam, water, snow, smoke or gas, may leak or flow into the Premises nor on account of any damage or annoyance occasioned by the condition or any damage or annoyance arising from any acts, omissions or negligence of co-tenants or other occupants of the building wherein the Premises are located, or the owners or occupants of adjacent properties, nor on account of the making of any repairs, alterations, improvements, additions or structural changes to the said building or any part thereof or any property adjacent thereto. Neither shall the Lessee be responsible for any damage to or loss of

any property, nor injury to any person or interruption of the Lessee's business, which occurs in, on or about the Premises, howsoever occurring.

The Sublessee shall indemnify and save the Lessee fully harmless from and against any claim, suit, liability or demand, and from any and all loss, cost, damage or expense arising out of or in respect of any failure by the Sublessee to observe or perform any of the covenants or obligations of the Lessee under this Sublease or the Lease and for all loss, cost, damage, expense and liability caused to or suffered by the Lessee by reason of any act or omission of the Sublessee or any of its employees or agents. Furthermore, the Sublessee shall indemnify and save the Lessee fully harmless from all claims and demands of every kind and nature made by any person to or against the Lessee for all costs, damages or expenses incurred by or injury or damage to such person or his, her or their property, if such claims or demands arise out of, or relate in any manner to, the use and occupation of the Premises by the Sublessee or any such person or from anything occurring in or about the Premises or in connection with this Sublease and from all costs, attorney's fees, expenses and liabilities incurred in connection with any such claims or any action or proceeding brought in respect thereto.

SECTION X - DEFAULT

- 10.1 In the event:
- the Sublessee fails to pay any amounts due hereunder on the day or dates appointed for the payment thereof and fails to remedy such default within five (5) days after Lessee delivers Notice of Default (or such shorter period of time, if any, provided under the Lease to cure such default);
- the Sublessee fails to observe or perform any other of the covenants agreements, provisions or conditions of this Sublease to be observed or performed by the Sublessee, (other than the terms, covenants and conditions set out below in subparagraphs 10.1.3 to 10.1.11 inclusive, for which no notice shall be required or, except as expressly provided otherwise in this Section 10.1, opportunity to cure granted) the Sublessee fails to cure such default within ten (10) days' notice thereof (or such shorter period of time, if any, provided under the Lease to cure such default);
- the Sublessee commits an act of bankruptcy or becomes insolvent, seeks, consents to or acquiesces in the appointment of any trustee, receiver or liquidator of the Sublessee, or takes or attempts to take the benefit of any act now or hereafter in force for bankrupt or insolvent debtors or files any proposal or makes any assignment for the benefit of creditors or any proceeding for a composition with creditors under any federal or state law is instituted by or against the Sublessee or a petition shall be filed against the Sublessee and not be discharged or bona fide disputed within seven (7) days from the filing thereof. The Sublessee acknowledges and agrees that, to the maximum extent permitted by law, it waives any rights it may have for itself or a trustee in bankruptcy under any bankruptcy or landlord-tenant law or any other applicable legislation to elect to retain the unexpired term of the Sublease, in the event of any bankruptcy or insolvency by the Sublessee:

- 10.1.4 the Sublessee sells or disposes of the goods, merchandise and/or chattels of the Sublessee or removes them from the Premises other than in the ordinary course of business;
- 10.1.5 the Sublessee abandons the Premises or fails to continually operate its Manchu Wok Restaurant business in the Premises in accordance with the provisions of this Sublease and the Franchise Agreement;
- the Sublease or any of the goods and chattels of Sublease shall be at any time seized or taken in execution or attachment by any creditor of the Sublessee;
- 10.1.7 a receiving order is made against the Sublessee, or a receiver is appointed by private instrument or by any court of competent jurisdiction for all or a portion of the Sublessee's property;
- 10.1.8 the Sublessee Subleases the Premises or any part thereof to a third party or assigns or transfers this Sublease to a third party or permits a third party the use or occupation of the Premises or grants any security or encumbrance on this Sublease, the Premises or any of its property located on the Premises other than in accordance with Section VII;
- 10.1.9 the Sublessee or any agent of the Sublessee falsifies any report required to be furnished to the Lessee pursuant to the terms of this Sublease or the Franchise Agreement, however, inadvertent or human error is excluded;
- 10.1.10 the Franchise Agreement is terminated or the Sublessee is in default or in breach of any provision contained in the Franchise Agreement and has not cured such default within the time period (if any) specified in the Franchise Agreement to cure such default or is in default or in breach of any other agreement entered into with the Lessee or with any affiliate of the Lessee;
- the Sublessee, within any twelve-consecutive-month period, commits three or more defaults under this Sublease or the Lease, whether or not such defaults are cured; then, in every such case, at the option of the Lessee, the then current month's and the next ensuing three (3) months' basic or minimum rent hereunder and all additional rent then due and owing hereunder shall immediately become due and payable and the Lessee may, at its option, terminate this Sublease and the term hereof shall then be considered to have reached its conclusion, in which event the Lessee may re-enter the Premises and take possession thereof and upon such re-entry the Lessee may remove all persons or property from the Premises and such property so removed shall be stored in a public warehouse or elsewhere at the cost of and for the account of the Sublessee, all without service of notice or resort to legal process and without the Lessee being deemed guilty of trespass or becoming liable for any loss or damage which may be occasioned thereby.
- 10.2 If the Lessee at any time terminates this Sublease for any breach, in addition to any of the remedies it may have, it may recover from the Sublessee all damages it incurs by reason of such breach, including the costs of recovering the Premises, attorney's fees and the worth at the time of such termination of the excess, if any, of the rent reserved in this Sublease for the remainder of the stated term hereof over the then reasonable rental value of the Premises

for the remainder of the stated term hereof, all of which amounts shall be immediately due and payable by the Sublessee to the Lessee.

- 10.3 If the Lessee does not exercise its option to terminate this Sublease pursuant to Section 10.1 hereof, the Lessee may at its option, without additional notice to the Sublessee and without effecting the termination, as agent of the Sublessee, take possession of the Premises together with all property therein, and, sublet all or part of the Premises together with all or any of such improvements, fixtures and furnishings located therein on whatever terms the Lessee may deem appropriate. The Sublessee hereby grants the Lessee an irrevocable mandate and power (coupled with an interest) to effect the foregoing (and to undertake all such acts, and make, execute, and deliver all such instruments, documents, and agreements, as it deems appropriate or advisable to consummate such taking of possession and subletting of the Premises), and in addition thereto, to make such alterations or repairs as are necessary in order to sublet the Premises. In such case, the Sublessee shall continue to be liable for the performance of all its obligations under this Sublease, including the payment of rent and all other amounts payable hereunder. The Lessee shall be entitled to enforce all of the obligations of the Sublessee against any such subtenant, to collect and receive all rent and other amounts payable by such subtenant and apply them against all indebtedness of the Sublessee to the Lessee from time to time existing. All deficiencies shall be payable by the Sublessee to the Lessee as and when and from time to time demanded. If the Lessee sublets without terminating as hereinabove provided, it may at any time thereafter elect to terminate this Sublease for any previous default of the Sublessee.
- 10.4 Without in any way creating an obligation on the Lessee, if the Sublessee shall fail to make any payments required to be made pursuant to the provisions of this Sublease or the Lease, as the case may be, then the Lessee, may, at its option, make all such payments on behalf of the Sublessee who shall forthwith reimburse the Lessee for all costs and expenses so incurred upon demand. If the Sublessee shall fail to observe or perform any of its other obligations under this Sublease, the Lessee may attempt to remedy such default and may enter the Premises for such purpose without thereby terminating this Sublease and the Sublessee shall reimburse the Lessee for its costs and expenses in so doing (plus an administration fee equal to fifteen percent (15%) of such costs and expenses). The Lessee shall not be liable for anything done or omitted to be done by the Lessee in attempting to remedy such default.
- 10.5 All arrears of rent and other amounts to be paid by the Sublessee to the Lessee shall bear interest from the date due at the rate per annum equal to the lesser of (i) 18% or (ii) the greatest legal rate.
- 10.6 If the Sublessee shall fail to perform any of its obligations herein, the Lessor shall have, without limiting the foregoing, all of the rights, remedies, privileges, protections and indemnities against the Sublessee which the Lessor has under the Lease for a breach of the obligations of the tenant/lessee thereunder.
- 10.7 Mention in this Sublease of any particular remedy or remedies of the Lessee in respect of any default by the Sublessee does not preclude the Lessee from exercising any other remedy in respect thereof, whether pursuant to this Sublease, the Lease or pursuant to law. No remedy shall be exclusive or dependent upon any other remedy, but the Lessee may, from time to time, exercise any one or more remedies independently or in combination, such remedies being cumulative and not alternative.

SECTION XI - LESSOR CONSENT

11.1 If required pursuant to the provisions of the Lease, the parties acknowledge that, this Sublease is subject to the Lessee obtaining the written consent of the Lessor to this Sublease. In the event that the said consent is not obtained, this Sublease shall be null and void and have no further effect, without any recourse one party against the other.

SECTION XII - TERMINATION OF LEASE

- 12.1 The Sublessee acknowledges and agrees that it has no greater interest in the Premises than the Lessee under the Lease and that, if the Lease is terminated for any reason whatsoever, this Sublease shall thereupon terminate forthwith.
- 12.2 The Sublessee acknowledges and agrees that it waives for itself and any trustee in bankruptcy, any rights it may have under any applicable state or federal landlord and tenant act, bankruptcy act, or any other applicable legislation, to elect to retain the unexpired term of the Sublease or to obtain any right to enter into any Lease or other agreement directly with the Lessor, if in the event of any bankruptcy or insolvency by the Sublessee, the Lessor terminates the Lease.

SECTION XIII – intentionally deleted

SECTION XIV - GENERAL PROVISIONS

- 14.1 The Sublessee agrees that it will not register or record this Sublease. Any registration or recording of this Sublease is void ab initio. Breach of this provision shall entitle the Lessee, at its option, to immediately terminate the Sublease.
- 14.2 Save and except as may be otherwise provided in this Sublease, all of the covenants, terms, agreements and conditions contained in the Lease are hereby incorporated mutatis mutandis into this Sublease as if the Lessee were the Landlord/lessor in the Lease and the Sublessee were the tenant/lessee in the Lease. If any term or condition of this Sublease is inconsistent with or at variance with any term or condition of the Lease, then the term or condition of this Sublease shall prevail.
- 14.3 The exercise by the Lessor of any of its rights contained in the Lease shall, upon written notice by the Lessor to the Sublessee of such exercise, be binding upon the Sublessee. The Lessee may also exercise its rights under this Sublease and at law notwithstanding the exercise by the Lessor of any of its rights contained in the Lease.
- 14.4 This Sublease, the Franchise Agreement and any agreements contemplated or referred to herein or thereby, including any schedules annexed, constitute the entire agreement between the parties pertaining to the subject matter hereof and supersede all prior agreements, understandings, negotiations, and discussions, whether oral or written, between the parties. There are no warranties, representations or other agreements between the parties in connection with the subject matter hereof except as specifically set forth herein.
- 14.5 The parties acknowledge and agree that:
- 14.5.1 the preamble hereof forms an integral part of this Sublease;
- 14.5.2 no amendment or other modification to this Sublease shall be valid or binding upon the parties unless the same is in writing and signed by the party as to whom enforcement of such amendment or modification is sought and such writing specifically refers to this Sublease;

- 14.5.3 the failure of the Lessee to insist upon the strict performance of any of the terms, covenants and conditions hereof shall not be deemed a waiver of any of the rights or be deemed a waiver of subsequent breach or default of any of such terms, covenants or conditions. The subsequent acceptance of rent or other payments hereunder by the Lessee shall not be deemed to be a waiver of any preceding breach by the Sublessee of any term, covenant or condition of this Sublease, regardless of the Lessee's knowledge of such preceding breach at the time of the acceptance of such rent or other monetary payment;
- 14.5.4 all notices, requests, demands, consents and other communications required or permitted under this Sublease must be in writing (including telex and telegraphic communication) and must be (as elected by the person giving such notice) hand delivered by messenger or courier service, telecommunicated, or mailed (airmail if international) by registered or certified mail (postage prepaid), return receipt requested, addressed to the appropriate party or parties at their respective addresses set forth above (and, if to the Franchisor, must be addressed to its President and C.E.O.) or to such other address as each such party may designate by notice complying with the terms of this Section; each such notice is deemed delivered:
 - a. on the date delivered if by personal delivery;
 - b. on the date of transmission, with confirmed answer back if by telefax or other telegraphic method; or
 - c. on the date upon which the return receipt is signed or delivery is refused or the notice is designated by the postal authorities or courier service as not deliverable, as the case may be, if mailed or couriered.
- 14.5.5 if there shall be any inconsistencies or conflicts between the terms of this Sublease and any of the provisions contained in the Franchise Agreement, the parties hereto acknowledge, agree and confirm that the provisions of the Franchise Agreement shall be read in priority to and shall supersede any of the provisions contained in this Sublease;
- 14.5.6 time shall be of the essence of this Sublease and every part hereof;
- 14.5.7 whenever a personal pronoun is used herein, it is understood that such usage shall include both singular and plural, masculine, feminine and neuter, and refer in appropriate cases to corporations or other legal entities as well as individuals;
- 14.5.8 the captions contained herein are for reference only and in no way affect this Sublease;
- 14.5.9 each obligation of any party hereof, even though not expressed as a covenant, is considered for all purposes to be a covenant;
- 14.5.10 if any covenant, obligation or agreement or part thereof or the application thereof to any person or circumstances shall be found to be invalid or unenforceable then such covenant, obligation or agreement or part thereof shall be deemed to be severable and shall not affect the validity or enforceability of any other provisions or covenant hereof;
- 14.5.11 each party shall from time to time hereafter and upon any reasonable request of the other party, execute, make or cause to be made, all such further acts, deeds, assurances,

certificates and things as may be required to more effectively implement the true intent of this Sublease;

14.5.12 all rights and liabilities herein given to or imposed upon the respective parties hereto shall extend to and bind the several respective permitted heirs, executors, administrators, successors and assigns of the said parties. No rights, however, shall enure to the benefit of any assignee or subtenant of the Sublessee unless such assignment or subletting has been approved by the Lessee in writing pursuant to the provisions of this Sublease; and

SECTION XV - JOINDER

All the Sublessee's legal and beneficial owners must enter the Joinder agreement set forth below. By doing so, each of such owners agrees that he or she is bound by all the terms of this Sublease as if he or she were the Sublessee thereunder and agrees that he or she is jointly and severally liable with the other owners and the Sublessee for all the Sublessee's obligations under this Sublease.

[signature page follows]

IN WITNESS WHEREOF each of the partie of the day hereinabove indicated.	es have	signed and sealed this Sublease effective as
Date Signed: 20_		Date Signed:20
MTY FRANCHISING USA, INC. ("Lessee")		("Sublessee")
per:	_	per
		per
		Date Signed:201 _
WITNESS	_	Name (Guarantor)
WITNESS		Name (Guarantor)
	JOINDE	:R
such parties agrees that he or she is join	tly and ns unde	ssee's beneficial and legal owners. Each of severally liable with the each other and the r this Sublease and is bound by all the terms der.
Print Name:	_	Print Name:
Ownership Percentage:	_%	Ownership Percentage:%
Print Name:	_	Print Name:
Ownership Percentage:	_%	Ownership Percentage:%

SCHEDULE "A"

1.	Amount pursuant to Section 3.6 (first month's rent)	\$
2.	Amount pursuant to Section 5.1 (security deposit)	\$

GUARANTEE

In consideration of M	TY Franchising US	A, Inc. ("Lessor	r") entering i	into the attache	d Sublease
with	(name of Sub	olessee) ("Suble	essee") and	in consideration	of the sum
of ten dollars (\$10.00)) paid to each of th	ne undersigned,	the receipt	and sufficiency	of which is
hereby by each of the	undersigned ackno	owledged, (each	of)	(and)	
undersigned, as Guar	antor(s) of the prov	isions of the atta	ched Suble	ase:	

- (a.) does hereby jointly and severally, irrevocably and unconditionally guarantee payment, observance and performance to the Lessee of all present and future obligations of any nature at any time owing by the Sublessee to the Lessee under the Sublease, whether direct or indirect, absolute or contingent, matured or unmatured, extended or renewed, wherever and however incurred, and whether the Sublessee is bound alone or with others and whether as principal or as surety (such obligations being herein after called the "Obligations"). This guarantee is a continuing guarantee of all of the Obligations and is a guaranty of payment, not collection. The Lessee shall not be bound to exercise or exhaust its recourse against the Sublessee or any other person, or against any other guaranties or any security it may at any time hold, before being entitled to full payment, observance and performance from the Guarantors of the Obligations. The Guarantors' liability to make payment, observance and performance of the Obligations under this guarantee shall arise upon the Lessee making demand in writing of any Guarantor, and such demand shall be deemed to have been effectively made when an envelope containing such demand addressed to a Guarantor at his address last known to the Franchisor is posted, postage prepaid or delivered by prepaid courier otherwise delivered to the Guarantor.
- (b.) Without in any way lessening the liability of the under this guarantee, and without obtaining the consent of or giving notice to the Guarantor, the Lessee may discontinue, reduce, increase or otherwise vary the credit of the Sublessee, may grant renewals, extensions, indulgences and releases to, and may accept compromises from or otherwise deal with the Sublessee and others in such manner as the Lessee may see fit, and the Lessee may apply all monies received from the Sublessee or others or from securities or guaranties toward such part of the Obligations as the Lessee may see fit. The Obligations of the Guarantor under this guarantee shall not be released, discharged or in any way be affected by any change, alteration or modification of this or any other agreement, or by the bankruptcy or insolvency of the Sublessee or of any Guarantor, or by any other loss of capacity of the Sublessee or any other Guarantor, or by any other act or proceeding in relation to the Sublessee or this or any other agreement, or by any other thing whereby the Guarantor might otherwise be released.
- (c.) Until payment, observance and performance in full of the Obligations, the Guarantor shall not claim any set-off or counterclaim against the Sublessee in respect of any liability of the Sublessee to the Guarantor, nor claim or prove in any bankruptcy or insolvency of the Sublessee in competition with the Lessee, nor have any right to be subrogated to the Lessee. All present and future debts and liabilities of the Sublessee to the Guarantor are hereby postponed to the Obligations, and all monies received by any Guarantor in respect thereof shall be received in trust for the Lessee and shall be paid over to the Lessee upon demand, the whole without in any way lessening or limiting the liability of the Guarantor under this guarantee. This postponement is independent of the quarantee herein and shall remain in full force and effect until payment, observance and

performance in full to the Franchisor of all of the Obligations is made, notwithstanding that the liability of the Guarantor under this guarantee may have been terminated.

(d.) The Lessee is not bound by any representations, warranties, promises or other inducements made by the Sublessee to the Guarantor or any of them and execution of this agreement by the Guarantor shall be conclusive evidence against the Guarantor that this agreement was not delivered in escrow or pursuant to any agreement that it should not be effective until any condition precedent or subsequent has been met. This guarantee shall not be discharged or affected by the death, disability, insolvency, or bankruptcy of any Guarantor.

The undersigned Guarantor(s) acknowledge that they have received, have had an ample time to read and have read this Agreement and fully understand its provisions. The undersigned Guarantor(s) further acknowledge that they have had an adequate opportunity to be advised by advisors of their own choosing (including without limitation their own lawyers, accountants, and bankers) regarding all pertinent aspects of the Sublease and the relationship created by the Sublease and that they have done so with the full understanding and knowledge of all of the terms of the Sublease.

In witness whereof the Guarantors have signed thisday of, 20				
(witness)	type name (Guarantor)			
(witness)	type name (Guarantor)			



EXHIBIT E

Manchu WOK

Table of Contents of Operating Manual

INTRODUCTION TO MANCHU WOK	INT-1
■ Welcome to Manchu WOK	INT-5
■ Employee Policies & Procedures	INT-7
■ Uniforms	INT-9
■ Personal Grooming Standards	INT-10
■ Employee Meal Policy	INT-11
■ Schedules	INT-12
■ Pay Periods	INT-12
■ Performance Reviews	INT-12
■ Probationary Period	INT-13
■ Discipline	INT-13
■ Complaint Procedures	INT-15
■ Customer Service Hotline	
■ Telephone	INT-17
■ HEALTH, SAFETY & SANITATION	⊔ \$&\$_1
■ Introduction	
■ Food Safety Hazards	
■ Food Contamination	
■ Biological Hazards	
■ Chemical Hazards	
■ Physical Hazards	
Preventing Food Contamination & Spoilage	
■ Preventing Food Contamination	
■ Preventing Food Spoilage	
■ Cleaning & Sanitizing	
■ Dishwashing Procedures	
■ Cleaning Checklists	
■ Dining Room	
■ Garbage Disposal	
■ Routine Health Inspections	
■ Mall Requirements	
■ Keeping Your Workplace Safe	HS&S-25
Occupational Health & Safety	HS&S-27
■ Basic Safety Procedures	HS&S-27
■ DAILY OPERATIONS	DO-1
■ Introduction	
Opening Procedures	
Afternoon Procedures	
Closing Procedures	
= 3100111g 1 1000dd100	

■ FRONT COUNTER	FC-1
■ Introduction	FC-2
■ 8 Keys To Good Service	FC-3
■ Sales & Service Routine / MSG	FC-8
Attract The Customer	FC-19
■ Sampling	FC-21
■ Close The Sale	FC-24
Serve The Customer	FC-26
Customer Serving Procedure	FC-28
■ Guide The Customer	FC-30
■ Front Counter Workstations	FC-31
■ Portion Control	FC-33
Suggestive Selling	FC-38
■ Unique Selling Points	FC-42
■ Complete The Transaction Process	FC-44
Cashier Procedures	
■ Cash Handling	FC-46
Register Policies & Procedures	FC-47
■ Handling Customer Complaints	FC-52
■ Steam Table Management	FC-54
■ CLEANING SCHEDULES & CHECKLISTS	CS&C-1
■ Introduction	CS&C-3
■ Daily Cleaning Schedules	CS&C-5
■ Kitchen	CS&C-7
■ Front Counter	CS&C-14
■ Cleaning Checklists	CS&C-19
■ Kitchen	CS&C-21
■ Front Counter	CS&C-23
Cleaning and Sanitation Program (Canada Only)	CS&C-25
■ EQUIPMENT & PREVENTIVE MAINTENANCE	E&M-1
■ Introduction	E&M-3
■ Equipment: Records, Warranty & Service	E&M-5
■ Equipment & Maintenance	E&M-11
	Total Number of Pages: 259



EXHIBIT E-1

Manchu WOK

In-Store Training Release and Waiver of Liability Agreement

In-Store Training Release and Waiver of Liability

READ THIS INSTRUMENT FULLY AND CAREFULLY PRIOR TO SIGNING. THIS IS A LEGALLY VALID AND BINDING OBLIGATION TO RELEASE A PARTY FROM ALL KNOWN AND UNKNOWN OBLIGATIONS. THIS DOCUMENT MUST BE SIGNED BEFORE TRAINING CAN BEGIN.

This In-Store Training Release and Waiver of Liability ("Release") executed on this	day
of, 20 ("Effective Date") by	("Trainee") is in
favor of , a corporation, a Limited Liability	company, or
as individual(s), and their respective directors, officers, members, partners, manage	gers, employees,
agents, insurers, successors, assigns, representatives, executors and heirs (individually	and collectively
the "Franchisee") and MTY Franchising USA, Inc., a Tennessee corporation, and its af	filiates and each
of their respective parents, subsidiaries, directors, officers, members, managers, em	ployees, agents,
successors, assigns and insurers (individually and collectively the "Franchisor").	Franchisor and
Franchisee shall collectively be referred to herein as the "Hosts."	

The Trainee desires to participate in a training program provided by the Hosts, which program involves engaging in activities related to being a restaurant worker ("Activities"), for the purpose of learning to operate a similar business. The Trainee understands that the Activities may take place at various restaurant facilities owned, operated or managed by Franchisor and/or Franchisee, and may include various restaurant activities, including, but not limited to, cooking, cleaning, lifting and climbing, all of which may take place in a facility where there is increased risk of bodily harm or personal injury, including but not limited to, slips, falls, burns, lacerations, exposure to hazardous materials, as well as many other risks. Trainee understands and agrees that no compensation shall be earned by or paid to Trainee in connection with Trainee's participation in the training program. In consideration of the training being provided to Trainee, in the Hosts' location, Trainee agrees that such training constitutes good and valuable consideration and hereby freely, voluntarily, and without duress executes this Release under the following terms:

Release and Waiver. Trainee does hereby release and forever discharge and hold harmless and promise to indemnify and not sue Hosts, both jointly and severally, from any and all liability, claims, actions, damages, judgments, obligations, expenses, including, without limitation, attorneys' fees, expert witness fees and costs, and demands of whatever kind or nature, either in law or in equity, which arise or may hereafter arise, in whole or in part, from Trainee's Activities with Hosts. Trainee understands that this Release discharges Hosts from any liability or claim that the Trainee may have against Hosts with respect to any bodily injury, personal injury, illness, death, or property damage that may result from Trainee's Activities with Hosts, whether caused by the negligence of Trainee or Hosts. Trainee also understands and hereby agrees that it is NOT an employee of Hosts and that Hosts do not assume any responsibility for or obligation to provide financial assistance or other assistance, including, but not limited to, medical, life, health, or disability, or workers' compensation insurance in the event of injury or illness.

<u>Medical Treatment</u>. Trainee does hereby release and forever discharge Hosts from any claim whatsoever which arises or may hereafter arise on account of any first aid, treatment, or service rendered in connection with the Trainee's Activities with Hosts.

<u>Insurance</u>. Trainee understands that, except as otherwise agreed to by Hosts in writing, Hosts do not carry or maintain health, medical, life, workers' compensation, or disability insurance coverage for any Trainee. Each Trainee is expected and encouraged to obtain his or her own medical, health, life, disability and workers' compensation insurance for his or her protection while conducting any Activities with Hosts.

1

<u>Photographic Release</u>. In consideration of the training provided to Trainee and without further compensation or royalties, Trainee hereby authorizes Hosts to capture and use in any form, format or media, whether now known or later developed for marketing, advertising and other business purposes, Trainee's name, image, likeness, voice and other personal information and hereby grants and conveys unto Hosts all right, title and interest in any and all photographic images and video or audio recordings captured or made by Hosts during the Trainee's Activities with Hosts, including but not limited to, any claims, damages, royalties, proceeds, or other benefits derived from such photographs or recordings.

<u>Prohibition on Hiring</u>. Trainee understands and agrees that during its Activities with Hosts, it will be afforded an opportunity to meet and work with experienced employees of Hosts, who could be of value to Trainee should they be employed by Trainee. Trainee hereby agrees that for a period of two (2) years following the Effective Date of this Release, Trainee shall NOT, directly or indirectly, recruit, solicit, offer, hire, or otherwise entice any employee or agent of Hosts to become a Trainee employee or to otherwise sever his or her employment with Hosts.

Other. Trainee expressly agrees that this Release is intended to be as broad and inclusive as permitted by the laws of the State of Arizona, and that this Release shall be governed by and interpreted in accordance with the laws of the State of Arizona. Trainee agrees for himself/herself and his or her heirs, executors and representatives, that the representations in this Release are contractually binding, and are not mere recitals, and that should Trainee or his/her heirs, executors or representatives assert a claim contrary to what Trainee has agreed to in this Release, the claiming party shall be liable for the expenses (including, without limitation attorneys' fees and costs) incurred by Hosts in defending the claims. This Release may not be modified orally, and a waiver or modification of any provision shall not be construed as a waiver or modification of any other provision herein or as consent to any subsequent waiver or modification. Trainee consents to the release by any third party to Hosts and their insurance carriers of Trainee's name and medical information that may relate solely to any injury or death he or she may suffer arising from the Activities. Trainee agrees that in the event that any clause or provision of this Release shall be held to be invalid by any court of competent jurisdiction, the invalidity of such clause or provision shall not otherwise affect the remaining provisions of this Release which shall continue to be enforceable and Hosts and Trainee agree to promptly negotiate the terms of a substitute clause or provision to replace the severed or unenforceable clause or provision, consistent with the parties' original intent and then-current law.

IN WITNESS WHEREOF, I certify that I am 18 years of age or older. I have carefully read, understand and have freely executed this Release as of the Effective Date.

Trainee Signature				
Printed Name				
Address				
street	city	state	zip	
Home Phone	Work Phone			
Witness Signature (must be 18 years of age or older)				
Printed Name				



EXHIBIT F

Manchu WOK

List of Current Franchisees of The Franchisor, MTY Franchising USA, Inc. As of November 30, 2024

CURRENT STORE LIST IN USA - MANCHU WOK

(As of November 30, 2024)

Franchise Company	Owners	Address	City	State	Zip	Phone	Owner Emails
NICKDBUN, INC.	Dara Bun,Chari ya Ty	2000 Riverchase Galleria, Suite D	Birmingham	AL	35244	2059853145	nickdbun@gmail.com ,chariyaty@yahoo.co m
Yreka Properties, LLC.	Avtar.Dhill on,Kanwar deep.Bhull ar,Sandee p.Bhullar	787 Montague Rd	Yreka	CA	96097	5302911913	Dhillon9655@gmail.c om,KANWERDEEP @GMAIL.COM,Yreka TravelPlaza@gmail.c om
SUNDANCE RESTAURANTS HOLDINGS LLC	Omar Bukhowa, Michael.M urray	1504 6th Street	Twentynine Palms	CA	92278	7603626420	omar@bukhowa.com, mike@sundancerh.co m
Global Miami Joint Venture	Global Miami Joint Venture	4200 NW 21 Street 2nd Level, Concourse D, D10A	Miami	FL	33123		marcia@globalmia.co m
Global Miami Joint Venture	Global Miami Joint Venture	4200 NW 21 Street	Miami	FL	33122		marcia@globalmia.co m
AirPancho's LLC.	AirPancho' s LLC.	Airside 3, 9702 Jeff Fuqua Blvd.	Orlando	FL	32827		tonysirica@att.net
O Hare Ventures II L.L.C.	Cortez Carter,Kev in Sotto,Matt Evans,Mar c Brooks	Fast Lane Food Court	Chicago	IL	60666	773.575.201	cortez@hydeparkhos pitality.com,kevin@hy deparkhospitality.com ,matt@hydeparkhosp itality.com,marc@hyd eparkhospitality.com
O Hare Ventures II L.L.C.	Kevin Sotto,Matt Evans,Mar c Brooks,Co rtez Carter	Rotunda Building, c/o Main Street Food Court	Chicago	IL	60666	773.575.201 1	kevin@hydeparkhosp itality.com,matt@hyd eparkhospitality.com, marc@hydeparkhosp itality.com,cortez@hy deparkhospitality.com

O Hare Ventures II L.L.C.	Kevin Sotto,Matt Evans,Mar c Brooks,Co rtez Carter	H/K Terminal Food Court, Bldg. 6, O'Hare Field	Chicago	IL	60666	773.575.201 1	kevin@hydeparkhosp itality.com,matt@hyd eparkhospitality.com, marc@hydeparkhosp itality.com,cortez@hy deparkhospitality.com
Asian Kitchen 168, Inc.	Jing.Dong	150th Ave & 147 St	Jamaica	NY	11430	7189178889	dongjing1688@gmail. com
Li Zhang	Li Zhang,Xuf an Chen	1500 Harvey Road	College Station	TX	77840		allenzha5658@gmail. com
The Army and Air Force Exchange Service ("AAFES")	Nicole Gladney	Building 1611 Marshall Road	Fort Bliss	TX	79916	214-312- 2435	gladneyng@aafes.co m
The Grove, Inc.	The Grove, Inc.	8000 Cedar Springs Road - Dallas Love Field Airport	Dallas	TX	75235	2143588692	ahernandez@tgiconc essions.com
Sun's Inc.	Sandy Sun,Nancy Lin Sun	17801 International Blvd., Space CA-4A, Concourse - A	Seatac	WA			,nancylinsun@gmail. com
Muckleshoot Indian Tribe	Virginia.Cr oss	2402 Auburn Way	South Auburn	WA	98002	235-804- 4444	Samuel.Askew@muc kleshootcasino.com

Current Franchisees for Outlets not yet Opened:

NONE



EXHIBIT G

Manchu WOK

Former Franchisees of the Franchisor

The names and last known addresses and telephone numbers of franchisees of the Franchisor, MTY Franchising USA, Inc., who have left the Manchu WOK® franchise system during the fiscal year ended November 30, 2024, or have not communicated with the Franchisor within 10 weeks of the issuance date of this Franchise Disclosure Document. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

For closed outlets

Company Name	Owner Names	City	State	Owner Phones/Email
Swinomish Indian Tribal Community dba Swinomish Casino & Lodge		Anacortes	WA	bcladoosby@swinomish.nsn.us,dwidma rk@swinomishcasino.com

For trans	ferred	outlets:
-----------	--------	----------

NONE



EXHIBIT H

Manchu WOK

State-Specific Disclosures and State-Specific Agreement Amendments

California Disclosure Addendum

In recognition of the requirements of the California Franchise Investment Law, Cal. Corp. Code §§ 31000-31516, and the California Franchise Relations Act, Cal. Bus. & Prof. Code §§ 20000-20043, the Franchise Disclosure Document for MTY Franchising USA, Inc. in connection with the offer and sale of franchises for use in the State of California shall be amended to include the following:

- 1. Our website, www.manchuwok.com, has not been reviewed or approved by the California Department of Business Oversight. Any complaints concerning the content of the website may be directed to the California Department of Business Oversight at www.dbo.ca.gov.
 - 2. Item 3, "Litigation," shall be amended by the addition of the following paragraphs:

Pursuant to California law, this Item does not include any information regarding the arrest of any person(s) that did not result in a conviction or plea of nolo contendere.

Neither we, nor any person identified in Item 2 above, is subject to any currently effective order of any national securities association or national securities exchange (as defined in the Securities and Exchange Act of 1934, 15 U.S.C. § 78a, et seq.) suspending or expelling such person from membership in such association or exchange.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

3. Item 5, "Initial Fees" shall be amended by the addition of the following:

"Payment of all initial fees is postponed until after all of franchisor's initial obligations are complete and franchisee is open for business."

4. Item 17, "Renewal, Termination, Transfer and Dispute Resolution," shall be amended by the addition of the following paragraph(s) at the conclusion of the Item:

The following notice is required to be inserted in this disclosure document by the state of California whenever an applicable provision is included in a Franchise Agreement.

California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer, or non-renewal of a franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, the law will control

The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C. § 101, et seq.).

The Franchise Agreement contains a covenant not to compete which extends beyond the termination of the franchise. This provision may not be enforceable under California law.

California Corporations Code § 31125, requires the franchisor to give the franchisee a disclosure document, approved by the Department of Business Oversight, before soliciting a proposed material modification of an existing franchise.

The Franchise Agreement requires the application of the laws of the State of New York. This provision may not be enforceable under California law.

- 5. THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE FRANCHISE DISCLOSURE DOCUMENT.
- 6. Section 3.1 of the Franchise Agreement, "Initial Franchise Fee" shall be amended by the addition of the following:

"Payment of all initial fees is postponed until after all of franchisor's initial obligations are complete and franchisee is open for business.

Illinois Disclosure

In recognition of the requirements of the Illinois Franchise Disclosure Act, III. Comp. Stat. §§ 705/1 to 705/44 and, more generally, Illinois law, the Disclosure Document for MTY Franchising USA, Inc. for use in the State of Illinois shall be amended as follows:

1. The "Summary" section of Item 17 (v), entitled Choice of Forum, is amended by adding the following language:

However, any provision in the Franchise Agreement that designates jurisdiction or venue in a forum outside of the State of Illinois is void under section 4 of the current Illinois Franchise Disclosure Act, although the Franchise Agreement may provide for arbitration in a forum outside of the State of Illinois.

2. The "Summary" section of Item 17 (w), entitled Choice of Law, is amended by adding the following language:

However, except for federal law, Illinois law applies if the jurisdiction requirements of the Illinois Franchise Disclosure Act of 1987 (as amended) are met.

2. Each provision of this addendum shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Illinois Franchise Disclosure Act, and, more generally, Illinois law, are met independently, without reference to this addendum.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Illinois Amendment to the Franchise Agreement

In recognition of the requirements of the Illinois Franchise Disclosure Act, III. Comp. Stat. §§ 705/1 to 705/44, and, more generally, Illinois law, the parties to the attached MTY Franchising USA, Inc. Franchise Agreement (the "Agreement") agree as follows:

1. Section 2.2 of the Agreement, under the heading "Renewal," shall be supplemented by the addition of the following new paragraph, which shall be considered an integral part of the Agreement:

If any of the provisions of this Section 2.2 are inconsistent with Section 20 of the Illinois Franchise Disclosure Act, the provisions of the Act shall apply. If Franchisor refuses to renew this Agreement, Franchisor shall compensate Franchisee if (and to the extent) such compensation is required under Section 20 of the Illinois Franchise Disclosure Act.

- 2. Section 10 of the Agreement, under the heading "Default and Termination of Franchise," shall be supplemented by the addition of the following new paragraph 10.13, which shall be considered an integral part of the Agreement:
 - 10.13 If any of the provisions of this Section 10 concerning termination are inconsistent with Section 19 of the Illinois Franchise Disclosure Act, then said Illinois law shall apply.
 - 3. Sections 14.1(c) and 14.1(d) of the Agreement are deleted.
- 4. Section 14.4 of the Agreement, under the heading "Dispute Resolution," shall be deleted in its entirety, and shall have no force or effect; and the following new paragraph shall be substituted in lieu thereof:

14.4 Reduced Time Within Which to Commence Actions

ANY AND ALL CLAIMS AND ACTIONS ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE RELATIONSHIP OF FRANCHISOR AND FRANCHISEE, OR FRANCHISEE'S OPERATION OF THE RESTAURANT, BROUGHT BY ANY PARTY HERETO AGAINST ANY OTHER SHALL BE COMMENCED WITHIN THREE (3) YEARS FROM THE OCCURRENCE OF THE FACTS GIVING RISE TO SUCH CLAIM OR ACTION. OR SUCH CLAIM OR ACTION SHALL BE FOREVER BARRED.

- 5. Section 14 of the Agreement, under the heading "Dispute Resolution," shall be supplemented by the addition of the following new Section 14.5, which shall be considered an integral part of the Agreement:
 - 14.5 Nothing contained in this Section 14 shall constitute a condition, stipulation, or provision purporting to bind any person to waive compliance with any provision of the Illinois Franchise Disclosure Act or any other Illinois law (as long as the jurisdictional requirements of the Illinois Franchise Disclosure Act are met).
 - 6. The Agreement shall be amended by the addition of the following language:

Nothing contained in this Agreement shall constitute a condition, stipulation, or provision purporting to bind any person to waive compliance with any provision of the Illinois Franchise Disclosure Act or any other Illinois law (as long as the jurisdictional requirements of the Illinois Franchise Disclosure Act are met).

- 7. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.
- 8. Each provision of this amendment shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Illinois Franchise Disclosure Act are met independently without reference to this amendment.

ACKNOWLEDGMENT:

inconsistent portion of the	e applicable foregoing state law addendum, if any, supe Multiple Unit Purchase Agreement or Franchise Agreeme , 20, and of the Franchise Disclosure Docum	nt dated the
	MTY FRANCHISING USA, INC.:	
	Ву:	
	Name:	
	Title:	
	Date:	
	FRANCHISEE:	
	Ву:	
	Date:	
	GUARANTOR:	
	Ву:	
	Date:	

Maryland Addendum to Franchise Disclosure Document and Franchise Agreement

ITEM 5 of the Franchise Disclosure Document, and the appropriate sections of the Franchise Agreement are amended to state "Based upon our financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed you to us and/or our affiliates shall be deferred until we complete all our preopening obligations to you under the Franchise Agreement."

ITEM 17 of the Franchise Disclosure Document, and the appropriate sections of the Franchise Agreement are amended to provide that the general release required, as a condition of renewal, sale and/or assignment transfer, shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

ITEM 17 of the Franchise Disclosure Document and Franchise Agreement are amended to state that you may bring a lawsuit in Maryland for claims arising under the Maryland Franchise Registration & Disclosure Law. Any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

ITEM 17 of the Franchise Disclosure Document and Franchise Agreement are amended to state that any limitation of claims provision will not act to reduce the amount of time afforded a franchise for bringing claims under the Maryland Franchise Registration and Disclosure Law.

ITEM 17 of the Franchise Disclosure Document and appropriate sections of the Franchise Agreement provide that certain disputes are resolved through arbitration. A Maryland franchise regulation states that it is an unfair or deceptive practice to require a franchisee to waive its right to file a lawsuit in Maryland claiming a violation of the Maryland Franchise Law. In light of the Federal Arbitration Act, there is some dispute as to whether this forum selection requirement is legally enforceable.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

[SIGNATURE PAGE FOLLOWS]

ACKNOWLEDGMENT:

inconsistent portion of the Multiple Uni	e foregoing state law addendum, if any, supersedes an it Purchase Agreement or Franchise Agreement dated the 20, and of the Franchise Disclosure Document.
	MTY FRANCHISING USA, INC.:
	Ву:
	Name:
	Title:
	Date:
	FRANCHISEE:
	Ву:
	Date:
	GUARANTOR:
	By:
	Date:

Michigan Disclosure

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU:

- 1. A PROHIBITION ON THE RIGHT OF A FRANCHISEE TO JOIN AN ASSOCIATION OF FRANCHISEES.
- 2. A REQUIREMENT THAT A FRANCHISEE ASSENT TO A RELEASE, ASSIGNMENT, NOVATION, WAIVER, OR ESTOPPEL WHICH DEPRIVES A FRANCHISEE OF RIGHTS AND PROTECTIONS PROVIDED IN THIS ACT. THIS SHALL NOT PRECLUDE A FRANCHISEE, AFTER ENTERING INTO A FRANCHISE AGREEMENT, FROM SETTLING ANY AND ALL CLAIMS.
- 3. A PROVISION THAT PERMITS A FRANCHISOR TO TERMINATE A FRANCHISE PRIOR TO THE EXPIRATION OF ITS TERM EXCEPT FOR GOOD CAUSE. GOOD CAUSE SHALL INCLUDE THE FAILURE OF THE FRANCHISEE TO COMPLY WITH ANY LAWFUL PROVISIONS OF THE FRANCHISE AGREEMENT AND TO CURE SUCH FAILURE AFTER BEING GIVEN WRITTEN NOTICE THEREOF AND A REASONABLE OPPORTUNITY, WHICH IN NO EVENT NEED BE MORE THAN 30 DAYS, TO CURE SUCH FAILURE.
- 4. A PROVISION THAT PERMITS A FRANCHISOR TO REFUSE TO RENEW A FRANCHISE WITHOUT FAIRLY COMPENSATING THE FRANCHISEE REPURCHASE OR OTHER MEANS FOR THE FAIR MARKET VALUE. AT THE TIME OF EXPIRATION. OF THE FRANCHISEE'S INVENTORY. SUPPLIES. EQUIPMENT. FIXTURES, AND FURNISHINGS. PERSONALIZED MATERIALS WHICH HAVE NO VALUE TO THE FRANCHISOR AND INVENTORY, SUPPLIES, EQUIPMENT, FIXTURES, AND FURNISHINGS NOT REASONABLY REQUIRED IN THE CONDUCT OF THE FRANCHISED BUSINESS ARE NOT SUBJECT TO COMPENSATION. THIS SUBSECTION APPLIES ONLY IF: (i) THE TERM OF THE FRANCHISE IS LESS THAN 5 YEARS; AND (ii) THE FRANCHISEE IS PROHIBITED BY THE FRANCHISE OR OTHER AGREEMENT FROM CONTINUING TO CONDUCT SUBSTANTIALLY THE SAME BUSINESS UNDER ANOTHER TRADEMARK, SERVICE MARK, TRADE NAME, LOGOTYPE, ADVERTISING, OR OTHER COMMERCIAL SYMBOL IN THE SAME AREA SUBSEQUENT TO THE EXPIRATION OF THE FRANCHISE OR THE FRANCHISEE DOES NOT RECEIVE AT LEAST 6 MONTHS ADVANCE NOTICE OF FRANCHISOR'S INTENT NOT TO RENEW THE FRANCHISE.
- 5. A PROVISION THAT PERMITS THE FRANCHISOR TO REFUSE TO RENEW A FRANCHISE ON TERMS GENERALLY AVAILABLE TO OTHER FRANCHISEES OF THE SAME CLASS OR TYPE UNDER SIMILAR CIRCUMSTANCES. THIS SECTION DOES NOT REQUIRE A RENEWAL PROVISION.
- 6. A PROVISION REQUIRING THAT ARBITRATION OR LITIGATION BE CONDUCTED OUTSIDE THIS STATE. THIS SHALL NOT PRECLUDE THE FRANCHISEE FROM ENTERING INTO AN AGREEMENT, AT THE TIME OF ARBITRATION, TO CONDUCT ARBITRATION AT A LOCATION OUTSIDE THIS STATE.

- 7. A PROVISION WHICH PERMITS A FRANCHISOR TO REFUSE TO PERMIT A TRANSFER OF OWNERSHIP OF A FRANCHISE, EXCEPT FOR GOOD CAUSE. THIS SUBDIVISION DOES NOT PREVENT A FRANCHISOR FROM EXERCISING A RIGHT OF FIRST REFUSAL TO PURCHASE THE FRANCHISE. GOOD CAUSE SHALL INCLUDE, BUT IS NOT LIMITED TO:
 - A. THE FAILURE OF THE PROPOSED FRANCHISEE TO MEET THE FRANCHISOR'S THEN CURRENT REASONABLE QUALIFICATIONS OR STANDARDS.
 - B. THE FACT THAT THE PROPOSED TRANSFEREE IS A COMPETITOR OF THE FRANCHISOR OR SUBFRANCHISOR.
 - C. THE UNWILLINGNESS OF THE PROPOSED TRANSFEREE TO AGREE IN WRITING TO COMPLY WITH ALL LAWFUL OBLIGATIONS.
 - D. THE FAILURE OF THE FRANCHISEE OR PROPOSED TRANSFEREE TO PAY ANY SUMS OWING TO THE FRANCHISOR OR TO CURE ANY DEFAULT IN THE FRANCHISE AGREEMENT EXISTING AT THE TIME OF THE PROPOSED TRANSFER.
- 8. A PROVISION THAT REQUIRES THE FRANCHISEE TO RESELL TO THE FRANCHISOR ITEMS THAT ARE NOT UNIQUELY IDENTIFIED WITH THE FRANCHISOR. THIS SUBDIVISION DOES NOT PROHIBIT A PROVISION THAT GRANTS TO A FRANCHISOR A RIGHT OF FIRST REFUSAL TO PURCHASE THE ASSETS OF A FRANCHISE ON THE SAME TERMS AND CONDITIONS AS A BONA FIDE THIRD PARTY WILLING AND ABLE TO PURCHASE THOSE ASSETS, NOR DOES THIS SUBDIVISION PROHIBIT A PROVISION THAT GRANTS THE FRANCHISOR THE RIGHT TO ACQUIRE THE ASSETS OF A FRANCHISE FOR THE MARKET OR APPRAISED VALUE OF SUCH ASSETS IF THE FRANCHISEE HAS BREACHED THE LAWFUL PROVISIONS OF THE FRANCHISE AGREEMENT AND HAS FAILED TO CURE THE BREACH IN THE MANNER PROVIDED IN SUBDIVISION (C).
- 9. A PROVISION WHICH PERMITS THE FRANCHISOR TO DIRECTLY OR INDIRECTLY CONVEY, ASSIGN, OR OTHERWISE TRANSFER ITS OBLIGATIONS TO FULFILL CONTRACTUAL OBLIGATIONS TO THE FRANCHISEE UNLESS PROVISION HAS BEEN MADE FOR PROVIDING THE REQUIRED CONTRACTUAL SERVICES.
- 10. NO STATEMENT, QUESTIONNAIRE, OR ACKNOWLEDGMENT SIGNED OR AGREED TO BY A FRANCHISEE IN CONNECTION WITH THE COMMENCEMENT OF THE FRANCHISE RELATIONSHIP SHALL HAVE THE EFFECT OF (I) WAIVING ANY CLAIMS UNDER ANY APPLICABLE STATE FRANCHISE LAW, INCLUDING FRAUD IN THE INDUCEMENT, OR (II) DISCLAIMING RELIANCE ON ANY STATEMENT MADE BY ANY FRANCHISOR, FRANCHISE SELLER, OR OTHER PERSON ACTING ON BEHALF OF THE FRANCHISOR. THIS PROVISION SUPERSEDES ANY OTHER TERM OF ANY DOCUMENT EXECUTED IN CONNECTION WITH THE FRANCHISE.

* * * *

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENDORSEMENT BY THE ATTORNEY GENERAL.

* * * *

IF THE FRANCHISOR'S MOST RECENT FINANCIAL STATEMENTS ARE UNAUDITED AND SHOW A NET WORTH OF LESS THAN \$100,000.00, THE FRANCHISOR MUST, AT THE REQUEST OF THE FRANCHISEE, ARRANGE FOR THE ESCROW OF INITIAL INVESTMENT AND OTHER FUNDS PAID BY THE FRANCHISEE UNTIL THE OBLIGATIONS TO PROVIDE REAL ESTATE, IMPROVEMENTS, EQUIPMENT, INVENTORY, TRAINING, OR OTHER ITEMS INCLUDED IN THE FRANCHISE OFFERING ARE FULFILLED. AT THE OPTION OF THE FRANCHISOR, A SURETY BOND MAY BE PROVIDED IN PLACE OF ESCROW.

* * * *

THE NAME AND ADDRESS OF THE FRANCHISOR'S AGENT IN THIS STATE AUTHORIZED TO RECEIVE SERVICE OF PROCESS IS: MICHIGAN DEPARTMENT OF COMMERCE, CORPORATION AND SECURITIES BUREAU, 6546 MERCANTILE WAY, P.O. BOX 30222, LANSING, MICHIGAN 48910.

ANY QUESTIONS REGARDING THIS NOTICE SHOULD BE DIRECTED TO:

DEPARTMENT OF THE ATTORNEY GENERAL'S OFFICE
CONSUMER PROTECTION DIVISION
ATTN: FRANCHISE
670 G. MENNEN WILLIAMS BUILDING
LANSING, MICHIGAN 48913

NOTE: NOTWITHSTANDING PARAGRAPH (F) ABOVE, WE INTEND TO, AND YOU AGREE THAT WE AND YOU WILL, ENFORCE FULLY THE PROVISIONS OF THE ARBITRATION SECTION OF OUR AGREEMENTS. WE BELIEVE THAT PARAGRAPH (F) IS UNCONSTITUTIONAL AND CANNOT PRECLUDE US FROM ENFORCING THE ARBITRATION PROVISIONS.

Minnesota Disclosure

In recognition of the requirements of the Minnesota Franchises Law, Minn. Stat. §§ 80C.01 through 80C.22, and of the Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce, Minn. Rules §§ 2860.0100 through 2860.9930, the Disclosure Document for MTY Franchising USA, Inc. for use in the State of Minnesota shall be amended to include the following:

1. Item 13, "Trademarks," shall be amended by the addition of the following paragraph at the end of the Item:

Pursuant to Minnesota Stat. Sec. 80C.12, Subd. 1(g), we are required to protect any rights which you have to use our proprietary marks.

2. Item 17, "Renewal, Termination, Transfer and Dispute Resolution," shall be amended by the addition of the following paragraphs:

With respect to franchisees governed by Minnesota law, we will comply with Minn. Stat. Sec. 80C.14, Subds. 3, 4, and 5 which require, except in certain specified cases, that a franchisee be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice of non-renewal of the Franchise Agreement, and that consent to the transfer of the franchise not be unreasonably withheld.

Pursuant to Minn. Rule 2860.4400D, any general release of claims that you or a transferor may have against us or our shareholders, directors, employees and agents, including without limitation claims arising under federal, state, and local laws and regulations shall exclude claims you or a transferor may have under the Minnesota Franchise Law and the Rules and Regulations promulgated thereunder by the Commissioner of Commerce.

Minn. Stat. § 80C.21 and Minn. Rule 2860.4400J prohibit us from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring you to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the disclosure document or agreement can abrogate or reduce any of your rights as provided for in Minnesota Statutes, Chapter 80C, or your rights to jury trial, any procedure, forum, or remedies as may be provided for by the laws of the jurisdiction.

3. Each provision of this addendum shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Minnesota Franchises Law or the Rules and Regulations promulgated thereunder by the Minnesota Commission of Commerce are met independently without reference to this addendum to the disclosure document.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Minnesota Amendment to the Franchise Agreement

In recognition of the requirements of the Minnesota Franchises Law, Minn. Stat. §§ 80C.01 through 80C.22, and of the Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce, Minn. Rules §§ 2860.0100 through 2860.9930, the parties to the attached MTY Franchising USA, Inc. Franchise Agreement (the "Agreement") agree as follows:

- 1. Section 2.2 (g) of the Agreement, under the heading "Renewal," shall be deleted in its entirety and shall have no force or effect, and the following paragraph shall be inserted in lieu thereof:
 - 2.2 (g) Franchisee (and all of its owners, including the Guarantor) shall each have executed a general release, in form and substance satisfactory to Franchisor, of any and all claims, demands, actions and causes of action that they may have or have had against Franchisor, its Affiliates or any of their respective officers, directors, shareholders, employees, agents or legal counsel, in their corporate or individual capacities, arising in connection with the execution or performance of the Agreement or any Contract; excluding only such claims as Franchisee may have that have arisen under the Minnesota Franchises Law and the Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce.
- 2. Section 2.2 of the Agreement, under the heading "Renewal," shall be supplemented by the addition of the following new paragraph:

Minnesota law provides franchisees with certain non-renewal rights. In sum, Minn. Stat. § 80C.14 (subd. 4) currently requires, except in certain specified cases, that a franchisee be given 180 days' notice of non-renewal of the Franchise Agreement.

3. Section 9.1 of the Agreement, under the heading "Marks," shall be amended by the addition of the following new paragraph:

Pursuant to Minnesota Stat. Sec. 80C.12, Subd. 1(g), Franchisor is required to protect any rights Franchisee may have to Franchisor Proprietary Marks.

- 4. Section 12.2 (c) of the Agreement, under the heading "Assignment or Transfer by Franchisee," shall be deleted in its entirety and shall have no force or effect, and the following paragraph shall be inserted in lieu thereof:
 - 12.2 (c) the Franchisee (and all of its owners, including Guarantor) shall have delivered to the Franchisor a complete release of its/their claims against the Franchisor, its Affiliates and their respective directors and officers, in form and substance satisfactory to the Franchisor; excluding only such claims as the Franchisee may have under the Minnesota Franchises Law and the Rules and Regulations promulgated thereunder by the Minnesota Commissioner of Commerce;
- 5. Section 12 of the Agreement, under the heading "Assignment and Transfer," shall be supplemented by the addition of the following new paragraph 12.10:
- 12.10 Minnesota law provides franchisees with certain transfer rights. In sum, Minn. Stat. §80C.14 (subd. 5) currently requires that consent to the transfer of the franchise may not be unreasonably withheld.

- 6. Section 10 of the Agreement, under the heading "Default and Termination of Franchise," shall be supplemented by the following new paragraph 10.13:
 - 10.13 Minnesota law provides franchisees with certain termination rights. In sum, Minn. Stat. § 80C.14 (subd. 3) currently requires, except in certain specified cases, that a franchisee be given 90 days' notice of termination (with 60 days to cure) of the Franchise Agreement.
- 7. Sections 14.1(f.) and (g.) of the Agreement, under the heading "Dispute Resolution," shall be deleted in their entirety and shall have no force or effect; and the following paragraphs shall be substituted in lieu thereof:
 - (f.) Nothing contained in this Agreement may bar or delay Franchisor's right to seek injunctive relief against threatened or actual conduct that may cause it loss or damages under the usual equity rules, including the applicable rules for obtaining restraining orders and preliminary injunctions. Franchisor need not prove actual or irreparable damages. No bond or other security is required to be posted (or if required, the sum of one thousand dollars [\$1000.00] is deemed sufficient). Nothing contained in this Agreement may be construed to limit or to preclude Franchisor from joining with any action for injunctive or provisional relief any monetary claims that Franchisor may have against the Franchisee or the Guarantor that arise out of the acts or omissions to act giving rise to the action for injunctive or provisional relief.
 - (g.) Without limiting the generality of Section 14.1(f) above, the Franchisee and the Guarantor agree that the Franchisor may seek injunctive relief in addition to such other relief as may be available to it at equity or at law for any dispute involving the proprietary marks, termination of this Agreement, or enforcement of the confidentiality provisions set forth in this Agreement, and any dispute involving enforcement of the covenants set out in Section 12 of this Agreement. The Franchisees' (and Guarantors') only remedy if an injunction is entered against any or all of them will be the dissolution of that injunction, if warranted, upon due hearing (all claims for damages by reason of the wrongful issuance of such injunction being expressly waived hereby).
- 8. Section 14 of the Agreement, under the heading "Dispute Resolution", shall be amended by the addition of the following paragraph 14.5, which shall be considered an integral part of the Agreement:
 - 14.5 Minn. Stat. §80C.21 and Minn. Rule 2860.4400J prohibit the Franchisor from requiring litigation to be conducted outside Minnesota, requiring waiver of a jury trial, or requiring you to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the disclosure document or agreement can abrogate or reduce any of the Franchisee's or Guarantor's rights as provided for in Minnesota Statutes, Chapter 80C, or any rights to jury trial, any procedure, forum, or remedies provided as may be for by the laws of the jurisdiction.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

9. Each	provision of this Agreement shall be effective only to the extent, with respec
to such provision, th	at the jurisdictional requirements of the Minnesota Franchises Law or the
Rules and Regulation	ns promulgated thereunder by the Minnesota Commissioner of Commerce
are met independent	y without reference to this addendum to the Agreement.

ACKNOWLEDGMENT:

inconsistent portion of the Multiple Unit F	oregoing state law addendum, if any, supersedes any Purchase Agreement or Franchise Agreement dated the D, and of the Franchise Disclosure Document.
1	MTY FRANCHISING USA, INC.:
ı	Зу:
,	Name:
	Title:
	Date:
1	FRANCHISEE:
1	Зу:
1	Date:
	GUARANTOR:
1	Зу:
1	Date:

New York Disclosure

1. The following information is added to the cover page of the Franchise Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN **EXHIBIT A** OR YOUR PUBLIC LIBRARY FOR SOURCES OF INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THE FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE, CONTACT THE FEDERAL TRADE COMMISSION AND NEW YORK STATE DEPARTMENT OF LAW, BUREAU OF INVESTOR PROTECTION AND SECURITIES, 120 BROADWAY, 23RD FLOOR, NEW YORK, NEW YORK, 10271. THE FRANCHISOR MAY, IF IT CHOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS WHICH ARE LESS FAVORABLE THAN THOSE SET FORTH IN T HIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is added at the end of Item 3:

Except as provided above, with regard to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge, or within the 10 year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.
- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective e order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.

3. The following is added to the end of Item 4:

Neither the franchisor, its affiliate, its predecessors, officers, or general partner during the 10-year period immediately before the date of the disclosure document: (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code; (b) obtained a discharge of its debts under the bankruptcy code; or (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code during or within 1 year after the officer or general partner of the franchisor held this position in the company or partnership.

4. The following is added to the end of Item 5:

The initial franchise fee constitutes part of our general operating funds and will be used as such in our discretion.

5. The following is added to the end of the "Summary" sections of Item 17(c), titled "Requirements for franchisee to renew," and Item 17(m), entitled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issues thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687.4 and 687.5 be satisfied.

6. The following language replaces the "Summary" section of Item 17(d), titled "**Termination** by franchisee":

You may terminate the agreement on any grounds available by law.

7. The following is added to the end of "Summary" section of Item 17(j), titled "Assignment of contract by franchisor":

However, no assignment will be made except to an assignee who in good faith and judgment of the franchisor, is willing and financially able to assume the franchisor's obligations under the Franchise Agreement.

8. The following is added to the end of the "Summary" sections of Item 17(v), titled "Choice of forum":, and Item 17(w), titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

9. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor,

franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

New York Amendment to the Franchise Agreement

In recognition of the requirements of the New York General Business Law, Article 33, Sections 680 through 695, and of the regulations promulgated thereunder (N.Y. Comp. Code R. & Regs., tit. 13, §§ 200.1 through 201.16), the parties to the attached MTY Franchising USA, Inc. Franchise Agreement (the "Agreement") agree as follows:

- 1. Section 2.2 (g) of the Agreement, under the heading "Renewal," shall be deleted in its entirety, and shall have no force or effect; and the following paragraph shall be substituted in lieu thereof:
 - 2.2 (g) the Franchisee (and all of its owners, including Guarantor) shall each have executed a general release, in form and substance satisfactory to Franchisor, of any and all claims, demands, actions and causes of action that they may have or have had against Franchisor, its Affiliates or any of their respective officers, directors, shareholders, employees, agents or legal counsel, in their corporate or individual capacities, arising in connection with the execution or performance of the Agreement or any Contract; provided, however, that all rights enjoyed by Franchisee and any causes of action arising in its favor from the provisions of New York General Business Law Sections 680-695 and the regulations issued thereunder, shall remain in force; it being the intent of this provision that the non-waiver provisions of N.Y. Gen. Bus. Law Sections 687.4 and 687.5 be satisfied; and
- 2. Section 12.2(c) of the Agreement, under the heading "Assignment or Transfer by Franchisee," shall be deleted in its entirety, and shall have no force or effect; and the following paragraph shall be substituted in lieu thereof:
 - 12.2(c) the Franchisee (and all of its owners, including Guarantor) shall have delivered to the Franchisor a complete release of its/their claims against the Franchisor, its Affiliates and their respective directors and officers, in form and substance satisfactory to the Franchisor; provided, however, that all rights enjoyed by the transferor and any causes of action arising in its favor from the provisions of New York General Business Law Sections 680-695 and the regulations issued thereunder, shall remain in force; it being the intent of this provision that the non-waiver provisions of N.Y. Gen. Bus. Law Sections 687.4 and 687.5 be satisfied;
- 3. Sections 14.1(f) and (g) of the Agreement, under the heading "Dispute Resolution," shall be deleted in their entirety, and shall have no force or effect; and the following paragraph shall be substituted in lieu thereof:
 - (f.) Nothing contained in this Agreement may bar or delay Franchisor's right to seek injunctive relief against threatened or actual conduct that may cause it loss or damages under the usual equity rules, including the applicable rules for obtaining restraining orders and preliminary injunctions. Franchisor need not prove actual or irreparable damages. No bond or other security is required to be posted (or if required, the sum of one thousand dollars [\$1000.00] is deemed sufficient). Nothing contained in this Agreement may be construed to limit or to preclude Franchisor from joining with any action for injunctive or provisional relief any monetary claims that Franchisor may have against the Franchisee or the Guarantor that arise out of the acts or omissions to act giving rise to the action for injunctive or provisional relief.

- (g.) Without limiting the generality of Section 14.1(f)) above, the Franchisee and the Guarantor agree that the Franchisor may seek injunctive relief in addition to such other relief as may be available to it at equity or at law for any dispute involving the proprietary marks, termination of this Agreement, or enforcement of the confidentiality provisions set forth in this Agreement, and any dispute involving enforcement of the covenants set out in Section 12 of this Agreement. The Franchisees' (and Guarantors') only remedy if an injunction is entered against any or all of them will be the dissolution of that injunction, if warranted, upon due hearing (all claims for damages by reason of the wrongful issuance of such injunction being expressly waived hereby).
- 4. Section 14 of the Agreement, under the heading "Dispute Resolution," shall be supplemented by the addition of Section 14.5:
 - 14.5 Nothing in this Agreement should be considered a waiver of any right conferred upon Franchisee by New York General Business Law, Sections 680-695.
- 5. There are circumstances in which an offering made by Franchisor would not fall within the scope of the New York General Business Law, Article 33, such as when the offer and acceptance occurred outside the state of New York. However, an offer or sale is deemed made in New York if Franchisee is domiciled in or the franchise will be opening in New York. Franchisor is required to furnish a New York prospectus to every prospective franchisee who is protected under the New York General Business Law, Article 33.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

[SIGNATURE PAGE FOLLOWS]

ACKNOWLEDGMENT:

inconsistent portion of the Multiple Uni	e foregoing state law addendum, if any, supersedes and Purchase Agreement or Franchise Agreement dated the 20, and of the Franchise Disclosure Document.
	MTY FRANCHISING USA, INC.:
	By:
	Name:
	Title:
	Date:
	FRANCHISEE:
	By:
	Date:
	GUARANTOR:
	Ву:
	Date:

North Dakota Amendment to the FDD and Franchise Agreement

ITEM 5 of the Franchise Disclosure Document and Section 3.1 of the Franchise Agreement are amended by the addition of the following language to the original language:

In lieu of an impound of franchise fees, the Franchisor will not require or accept the payment of any initial franchise fees until the franchisee (a) has received all initial training that it is entitled to under the franchise agreement or offering circular, and (b) is open for business.

Sections of the Franchise Disclosure Document and Franchise Agreement requiring that you sign a general release, estoppel or waiver as a condition of renewal and or assignment, may not be enforceable as they relate to releases of the North Dakota Franchise Investment Law.

The Franchise Agreement contains a covenant not to compete which may not be enforceable under North Dakota law.

Sections of the Franchise Disclosure Document and Franchise Agreement requiring resolution of disputes to be outside North Dakota may not be enforceable under Section 51-19-09 of the North Dakota Franchise Investment Law, and are amended accordingly to the extent required by law.

Sections of the Franchise Disclosure Document and Franchise Agreement relating to choice of law, may not be enforceable under Section 51-19-09 of the North Dakota Franchise Investment Law, and are amended accordingly to the extent required by law.

Sections of the Franchise Disclosure Document and Franchise Agreement requiring you to consent to liquidated damages and/or termination penalties, may not be enforceable under Section 51-19-09 of the North Dakota Franchise Investment Law, and are amended accordingly to the extent required by law.

Sections of the Franchise Disclosure Document and Franchise Agreement requiring you to consent to a waiver of trial by jury, may not be enforceable under Section 51-19-09 of the North Dakota Franchise Investment Law, and are amended accordingly to the extent required by law.

Sections of the Franchise Disclosure Document and Franchise Agreement requiring you to consent to a waiver of exemplary and punitive damages, may not be enforceable under Section 51-19-09 of the North Dakota Franchise Investment Law, and are amended accordingly to the extent required by law.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

STATEMENT OF DISCLOSURE DOCUMENT ACCURACY

THE FRANCHISOR REPRESENTS THAT THIS DISCLOSURE DOCUMENT DOES NOT KNOWINGLY OMIT ANY MATERIAL FACT OR CONTAIN ANY UNTRUE STATEMENT OF A MATERIAL FACT.

Virginia Disclosure

In recognition of the restrictions contained in Section 13.1-564 of the Virginia Retail Franchising Act, the Franchise Disclosure Document for MTY Franchising USA, Inc. for use in the Commonwealth of Virginia shall be amended as follows:

1. <u>Item 17</u>. The following statement is added to Item 17.h. of the table:

Pursuant to Section 13.1-564 of the Virginia Retail Franchising Act, it is unlawful for a franchisor to cancel a franchise without reasonable cause. If any grounds for default or termination stated in the Franchise Agreement does not constitute "reasonable cause," as that term may be defined in the Virginia Retail Franchising Act or the laws of Virginia, that provision may not be enforceable.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

Washington Disclosure

In recognition of the requirements of the Washington Franchise Investment Protection Act, Wash. Rev. Code §§ 19.100.180, the Franchise Disclosure Document for MTY Franchising USA, Inc. in connection with the offer and sale of franchises for use in the State of Washington shall be amended to include the following:

1. Item 17, "Renewal, Termination, Transfer and Dispute Resolution," shall be amended by the addition of the following paragraphs at the conclusion of the Item:

The state of Washington has a statute, RCW 19.100.180, which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration involving a franchise purchased in Washington, the arbitration site shall be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration, or as determined by the arbitrator.

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW shall prevail.

A release or waiver of rights executed by a franchisee shall not include rights under the Washington Franchise Investment Protection Act except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, rights or remedies under the Act such as a right to a jury trial may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

2. Item 17, "Renewal, Termination, Transfer and Dispute Resolution," shall be amended by deleting "d" and the following new "d." shall be substituted in lieu thereof:

Selection in Franchise

d. Termination by you None Pursuant to Washington State Law, the franchisee may terminate the Agreement upon any grounds available by law.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

3. Each provision of this addendum to the disclosure document shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Washington Franchise Investment Protection Act, Wash. Rev. Code §§ 19.100.180, are met independently without reference to this addendum to the disclosure document.

Washington Amendment to the Franchise Agreement

In recognition of the requirements of the Washington Franchise Investment Protection Act, Wash. Rev. Code §§ 19.100.010 through 19.100.940, the parties to the attached MTY Franchising USA, Inc. Franchise Agreement agree as follows:

The state of Washington has a statute, RCW 19.100.180, which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration involving a franchise purchased in Washington, the arbitration site shall be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration, or as determined by the arbitrator.

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW shall prevail.

A release or waiver of rights executed by a franchisee shall not include rights under the Washington Franchise Investment Protection Act except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, rights or remedies under the Act such as a right to a jury trial may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Each provision of this amendment shall be effective only to the extent, with respect to such provision, that the jurisdictional requirements of the Washington Franchise Investment Protection Act, Wash. Rev. Code §§ 19.100.010 through 19.100.940, are met independently without reference to this amendment.

ITEM 5 of the Franchise Disclosure Document and Section 3.1 of the Franchise Agreement are amended by the addition of the following language to the original language:

In lieu of an impound of franchise fees, the Franchisor will not require or accept the payment of any initial franchise fees until the franchisee (a) has received all initial training that it is entitled to under the franchise agreement or offering circular, and (b) is open for business.

No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.

[SIGNATURE PAGE FOLLOWS]

ACKNOWLEDGMENT:

inconsistent portion of the Multiple Uni	e foregoing state law addendum, if any, supersedes an it Purchase Agreement or Franchise Agreement dated the 20, and of the Franchise Disclosure Document.
	MTY FRANCHISING USA, INC.:
	Ву:
	Name:
	Title:
	Date:
	FRANCHISEE:
	Ву:
	Date:
	GUARANTOR:
	By:
	Date:



EXHIBIT I

Manchu WOK

Required Lease Terms

REQUIRED LEASE TERMS

The Terms and Conditions in the attached Lease Addendum to Lease Agreement ("Lease Addendum") must be included in the Franchisee's Lease for the location of the Franchised Business via execution of the attached Lease Addendum or through modifications to the actual Lease

LEASE ADDENDUM TO LEASE AGREEMENT

("Lease Addendum")

Dated:	, 20	
be	tween	
;	and	
(Landlord Name)	(Tenant Name)	
(Address of "Premises")		

1. Use of Premises.

During the term of the Lease, the Premises may be used only for the operation of a quick service restaurant under the Manchu WOK, trademarks, trade names, and logos, which specialize in the sale of Chinese and Asian-style food and drinks and any other items sold under the Manchu WOK system. Landlord consents to Tenant's use of such trademarks, trade names, logos, tag lines, signs, décor items, color schemes, and related components of the Manchu WOK franchise system as MTY Franchising USA, Inc., franchisor of the Manchu WOK brand ("Franchisor"), or any of its affiliates, may prescribe for Manchu WOK franchisees and which may be altered or changed by Franchisor from time to time.

2. Assignment and Notices.

- a. Notwithstanding anything to the contrary in the Lease, Tenant shall have the right to assign the Lease and all amendments thereto, as applicable, including this Lease Addendum (collectively, "Lease"), and all rights hereunder to Franchisor or any of its affiliates, or to a third-party Manchu WOK franchisee approved by Franchisor upon the expiration or termination of the current franchise agreement by and between Franchisor and Tenant ("Franchise Agreement"), without obtaining Landlord's consent and without the imposition of any assignment fee or similar charge. Landlord shall not accelerate the rent owed hereunder in connection with such assignment(s), so long as Franchisor, its affiliate(s) or its third-party franchisee(s) assumes in writing the obligations of Tenant under the Lease. Nothing in this Section 2.a shall serve to extend the term of the Lease or provide Franchisor any occupancy rights, options to renew or other rights not expressly set forth to Tenant in the Lease.
- b. Landlord agrees to furnish Franchisor with copies of any and all letters and notices to Tenant pertaining to the Lease and the Premises at the same time that such letters and notices are sent to Tenant. Landlord further agrees that, if it intends to terminate the Lease, the Landlord will give Franchisor the same advance written notice of such intent as provided to Tenant, specifying in such notice all defaults that are the cause of the proposed termination. Franchisor shall have the right to cure, at its sole option, any such default within the time periods granted to Tenant under the Lease. If neither Tenant nor Franchisor cures all such defaults within said time periods (or such longer cure periods as may be specifically permitted by the Lease), then the Landlord may terminate the Lease, re-enter the Premises and/or exercise all other rights as set forth in the Lease.
- c. Prior to the expiration or termination of the Lease, Franchisor shall have the right to enter the Premises to make any reasonable modifications or reasonable alterations

necessary to protect Franchisor's interest in the Manchu WOK brand and its trademarks, trade names, logos, tag lines, signs, décor items, color schemes, and related components of the Manchu WOK franchise system, or to cure any default under the Lease, and Landlord and Tenant agree that Franchisor shall not be liable for trespass or any other crimes or tort.

Notices.

All notices and demands required to be given hereunder shall be in writing and shall be sent by personal delivery, expedited delivery service, certified or registered mail, return receipt requested, first-class postage prepaid, email or facsimile (provided that the sender confirm the facsimile by sending an original confirmation copy by certified transmission), to the respective parties at the following addresses unless and until a different address has been designated by written notice to the other parties.

If directed to <u>Tenant</u> , the notice shall be addressed to:		
Attn:		
Facsimile:		
Email:		
If directed to <u>Landlord</u> , the notice shall be address to:		
Attn:		
Facsimile:		
Email:		
If directed to <u>Franchisor</u> , the notice shall be addressed to		
MTY Franchising USA, Inc.		
9311 E. Via De Ventura		
Scottsdale, AZ 85258		
Attn: Real Estate Department		

Facsimile: (480) 362-4792 Email: leases@kahalamgmt.com

Any notices sent by personal delivery shall be deemed given upon receipt. Any notices given by email or facsimile shall be deemed given on the business day of transmission, provided confirmation is made as provided above. Any notice sent by expedited delivery service or registered or certified mail shall be deemed given three (3) business days after the time of mailing. Any change in the foregoing addresses shall be effected by giving fifteen (15) days written notice of such change to the other parties.

4. Amendments.

Landlord and Tenant will not amend, renew, extend or otherwise modify the Lease in any manner which would materially affect any of the provisions in this Lease Addendum without Franchisor's prior written consent.

5. **Right of First Refusal.**

Landlord and Tenant agree that should Landlord and Tenant desire to terminate the Lease prior to the scheduled expiration date, Landlord will notify Franchisor of the proposed termination not less than thirty (30) days in advance of the proposed termination date. Franchisor shall have a right of first refusal to assume the Lease whereby Tenant will assign the Lease to Franchisor and Franchisor will assume all rights and obligations of Tenant under the Lease.

6. Miscellaneous.

The terms and conditions of this Lease Addendum will supersede any conflicting terms of the Lease. Any capitalized term not specifically defined in this Lease Addendum shall have the meaning ascribed to such term in the Lease.

IN WITNESS WHEREOF, the parties hereto, by and through their respective representatives authorized to enter into and bind each respective party without further consent or authorization, have duly executed and delivered this Lease Addendum in duplicate as of the date set forth above.

LANDLORD:	, a(n)
	By:
	Name:
	Title:
TENANT:	, a(n)
	By:
	Name:
	Title:
	D. a
	By:
	Name:
	Title:



EXHIBIT J

Manchu WOK

State Effective Dates

Manchu Wok STATE EFFECTIVE DATES

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration:

California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

California	March 28, 2025
Hawaii	Not Registered
Illinois	March 28, 2025
Indiana	March 28, 2025
Maryland	Not Registered
Michigan	Pending
Minnesota	Not Registered
New York	March 28, 2025
North Dakota	Pending
Rhode Island	Not Registered
South Dakota	Not Registered
Virginia	Pending
Washington	Pending
Wisconsin	Pending

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.



EXHIBIT K

Manchu WOK

Receipt

EXHIBIT K RECEIPT

This Disclosure Document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If MTY Franchising USA, Inc. offers you a franchise, it must provide this Disclosure Document to you (a) 14 calendar days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale, or (b) Under New York and Rhode Island law at the earlier of (i) your first personal meeting to discuss the franchise, or (ii) 10 business days before you sign a binding agreement with, or make payment to us or an affiliate in connection with the proposed franchise sale, or (c) lowa requires that we give you this disclosure document at the earlier of the first personal meeting or 14 days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship, or (d) Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.

If MTY Franchising USA, Inc. does not deliver this Disclosure Document on time or if it contains a false or misleading statement or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and to the appropriate state agency listed in **Exhibit A**.

The franchisor is MTY Franchising USA, Inc., located at 9311 E. Via De Ventura, Scottsdale, Arizona 85258. Its telephone number is (480) 362-4800. The franchise seller for this offering is A list of any additional franchise sellers, if any, will be attached to this Receipt at the time of sale. MTY Franchising USA, Inc. authorizes the respective state agencies identified on Exhibit A to receive service of process for it in the particular state. I received a Franchise Disclosure Document dated (and with effective dates of state registration as listed in the State Effective Dates page) that included the following Exhibits: List of State Regulators/Agents for Service of F Current Franchisees of MTY Franchising USA, **Process** Inc. F Current Franchisees of MTY Franchising Financial Statements G Former Franchisees of MTY Franchising USA В State-Specific Disclosure and State-Specific С Franchise Agreement Н Agreement Amendments D Sublease Agreement (subject to amendment Required Lease Terms to accommodate local laws) Table of Contents of Operating Manual State Effective Dates Ε J In-Store Training Release and Waiver of Receipt Liability Agreement Prospective Franchisee **Date Received** Name (Please print)

Issuance date: March 28, 2025

Address

(Retain this copy)

EXHIBIT K RECEIPT

This Disclosure Document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If MTY Franchising USA, Inc. offers you a franchise, it must provide this Disclosure Document to you (a) 14 calendar days before you sign a binding agreement with, or make a payment to, us or an affiliate in connection with the proposed franchise sale, or (b) Under New York and Rhode Island law at the earlier of (i) your first personal meeting to discuss the franchise, or (ii) 10 business days before you sign a binding agreement with, or make payment to us or an affiliate in connection with the proposed franchise sale, or (c) lowa requires that we give you this disclosure document at the earlier of the first personal meeting or 14 days before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship, or (d) Michigan requires that we give you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration. whichever occurs first.

If MTY Franchising USA, Inc. does not deliver this Disclosure Document on time or if it contains a false or misleading statement or a material omission, a violation of federal and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580 and to the appropriate state agency listed in Exhibit A.

The franchisor is MTY Franchising USA, Inc., located Its telephone number is (480) 362-4800. The franchism	at 9311 E. Via De Ventura, Scottsdale, Arizona 85258. anchise seller for this offering is
. ,	_, () A list of any additional franchise
sellers, if any, will be attached to this Receipt at the t	
MTY Franchising USA, Inc. authorizes the respective of process for it in the particular state.	state agencies identified on Exhibit A to receive service
I received a Franchise Disclosure Document dated (the State Effective Dates page) that included the following	and with effective dates of state registration as listed in owing Exhibits:
A List of State Regulators/Agents for Service of Process	F Current Franchisees of MTY Franchising USA, Inc.
B Financial Statements	G Former Franchisees of MTY Franchising USA
C Franchise Agreement	H State-Specific Disclosure and State-Specific Agreement Amendments
D Sublease Agreement (subject to amendment to accommodate local laws)	I Required Lease Terms
E Table of Contents of Operating Manual	J State Effective Dates
E-1 In-Store Training Release and Waiver of Liability Agreement	K Receipt
Date Received Pro	spective Franchisee
Nar	me (Please print)
Ado	lress

(Return to MTY Franchising USA)

Issuance date: March 28, 2025